

University of Trento

DOCTORAL THESIS

Geometric applications of Linear and Nonlinear Potential Theory

Author: Mattia FOGAGNOLO

Supervisor: Prof. Lorenzo MAZZIERI

A thesis submitted in fulfillment of the requirements for the degree of Doctor of Philosophy

in the

Department of Mathematics

Quis custodiet ipsos custodes?

Decimus Iunius Iuvenalis

UNIVERSITY OF TRENTO

Abstract

Department of Mathematics

Doctor of Philosophy

Geometric applications of Linear and Nonlinear Potential Theory

by Mattia FOGAGNOLO

We provide geometric inequalities on \mathbb{R}^n and on general manifolds with nonnegative Ricci curvature by employing suitable monotone quantities along the flow of capacitary and p-capacitary potentials, as well as through related boundary value problems. Among the main achievements, we cite

- (i) a Willmore-type inequality on manifolds with nonnegative Ricci curvature leading in turn to the sharp Isoperimetric Inequality on 3-manifolds with nonnegative Ricci curvature;
- (ii) enhanced Kasue/Croke-Kleiner splitting theorems;
- (iii) a generalised Minkowski-type inequality in \mathbb{R}^n holding with no assumptions on the boundary of the domain considered except for smoothness;
- (iv) a complete discussion of maximal volume solutions to the least area problem with obstacle on Riemannian manifolds and its relation with the variational *p*-capacity.

Ringraziamenti

Vorrei intanto ringraziare l'Università di Trento, in particolare la commissione che mi ha dato la possibilità di perseguire un Dottorato di Ricerca in Matematica. Ringrazio calorosamente il mio supervisore Prof. Lorenzo Mazzieri, innanzitutto per il coraggio di avermi accettato come dottorando, nonostante le mie iniziali enormi lacune in Geometria Differenziale e non solo, poi per avermi introdotto all'Analisi Geometrica, ramo che non conoscevo prima ma che ho scoperto racchiudere molti degli aspetti che più mi piacciono della Matematica, e soprattutto per aver portato pazienza e cercato di farmi crescere come matematico durante questi anni, sempre con simpatia e leggerezza. Il suo stile, gusto e le sue capacità sono e credo che rimarranno sempre un modello per me. Ringrazio anche Virginia e Andrea con i quali ho collaborato, assieme a Lorenzo, nell'ottenere la maggior parte dei risultati raccolti in questa tesi.

Ringrazio i miei colleghi di dottorato, molti dei quali sono diventati per me ottimi amici e i coinquilini con cui ho convissuto questi anni, per cui vale lo stesso discorso. Ringrazio anche tutte le persone conosciute in giro per varie conferenze con cui ho fatto amicizia, e più in generale tutti coloro che sono stati simpatici con me in questi anni.

Ringrazio ovviamente anche i miei amici di sempre, e la mia famiglia, che supporta tutto quello che faccio nonostante non le sia forse sempre molto chiaro.

Ringrazio infine Elisa, per la quale non basterebbe certo una pagina, per tutto quello che è.

Contents

| Al | ostrac | ct | | V |
|----|---------------------|-------------------|--|--------------------------|
| Ri | ngra | ziamen | ıti | vii |
| N | otatio | on and | conventions | 1 |
| In | Shar Geo Inve | metric erse Me | metric inequalities in nonnegative Ricci curvature | 3 6 10 13 15 |
| 1 | Geo | metric | inequalities in nonnegative Ricci curvature | 19 |
| | 1.1 | Struct | ture of the chapter | 19 |
| | 1.2 | Harm | onic functions in exterior domains | 19 |
| | | 1.2.1 | Green's functions and parabolicity | 20 |
| | | 1.2.2 | Harmonic functions on manifolds with nonnegative Ricci curvature. | |
| | | 1.2.3 | Ends of manifolds with nonnegative Ricci curvature | 23 |
| | | 1.2.4 | The exterior problem on nonparabolic manifolds | 24 |
| | | 1.2.5 | The exterior problem on parabolic manifolds | 27 |
| | 1.3 | | Monotonicity-Rigidity Theorems for nonparabolic manifolds | 29 |
| | | 1.3.1 | Preparation of the proof of the Monotonicity-Rigidity-Theorems . | 31 |
| | | 1.3.2 | Proof of Theorems 1.25 and 1.26 | 37 |
| | 1.4 | _ | time behaviour of the electrostatic potential | 44 |
| | | 1.4.1 | Manifolds with Euclidean volume growth | 45 |
| | | 1.4.2 | Asymptotics for <i>u</i> and its gradient | 47 |
| | | 1.4.3 | Limits of the monotone quantities | 51 |
| | 1 - | 1.4.4 | The Willmore-type inequalities for nonnegative Ricci | 56 |
| | 1.5 | manif | consequences of the Monotonicity-Rigidity Theorem for nonparabolic | (0 |
| | | manıı 1.5.1 | | 60 60 |
| | | 1.5.1 | Capacity estimates | 61 |
| | 1.6 | | olic manifolds and Enhanced Kasue's Theorem | 63 |
| | 1.0 | 1.6.1 | | 65 |
| | | 1.6.2 | A Maximum Principle/Splitting Theorem for manifolds with two | 03 |
| | | 1.0.2 | boundary components | 67 |
| | 1.7 | The is | soperimetric inequality for 3-manifolds | 68 |
| | 1.7 | 1.7.1 | Huisken's argument. | 68 |
| | | 1.7.1 | Tools from the Mean Curvature Flow of mean-convex domains | 69 |
| | | 1.7.3 | Some properties of the strictly outward minimising hull | 70 |
| | | 1.7.3 | Perspectives and applications of the Isoperimetric Inequalities | 75 |

| 2 | Min | kowsk | i inequalities via Nonlinear Potential Theory | 79 | | |
|----|------------------|----------|--|------------|--|--|
| | 2.1 | Structi | are of the chapter | <i>7</i> 9 | | |
| | 2.2 | Effecti | ve Monotonicity. Statements and preparatory material | <i>7</i> 9 | | |
| | | 2.2.1 | Preliminaries on <i>p</i> -capacitary potentials | 79 | | |
| | | 2.2.2 | A Kato-type identity for <i>p</i> -harmonic functions | 81 | | |
| | | 2.2.3 | The effectively monotone quantities | 85 | | |
| | | 2.2.4 | The conformal setting | 86 | | |
| | 2.3 | Proof | of the Effective Monotonicity Theorems | 93 | | |
| | | 2.3.1 | The Effective Monotonicity of Φ^p_{∞} | 93 | | |
| | | 2.3.2 | The Effective Monotonicity of Φ_{β}^{p} | 96 | | |
| | | 2.3.3 | Full monotonicity in absence of critical points | 105 | | |
| | 2.4 | 1 11 | | | | |
| | | 2.4.1 | From the L^p -Minkowski inequality to the Extended Minkowski In- | | | |
| | | | equality | 107 | | |
| | | 2.4.2 | Applications of the Extended Minkowski Inequality | 111 | | |
| | 2.5 | | consequences of the Effective Monotonicity Theorems | 113 | | |
| | | 2.5.1 | <i>p</i> -capacitary inequalities and the relations with Willmore-type func- | | | |
| | | | tionals | 114 | | |
| | | 2.5.2 | Spherical symmetry under pinching conditions | 115 | | |
| | | 2.5.3 | Relations with a classical overdetermined problem | 116 | | |
| 3 | On | strictly | outward minimising hulls and p-capacity | 119 | | |
| | 3.1 | • | result and structure of the chapter | 119 | | |
| | 3.2 | | rictly outward minising hull of a set with finite perimeter | 120 | | |
| | | 3.2.1 | Preliminaries | 120 | | |
| | | 3.2.2 | Basic properties of outward minimising sets | 121 | | |
| | | 3.2.3 | Maximal volume solutions to the least area problem with obstacle | | | |
| | | | in Riemannian manifolds | 124 | | |
| | | 3.2.4 | The strictly outward minimising hull of a set with finite perimeter | 126 | | |
| | 3.3 | Famili | es of manifolds admitting the strictly outward minimising hull | 129 | | |
| | | 3.3.1 | Euclidean-like isoperimetric inequality and strictly outward min- | | | |
| | | | imising hull | 130 | | |
| | | 3.3.2 | Weak Inverse Mean Curvature Flow and strictly outward minimis- | | | |
| | | | ing hulls | 131 | | |
| | 3.4 | Conve | rgence of <i>p</i> -capacities to $ \partial \Omega^* $ | 135 | | |
| | | 3.4.1 | <i>p</i> -nonparabolicity and <i>p</i> -capacitary potentials | 136 | | |
| A | Con | npariso | n with Colding's monotonicity formulas | 143 | | |
| В | Exis | tence o | f p-capacitary potentials | 147 | | |
| Bi | Bibliography 15: | | | | | |

Notation and conventions

We briefly discuss some of the conventions adopted. Most of them will be readily recalled along the thesis.

Riemannian manifolds and submanifolds

Riemannian manifolds will be usually denoted by (M,g). We address the reader to [Pet06] for the Riemannian geometric notions we employ. We denote the Levi-Civita connection by D, the Riemann curvature by Riem and the Ricci curvature by Ric. Let $N \hookrightarrow M$ be a codimension-1 submanifold. In the present work, submanifolds will mostly appear as boundaries of open subsets of Ω or smooth level sets of functions. We denote by h its second fundamental form, and by H its mean curvature, that we define just as the trace of h. No normalising constants are considered in the definition. Sometimes, in order to highlight the submanifold involved, we will add its name as a subscript. For example, H_N denotes the mean curvature of N.

We largely use the following computational convention. Let N be a codimension-1 submanifold of M, and let f be a smooth function on an open neighbourhood of $x \in N$. Submanifolds are always endowed with the Riemannian metric inherited by the ambient metricg, and we denote it by g_N or by g^\top , where in the last notation the submanifold is understood. Then, we denote the normal derivative of f with respect to N on x as

$$D^{\perp}f(x) = \langle Df \mid \nu \rangle (x)\nu(x),$$

where v(x) is a unit normal vector to N in x, and where by $\langle X | Y \rangle$ we are denoting g(X,Y). The orientation chosen for v is always the one pointing towards the (usually unique) unbounded component of $M \setminus N$. The dependence on x is usually omitted. Accordingly, we define the tangential derivative of f with respect to N as

$$\mathbf{D}^{\top} f = \mathbf{D} f - \mathbf{D}^{\perp} f,$$

where we omitted the point where the computation is performed. It is straightforward to observe that

$$|Df|^2 = |D^{\top}f|^2 + |D^{\perp}f|^2.$$

With |T|, we denote the norm of the tensor T with respect to the underlying metric g. When different metrics are considered, a suitable subscript appears. The submanifold N the above formalism refers to is often understood, the typical situation being the following. Let f be a function of the norm of some derivative tensor of a function u, such as the gradient Du or the Hessian tensor DDu. In this case, when no other information is added, the tangential and normal computations at a point x take place with respect to the level set of u the point x belongs to.

Volumes, areas and perimeter

We are denoting by $d\mu$ the volume element of the ambient metric g, and, given a submanifold N of M, by $d\sigma$ the volume element of the metric g_N induced by M on N. We

are referring to it as area element induced on N. Given a measurable set $\Omega \subseteq M$, we denote by $|\Omega|$ its volume measure, that we are going to confuse with the Lebesgue measure induced by g. Similarly, for a smooth submanifold N we let

$$|N| = \int_{N} d\sigma.$$

Occasionally, in Chapters 1 and 2, when N is not smooth, but still has finite n-1-dimensional Hausdorff measure \mathcal{H}^{n-1} , we let $|N| = \mathcal{H}^{n-1}(N)$.

Sets with nonsmooth boundary are important mostly in Chapter 3, where we are using the theory of sets with finite perimeter. We decided to collect definitions and notions employed in Subsection 3.2.1. Here, we just point that our main source about tools of Geometric Measure Theory is [Mag12].

Dimension

Except when differently indicated, the dimensions of the ambient metric this thesis consider are the following. In Chapter 1, the dimension of the underlying metric is $n \geq 3$. Chapter 2, that takes place in the ambient \mathbb{R}^n , is instead valid in dimension $n \geq 2$. However, we remark that the conclusions for n = 2 are somehow different, and we are never going to consider them. Chapter 3 involves Riemannian manifolds of dimension $n \geq 2$. Appendix A, being related to Chapter 1 holds for $n \geq 3$ while Appendix B, that we apply in Chapter 2 and 3 makes sense for $n \geq 2$.

Geometric inequalities are from long ago thoroughly studied in both Analysis and Geometry, with also ground-breaking incursions in General Relativity. To name the maybe most famous one, let us mention the Isoperimetric Inequality in \mathbb{R}^n , the ultimate version of which, holding true for sets with finite perimeter, was obtained, according to [Mir97], by adding Federer's insights of [Fed58] to De Giorgi's [DG58]. In curved Riemannian settings, we cite the celebrated version on compact Riemannian manifolds with positive Ricci curvature known as Levý-Gromov Isoperimetric Inequality, obtained in [Gro80] building on [Lev22], and the one on noncompact Cartan-Hadamard manifolds of dimension 3 and 4 due respectively to Kleiner [Kle92] and Croke [Cro84]. Let us briefly point out that the general case, known in literature as Cartan-Hadamard conjecture is still object of ongoing research, and that a maybe decisive attempt at its solution has been proposed in [GS19]. Among the countless generalisations and applications of the Isoperimetric Inequality developed through more of a century of Mathematics, let us highlight now the family of inequalities known in literature as Alexandrov-Fenchel Inequalities, involving quantities called quermassintegrals, that are suitable integrals of elementary functions of the principal curvature of the boundary $\partial\Omega$ of an open bounded sets with smooth boundary Ω . These inequalities originated in [Ale37; Ale38; Fen29], where they were shown to hold for convex sets Ω . The simplest inequality in this family is often referred to as Minkowski Inequality, and was first obtained for convex sets in [Min03], and can be written as follows

$$\left(\frac{|\partial\Omega|}{|\mathbb{S}^{n-1}|}\right)^{\frac{n-2}{n-1}} \le \frac{1}{|\mathbb{S}^{n-1}|} \int_{\partial\Omega} \frac{H}{n-1} \, d\sigma, \tag{1}$$

where by \mathbb{S}^{n-1} we denote the unit sphere of \mathbb{R}^n . In recent years, effort has been put in order to go beyond the convex settings, as well as to establish this inequality on relevant Riemannian manifolds. We are mentioning some of these extensions available in literature later on in this Introduction. For now, let us just point out that some motivations behind this inequality are the challenge this problem offers and its importance in General Relativity. For what it concerns the latter point, we refer to the thorough survey [Mar09], where, precisely in Subsection 7.1, the author clearly explains how the Penrose inequality for shells of matter collapsing at the speed of light in flat Minkowski space, conjectured in [Pen73], is related to (1), as first observed by Gibbons in his Ph. D. thesis [Gib73].

We focus now on a mathematical technique that has been recently employed in the context of the Minkowski inequality. In order to best describe the approach adopted in this thesis, we explain how the Minkowski Inequality can be deduced through the Inverse Mean Curvature Flow (IMCF), a geometric evolution equation whose weak formulation led, just to mention the most striking application, to the proof by Huisken and Ilmanen, [HI01], of the Riemannian Penrose Inequality, constituting a fundamental step in the realisation of the conjectures of the aforementioned [Pen73]. Its classical, smooth formulation consists in evolving the immersion $F_0: \partial\Omega \hookrightarrow \mathbb{R}^n$ by

$$\frac{\partial}{\partial t}F(t,x) = \frac{1}{H}(t,x)\nu(t,x), \qquad F(0,x) = F_0(x),$$

yielding, at least for a small T>0, a sequence of sets $\{\Omega_t\}_{t\in[0,T)}$ with boundaries $\partial\Omega_t$ given by the immersions $F(t,\cdot):\partial\Omega\hookrightarrow\mathbb{R}^n$, and where H(t,x) and v(t,x) are respectively the mean curvature and the outward pointing unit normal of $\partial\Omega_t$. Clearly, what just described makes sense for a strictly mean-convex initial hypersurface $\partial\Omega$, that is, with strictly positive mean curvature, an assumption obviously satisfied by convex initial sets. Along this evolution, one can then define the quantity

$$\mathscr{Q}(t) = |\partial \Omega_t|^{-\frac{n-2}{n-1}} \int_{\partial \Omega_t} \mathsf{H}(t, \cdot) \, \mathsf{d}\sigma_t, \tag{2}$$

where $d\sigma_t$ is the area element induced on $\partial\Omega_t$ by the ambient flat metric. A direct computation then shows that $\mathcal{Q}(t)$ is a monotone nonincreasing function. In particular, if $T=+\infty$ and, as geometrically desirable, $\partial\Omega_t$ converges smoothly enough to expanding spheres so that the scaling invariant $\mathcal{Q}(t)$ converges to its value on spheres, one gets (1). These strong requirements are actually shown to be satisfied in the class of bounded starshaped sets with smooth strictly mean-convex boundary in [Ger90; Urb90], yielding, through the argument just outlined the extension of the classical Minkowski inequality to this broader class of sets. This scheme, together with the derivation of other Alexandrov-Fenchel inequalities, is carried out in details in [GL09]. We point out, on the other hand, that this approach is suitable only for those hypersurfaces that do not change topology along their evolution. For example, in \mathbb{R}^n , there is no chance to employ it if the initial boundary does not have spherical topology. These topological restrictions can be overtaken considering the delicate notion of weak solutions introduced by Huisken-Ilmanen in the aforementioned [HI01], that allows, as lectured in [Hui], to prove (1) in the broader class of outward minimising sets with smooth boundary. Leaving the details of this approach to the sequel, where we are drawing a systematic comparison between that and our methods, we just say here that a bounded set E with finite perimeter is outward minimising if $P(E) \leq P(F)$ for any $E \subseteq F$, and that, if the boundary of an outward minimising set is smooth, then it is mean-convex, as immediately seen through the standard variational argument. The inclusion of the class of strictly starshaped sets with smooth strictly mean-convex boundary into this one substantially follows from the combination of [Ger90; Urb90] and [HI01], and we refer the reader to Proposition 3.25 for a complete proof of (a generalised version of) this result.

Finally reaching the starting point of this work, we do now recall another celebrated geometric inequality, that along this work will be referred to as Willmore-type inequality, asserting that a bounded set $\Omega \subset \mathbb{R}^n$ with smooth boundary satisfies

$$\left|\mathbb{S}^{n-1}\right| \le \int\limits_{\partial\Omega} \left| \frac{\mathsf{H}}{n-1} \right|^{n-1} \!\!\!\! \mathrm{d}\sigma. \tag{3}$$

This inequality was first obtained for surfaces inside \mathbb{R}^3 in [Wil68], where it is usually called Willmore inequality, and later extended to all dimensions in [Che71]. We remark that the Willmore energy, that is the right hand side of (3) for n=3, arises also in the context of cell membranes, as a first approximation of the Helfrich energy introduced in [Hel73]. Despite, on those sets satisfying the Minkowski Inequality, the latter implies (3) as a straightforward application of the Hölder inequality immediately shows, the Willmore-type inequality holds true with no assumptions on Ω other than smoothness of its boundary. Moreover, equality is achieved in (3) only on balls. Observe that an Inverse Mean Curvature Flow-proof is viable also for (3), considering, along the evolution of a

set with strictly mean-convex boundary, the function

$$\mathscr{W}(t) = \int_{\partial \Omega_t} H^{n-1}(t, x) \, d\sigma_t \tag{4}$$

and finally obtaining (3) as a consequence of its monotonicity. As before, this scheme allows to prove the Willmore-type inequality for starshaped sets with smooth strictly mean-convex boundary, by appealing to [Ger90; Urb90], and most likely also for outward minimising sets by combining the deep weak notion of IMCF of [HI01] with the observations performed in [Hui].

A different approach to (3) was introduced some years ago by Agostiniani and Mazzieri in [AM20]. They considered the boundary value problem

$$\begin{cases} \Delta u = 0 & \text{in } \mathbb{R}^n \setminus \overline{\Omega} \\ u = 1 & \text{on } \partial\Omega \\ u(y) \to 0 & \text{as } |y| \to +\infty, \end{cases}$$
 (5)

and showed that the function

$$U(t) = t^{-(n-1)} \int_{\{u=t\}} |Du|^{n-1} d\sigma, \qquad t \in (0,1]$$
 (6)

is monotone nondecreasing, and in particular by coupling

$$U'(1) = (n-2) \int_{\partial\Omega} |\mathrm{D}u|^{n-2} \left[H - \left(\frac{n-1}{n-2} \right) |\mathrm{D}u| \right] \, \mathrm{d}\sigma \ge 0$$

with

$$|\mathbb{S}^{n-1}| = \lim_{t \to 0^+} U(t) \le U(1),$$
 (7)

they showed, with the aid of the Hölder inequality, (3). Moreover, since the monotonicity is strict unless Ω is a ball, this also shows that equality in (3) is achieved just on balls, recovering the full statement of [Che71]. This approach immediately shows an analogy with the IMCF's one, since it is based on a monotonic quantity along a suitable evolution of the set considered, but it also displays the power arising from the simplicity of (5). Indeed, being the solution of this exterior boundary value problem well known to exist, and to be smooth, thus classical, both long-time existence results as that of [Ger90; Urb90] for the IMCF, or the necessity of a delicate notion of weak solution as that of Huisken-Ilmanen's weak IMCF are immediately by-passed. Moreover, no starshapedness or outward minimising assumptions are necessary, nor mean-convexity of the boundary. Let us also observe that also (7) follows from very classical asymptotic expansions at spatial infinity of the solution to (5).

The starting point of the present thesis is to generalise and better understand this new approach to geometric inequalities. This is done essentially in two directions. The first one concerns the application of this method to complete noncompact manifolds with nonnegative Ricci curvature. The second one, concerning instead flat \mathbb{R}^n , regards the nonlinear generalisation of it, that is, considering the evolution given by the level sets of the solution of the nonlinear version of (5), where the laplacian is replaced by the p-laplacian. The following two Sections are devoted to illustrate the main results arising from these lines of research. They will be developed in Chapter 1 and Chapter 2.

Sharp geometric inequalities in nonnegative Ricci curvature

Looking at [AM20], one can realise that the main inequalities, actually based on the Bochner identity, *formally* hold true even if the ambient is a nonnegatively Ricci curved complete noncompact Riemannian manifold (M,g). We do actually show that this is truly the case, and prove that the function U defined as in (6) is monotone nondecreasing, everytime a solution to

$$\begin{cases} \Delta u = 0 & \text{in } M \setminus \overline{\Omega} \\ u = 1 & \text{on } \partial\Omega \\ u(y) \to 0 & \text{as } d(O, y) \to +\infty, \end{cases}$$
 (8)

with O being a fixed point in Ω and d the distance on M, exists. Clearly, the definition of U is given in terms of such solution. It is known, and re-proved in Theorem 1.14, that a solution to (8) exists if and only if (M,g) is *nonparabolic*, that is, if it admits a positive Green's function. As in [AM20] we show that the monotonicity property of U is in fact shared by a family of functions $U_{\beta}:(0,1]\to\mathbb{R}$ defined by

$$U_{\beta}(t) = t^{-\beta \left(\frac{n-1}{n-2}\right)} \int |Du|^{\beta+1} d\sigma,$$

$$\{u=t\}$$

if $\beta \geq (n-2)/(n-1)$. If, for β in this range, $U'_{\beta}(t_0) = 0$ for some $t_0 \in (0,1]$, we do show that $(\{u \leq t_0\}, g)$ must be isometric to a truncated cone. The statement of this Monotonicity-Rigidity Theorem for U_{β} in nonparabolic manifolds with nonnegative Ricci curvature is to be found in Theorem 1.19.

Recalling that the monotonicity of U_{β} , and precisely that of $U_{n-2}=U$, constituted the main ingredient in the recent potential-theoretic proof of the Willmore-type inequality outlined above, we are committed to apply that in order to derive a Willmore-type inequality for manifolds with nonnegative Ricci curvature. However, a serious issue arises in the characterisation of the limit of U_{β} as $t \to 0^+$, since, as we detail in Remark 1.39, there is no hope to derive pointwise Euclidean-like asymptotic expansions of the gradient in our general setting, ultimately due to the possibility of infinite topology. On the other hand, we overcome this problem by working out suitable delicate integral asymptotic expansions, in the spirit of [CM97b]. This is an important technical difference from the Euclidean case. This will be carried out in Section 1.4, that culminates in the computation

$$\lim_{t \to 0^+} U_{\beta}(t) = \operatorname{Cap}(\Omega)^{1-\beta/(n-2)} \operatorname{AVR}(g)^{\beta/(n-2)} (n-2)^{\beta+1} |S^{n-1}|, \tag{9}$$

see Proposition 1.40 and (1.4.28). With AVR(g), appeared above, we denote the Asymptotic Volume Ratio of g, defined as

$$AVR(g) = \lim_{r \to +\infty} \Theta(r)$$

and where

$$(0,+\infty)\ni r\longmapsto \Theta(r)=\frac{|B(x,r)|}{|\mathbb{B}^n|r^n},$$

with \mathbb{B}^n denoting the unit ball of \mathbb{R}^n . With B(x,r) we indicate the geodesic ball of radius r centered at $x \in M$. The classical Bishop-Gromov Theorem [Bis63; Gro81] ensures that Θ is monotone nonincreasing and thus admitting a limit as $r \to +\infty$, that actually does not depend on the point x. Such limit AVR(g) is then included in [0,1], and is 1 only on flat

 \mathbb{R}^n . We say that (M,g) has Euclidean volume growth if $\mathrm{AVR}(g) > 0$. The limit in (9) actually tells that a nontrivial Willmore-type inequality is likely to be deduced only if (M,g) has Euclidean volume growth. We notice that by a known characterisation, see [Var82], a complete noncompact manifold with nonnegative Ricci curvature is in particular nonparabolic. In fact, such characterisation states that a complete noncompact Riemannian manifold with $\mathrm{Ric} \geq 0$ is nonparabolic if and only if, roughly speaking, the volume of geodesic balls grows faster than quadratically, see Theorem 1.4.

The first main result of this thesis asserts that on a complete noncompact Riemannian manifold (M,g) with Ric ≥ 0 and Euclidean volume growth it holds

$$AVR(g)|S^{n-1}| \le \int_{\partial\Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma, \tag{10}$$

with equality achieved only if $(M \setminus \Omega, g)$ is isometric to a truncated cone. This is the content of Theorem 1.44. This result encompasses a variety of manifolds of interest. For example, for $n \geq 4$, there exists an important class of complete noncompact Ricci flat Riemannian manifolds with 0 < AVR(g) < 1, that is the class of Ricci flat *Asymptotically* Locally Euclidean (ALE for short) manifolds. We refer the reader to Definition 1.48 for the precise notion. For the time being, we just recall that an *n*-dimensional Riemannian manifold is ALE if it is asymptotic to $((\mathbb{R}^n \setminus \{0\})/\Gamma, g_{\mathbb{R}^n})$, where Γ is a finite subgroup of SO(n) acting freely on $\mathbb{R}^n \setminus \{0\}$. This family of Riemannian manifolds is widely studied. In this regard, we first mention that in [BKN89] it is proved that any Ricci flat manifold with Euclidean volume growth and strictly faster than quadratic curvature decay is actually ALE. Moreover, we point out that 4-dimensional Ricci flat ALE manifolds appear as important examples of gravitational instantons, that are noncompact hyperkhäler 4manifolds with decaying curvature at infinity, introduced by Hawking in [Haw77] in the framework of his Euclidean quantum gravity theory. An explicit example is given by the famous Eguchi-Hanson metric, introduced in [EH79], where n = 4, Ric = 0 and $\Gamma = \mathbb{Z}_2$. We remark that ALE gravitational instantons are completely classified in [Kro89b] and [Kro89a]. Concerning the general class of gravitational instantons, let us cite, after the important works of Minerbe [Min09a; Min10; Min11], the recent Ph.D. thesis [Che17], where gravitational instantons with strictly faster than quadratic curvature decay are classified. We refer the reader to the latter work and to the references therein for a more complete picture on this subject. Our Willmore-type inequality actually applies to ALE manifolds with nonnegative Ricci curvature, where it improves to

$$\inf \left\{ \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma \right| \Omega \subset M \text{ bounded with smooth boundary} \right\} = \frac{|S^{n-1}|}{\operatorname{card} \Gamma}, \quad (11)$$

with equality achieved on truncated cones with link homothetic to $(S^{n-1}/\Gamma, g_{S^{n-1}/\Gamma})$. Notice in particular that if Γ is trivial one recovers the classical Willmore-type inequality described above with its rigidity statement. The lower bound of the Willmore-type functional shown in (10) being actually an infimum is a fact that holds true for a larger class of manifolds. Indeed, as proved in Theorem 1.47, it is sufficient to assume the hypotheses of Theorem 1.44 together with a quadratic curvature decay condition. This will immediately yield (11) as shown in Corollary 1.49. Understanding metric and topological consequences of curvature decay conditions is a very interesting and widely studied problem in geometric analysis. Dropping any attempt of being complete, we refer the interested reader to the aforementioned [BKN89], to the seminal [CGT82], to [Rei15], where the case n=3 is considered, and to [Yeg09] and the references therein.

To make the picture more complete, let us also mention that Willmore-type inequalities are proven in [AM17] for asymptotically flat (AE) static metrics in the framework of General Relativity, and in [Sch18] for integral 2-varifolds in Cartan-Hadamard manifolds.

As for the Euclidean case, Theorem 1.19 will actually be proved working in the manifold $(M \setminus \Omega, \tilde{g})$ where \tilde{g} is conformally related to g by

$$\tilde{g} = u^{\frac{1}{n-2}}g,$$

where u is a solution to (8). In this setting, integral identities and splitting techniques are employed to infer the monotonicity of (the conformal analogue of) U_{β} and the related rigidity. We point out that these techniques can easily produce a more general version of Theorem 1.19 for nonparabolic ends with Ric ≥ 0 of a noncompact Riemannian manifold. This *conformal splitting method* was introduced in [AM15] and has proved to be fruitful in various other situations, such as [AM17; BM18] and [BM17], where it has been applied to the relativistic setting. A different approach to exterior problems, but still relying on a conformal change of metric, has been introduced in [BMM19]. As a consequence of that approach, the authors were able to prove that in \mathbb{R}^n , suitable, new pinching conditions on the mean curvature of $\partial\Omega$ or on the normal derivative of u at $\partial\Omega$ force Ω to be a ball. Here, we show that those results can be achieved also in nonparabolic manifolds with nonnegative Ricci curvature. To do so, we propose an alternative, easier tool, based on a maximum principle for $|\mathrm{D}u|/u^{(n-1)/(n-2)}$ and the Hopf boundary lemma, that we interpret as a Monotonicity-Rigidity Theorem for the function

$$U_{\infty}(t) = \sup_{\{u=t\}} \frac{|\mathrm{D}u|}{u^{\frac{n-1}{n-2}}}$$

Two main consequences of the monotonicity-rigidity properties of U_{∞} are described in Section 1.5, together with some other consequences of the monotonicity of U_{β} , and, as just hinted, constitute the analogues of the main results of [BMM19] in our general Riemannian setting. As for the Willmore-type inequality, the main difficulty in reaching them lies in the asymptotic behaviour of U_{∞} as $t \to 0^+$, described in Proposition 1.43.

So far, we have considered *nonparabolic* Riemannian manifolds with Ric ≥ 0 . We now turn our attention to *parabolic* Riemannian manifolds with nonnegative Ricci curvature. As we will see in Section 1.2, for this class of manifolds, problem (8) does not admit a solution, while the following problem does

$$\begin{cases} \Delta \psi = 0 & \text{in } M \setminus \overline{\Omega} \\ \psi = 0 & \text{on } \partial \Omega \\ \psi(y) \to +\infty & \text{as } d(O, y) \to +\infty, \end{cases}$$
 (12)

where $\Omega \subset M$ is any bounded and open subset with smooth boundary. Inspired by the fact that problem (12) presents strong formal analogies with the conformal reformulation of problem (8) in terms of \tilde{g} (see problem (1.3.13) below), we also provide a Monotonicity-Rigidity Theorem for parabolic manifolds with Ric ≥ 0 involving ψ in place of u. For $\beta \geq 0$, we define the function $\Psi_{\beta}: [0, \infty) \to \mathbb{R}$ as

$$\Psi_{eta}(s) = \int\limits_{\{\psi=s\}} |\mathrm{D}\psi|^{eta+1} \,\mathrm{d}\sigma$$
 ,

and we prove that it is monotone nonincreasing if $\beta \geq (n-2)/(n-1)$, with derivative vanishing at $s_0 \in [0,\infty)$ only if $(\psi \geq s_0,g)$ is isometric to a truncated cylinder, see Theorem 1.54. In complete analogy with the nonparabolic case, we may also define the function

$$\Psi_{\infty}(s) = \sup_{\{\psi = s\}} |\mathcal{D}\psi|,\tag{13}$$

and prove that it is nonincreasing, as stated in Theorem 1.55. The coupling of the Monotonicity - Rigidity Theorems for nonparabolic and parabolic manifolds of nonnegative Ricci curvature yields as a straightforward consequence an enhanced version of a theorem of Kasue, [Kas83, Theorem C (2)], see also [CK92], asserting that if a smooth boundary $\partial\Omega\subset M$ has mean curvature $H\leq 0$ on the whole $\partial\Omega$, then H=0 on $\partial\Omega$ and $M\setminus\Omega$ is isometric to a half cylinder. Our result actually gives precise lower bounds for the supremum of H in terms of our monotone quantities and their derivatives, see Theorem 1.56. We point out that Kasue/Croke-Kleiner result holds true also in a compact version with a two-connected-components boundary. We recover, through our techniques, also (a slightly weaker form of) such statement in Theorem 1.59. We inform the reader that similar splitting theorems have been recently provided in dimension 3 under a mild scalar curvature lower bound in [GJ19].

Finally, we combine our sharp Willmore-type inequality (10) with curvature flow techniques along the lines of an argument presented by Huisken in [Hui]. We obtain a characterisation of the infimum of the Willmore functional in terms of the isoperimetric ratio of complete noncompact 3-manifolds with nonnegative Ricci curvature, refining the analogous result stated in the aforementioned contribution. We get in fact

$$\inf \frac{|\partial \Omega|^3}{36\pi |\Omega|^2} = \inf \frac{\int_{\partial \Omega} H^2 d\sigma}{16\pi} = AVR(g), \tag{14}$$

on any complete noncompact Riemannian manifold (M, g) with Ric ≥ 0 and Euclidean volume growth, yielding in turn

$$\frac{|\partial\Omega|^3}{|\Omega|^2} \ge 36\pi \,\text{AVR}(g). \tag{15}$$

The key argument lectured by [Hui] involves the computation of the evolution along the Mean Curvature Flow of a quantity we call Isoperimetric Difference, namely

$$D(t) = |\partial \Omega_t|^{3/2} - C|\Omega_t|,$$

with $\Omega_0 = \Omega$ and some positive constant C. Such quantity is indeed shown to be monotone if along the evolution some lower bound on the Willmore energy is preserved, and if the flow vanishes as $t \to T^-$ for some T > 0. In this case, consequently, an Isoperimetric Inequality is deduced with constant related to such lower bound. So, we plug our new Willmore inequality (10) for n=3 into this argument, combine it with known important results in the weak Mean Curvature Flow of mean-convex sets and with the Kasue/Croke-Kleiner Theorem we have re-discovered to get (14). We are also able to show that equality holds in (15) if and only if Ω is isometric to a ball of flat \mathbb{R}^n . Beside the characterisation of the isoperimetric constant in terms of the Asymptotic Volume Ratio, the novelties with respect to [Hui] lie in the rigidity statement and in the fact that the infimum of the Willmore functional is taken over the whole class of bounded open subsets Ω with smooth boundary, and not just over *outward minimising* subsets. All of these

improvements substantially come from our optimal Willmore-type inequality (10). We wish to remark that the structure of complete noncompact 3-manifolds with nonnegative Ricci curvature is well understood, see [Liu13], although this knowledge never enters in the argument employed. Actually, as pointed out in [Heb99, Theorem 8.4], a positive nonsharp isoperimetric constant for complete noncompact Riemannian manifolds with Ric ≥ 0 and Euclidean volume growth is known to exist. This was realised for the first time most likely in [Var95], and can be achieved rigorously through the methods employed in [Car94].

The Isoperimetric Inequality is well known to be related to a huge variety of issues. We discuss some of them in Subsection 1.7.4. Briefly, we consider the Sobolev Inequality and the Faber-Krahn Inequality, both achieving optimal constants deduced from (23) on three-dimensional ambient manifolds, we state and comment the natural higher dimensional conjecture and compare the Isoperimetric Inequality with the Isoperimetric problem.

Let us finally observe that relations between isoperimetry and mean curvature functionals date back to Almgren [Alm86], while a first derivation of isoperimetric inequalities through a curvature flow has been obtained by Topping in the case of curves [Top98]. Isoperimetric inequalities in \mathbb{R}^n and in Cartan-Hadamard manifolds through curvature flows have been established by Schulze in [Sch08] and [Sch18], while the application to manifolds with nonnegative Ricci curvature is suggested in the already mentioned [Hui]. The techniques lectured in [Hui] have interesting applications also in connection with the relativistic ADM mass, see [JL17] for the details.

Geometric aspects of p-capacitary potentials

The natural nonlinear generalisation of problem (8) is

$$\begin{cases} \Delta_{p}u = 0 & \text{in } \mathbb{R}^{n} \setminus \overline{\Omega} \\ u = 1 & \text{on } \partial\Omega \\ u(y) \to 0 & \text{as } |y| \to +\infty, \end{cases}$$
 (16)

for $\Omega \subset \mathbb{R}^n$ a bounded set with smooth boundary and where the *p*-laplacian operator Δ_p acts on smooth functions f as $\Delta_p f = \operatorname{div}(|\mathrm{D} f|^{p-2}\mathrm{D} f)$. In general, a Sobolev-type notion of p-harmonic functions is available, and it allows to show existence of a function $u \in W^{1,p}$ weakly solving (16), and where the boundary value is reached smoothly. In Chapter 2, we are committed to study monotonicity formulas along the level set flow of the solution u of the problem above, that we refer to as the p-capacitary potential of Ω . However, after guessing the correct quantities that should be formally smooth, a major problem appears at once. Indeed, the lack of smoothness for solutions to (16), the optimal regularity being mild $\mathscr{C}^{1,\alpha}$, invalidates most of the arguments employed in the linear case to infer monotonicity, the most dramatic issue being the lack of a satisfactory Sard-type property. On the other hand, it is evident that all of the consequences we were able to draw from the monotonicity formulas in the linear setting just followed from $U'_{\beta}(1) \geq 0$ and $\lim_{t\to 0^+} U_{\beta}(t) \leq U_{\beta}(1)$. Some reason for optimism can then arise, since Tolksdorf's Hopf lemma [Tol83] for p-harmonic functions and Kichenassamy-Veron's [KV86] asymptotic expansion ensures that the solution to (16) is actually smooth in a neighbourhood of Ω as well as sufficiently far away. What we succeed to prove in Theorem 2.7 is in fact an

effective monotonicity property of the function

$$U_{\beta}^{p}(t) = t^{-\beta(p-1)\frac{(n-1)}{(n-p)}} \int_{\{u=t\}} |\mathrm{D}u|^{(\beta+1)(p-1)} d\sigma,$$

defined only on regular values of u, for $\beta \geq (n-p)/[(p-1)(n-1)]$. By effective monotonicity, we substantially mean that $(U^p_\beta)'(1) \geq 0$ and $\lim_{t\to 0^+} U^p_\beta(t) \leq U^p_\beta(1)$. Besides their geometric implications, these formulas have a technical relevance on their own, as they persist through all the possible singularities of the flow. It is worth noticing that, in the present framework, the flow singularities correspond to the critical points of u, and these might in principle be arranged in sets with full measure.

The main geometric implication of such effective monotonicity is an extension of the Minkowski Inequality (1), holding for every bounded and smooth subset of \mathbb{R}^n , in which the total mean curvature of the boundary is replaced by the L^1 -norm of the mean curvature, whereas the perimeter of the set Ω is replaced by the one of its *strictly outward minimising hull* Ω^* . For the reader's convenience we briefly recall that a set is called *outward minimising* if it minimises the perimeter among all the sets containing it; moreover, an outward minimising set is called *strictly outward minimising* if it coincides almost everywhere with any outward minimising set containing it and having the same perimeter. Loosely speaking, Ω^* is, up to negligible components, the smallest strictly outward minimising set that contains Ω , and it happens to be the set containing Ω with smallest perimeter. We are actually dedicating Chapter 3 to describe various aspects of the strictly outward minimising hull in Riemannian geometry, and then we address the reader to its presentation below for further discussions. Such Extended Minkowski Inequality, content of Theorem 2.25, reads

$$\left(\frac{\left|\partial\Omega^*\right|}{\left|\mathbb{S}^{n-1}\right|}\right)^{\frac{n-2}{n-1}} \le \frac{1}{\left|\mathbb{S}^{n-1}\right|} \int\limits_{\partial\Omega} \left|\frac{\mathbf{H}}{n-1}\right| \, \mathrm{d}\sigma. \tag{17}$$

Let us point out that (17) leaves the long-standing question about the validity of the Minkowski Inequality (1) under the sole assumption of mean-convexity for $\partial\Omega$ still open. To this end, a straightforward consequence of the celebrated Michael-Simon Sobolev inequality [MS73] yields the existence of a constant C(n) depending only on the dimension such that

$$C(n) \leq |\partial\Omega|^{-\frac{n-2}{n-1}} \int_{\partial\Omega} |H| \, d\sigma \tag{18}$$

is true for any bounded set with smooth boundary Ω . The question, then, can be extended in terms of the optimal constant in (18), at least under a mean-convexity assumption. We observe that an attempt to answer in the positive, that is, to show that one can take as C(n) the one given by (17) replacing $\partial\Omega^*$ with $\partial\Omega$ when the boundary is just mean-convex was made by Trudinger in [Tru94]. However, such proof is acknowledged to be incomplete in [Gua+10]. On the other hand, in [Dal+16] it is shown that (18) holds true with such conjectured constant in the class of axisymmetric sets with $\mathcal{C}^{1,1}$ boundary regularity of \mathbb{R}^3 .

As a matter of fact, the above Extended Minkowski Inequality is deduced as the limit, for $p \to 1^+$, of the following geometric *p*-capacitary inequality, which we believe of

independent interest

$$C_p(\Omega)^{\frac{n-p-1}{n-p}} \le \frac{1}{|\mathbb{S}^{n-1}|} \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^p d\sigma, \tag{19}$$

stated in Theorem 2.24. Here, by $C_p(\Omega)$ we denote a normalised version of the classical variational p-capacity of Ω denoted by $\operatorname{Cap}_p(\Omega)$. Briefly we remark that (19) follows from the effective monotonicity of U^p_β with value $\beta = 1/(p-1)$ with a derivation very similar to that of the Willmore-type inequality from the monotonicity of U_β with p=2.

On the other hand, in order to deduce (17) from (19), one needs to compute the limit of the p-capacity of a bounded set with smooth boundary as $p \to 1^+$. We have established that

$$\lim_{p \to 1^+} C_p(\Omega) = \frac{|\partial \Omega^*|}{|S^{n-1}|}.$$
 (20)

This relation, that we observe at once to be equivalent to $\lim_{p\to 1^+} \operatorname{Cap}_p(\Omega) = |\partial\Omega^*|$, to the author's knowledge has never been explicitly considered in literature. Actually, it happens to be intimately related to the geometry and to the potential theoretic properties of the underlying ambient, and this is the reason we prove it in a general setting in Theorem 3.1 of Chapter 3.

As an immediate corollary of (17), we recover the Minkowski Inequality for *outward minimising* sets, since for every Ω in this class it holds $|\partial\Omega|=|\partial\Omega^*|$ (see Remark 3.16) and $H\geq 0$, as a standard variational computation readily shows. Such inequality was originally conceived by Huisken in [Hui], exploiting the theory of weak solutions to the IMCF, previously developed in [HI01] (see also [FS14, Theorem 2–(b)] for a published version of the argument in the case of outward minimising sets with strictly mean-convex boundary, or [Wei18] in the more general Schwarzschild setting). We remark here that the class of outward minimising sets contain that of strictly starsharped sets with smooth strictly mean-convex boundary. This substantially follows from [Ger90; Urb90] together with the area minimising properties of the Inverse Mean Curvature Flow observed and exploited with dramatic success in [HI01]. We give a self contained proof and slightly improve this fact in Subsection 3.3.2.

It is worth pointing out that an approximation argument through Mean Curvature Flow, pointed out in [HI01] and refined in [HI08] allows to deduce the Extended Minkowski Inequality from its version for outward minimising sets if $n \le 7$, as we show in Subsection 2.4.1. On the other hand, this argument breaks down in higher dimension due to minimal surfaces regularity issues, and then in higher dimensions inequality (17) seems to be actually stronger.

A simple and very nice application of inequality (17) is a nearly umbilical estimate for *outward minimising* surfaces in \mathbb{R}^3 with *optimal constant*. The relation between the Minkowski Inequality and the nearly umbilical estimates was suggested by Huisken in [Hui]. Here, for the sake of reference, we included a proof of this fact in Section 2.5 (see Theorem 2.28). The general nearly umbilical estimate for surfaces in \mathbb{R}^3 with an implicit dimensional constant is a very remarkable theorem, proved in [DLM05] by De Lellis and Müller. We refer the reader to the original paper [DLM05] as well as to the Ph.D. thesis [Per11] and the references therein for a complete account about the geometric features and implications of such a deep result.

Another application of the Extended Minkowski Inequality is the following inequality that we call Volumetric Minkowski Inequality,

$$\left(\frac{|\Omega|}{|\mathbb{B}^n|}\right)^{\frac{n-2}{n}} \le \frac{1}{|\mathbb{S}^{n-1}|} \int_{\partial \Omega} \left| \frac{H}{n-1} \right| d\sigma, \tag{21}$$

again holding true for any bounded open set $\Omega \subset \mathbb{R}^n$ with smooth boundary. Inequality (21) follows from (17) through a direct application of the Isoperimetric Inequality to Ω^* , that actually also yields the isometry of $\partial\Omega$ with a sphere if equality is achieved in (21). To the authors' knowledge, the Volumetric Minkowski Inequality was previously known to hold for domains with a strictly mean-convex boundary of positive scalar curvature (for short $\partial\Omega\in\Gamma_2^+$). On this regard, we refer the reader to the paper [CW13] and the subsequent [Qiu15], where the inequality was proved with methods based on Optimal Transport.

To complete the analogy with the linear case treated in Chapter 1, we also prove in Theorem 2.8 that the function

$$U_{\infty}^{p}(t) = \sup_{\{u=t\}} \frac{|Du|}{u^{\frac{n-1}{n-p}}},$$

is, in a suitable sense, effectively monotone. Together with some nonlinear generalisations of results appeared in [AM20] and [BMM19], we observe in Section 2.5 how the monotonicity of U_{∞}^p is related to modern techniques in overdetermined boundary value problems, with particular mention for [GS99] and [Pog18].

Before going on, let us recall, as anticipated before, that Minkowski-type inequalities as well as many other related inequalities, like for example Penrose-type inequalities, are provided in literature by using the Inverse Mean Curvature Flow or suitable generalisations of it. Without any attempt to be complete, in [GL09; LW17; BHW16; GWW13; GWW14; Ge+15; LG16; MS16] suitable nontrivial generalisations of the long time existence result of [Ger90; Urb90] under more or less restrictive requirements on the initial sets are derived and applied to get the desired geometric applications. On the other hand, in [BM08; FS14; Wei18; McC17; LN15] the Huisken-Ilmanen's weak notion of Inverse Mean Curvature Flow is employed for these purposes. We then find convenient to yield some other insights on the relations between the method employed here and that of the IMCF.

Inverse Mean Curvature Flow VS Nonlinear Potential Theory

The relation with the Inverse Mean Curvature Flow, in particular to its weak formulation, is quite transparent when dealing with the nonlinear approach to the Minkowski Inequality developed in Chapter 2. Actually, in light of the discussion that follows, the success of the linear potential theoretic argument of [AM20] and of the extensions worked out in Chapter 1 will appear somewhat surprising.

The key point in our approach is to replace the delicate elliptic regularisation procedure of Huisken-Ilmanen with a novel analysis of a very natural family of approximate solutions, namely the p-capacitary potentials of Ω , with $p \to 1^+$. In fact, a well known result due to Moser [Mos07], subsequently extended by Kotschwar-Ni [KN09] and recently

in [MRS19], says that if u_p is a weak solution to problem

$$\begin{cases} \Delta_p u_p = 0 & \text{in} \quad \mathbb{R}^n \setminus \overline{\Omega}, \\ u_p = 1 & \text{on} \quad \partial \Omega, \\ u_p(y) \to 0 & \text{as} \quad |y| \to \infty, \end{cases}$$

then, as $p \to 1^+$, the functions $w_p = -(p-1) \log u_p$ converge locally uniformly in $\mathbb{R}^n \setminus \Omega$ to a weak solution of the IMCF, the definition of which is recalled in Section 3.3.2. It must be noted that w_p satisfies the identity

$$\Delta_p w_p = |\mathrm{D} w_p|^p,$$

which is formally converging to

$$\operatorname{div}\left(\frac{\mathrm{D}w}{|\mathrm{D}w|}\right) = |\mathrm{D}w|\,,\tag{22}$$

and the latter equation is known to rule the level sets formulation of the IMCF. Albeit its simplicity, this very clean approximation scheme has found applications so far only to the existence theory for the weak IMCF. Very roughly, what we discover in Chapter 2 is that the p-harmonic approximators u_p possess self consistent monotonicity formulas, interpolating the linear ones of [AM15] and Chapter 1 with that of the function $\mathcal Q$ defined in (2) considered above. In light of this achievement the techniques worked out in [HI01] allowing, as discussed in [Hui; FS14; Wei18], to extend the validity of the Minkowski Inequality to outward minimising sets are fully replaced by new ones in the context of nonlinear potential theory. In order to understand the relation between the monotonicity of U_{β}^{p} and the one of $\mathcal Q$ defined by (2), we proceed formally. Setting as above $w_p = -(p-1)\log u_p$ and $t = -(p-1)\log \tau$, the monotonicity of the functions $U_{\beta}^{p}(\tau)$ with $\beta = 1/(p-1)$ is equivalent to say that for every 1 the function

$$[0,+\infty) \ni t \longmapsto e^{-\frac{n-p-1}{n-p}t} \int |Dw_p|^p d\sigma$$

is nonincreasing. Taking the formal limit as $p \to 1^+$, one would get the same monotonicity statement for the function

$$[0,+\infty)
ightarrow t \longmapsto e^{-rac{n-2}{n-1}t} \int |\mathrm{D}w| \,\mathrm{d}\sigma$$
 ,

where w solves (22), and thus |Dw|(x) coincides with the mean curvature of the level set passing through x. Recalling that $|\{w=t\}|=|\partial\Omega_t|=|\partial\Omega|\,\mathrm{e}^t$ along the IMCF, it is easy to realise that the latter monotonicity is equivalent to the one in (2). We stress that, at least with the technology available so far, such a computation is purely formal, since w_p is converging to w only locally uniformly and w itself is nothing more than a weak solution to the IMCF. This is the best result available so far also under assumptions on Ω ensuring both w_p and w to be smooth, such as convexity. In particular, we emphasise that the techniques employed in Chapters 1 and 2 are fully self contained and free from any argument involving the IMCF or its weak formulation. The approximation of the IMCF through the p-capacitary potentials is somehow replaced by the simple (20), the proof of which, in \mathbb{R}^n , still will not use any of the IMFC approximation results of [Mos07; KN09; MRS19].

Coming back for a while to the linear setting, the above formal computations show that the achievement of the Willmore-type inequality (3) through the monotonicity of U_{β} along the level set flow of the solution u to (5), carried out in [AM20], as well as, consequently, the extension performed in Chapter 1, is quite surprising. Indeed, proceeding formally as above, the monotonicity of U_{β} with $\beta = (n-2)$ is equivalent to that of

$$t \to \int\limits_{\{w=t\}} |\mathrm{D}w|^{n-1} \,\mathrm{d}\sigma,$$

where $w = -\log u$. On the other hand, the monotonicity of \mathscr{W} defined in (4) along the IMCF would be formally recovered by letting $p \to 1^+$ in the monotonicity of U^p_β along the solution u_p to (16) with $\beta = (n-p)/(p-1)$ that is equivalent to the monotonicity of

$$t \to \int_{\{w_p = t\}} |\mathsf{D}w_p|^{n-1} \, \mathrm{d}\sigma,$$

where $w_p = -(p-1) \log u_p$, so that $w = w_2$. This sort of uniformity in p, that makes a limit as $p \to 1^+$ useless when looking after the Willmore-type inequality, is rigorously partially explained in Subsection 2.5.1, where we show, among some other things, that the Willmore-type inequality in \mathbb{R}^n can actually be deduced from *any* value of p, the value p = 2 remaining thus the most convenient.

In Chapter 1 and 2, we are using the notion of strictly outward minimising hull outlined above twice. In the latter, we have already explicitly illustrated how it (actually, its area) will appear in the limit of the p-capacity of a bounded set with smooth boundary Ω . In Chapter 1, it will allow us to reduce the proof of the Isoperimetric Inequality (15) to the mean-convex sets. Chapter 3 is devoted to discuss this notion in considerable generality, in particular justifying the previous applications.

The Strictly Outward Minimising Hull in Riemannian manifolds

The main property one would like the strictly outward minimising hull Ω^* of a bounded set Ω to satisfy is that of being a bounded set minimising the perimeter among all of those sets containing Ω . This is actually the property needed in the argument of the proof of the Isoperimetric Inequality (15). However, easy examples show that in a general Riemannian ambient this is not always quite so, since Ω^* , roughly defined as the intersection of all the strictly outward minimising sets containing Ω , can in general happen to be the empty set. Indeed, manifolds with cuspidal or cylindrical ends, discussed in more details in Examples 3.8 and 3.9 provide examples of spaces containing bounded sets Ω that are not contained in any strictly outward minimising set, yielding no admissible sets in the intersection defining Ω^* . Moreover, cuspidal manifolds do not even admit solution to the least area problem with obstacle Ω , for any bounded Ω with finite perimeter contained in it, no matter the regularity of the boundary. On the other hand, the cylindrical manifold of Example 3.9 do admit solutions for the least area problem with obstacle Ω , but it happens that one cannot select, among these solutions, a bounded one with maximal volume. We realise in fact, in Theorem 3.13, that the well posedness of Ω^* is equivalent to the solvability of the maximum volume-least area problem with obstacle, and that this is in turn equivalent to the existence of an exhausting sequence of bounded strictly outward minimising sets.

If, on the one end, the condition of having such an exhausting sequence is not difficult to check on explicit metrics, such as warped products , on the other hand this general criterion does not seem easy to be directly applied to relevant classes of manifolds, for example, it is still not clear whether in complete noncompact manifolds with nonnegative Ricci curvature with Euclidean volume growth we can define the strictly outward minimising hull. The first achievement of Theorem 3.1 is that Ω^* is an open bounded maximal volume solution to the least area problem with obstacle a bounded set with finite perimeter Ω if one of the following two conditions are satisfied.

(i) The ambient manifold satisfies an Euclidean-like Isoperimetric Inequality, namely, there exists $C_{\rm iso} > 0$ such that

$$\frac{|\partial\Omega|^n}{|\Omega|^{n-1}} \ge C_{\text{iso}} \tag{23}$$

for any bounded set Ω with smooth boundary;

(ii) the ambient manifold has nonnegative Ricci curvature and the *superlinear uniform* volume growth condition holds

$$C_{\text{vol}}^{-1} r^b \le |B(O, r)| \le C_{\text{vol}} r^b \tag{24}$$

for some b > 1 and $C_{\text{vol}} > 0$, for any $r \ge R$ for some R > 0.

As we observe in Examples 3.8 and 3.9, the cylindrical and cuspidal manifolds described above coherently do not satisfy (i) nor (ii). However, the above assumptions are satisfied by a great variety of manifolds. First of all, let us remark that complete noncompact manifolds with nonnegative Ricci curvature satisfy condition (23) as pointed out above (see [Heb99, Theorem 8.4]), as well as Cartan-Hadamard manifolds, as a consequence of [HS74] (see [Heb99, Theorem 8.3] for an explicit derivation). We observe that the strictly outward minimising hull in Cartan-Hadamard manifolds was considered, for reasons that were analogue to ours in Chapter 1, in [Sch08]. Condition (23) is verified also on Asymptotically Flat ambient metrics, object of [HI01], and Asymptotically Locally Hyperbolic Riemannian manifolds, considered in [LN15]. This is a consequence of the fact that by the aforementioned [HS74] and the assumed asymptotic expansion of the metric (23) is valid outside a compact set, and thus, by [PST14, Theorem 3.2], this is true on the whole manifold. For what it concerns the assumptions in (ii) observe that they are naturally satisfied by complete noncompact Riemannian manifolds with nonnegative Ricci cuvature that are (locally) asymptotic to warped products $d\rho \otimes d\rho + \rho^{2\alpha}g_N$ for some smooth n-1 dimensional manifold N. In particular (ii) encompasses the case of the so called ALF and ALG manifolds, arising in the classification issue of gravitational instantons already hinted above. Moreover, the above warped products arise as models in the celebrated [CC96]. Concerning the relation between (i) and (ii), we point out that if b < n (observe that by Bishop-Gromov $b \le n$) they are mutually exclusive. Indeed, if the Isoperimetric Inequality holds, then it is known that so does the related Sobolev Inequality and thus by the arguments of [Car95] one can conclude that the volume of geodesic balls |B(x,r)| grow at least as r^n , see [PST14, Proposition 3.1] for a self contained proof.

We emphasise that also the description of Ω^* , although very briefly outlined in [HI01, Section 1], was not, to the author's knowledge, rigorously carried out in literature even in the easier case of flat \mathbb{R}^n . In this framework, the most close study is that performed in [BT84], where similar properties were derived for a related notion the authors refer to as *minimal hull*. This notion differs subtly but fundamentally from that of the strictly outward minimising hull, since, in the present terminology, it coincides with the intersection of all the outward minimising sets containing Ω , not just the strictly outward minimising ones. Despite, as it will be clear by the arguments used in Chapter 3, the area of the

resulting set will be the same, the volume in general differs, since Bassanezi-Tamanini's minimal hull will not in general yield a *maximal volume* solution to the least area problem with obstacle Ω . However, we acknowledge that some of the techniques presented [BT84] will be very useful in our arguments.

The proof differs sensibly when dealing with assumption (i) or (ii). Indeed, in the first case, we prove through the Direct Method in Calculus of Variations that the least area problem with obstacle Ω admits a maximal volume solution, and then obtaining, through the general criterion given in Theorem 3.13, the desired properties of Ω^* . The assumed Isoperimetric Inequality enters in an ODE argument 1 that shows uniform boundedness of the minimising sequence considered. On the other hand, the proof in case (ii) is an application of the existence theorem for weak IMCF developed in [MRS19, Theorem 1.8]. Indeed, once existence is established, the Huisken-Ilmanen's theory applies and show that level sets of this flow are strictly outward minimising, easily yielding the exhausting sequence of strictly outward minimising sets required by Theorem 3.13. It is important to notice that, anyway, in the case of nonnegative Ricci curvature with Euclidean volume growth, being (23) in force, the weak IMCF still does not enter the game, leaving the proof of the Isoperimetric Inequality free from such an expanding flow.

The last issue we face in Chapter 3 is that of the convergence as $p \to 1^+$ of the variational p-capacity of a bounded set with smooth boundary Ω to the area of its strictly outward minimising hull. In general this is not quite so. Indeed, we observe in Example 3.29 that the natural higher dimensional version of the 2-dimensional famous Hamilton's cigar [Ham88] admits a satisfactory notion of strictly outward minimising hull in the variational sense explained above for any bounded set Ω , but on the other hand the variational p-capacity of Ω is zero for any p > 1. The manifold in this example has nonnegative Ricci curvature and *linear* volume growth. In particular, it does not satisfy (i), and neither, sharply, (ii), since b in (24) is not allowed to be 1, that corresponds to linear volume growth. On the other hand we prove in Theorem 3.1 that if (i) or (ii) is satisfied then the claimed convergence takes place for any bounded set with smooth boundary. Again, the techniques in the most delicate part of the proof, that is $|\partial \Omega^*| \leq \liminf_{n \to 1^+} \operatorname{Cap}_n(\Omega)$, differs in relation to the assumption satisfied. If the ambient metric comes with an Isoperimetric Inequality, then an argument inspired by [Xu96] involving the Sobolev Inequality following from the Isoperimetric one does the job, yielding a self-contained proof. Assumption (i) is obviously satisfied in \mathbb{R}^n , that is what we needed when passing to the limit as $p \to 1^+$ in Chapter 2 in reaching the Extended Minkowski Inequality (17). If, on the other hand, we are assuming (ii), then we mostly rely on the decay estimates of the p-Green's function provided in [MRS19], in turn leading to the existence of the weak IMCF.

The present work ends with two appendices. Briefly, in Appendix A we systematically compare the monotonicity formulas for nonparabolic manifolds with nonnegative Ricci curvature of Chapter 1 with those for the Green's function obtained by Colding in [Col12] and subsequently extended in [CM14b]. In particular, in light of this, observe that Chapter 2 can be interpreted as a bridge between such monotonicity formulas, that according to [Col12] and [CM14a] can be understood as a sort of regularised version of the Bishop-Gromov monotonicity, and that of the function $\mathcal Q$ defined in (2) leading to the Minkowski inequality through Inverse Mean Curvature Flow.

In Appendix B, we give a proof of the existence of the *p*-capacitary potential under the sole hypothesis of *p*-nonparabolicity and decay at infinity of the *p*-Green's function,

¹We thank Prof. G. P. Leonardi for having outlined this idea.

together with basic properties of the p-capacity. We are applying this result in Chapter 2 and 3.

Most of the achievements of Chapter 1 are contained in [AFM18], those of Chapter 2 are obtained through the combination of [FMP19] with [AFM19], while the main result of Chapter 3 is not yet available in literature. It will be the content of a forthcoming joint paper with L. Mazzieri.

Chapter 1

Geometric inequalities in nonnegative Ricci curvature via Linear Potential theory

1.1 Structure of the chapter

In Section 1.2, we review, for ease of the reader, the theory of harmonic functions on Riemannian manifolds with nonnegative Ricci curvature we are going to employ along this chapter. The most important results are the existence Theorems for exterior boundary value problems characterising respectively nonparabolic and parabolic manifolds with nonnegative Ricci curvature, namely Theorems 1.14 and 1.17. Although Theorem B.1 in Appendix B fully encompasses Theorem 1.14, we propose here a proof that in some sense is best suited for Ric ≥ 0 . In Section 1.3 we introduce the conformal formulation of problem (1.2.15) and we prove, in this setting, (the conformal version of) Theorems 1.19 and 1.21, that are the Monotonicity-Rigidity Theorems in nonparabolic manifolds with nonnegative Ricci respectively for the function U_{β} and U_{∞} already encountered in the Introduction. In Section 1.4 we work out the integral asymptotic estimates for the electrostatic potential on manifolds with nonnegative Ricci curvature. With these estimates at hand, we conclude the proof of Theorem 1.44, that is the Willmore-type inequality for manifolds with nonnegative Ricci curvature and Euclidean volume growth, and discuss its Corollary 1.49 for ALE manifolds. In Section 1.5, we consider some other consequences of the Monotonity-Rigidity Theorems, generalising in the present general setting results of [AM20] and [BMM19]. In Section 1.6 we turn our attention to parabolic manifolds, and prove a suitable Monotonicity-Rigidity Theorem in this setting. We then combine it with that for nonparabolic manifolds to get enhanced versions of the noncompact splitting theorem by Kasue/Croke-Kleiner, see Theorem 1.56. At the end of the section, we also show how to recover its compact version, see Theorem 1.59. Finally, in Section 1.7, we prove the Isoperimetric Inequality on complete noncompact Riemannian 3-manifolds with Ric ≥ 0 . Finally, we give some perspectives and applications of such result, already mentioned in the Introduction.

1.2 Harmonic functions in exterior domains

In this section we are mainly concerned with characterising Riemannian manifolds for which problems (5) and (12) admit a solution. We are going to see that complete non-compact nonnegatively Ricci curved manifolds for which a solution to (5) exists are the *nonparabolic* ones, namely, manifolds admitting a positive Green's function, while those admitting a solution to (12) are the *parabolic* ones.

Nothing substantially new appears in this section. We are just collecting, re-arranging and applying classical results contained in [LT87; LT92; LT95; LY86; Yau75], and [Var82]. The interested reader might also refer to the nice survey [Gri99], where the relation with the Brownian motion on manifolds is also explored, or, for a more general account on the vast subject of harmonic functions on manifolds, to the lecture notes [Li] and the references therein. Other important works in this field will be readily cited along the chapter. Before starting, let us mention that the results gathered in this preliminary section are spread in a huge literature, and frequently they do not appear exactly in the form we need, or comes without a with a detailed proof. For this reason, we include the most relevant ones.

1.2.1 Green's functions and parabolicity

Let us begin with the definition of Green's functions on Riemannian manifolds.

Definition 1.1 (Green's function). A smooth function

$$G: (M \times M) \setminus \text{Diag}(M) \to \mathbb{R}$$
,

where $Diag(M) = \{(O, O), O \in M\}$, is said to be a Green's function for the Riemannian manifold (M, g) if the following requirements are satisfied.

- (i) G(x,y) = G(y,x) for any $x,y \in M$, $x \neq y$.
- (ii) $\Delta G(O, \cdot) = 0$ on $M \setminus \{O\}$, for any $O \in M$.
- (iii) The following asymptotic expansion holds for $x \to O$:

$$G(O,x) = (1 + o(1)) d^{2-n}(O,x).$$
 (1.2.1)

It is well known that on a complete noncompact Riemannian manifold there always exists a Green's function. This result has been obtained for the first time by Malgrange in [Mal55], while a constructive proof, best suited for applications, has been given by Li-Tam in [LT87]. Complete noncompact Riemannian manifolds are then divided into two classes.

Definition 1.2 (Parabolicity). *Complete noncompact Riemannian manifolds which support a* positive *Green's function are called* nonparabolic. *Otherwise they are called* parabolic.

A by-product of Li-Tam's construction of Green's function gives the following very useful characterisation of parabolicity, see for example [Li, Theorem 2.3] for a proof.

Theorem 1.3 (Li-Tam). Let (M,g) be a complete noncompact Riemannian manifold. Then, it is nonparabolic if and only if there exists a positive super-harmonic function f defined on the complement of a geodesic ball B(O,R) such that

$$\liminf_{d(O,x)\to\infty} f(x) < \inf_{\partial B(O,R)} f.$$
(1.2.2)

Notice that if (M, g) is a nonparabolic Riemannian manifold then a barrier function f as in Theorem 1.3 is just the function $G_{|M\setminus B(p,R)}$. A positive Green's function G is called *minimal* if

$$G(p,q) \leq \tilde{G}(p,q)$$

for any other positive Green's function \tilde{G} . The construction of the Green's function in [LT87] actually provides the minimal one.

The following theorem is a fundamental characterisation of parabolicity for manifolds with Ric ≥ 0 in terms of the volume growth of geodesic balls, first appeared in a complete version in [Var82].

Theorem 1.4 (Varopoulos). Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 . Then (M,g) is nonparabolic if and only if

$$\int_{1}^{+\infty} \frac{r}{|B(O,r)|} \, \mathrm{d}r < +\infty, \tag{1.2.3}$$

for any $O \in M$, where B(O, r) is a geodesic ball centered at O with radius $r \ge 0$.

The above characterisation roughly says that on nonparabolic manifolds volumes are growing faster than quadratically, while on the parabolic ones they grow at most quadratically. On the other hand, on a complete noncompact Riemannian manifold with Ric ≥ 0 Bishop-Gromov's Theorem and a result of Yau [Yau76] respectively show that the growth of volumes of geodesic balls B(p,r) is controlled from above by r^n and from below by r.

From now on we focus our discussion only on complete noncompact manifolds with nonnegative Ricci curvature. In the following two subsections we collect some basic though fundamental facts in this context, for the ease of references.

1.2.2 Harmonic functions on manifolds with nonnegative Ricci curvature.

A basic tool in the study of the potential theory on Riemannian manifolds is the following celebrated gradient estimate, first provided by Yau in [Yau75] (see also the nice presentation given in [SY94]).

Theorem 1.5 (Yau's Gradient Estimate). Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 . Let u be a positive harmonic function defined on a geodesic ball B(O,2R) of center $O \in M$ and radius 2R. Then, there exists a constant C = C(n) > 0 such that

$$\sup_{x \in B(O,R)} \frac{|\mathrm{D}u|}{u} \le \frac{C}{R}. \tag{1.2.4}$$

We now apply the above inequality to a harmonic function v defined in a geodesic annulus $B(O, R_1) \setminus \overline{B(O, R_0)}$. We obtain a decay estimate on the gradient of u that we will employ several times along this paper.

Proposition 1.6. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 , and let u be a positive harmonic function defined in a geodesic annulus $B(O,R_1)\setminus \overline{B(O,R_0)}$, with $R_1>3R_0$. Then, there exists a constant C=C(n) such that

$$|Du|(x) \le C \frac{u(x)}{d(O, x)}, \tag{1.2.5}$$

for any point q such that $2R_0 \le d(O, x) < \frac{R_1 + R_0}{2}$. In particular, if u is a harmonic function defined in $M \setminus \overline{B(O, R_0)}$, then

$$|Du|(x) \le C \frac{u(x)}{d(O, x)} \tag{1.2.6}$$

for any point x with $2R_0 \le d(O, x)$.

Proof. Let q be such that $2R_0 \le d(O,x) < \frac{R_1 + R_0}{2}$. Then the ball $B(x, d(O,x) - r_0)$ is all contained in the annulus $B(O,R_1) \setminus \overline{B(O,R_0)}$. In particular, by Yau's inequality (1.2.4) we have

$$|Du|(x) \le C \frac{u(x)}{d(O,x) - r_0} \le 2C \frac{u(x)}{d(O,x)}.$$

Letting $R_1 \to \infty$, we get also (1.2.6).

A fundamental and classical application of Yau's gradient inequality is the Harnack's inequality, that we are going to apply several times in the sequel. The explicit dependencies of the Harnack constant on large annuli will often play an important role. As usual we state and prove it in the form and in the setting we are interested, mainly following [Li]

Proposition 1.7 (Harnack's inequality). Let (M, g) be a complete noncompact manifold with nonnegative Ricci curvature, and let $K \subset M$ be a connected compact subset, and let it be covered by a finite number of geodesic balls $B(x_i, R_i)$ with $x_i \in M$. Then, for u a harmonic function defined on an open neighbourhood of K, we have

$$u(x) \le e^{C_{\text{har}}|\gamma_{x,y}|} u(y) \tag{1.2.7}$$

for any $x,y \in K$, where $\gamma_{x,y}$ is a curve joining x and y fully contained in K, and where the constant C_{har} satisfies

$$C_{\text{har}} = \sup_{i} C_{i}$$
,

with C_i being such that $|D \log u| \le C_i$ on $B(x_i, R_i)$.

Proof. The nonnegative Ricci curvature, ensures, by Yau's Theorem 1.5, the existence of C_i as in the statement. Let then $x, y \in K$, and let $\gamma_{x,y}$ be a curve joining x and y fully contained in K (it obviously exists by connectedness). Then, we have

$$\log u(x) - \log u(y) = \int_{\gamma_{x,y}} D\log u \, ds \le C_{\text{har}} |\gamma_{x,y}|.$$

Exponentiating, we obtain (1.2.7).

An immediate application of the above Harnack inequality yields at once the following compactness theorem for sequences of harmonic functions. Again, the source of the proof is [Li]

Lemma 1.8. Let (M,g) be a complete Riemannian manifold with Ric ≥ 0 , and let $D \subset M$ be an open connected subset. Let $\{f_i\}$ be a sequence of positive harmonic functions defined on D, and suppose there exists a constant C such that $f_i(x) \leq C$ at some point $x \in D$ for any $i \in \mathbb{N}$. Then, there exists a subsequence $\{f_{i_j}\}$ converging to a positive harmonic function f uniformly on any compact set $K \subset U$.

Proof. It clearly suffices to prove the statement for connected compact $K \subset D$ containing x. By Proposition 1.7, we have

$$f_i(y) \le f_i(x)e^{C_1|\gamma|} \tag{1.2.8}$$

for some constant C_1 not depending on i. By the uniform bound on $f_i(x)$ and the compactness of K we then deduce that the sequence f_i is uniformly bounded in K. Combining (1.2.8) with the Yau's inequality (1.2.4) we also get that $|Df_i|$ are uniformly bounded in K, and then by Ascoli-Arzelà a subsequence f_{i_j} converges uniformly on K to a continuous function f.

Let now $\psi \in \mathscr{C}_c^{\infty}(D)$ be a test function. Then, by harmonicity of f_{i_j} and uniform convergence on compact sets,

$$0 = \lim_{j \to \infty} \int_D f_{i_j} \Delta \psi \, \mathrm{d}\mu = \int_D f \Delta \psi \, \mathrm{d}\mu,$$

that is, f is harmonic in the sense of distribution. Standard regularity theory then implies that f is classically harmonic, completing the proof.

1.2.3 Ends of manifolds with nonnegative Ricci curvature

It is a well-known and largely exploited fact that a complete noncompact Riemannian manifold (M,g) with nonnegative Ricci curvature that is not a Riemannian cylinder has just one end. However, since a complete proof of this fact is hard to find in standard literature, we discuss the details below.

We employ the following definition of end, that is, the one used in the works by Li and Tam, see for example [LT92, Definition 0.4 and discussion thereafter].

Definition 1.9 (Ends of Riemannian manifolds). *An end of a Riemannian manifold* (M, g) with respect to a compact subset $K \subset M$ is an unbounded connected component of $M \setminus K$. We say that (M, g) has a finite number of ends if the number of ends with respect to any compact subset $K \subset M$ is bounded by a natural number k independent of K. In this case, we say that (M, g) has k ends if it has k ends with respect to a compact subset $K \subset M$ and to any other compact subsets of M containing K.

To state the result, we quickly recall some terminology. A *line* in (M,g) is a curve $\gamma: \mathbb{R} \to M$ which is a minimal geodesic between any two points lying on it. A *ray* is half a line. We also recall that a complete Riemannian manifold (M,g) is called a *Riemannian cylinder* if it is isometric to the Riemannian product $(\mathbb{R} \times N^{n-1}, dt \otimes dt + g_{N^{n-1}})$, where N^{n-1} is a compact manifold.

Remark 1.10. Using the above definition and terminology it is clear that if a Riemannian manifold has at least two ends, then it contains a line.

Proposition 1.11. *Let* (M,g) *be a complete noncompact Riemannian manifold with* Ric ≥ 0 . *If* (M,g) *is not a Riemannian cylinder, then it has just one* end.

Proof. Since (M^n, g) has nonnegative Ricci curvature by hypothesis, then the Cheeger-Gromoll Splitting Theorem [CG71] implies that

$$(M,g)$$
 is isometric to $(\mathbb{R}^m \times N^{n-m}, g_{\mathbb{R}^k} + g_{N^{n-m}}),$ (1.2.9)

for some $m \in \{0, ..., n\}$, where the manifold $(N^{n-m}, g_{N^{n-m}})$ has nonnegative Ricci curvature and does not contain any line. The Riemannian manifold (M, g) is a Riemannian cylinder if m = 1 and N^{n-1} is compact. Let us then suppose that (M, g) is not a Riemannian cylinder. We consider the cases m = 0, m = 1 and $m \ge 2$.

Case m = 0. In this case, (M, g) does not contain any line. Then there is no more than one end, in view of Remark 1.10.

Case m = 1. Since (M, g) is not a cylinder, we have that N^{n-1} is a noncompact Riemannian manifold that contains no lines. Then again by Remark 1.10, it has at most one end. Thus, also $\mathbb{R} \times N^{n-1}$ has at most one end.

Case $m \geq 2$. We show that

$$M \setminus K$$
 is connected for every compact $K \subset M$. (1.2.10)

In view of Definition 1.9, this readily implies that M has at most one end. Now, to check (1.2.10) in view of (1.2.9), it is sufficient to check that for every compact $Q \in \mathbb{R}^m$ and every compact $P \in N^{n-m}$, we have that $(\mathbb{R}^k \times N^{n-m}) \setminus (Q \times P)$ is connected. Let then $(x,q), (y,p) \in (\mathbb{R}^m \times N^{n-m}) \setminus (Q \times P)$ and suppose for the moment that $x,y \in Q$, so that, in turn, $q,p \notin P$. Choose $z \in \mathbb{R}^m \setminus Q$ and define the curves

$$\alpha(t) = (tx + (1-t)z, q), \qquad \beta(t) = (ty + (1-t)z, p),$$

 $t \in [0,1]$, connecting (x,q) to (z,q) and (y,p) to (z,p), respectively. Note that $\alpha(t), \beta(t) \in (\mathbb{R}^m \times N^{n-m}) \setminus (Q \times P)$ for every $t \in [0,1]$, because $q,p \notin Q$. Now, let $\gamma(t)$ be a continuous curve in N^{n-m} connecting q and p. Then the curve $(z,\gamma(t)) \in ((\mathbb{R}^m \setminus Q) \times N^{n-m})$ connects (z,q) to (z,p). Gluing together the curves α,β , and (z,γ) , we obtain a continuous path lying in $(\mathbb{R}^m \times N^{n-m}) \setminus (Q \times P)$ and connecting (x,q) to (y,p). Obtaining such a curve in the case where either $x \notin Q$ or $y \notin Q$ requires a similar simpler construction. We have thus proved that $(\mathbb{R}^m \times N^{n-m}) \setminus (Q \times P)$ is path-connected, hence connected.

We proved that (M, g) has at most one end, if it is not a cylinder. Then, it has exactly one end, because it is noncompact.

Let us now describe separately some aspects of harmonic functions on nonparabolic and parabolic manifolds. In particular, we are going to characterise these two classes of manifolds through a couple of existence results for solutions of suitable boundary value problems in exterior domains (see Theorems 1.14 and 1.17 below). The monotone quantities analysed in Theorems 1.19 and 1.54 are defined along the level sets of these solutions.

1.2.4 The exterior problem on nonparabolic manifolds

The following is a fundamental estimate proved by Li-Yau in [LY86].

Theorem 1.12 (Li-Yau). Let (M,g) be a nonparabolic Riemannian manifold with Ric ≥ 0 . Then, its minimal Green's function G satisfies

$$C^{-1} \int_{d(O,x)}^{+\infty} \frac{r}{|B(O,r)|} dt \le G(O,x) \le C \int_{d(O,x)}^{+\infty} \frac{r}{|B(O,r)|} dt, \qquad (1.2.11)$$

for some C = C(n) > 0.

Combining (1.2.3) with (1.2.11), we get that the minimal Green's function goes to 0 at infinity, i.e. for any fixed O in M

$$\lim_{d(O,y)\to\infty} G(O,y) = 0.$$
 (1.2.12)

An easy application of Laplace Comparison Theorem then gives the following well known fact.

Lemma 1.13. Let (M,g) be a nonparabolic Riemannian manifold with $Ric \ge 0$, and let G be its minimal Green's function. Then, for any fixed pole $O \in M$ we have

$$d^{2-n}(O,x) \le G(O,x). \tag{1.2.13}$$

for any $x \neq 0$ in M.

Proof. Let r be the function mapping a point x in M to d(O, x). By the Laplacian Comparison Theorem, we have

$$\Delta r \leq \frac{n-1}{r}$$

in the sense of distributions (see e.g. [CLN06, Theorem 1.128]). Therefore, we have, in the sense of distributions,

$$\Delta r^{2-n} = (n-2) \left[(n-1)r^{-n} - r^{1-n}\Delta r \right] \ge 0,$$
 (1.2.14)

and then the function $r^{2-n}-G(O,\cdot)$ is sub-harmonic. By the maximum principle, for any $\varepsilon>0$ and $R>\varepsilon$

$$\max_{\overline{B(O,R)}\setminus B(O,\varepsilon)}(r^{2-n}-G(O,\cdot)) = \max_{\partial B(O,R)\cup \partial B(O,\varepsilon)}(r^{2-n}-G(O,\cdot)).$$

We conclude by passing to the limit as $\varepsilon \to 0$ and $R \to \infty$, taking into account the asymptotic behaviour at the pole O given by (1.2.1) and that $G \to 0$ at infinity, as observed in (1.2.12).

Now, we characterise the existence of a solution to problem (8) with the nonparabolicity of the ambient manifold. Let us recall that, with respect to a bounded open subset $\Omega \subset M$ with smooth boundary, and denoting by O a generic reference point taken inside Ω , we consider

$$\begin{cases} \Delta u = 0 & \text{in } M \setminus \overline{\Omega} \\ u = 1 & \text{on } \partial\Omega \\ u(y) \to 0 & \text{as } d(O, y) \to +\infty, \end{cases}$$
 (1.2.15)

The proof we give below uses some aspects of the nonnegative Ricci curvature, that we believe of independent interest. We refer the reader to Theorem B.1 for a more general argument involving boundary regularity estimates for solutions of elliptic equations.

Theorem 1.14. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 , and let $\Omega \subset M$ be a bounded open subset with smooth boundary. Then, there exists an unique solution to problem (1.2.15) if and only if (M,g) is nonparabolic.

Proof. Most of the arguments here are suitable adaptation of some already presented in [LT92]. By Theorem 1.3 the existence of a solution to problem (1.2.15) implies nonparabolicity of M, since the restriction of u to $M \setminus B(O, R)$ with $\Omega \subset B(O, R)$ clearly satisfies condition (1.2.2).

Conversely, assume that M is nonparabolic, and consider an increasing sequence of radii $\{R_i\}_{i\in\mathbb{N}}$ such that $\Omega\subset B(O,R_1)$ and $R_i\to\infty$. Let, for any $i\in\mathbb{N}$, u_i be the solution to the following problem:

$$\begin{cases} \Delta u = 0 & \text{in } B(O, R_i) \setminus \overline{\Omega} \\ u = 1 & \text{on } \partial\Omega \\ u = 0 & \text{on } \partial B(O, R_i). \end{cases}$$
 (1.2.16)

Let now G be the minimal positive Green's function, and consider the function $G(O, \cdot)$. Due to the Maximum Principle for harmonic functions and the boundary conditions in problem (1.2.16) we have that

$$0 \le u_{Ri}(x) \le \frac{G(O, x)}{\min_{\partial \Omega} G(O, \cdot)}, \tag{1.2.17}$$

for $x \in B(O, R_i)$. Let then K be a compact set contained in $M \setminus \overline{\Omega}$. We can clearly suppose without loss of generality that K is contained in $B(O, R_i) \setminus \overline{\Omega}$ for any i. Then, (1.2.17) and Lemma 1.8 give that u_i converges up to a subsequence to a harmonic function u on K. We now need to check the convergence of (a subsequence of) u_i on compact subsets K of $M \setminus \Omega$, that, is possibly containing portions of $\partial\Omega$. To see this, we show that $|Du_i|$ is bounded uniformly in i also on these compact subsets. To see this, consider the P-function (see [PP79])

$$f_i = \frac{|\mathrm{D}u_i|}{u_i^{\frac{(n-1)}{(n-2)}}}.$$

It can be directly checked that the Bochner formula, and it is actually the content of (1.3.35) written in terms of the current metric g, that f_i is subsolution of an elliptic equation. In particular, once restricted on the annulus $B(O,R)\setminus \overline{\Omega}$ for some fixed R>0 such that the ball contains $\overline{\Omega}$, we deduce from the Maximum Principle that the maximum value of f_i is assumed, for i big enough, on $\partial\Omega\cup B(O,R)$. Then, if K is a compact set of $M\setminus \Omega$, it suffices to consider separately the case where the maximum values of (a subsequence of) f_i are assumed on interior points x_i of $M\setminus \Omega$ uniformly away from $\partial\Omega$ and the case where the points x_i lie on $\partial\Omega$. In the first case, Du_i is uniformly bounded on K as an immediate consequence of Yau's (1.2.4) together with the uniform convergence of u_i on the compact subsets of $M\setminus \overline{\Omega}$. Assume then that the points x_i lie on $\partial\Omega$. Then, letting $v=-Du_i/|Du_i|$ the unit normal to $\partial\Omega$, we have $(\partial f_i/\partial v)(x_i)\leq 0$. On the other hand, recalling that $u_i=1$ on $\partial\Omega$, it holds

$$\frac{\partial f_i}{\partial \nu} = 2 \left[\frac{n-1}{n-2} |Du_i|^3 - 2H |Du_i|^2 \right],$$

on x_i , where we wrote the mean curvature H of $\partial\Omega$ in terms of the Hessian of u_i exactly as in (1.3.17). Using the Hopf's lemma, that ensures that $|Du_i|$ does not vanish on $\partial\Omega$, we then get

$$\sup_{K} |\mathrm{D}u_i| \le \frac{n-1}{n-2} \sup_{\partial \Omega} \mathrm{H},$$

that, is $|Du_i|$ is uniformly bounded also in this case.

We have built a harmonic function defined on $M \setminus \Omega$ taking the value 1 on $\partial \Omega$. Finally, by (1.2.17), the uniform convergence of u_i to u and (1.2.12), we get $u(y) \to 0$ as $d(O, y) \to +\infty$, completing the proof of existence.

Uniqueness is checked as usual, considering for another solution v the harmonic function u-v, and satisfying $u-v \le \varepsilon$ on (a smoothed out approximation of) $\partial\Omega \cup \partial B(O,R_{\varepsilon})$ for R_{ε} big enough, for any $\varepsilon > 0$. The Maximum Principle then yields $u \le v + \varepsilon$ on $B(O,R_{\varepsilon})\setminus\overline{\Omega}$, that implies $u \le v$ letting $\varepsilon \to 0$. Reversing the roles of u and v yields uniqueness.

We conclude this section with the following easy lemma, which shows that we can control function u by the minimal Green's function G.

Lemma 1.15. Let (M,g) be a nonparabolic Riemannian manifold with Ric ≥ 0 , and let G be its minimal Green's function. Let u be a solution to (1.2.15) for some open and bounded set Ω with smooth boundary, and let $O \in \Omega$. Then, there exist constants $C_1 = C_1(\Omega) > 0$ and $C_2 = C_2(\Omega) > 0$ such that

$$C_1 G(O, y) \le u(y) \le C_2 G(O, y)$$
 (1.2.18)

on $M \setminus \Omega$. In particular

$$C_1 d(O, y)^{2-n} \le u(y).$$
 (1.2.19)

on $M \setminus \Omega$.

Proof. Just set $0 < C_1 < 1/\max_{\partial\Omega} G(O, \cdot)$, and $C_2 > 1/\min_{\partial\Omega} G(O, \cdot)$. The claim follows from the Maximum principle and the observation that both u and G are vanishing at infinity. The inequality (1.2.19) is obtained combining the lower estimate on u by (1.2.18) with (1.2.13).

1.2.5 The exterior problem on parabolic manifolds

The following inequalities, proved in [LT95, Theorem 2.6], can be interpreted as a version for parabolic manifolds of the Li-Yau inequalities recalled in Theorem 1.12. We point out that we are always dealing with Green's functions obtained by the Li-Tam's construction.

Theorem 1.16. Let (M,g) be a parabolic manifold with Ric ≥ 0 , and let $O \in M$. Let G be a Green's function. Then, for any fixed $r_0 > 0$ and for any x with $d(O,x) > 2r_0$ there holds

$$-G(O,x) \leq C_1 \int_{r_0}^{d(O,x)} \frac{r}{|B(O,r)|} dr + C_2, \qquad (1.2.20)$$

for some constants C_1 and C_2 depending only on n, r_0 and the choice of G. Moreover, for any $R > r_0$, there holds

$$C_3 \int_{r_0}^{R} \frac{r}{|B(O,r)|} dr + C_4 \le \sup_{\partial B(O,R)} -G(O,\cdot),$$
 (1.2.21)

for some constants C₃ and C₄ depending only on n, r₀ and the choice of G.

When (M, g) is parabolic, Li-Tam substantially proved in [LT92, Lemma 1.2] that the exterior problem

$$\begin{cases} \Delta \psi = 0 & \text{in } M \setminus \overline{\Omega} \\ \psi = 0 & \text{on } \partial \Omega \\ \psi(y) \to +\infty & \text{as } d(O, y) \to +\infty, \end{cases}$$
 (1.2.22)

admits a solution. The construction of such a solution ψ , combined with Yau's inequality and Theorem 1.16, readily implies a uniform gradient bound on ψ .

Theorem 1.17. Let (M,g) be a parabolic Riemannian manifold with Ric ≥ 0 , and let $\Omega \subset M$ be a bounded and open subset with smooth boundary. Then, there exists a solution to problem (1.2.22). Moreover, $|D\psi|$ is uniformly bounded in $M \setminus \Omega$.

Remark 1.18. Recall from Subsection 1.2.3 that if (M,g) is a Riemannian cylinder, then $M \setminus \Omega$ might have two connected components. If this is the case, it will be understood that we consider problem (1.2.22) on a connected component of $M \setminus \Omega$. All the proofs work unchanged in this case.

Proof of Theorem 1.17. Let $O \in \Omega$, let $U \subset \Omega$ be an open neighbourhood of O and let K be the compact set defined by $K = \overline{\Omega} \setminus U$. Consider, for a sequence $B(O, R_i)$ of geodesic balls with increasing radii containing Ω , a corresponding sequence of positive Green's

functions $G_i(O, \cdot)$ of $B(O, R_i)$ with pole in O such that $G_i(O, x) = 0$ for $x \in \partial B(O, R_i)$. We then consider the sequence of functions defined in $B(O, R_i) \setminus \{O\}$ by

$$h_i(y) = \sup_{x \in K} G_i(O, x) - G_i(O, y).$$

The construction in [LT87] implies that there exists a Green's function G on M such that h_i converges to $-G(O,\cdot)$ uniformly on compact subsets of $M\setminus\{O\}$ (compare with the discussions around Lemma 1.2 in [LT92]). Observing that $h_i = \sup_K G_i(O,\cdot)$ on $\partial B(O,R_i)$, we set

$$a_i = \sup_{x \in K} G_i(O, x)$$

and consider the solution ψ_i to the problem

$$\begin{cases} \Delta \psi = 0 & \text{in } B(O, R_i) \setminus \overline{\Omega} \\ \psi = 0 & \text{on } \partial \Omega \\ \psi = a_i & \text{on } \partial B(O, R_i). \end{cases}$$

Since $\sup_K G_i(O, \cdot) \ge \sup_{\partial\Omega} G_i(O, \cdot)$, the Maximum Principle immediately gives

$$h_i - \sup_{\partial \Omega} h_i \le \psi_i \le h_i \tag{1.2.23}$$

on $B(O,R_i)\setminus\Omega$. Since the sequence h_i is converging (uniformly on compact sets) to the Li-Tam Green's function, the second inequality in (1.2.23) combined with Lemma 1.8 shows that ψ_i converges uniformly on the compact subsets of $M\setminus\overline\Omega$ to a harmonic function ψ . Moreover, since for every $y\in M\setminus\{O\}$ the sequence $h_i(y)$ converges to -G(O,y) and since by (1.2.21) we have that $-G(O,y_j)\to +\infty$ along a sequence of points y_j such that $d(O,y_j)\to +\infty$, we use the first inequality in (1.2.23), to deduce that $\psi(y_j)\to +\infty$, as $j\to +\infty$. In particular, since by [CM97b, Lemma 3.40] ψ must admit a limit at infinity, we infer that $\psi(y)\to +\infty$, as $d(O,y)\to +\infty$ As in the proof of Theorem 1.14, we need to check the convergence of ψ_i also on compact $K\subset M\setminus\Omega$ possibly containing portions of the boundary $\partial\Omega$, by showing that the sequence of $|D\psi_i|$ is uniformly bounded on such compact sets. To see this, let B(O,R) big enough so that it contains $K\cup\Omega$, and consider the harmonic functions

$$v_i = 1 - \frac{\psi_i}{\sup_{B(O,R)} \psi_i + 1},$$

and define, as in the proof of Theorem 1.14, the P-functions

$$f_i = \frac{|\mathrm{D}v_i|}{v_i^{\frac{n-1}{n-2}}}.$$

Then, again, f_i is subsolution of an elliptic equation (that again, corresponds exactly to (1.3.35) in the original metric), and thus the maximum values of f_i are achieved on $x_i \in \partial\Omega \cup \partial B(O,R)$. If $x_i \in \partial B(O,R)$, we easily see that $|D\psi_i|$ is uniformly bounded by Yau's inequality (1.2.4) and the uniform convergence of ψ_i on compact subsets of $M \setminus \overline{\Omega}$. If, on the other hand, $x_i \in \partial\Omega$, the very same computation performed in the proof of Theorem 1.14 involving the Hopf's lemma shows that

$$|\mathrm{D}\psi_i| \le \mathrm{H}\left(1 + \sup_{B(O,R)} \psi_i\right). \tag{1.2.24}$$

Since $\psi \to \infty$ at infinity, we can infer from the uniform convergence of ψ_i to ψ on compact

subsets of $M \setminus \overline{\Omega}$ that the points where the functions ψ_i achieve the maximum value inside B(O,R) are located, for i big enough, in a compact subset K' away from $\partial\Omega$, and then, again by the uniform convergence of ψ_i on K', (1.2.24) allows to conclude that $|D\psi_i|$ is uniformly bounded in this case too.

We are left to show that $|D\psi|$ is uniformly bounded. Observe that, again by (1.2.23), $\psi \leq -G(O,\cdot)$. Inequality (1.2.6) then yields

$$|D\psi|(x) \le C \frac{\psi(x)}{d(O,x)} \le C \frac{-G(O,x)}{d(O,x)}$$
 (1.2.25)

for some constant C and any x outside some big geodesic ball $B(O, r_0)$. Combining now (1.2.20) with Yau's lower bound on the growth of geodesic balls, saying that $|B(O, r)| \ge Cr$ for any $r \ge 1$ and for some constant C, we also have

$$\frac{-G(O,x)}{d(O,x)} \le C_1 \frac{d(O,x) + C_2}{d(O,x)}$$

for x with $d(O, x) > 2r_0$ and constants C_1 and C_2 . Plugging it in (1.2.25), this shows that $|D\psi|$ is uniformly bounded, as claimed.

1.3 The Monotonicity-Rigidity Theorems for nonparabolic manifolds

The goal of this section is to prove the monotonicity of the function $U_{\beta}:(0,1]\to\mathbb{R}$ for $\beta\geq (n-2)/(n-1)$ and $U_{\infty}:(0,1]\to\mathbb{R}$ already presented in the Introduction, defined respectively by

$$U_{\beta}(t) = t^{-\beta \left(\frac{n-1}{n-2}\right)} \int |Du|^{\beta+1} d\sigma$$

$$\{u=t\}$$
(1.3.1)

and

$$U_{\infty}(t) = \sup_{\{u=t\}} \frac{|Du|}{u^{\frac{n-1}{n-2}}}.$$
 (1.3.2)

The following is the statement of the Monotonicity-Rigidity Theorem for U_{β} .

Theorem 1.19 (Monotonicity-Rigidity Theorem for nonparabolic manifolds). Let (M,g) be a nonparabolic Riemannian manifold with Ric ≥ 0 . Given a bounded and open subset $\Omega \subset M$ with smooth boundary, let u be the solution to problem (1.2.15) and let $U_{\beta}:(0,1] \to \mathbb{R}$ be the function defined in (1.3.1). Then, for every $\beta \geq (n-2)/(n-1)$, the function U_{β} is differentiable, with derivative

$$\frac{\mathrm{d}U_{\beta}}{\mathrm{d}t}(t) = \beta t^{-\beta \left(\frac{n-1}{n-2}\right)} \int |\mathrm{D}u|^{\beta} \left[H - \left(\frac{n-1}{n-2}\right) |\mathrm{D}\log u| \right] d\sigma, \tag{1.3.3}$$

where H is the mean curvature of the level set $\{u = t\}$ computed with respect to the unit normal vector field v = -Du/|Du|. The derivative of U_{β} fulfils

$$\frac{dU_{\beta}}{dt}(t) = \frac{\beta}{t^{2}} \int_{\{u < t\}} u^{2-\beta \left(\frac{n-1}{n-2}\right)} |Du|^{\beta-2} \left\{ \text{Ric}(Du, Du) + |Du|^{2} \left| h - \frac{H}{n-1} g^{\top} \right|^{2} + \beta |D^{\top}|Du||^{2} + (\beta - \frac{n-2}{n-1}) |Du|^{2} \left[H - \left(\frac{n-1}{n-2}\right) |D\log u| \right]^{2} \right\} d\mu,$$
(1.3.4)

where H is the mean curvature of the level sets of u computed with respect to the unit normal vector field ν . In particular, U_{β} is nondecreasing. Moreover, $(dU_{\beta}/dt)(t_0) = 0$ for some $t_0 \leq 1$ and some $\beta \geq (n-2)/(n-1)$ if and only if (M,g) has Euclidean volume growth and $(\{u \leq t_0\}, g)$ is isometric to

$$\left(\left[r_0, +\infty \right) \times \left\{ u = t_0 \right\}, \ dr \otimes dr + \left(\frac{r}{r_0} \right)^2 g_{\{u = t_0\}} \right), \quad \text{with} \quad r_0 = \left(\frac{\left| \left\{ u = t_0 \right\} \right|}{\text{AVR}(g) |\mathbb{S}^{n-1}|} \right)^{\frac{1}{n-1}}.$$
(1.3.5)

In this case, in particular, $\{u = t_0\}$ is a connected totally umbilic submanifold with constant mean curvature.

Remark 1.20 (Meaning of (1.3.3) at singular values). It is important to point out that when t is a singular value for u, (1.3.3) has to be interpreted as the (unique) extension of such quantity on regular values. This will turn to be allowable by Sard Theorem, and it is an important technical point of the proof. This agreement will be assumed throughout the whole chapter.

Analogously, the following is the Monotonicity-Rigidity Theorem for U_{∞} .

Theorem 1.21 (Monotonicity-Rigidity Theorem for U_{∞}). Let (M,g) be a complete non-parabolic Riemannian manifold with Ric ≥ 0 . Then, the function $U_{\infty}:(0,1] \to \mathbb{R}$ defined by (1.3.2) is monotone nondecreasing. Moreover, if $x_t \in \{u=t\}$ is such that

$$\frac{|Du|}{u^{\frac{n-1}{n-2}}}(x_t) = \sup_{\{u=t\}} \frac{|Du|}{u^{\frac{n-1}{n-2}}},$$

then

$$H - \frac{n-1}{n-2} |D\log u|(x_t) = -\frac{\partial}{\partial \nu_t} \log \frac{|Du|}{u^{\frac{n-1}{n-2}}}(x_t) \ge 0, \tag{1.3.6}$$

where v_t is the unit normal to $\{u=t\}$ given by $v_t=-Du/|Du|$. Moreover, $U_\infty(T)=U_\infty(t_0)$ for some $T< t_0$ or inequality (1.3.6) holds with equality sign a $t=t_0$ with $t_0\in (0,1]$ if only if (M,g) has Euclidean volume growth and and $(\{u\leq t_0\},g)$ is isometric to

$$\left(\left[r_0, +\infty \right) \times \left\{ u = t_0 \right\}, \ dr \otimes dr + \left(\frac{r}{r_0} \right)^2 g_{\{u = t_0\}} \right), \quad \text{with} \quad r_0 = \left(\frac{|\{u = t_0\}|}{\text{AVR}(g)|\mathbb{S}^{n-1}|} \right)^{\frac{1}{n-1}}.$$

In this case, in particular, $\{u = t_0\}$ is a connected totally umbilic submanifold with constant mean curvature.

1.3.1 Preparation of the proof of the Monotonicity-Rigidity-Theorems

Before providing the proof of Theorems 1.19 and 1.21, we give a sharp and complete statement the refined Kato's inequality for harmonic functions, that is one of the main responsible of the monotonicity, and describe the conformal setting where we are actually going to work out the proofs.

A Kato-type identity and a related splitting principle

The classical Kato inequality for functions on a Riemannian manifold asserts that if f is a smooth function defined somewhere on a Riemannian manifold, then

$$\left| \mathbf{D} | \mathbf{D} f | \right|^2 \le |\mathbf{D} \mathbf{D} f|^2.$$

When, in addition, *f* satisfies some differential equation, the above inequality can be crucially improved. In the case of harmonic functions, the refined Kato inequality says that

$$\left| \mathrm{D} |\mathrm{D} f| \right|^2 \leq \left(\frac{n-1}{n} \right) |\mathrm{D} \mathrm{D} f|^2.$$

Other refined Kato inequalities, suited for various elliptic equations, have been derived and applied successfully in the literature. We address the reader to [Her00] for an account on this subject.

In this work, and dramatically in the next chapter, we will not only use the inequality, but also the deficit from being an inequality will be fundamental. The following is the Kato-type identity for harmonic functions in Riemannian manifolds.

Proposition 1.22 (Kato-type identity for harmonic function). Let (M,g) be a Riemannian manifold, and let f be a harmonic function defined on some subset of M. Then, in an open neighbourhood of point where $|\nabla v| \neq 0$, the following identity holds true

$$|DDf|^{2} - \left(\frac{n}{n-1}\right) |D|Df|^{2} = |Df|^{2} \left| h - \frac{H}{n-1} g^{\top} \right|^{2} + \left(\frac{n-2}{n-1}\right) |D^{\top}|Df|^{2}.$$
 (1.3.7)

Moreover, if f is nonconstant and the right hand side of (1.3.7) vanishes in $\{\rho_0 \leq f \leq \rho_1\}$ for some $\rho_0 < \rho_1$ with ρ_1 possibly infinite, then the Riemannian manifold $(\{\rho_0 \leq f \leq \rho_1\}, g)$ is isometric to the warped product $([\rho_0, \rho_1] \times \{f = \rho_0\}, d\rho \otimes d\rho + \eta^2(\rho)g_{\{f = \rho_0\}})$, where f, η and the coordinate ρ are related as

$$\eta(\rho) = \left(\frac{f'(\rho_0)}{f'(\rho)}\right)^{\frac{1}{n-1}} \tag{1.3.8}$$

We will not prove Proposition 1.22 here, since a more general statement will be proved in the next Chapter, see Proposition 2.5. It is anyway substantially known, and can be deduced for example by tracking the proof of [BC12, Proposition 5.1]. The splitting result arising from the vanishing of the right hand side of (1.3.7) is actually an interesting generalisation of a well known splitting result asserting that a function with null hessian in some region of a Riemannian manifold forces such a region to be a cylinder. This fact can be deduced from the more general principle discussed in [CC96, Section 1] and [CMM12, Theorem 1.1] or it can be directly proved as done in [AM15, Theorem 4.1 (i)]. Here, we derive it from Proposition 1.22.

Corollary 1.23. Let (M,g) be a Riemannian manifold, and let f be a function defined on some subset of M that is nonconstant and satisfies |DDf| = 0 on $\{\rho_0 \le f \le \rho_1\}$ for some $\rho_0 < \rho_1$

with ρ_1 possibly infinite. Then, the Riemannian manifold $(\{\rho_0 \leq f \leq \rho_1\}, g)$ is isometric to the cylinder $([\rho_0, \rho_1] \times (\{f = \rho_0\}, d\rho \otimes d\rho + g_{|\{f = \rho_0\}})$ and, in this case, f is an affine function of ρ .

Proof. Clearly the function f is harmonic. Since $|D|Df|| \le |DDf|$, the function |Df| is constant in $\{\rho_0 \le f \le \rho_1\}$, and this constant must be positive by assumption. Then, (1.3.7) holds, and since the left hand side is zero so is the right hand one, and in particular the rigidity statement of Proposition 1.22 applies. Moreover, deriving (1.3.8) and taking into account that $f''(\rho) = 0$, we find that $\eta'(\rho) = 0$. Since $\eta(\rho_0) = 1$, we conclude that η is the constant 1, that means cylindrical splitting. Moreover, again by (1.3.8), this implies that $f'(\rho) = f'(\rho_0)$, implying that f is an affine function of ρ .

The conformal setting

Let (M, g) be a nonparabolic Riemannian manifold with nonnegative Ricci curvature. Let $\Omega \subset M$ be a bounded and open set with smooth boundary, and let u be the solution to problem (1.2.15). We introduce, in $M \setminus \Omega$ the metric

$$\tilde{g} = u^{\frac{2}{n-2}}g. \tag{1.3.9}$$

The expression for \tilde{g} is formally the same as in [AM15] and [AM20]. Let us explain why such a conformal change of metric is natural also in the current setting. Our model geometry is that of a truncated cone

$$(M \setminus \Omega, g) \cong \left(\left[r_0, +\infty \right) \times \partial \Omega, \ dr \otimes dr + \left(\frac{r}{r_0} \right)^2 g_{\partial \Omega} \right),$$
 (1.3.10)

for some positive constant r_0 , and where $g_{\partial\Omega}$ is the metric induced by g on $\partial\Omega$. We also assume that $\partial\Omega$ is a smooth closed sub-manifold with $\mathrm{Ric}_{\partial\Omega} \geq (n-2)g_{\partial\Omega}$. Such a curvature assumption on $\partial\Omega$ is equivalent to suppose that the conical region in (1.3.10) has nonnegative Ricci curvature. In this model setting, the solution to problem (1.2.15) is $u(r) = (r/r_0)^{2-n}$. With this specific u, the metric \tilde{g} becomes

$$\tilde{g} = d\rho \otimes d\rho + g_{\partial\Omega}$$
,

where $\rho = \log r$. In other words \tilde{g} is a (half) Riemannian cylinder over $(\partial \Omega, g_{\partial \Omega})$. In parallel, as the rigidity statement in Theorem 1.19 gives a characterisation of the truncated cone metrics (1.3.10), so its conformal version in Theorem 1.25 characterises cylindrical metrics.

Having this in mind, we are now going to describe the general features of $(M \setminus \Omega, \tilde{g})$ in more details. Letting

$$\varphi = -\log u, \tag{1.3.11}$$

we have that $\tilde{g} = e^{-\frac{2\varphi}{n-2}}g$.

We denote by ∇ , the Levi-Civita connection of the metric \tilde{g} , by $\nabla\nabla$ its Hessian, and we put the subscript \tilde{g} on any other quantity induced by \tilde{g} . We have, for a smooth function w

$$abla_{lpha}
abla_{eta}w \,=\, \mathrm{D}_{lpha}\mathrm{D}_{eta}w + rac{1}{n-2}\Big(\partial_{lpha}w\partial_{eta}\varphi + \partial_{eta}w\partial_{lpha}\varphi - \langle \mathrm{D}w\,|\,\mathrm{D}\varphi
angle g_{lphaeta}\Big),$$

where by $\langle \cdot, \cdot \rangle$ we denote the scalar product induced by g. In particular,

$$\Delta_{\tilde{g}}\varphi = 0. \tag{1.3.12}$$

Moreover, the Ricci tensor $Ric_{\tilde{g}}$ of \tilde{g} and the Ricci tensor Ric of g satisfy

$$\operatorname{Ric}_{\tilde{g}} = \operatorname{Ric} + \nabla_{\alpha} \nabla_{\beta} \varphi - \frac{d\varphi \otimes d\varphi}{n-2} + \frac{|\nabla \varphi|_{\tilde{g}}^{2}}{n-2} \tilde{g}. \tag{1.3.13}$$

Finally, by (1.3.12) and (1.3.13) problem (1.2.15) becomes

$$\begin{cases} \Delta_{\tilde{g}}\varphi = 0 & \text{in } M \setminus \overline{\Omega} \\ \operatorname{Ric}_{\tilde{g}} - \nabla \nabla \varphi + \frac{\operatorname{d}\varphi \otimes \operatorname{d}\varphi}{n-2} = \frac{|\nabla \varphi|_{\tilde{g}}^2}{n-2} \tilde{g} + \operatorname{Ric} & \text{in } M \setminus \overline{\Omega} \\ \varphi = 0 & \text{on } \partial \Omega \\ \varphi(x) \to +\infty & \text{as } d(O, x) \to +\infty. \end{cases}$$
 (1.3.14)

The classical Bochner identity applied to φ in $(M \setminus \Omega, \tilde{g})$, combined with the first two equations of the above system, immediately yields the following identity

$$\Delta_{\tilde{g}} |\nabla \varphi|_{\tilde{g}}^{2} - \langle \nabla |\nabla \varphi|_{\tilde{g}}^{2} |\nabla \varphi\rangle_{\tilde{g}} = 2 \left[\text{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|_{\tilde{g}}^{2} \right], \tag{1.3.15}$$

where Ric is the Ricci tensor of the background metric g. Such a relation is at the heart of this work. As a first application, we state the following slight generalisation of Corollary 1.23 for solutions of (1.3.14) arising from solutions of (1.2.15).

Lemma 1.24. Let (M,g) be a nonparabolic Riemannian manifold with $\mathrm{Ric} \geq 0$, let $\Omega \subset M$ be a bounded open subset with smooth boundary. Assume that $\nabla |\nabla \varphi|_{\tilde{g}} = 0$ on $\{\varphi \geq s_0\}$ for some $s_0 \in [0,+\infty)$. Then the Riemannian manifold $(\{\varphi \geq s_0\},\tilde{g})$ is isometric to the Riemannian product $([s_0,+\infty) \times \{\varphi = s_0\}, d\rho \otimes d\rho + \tilde{g}_{\{|\varphi = s_0\}})$. In particular, $\partial\Omega$ is a connected totally geodesic submanifold inside $(M \setminus \Omega,\tilde{g})$.

Proof. It suffices to observe that plugging $\nabla |\nabla \varphi|_{\tilde{g}} = 0$ in (1.3.15) readily implies, since Ric ≥ 0 , that

$$\nabla \nabla \varphi \equiv 0$$
 in $\{\varphi \geq s_0\}$.

The isometry of $(M \setminus \Omega, \tilde{g})$ with the claimed Riemannian product then follows from Corollary 1.23.

We now briefly record some of the main relations among geometric quantities induced by the two metrics. We omit the computations, since they are straightforward and completely analogous to those carried out in [AM15] and [AM20]. First, observe that

$$|\nabla \varphi|_{\tilde{g}} = \frac{|\mathrm{D}u|}{u^{\frac{n-1}{n-2}}}.\tag{1.3.16}$$

Let H and $H_{\tilde{g}}$ be the mean curvatures of the level sets of u, that coincide with those of φ , respectively in the Riemannian manifold $(M \setminus \Omega, g)$ and in $(M \setminus \Omega, \tilde{g})$. They are computed using the unit normal vectors -Du/|Du| and $\nabla \varphi/|\nabla \varphi|_{\tilde{g}}$, respectively. Exploiting the g-harmonicity and \tilde{g} -harmonicity of u and φ , we obtain that

$$H = \frac{DDu(Du, Du)}{|Du|^3}, \qquad H_{\tilde{g}} = -\frac{\nabla\nabla\varphi(\nabla\varphi, \nabla\varphi)}{|\nabla\varphi|_{\tilde{g}}^3}.$$
 (1.3.17)

These quantities are related as follows

$$H_{\tilde{g}} = u^{-\frac{1}{n-2}} \left[H - \left(\frac{n-1}{n-2} \right) \frac{|Du|}{u} \right].$$
 (1.3.18)

Letting $d\sigma_{\tilde{g}}$ and $d\mu_{\tilde{g}}$ denote respectively the area and the volume elements naturally induced by \tilde{g} on $M \setminus \Omega$, we have

$$d\sigma_{\tilde{g}} = u^{\frac{n-1}{n-2}} d\sigma, \qquad d\mu_{\tilde{g}} = u^{\frac{n}{n-2}} d\mu. \qquad (1.3.19)$$

Clearly, the first identity in (1.3.19) has to be understood with respect to a hypersurface N in M. Finally, for every $\beta \geq 0$, define the conformal analogue of the U_{β} as the function $\Phi_{\beta}: [0, +\infty) \longrightarrow \mathbb{R}$ mapping

$$\Phi_{\beta}(s) = \int_{\{\varphi = s\}} |\nabla \varphi|_{\tilde{g}}^{\beta+1} d\sigma_{\tilde{g}}. \tag{1.3.20}$$

The functions U_{β} and Φ_{β} , and their derivatives are related to each other as follows

$$U_{\beta} = \Phi_{\beta}(-\log t),$$

$$-tU_{\beta}'(t) = \Phi_{\beta}'(-\log t),$$
(1.3.21)

for $0 < t \le 1$. The following theorems are the conformal versions of the Monotonicity-Rigidity Theorem 1.19.

Theorem 1.25. Let (M, g) be a nonparabolic Riemmanian manifold with Ric ≥ 0 . Let $\Omega \subset M$ be a bounded and open subset with smooth boundary, and let Φ_{β} be defined as in (1.3.20). Then, for every $\beta \geq (n-2)/(n-1)$, the function Φ_{β} is differentiable with derivative

$$\frac{\mathrm{d}\Phi_{\beta}}{\mathrm{d}s}(s) = -\beta \int_{\{\varphi=s\}} |\nabla \varphi|_{\tilde{g}}^{\beta} H_{\tilde{g}} \, \mathrm{d}\sigma_{g}, \qquad (1.3.22)$$

where H_g is the mean curvature of the level set $\{\varphi = s\}$ computed with respect to the unit normal vector field $v_{\tilde{g}} = \nabla \varphi / |\nabla \varphi|_{\tilde{g}}$. Moreover, for every $s \geq 0$, the derivative fulfils

$$\frac{d\Phi_{\beta}}{ds}(s) = -\beta e^{s} \int_{\{\varphi \geq s\}} \frac{|\nabla \varphi|_{\tilde{g}}^{\beta-2} \left\{ \operatorname{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|_{\tilde{g}}^{2} + (\beta - 2) |\nabla |\nabla \varphi|_{\tilde{g}}|_{\tilde{g}}^{2} \right\}}{e^{\varphi}} d\mu_{\tilde{g}}$$

$$= -\beta e^{s} \int_{\{\varphi \geq s\}} e^{-\varphi} |\nabla \varphi|_{\tilde{g}}^{\beta-2} \left\{ \operatorname{Ric}(\nabla \varphi, \nabla \varphi) + \left(\beta - \frac{n-2}{n-1}\right) |\nabla |\nabla \varphi|_{\tilde{g}}|_{\tilde{g}}^{2} + |\nabla \varphi|_{\tilde{g}}^{2} \right\} + |\nabla \varphi|^{2} \left| h_{\tilde{g}} - \frac{H}{n-1} \tilde{g}^{\top} \right|^{2} + \left(\frac{n-2}{n-1}\right) |\nabla^{\top} |\nabla \varphi||^{2} \right\} d\mu_{\tilde{g}}, \tag{1.3.23}$$

where the tangential elements are referred to the level sets of φ . In particular, $d\Phi_{\beta}/ds$ is always nonpositive. Moreover, $(d\Phi_{\beta}/ds)(s_0)=0$ for some $s_0\geq 0$ and some $\beta\geq (n-2)(n-1)$ if and only if $(\{\varphi\geq s_0\},\tilde{g})$ is isometric to the Riemannian product $([s_0,+\infty)\times\{\varphi=s_0\},d\rho\otimes d\rho+\tilde{g}_{\{|\varphi=s_0\}})$. In particular, $\{\varphi=s_0\}$ is a connected totally geodesic submanifold.

On the other hand, letting $\Phi_{\infty} : [0, +\infty) \to \mathbb{R}$ be defined by

$$\Phi_{\infty}(s) = \sup_{\{\varphi = s\}} |\nabla \varphi|, \tag{1.3.24}$$

the conformal version of Theorem 1.21 reads as follows.

Theorem 1.26. Let (M,g) be a nonparabolic Riemmanian manifold with $\text{Ric} \geq 0$. Let $\Omega \subset M$ be a bounded and open subset with smooth boundary. Then, the function $\Phi_{\infty} : [0,\infty) \to \mathbb{R}$ defined by (1.3.24) is monotone nonincreasing. Moreover, if $x_s \in \{\varphi = s\}$ is such that

$$|
abla arphi|(x_s) = \sup_{\{arphi = s\}} |
abla arphi|,$$

then

$$H(x_s) = -\frac{\partial}{\partial \nu_s} \log |\nabla \varphi|(x_s) \ge 0, \tag{1.3.25}$$

where v_s is the unit normal to $\{\varphi = s\}$ given by $v_s = D\varphi/|D\varphi|$. Moreover, equality holds in the inequality of (1.3.25) for some $s = s_0 \in [0, \infty)$ or $\Phi_\infty(S) = \Phi_\infty(s_0)$ for some $S > s_0$ if and only if $\{\varphi \geq s_0\}, \tilde{g}\}$ is isometric to the Riemannian product $\{s_0, +\infty\} \times \{\varphi = s_0\}, d\varphi \otimes d\varphi + \tilde{g}_{\{\varphi = s_0\}}\}$. In this case, in particular, $\partial\Omega$ is a connected totally geodesic submanifold.

We would like to highlight the full and striking formal analogy between the two statements above and Theorems 1.54 and 1.55, that are the Monotonicity-Rigidity Theorems for parabolic manifolds. As we are going to see later, the proofs also will follow the very same scheme.

Proofs of Theorems 1.19 and 1.21 after Theorems 1.25 and 1.26. Deducing formulas (1.3.3) and (1.3.4) from formulas (1.3.22) and (1.3.23) is just a matter of lengthy but straightforward computations carried out using the relations between g and \tilde{g} recalled above. We sketch the main steps. First, compute

$$|\nabla\nabla\varphi|_{\tilde{g}}^{2} = u^{-\frac{4}{n-2}} \left\{ \left| \frac{\mathrm{DD}u}{u} \right|^{2} + \frac{n(n-1)}{(n-2)^{2}} \left| \frac{\mathrm{D}u}{u} \right|^{4} - \frac{2n}{n-2} \left| \frac{\mathrm{D}u}{u} \right|^{3} \mathrm{H} \right\}, \tag{1.3.26}$$

where H is defined as in (1.3.17), and

$$\left|\nabla|\nabla\varphi|_{\tilde{g}}\right|_{\tilde{g}}^{2} = u^{-\frac{4}{n-2}} \left\{ \left|\frac{\mathrm{D}|\mathrm{D}u|}{u}\right|^{2} + \left(\frac{n-1}{n-2}\right)^{2} \left|\frac{\mathrm{D}u}{u}\right|^{4} - 2\left(\frac{n-1}{n-2}\right) \left|\frac{\mathrm{D}u}{u}\right|^{3} \mathrm{H} \right\}. \quad (1.3.27)$$

By (1.3.26) and (1.3.27), we can write

$$|\nabla\nabla\varphi|_{\tilde{g}}^{2} + (\beta - 2)|\nabla|\nabla\varphi|_{\tilde{g}}|_{\tilde{g}}^{2} = u^{-\frac{4}{n-2}} \left\{ \frac{|\mathrm{DD}u|^{2} - (\frac{n}{n-1})|\mathrm{D}|\mathrm{D}u||^{2}}{u^{2}} + \left(\beta - \frac{n-1}{n-2}\right) \left[\left| \frac{\mathrm{D}|\mathrm{D}u|}{u} \right|^{2} + \left(\frac{n-1}{n-2}\right)^{2} \left| \frac{\mathrm{D}u}{u} \right|^{4} - 2\left(\frac{n-1}{n-2}\right) \left| \frac{\mathrm{D}u}{u} \right|^{3} \mathrm{H} \right] \right\}.$$
(1.3.28)

Now, considering a orthonormal frame as $\{e_1, \dots, e_{n-1}, Du/|Du|\}$, where the first n-1 vectors are tangent to the level sets of u, we can decompose

$$\left|\frac{\mathbf{D}|\mathbf{D}u|}{u}\right|^2 = \left|\frac{\mathbf{D}u}{u}\right|^2 \mathbf{H}^2 + \sum_{j=1}^{n-1} \left\langle \frac{\mathbf{D}|\mathbf{D}u|}{u} \middle| e_j \right\rangle^2. \tag{1.3.29}$$

Plugging the above decomposition into (1.3.28), we obtain, with the aid of some algebra,

$$\begin{split} |\nabla\nabla\varphi|_{\tilde{g}}^{2} + (\beta - 2)|\nabla|\nabla\varphi|_{\tilde{g}}|_{\tilde{g}}^{2} &= u^{-\frac{4}{n-2}} \left\{ \left[|DDu|^{2} - \left(\frac{n}{n-1}\right)|D|Du||^{2} \right] + (\beta - \frac{n-2}{n-1}) |D^{\top}|Du||^{2} + (\beta - \frac{n-2}{n-1}) |Du|^{2} \left[H - \left(\frac{n-1}{n-2}\right)|D\log u| \right]^{2} \right\}, \\ &+ (\beta - \frac{n-2}{n-1}) |Du|^{2} \left[H - \left(\frac{n-1}{n-2}\right)|D\log u| \right]^{2} \right\}, \end{split}$$

$$(1.3.30)$$

where

$$\left| \mathbf{D}^{\top} |\mathbf{D}u| \right|^2 = \sum_{j=1}^{n-1} \left\langle \frac{\mathbf{D}|\mathbf{D}u|}{u} \middle| e_j \right\rangle^2.$$

The monotonicity formula (1.3.4) now follows easily by pluggin (1.3.7) into the right hand side of (1.3.30). On the other hand, (1.3.3) follows from (1.3.22) by an easy computation using directly (1.3.16), (1.3.18) and (1.3.19).

With regard to the relation between the monotonicity U_{∞} and Ψ_{∞} , they are obviously equivalent since $|\nabla \varphi|_{\tilde{g}} = |\mathrm{D}u|/u^{(n-1)/(n-2)}$. On the other hand, (1.3.6) follows from (1.3.25) using directly (1.3.16) and (1.3.18).

Assume now that $U_{\beta}'(t_0)=0$ or the inequality in (1.3.6) holds with equality sign at x_{t_0} for some $t_0\in(0,1]$ and some $\beta\geq(n-2)/(n-1)$. Then, by (1.3.21) or by (1.3.16) respectively, we deduce that $\Phi_{\beta}'(-\log t_0)=0$ or equality holds in the inequality of (1.3.25) respectively. Similarly, if $U_{\infty}(t_1)=U_{\infty}(t_0)$ then $\Phi_{\infty}(-\log t_0)=\Phi_{\infty}(-\log t_1)$. In all of these cases, the cylindrical rigidity statements of Theorems 1.25 and 1.26 are in force, and we are thus left to show that they imply the conical splitting of Theorems 1.19 and 1.21. To see this, observe first that that φ is an affine function of the coordinate s, and we write it as $\varphi=as+b$, for some constants a and b. Observe that $a\neq 0$ due to the nonconstancy of φ (that follows from (1.3.14)). By the definition of \tilde{g} given by (1.3.9), we have that the metric g on $\{u\leq t_0\}$, with $t_0=e^{-s_0}$ is

$$g = A e^{\frac{2s}{n-2}} \left(ds \otimes ds + \tilde{g}_{|\{\varphi=s_0\}} \right),$$

for some positive constant A. Define now the coordinate r by

$$\mathrm{d}r = \sqrt{\mathrm{A}}\,\mathrm{e}^{\frac{s}{n-2}}\,\mathrm{d}s,$$

so that

$$g = dr \otimes dr + \left(\frac{r}{r_0}\right)^2 g_{\{u=t_0\}}, \tag{1.3.31}$$

with $\{u \le t_0\} = \{r \ge r_0\}$, so that $\{r = r_0\} = \{u = t_0\}$, where we also used that by the constancy of the conformal factor in the definition of \tilde{g} on the level set of φ , the metric \tilde{g} and g are proportional when restricted to a level set of φ . The structure of g displayed in (1.3.31) is that of a (truncated) cone with cross section given by $\{u = t_0\}$, that in particular has Euclidean volume growth. We are then left to show that r_0 assumes the value claimed in (1.3.5). To compute it, observe that for $r_0 \le r \le R$, the set $\{r_0 \le r \le R\}$ is a geodesic annulus, and so we can treat $\{r = R\}$ as a geodesic sphere $\partial B(O, R)$ centered at some $O \in M$. We have

$$|\partial B(O,R)| = \int_{\partial B(O,R)} d\sigma = \int_{\{u=t_0\}} \left(\frac{R}{r_0}\right)^{n-1} \sqrt{\det g_{ij}^{\Sigma}} d\vartheta^1 \dots d\vartheta^{n-1} = \left(\frac{R}{r_0}\right)^{n-1} |\{u=t_0\}|,$$

so that we can explicitly express r_0 in terms of AVR(g) as in (1.3.5) by

$$AVR(g) = \lim_{R \to +\infty} \frac{|\partial B(O, R)|}{R^{n-1}|S^{n-1}|} = \frac{|\{u = t_0\}|}{r_0^{n-1}|S^{n-1}|},$$

where the first identity is a well known characterisation of AVR(g) in terms of areas of geodesic spheres instead of volumes of geodesic balls (see Subsection 1.4.1 , it follows easily by the standard proof of the Bishop-Gromov Theorem (see e.g. [Pet06]). The proof of the second claim is completed.

The computations we performed above show that the derivative of U_{β} (1.3.4) can also be written displaying the quantity on the right hand side of (1.3.30), that manifests the role of the Kato inequality in the monotonicity. This is actually how the monotonicity formula was presented in [AFM18]. Here, we preferred to plug (1.3.7) in order to ease the comparison with the (effective) monotonicity of the nonlinear counterpart worked out in Chapter 2. See in particular Theorem 2.23.

1.3.2 Proof of Theorems 1.25 and 1.26

In what follows, we are always referring to a background nonparabolic Riemannian manifold (M,g) with Ric ≥ 0 . Outside a bounded and open subset $\Omega \subset M$ with smooth boundary, we define the conformal metric \tilde{g} as in (1.3.9). In particular, with the same notations as in the previous subsection, $(M \setminus \Omega, \tilde{g}, \varphi)$ is a solution to (1.3.14).

We first prove (the conformal version of) Theorem 1.21, that is the Monotonicity-Rigidity Theorem for Φ_{∞} , since it is considerably easier and gives as a corollary the uniform boundedness of $|\nabla \varphi|$, a property that will be important also in the proof of (the conformal version of) Theorem 1.19.

Proof of Theorems 1.26. We first show that there exists a constant $C = C(g, \Omega) > 0$ such that $|\nabla \varphi|_{\tilde{g}} \leq C$ in $M \setminus \Omega$. Fixing a reference point O inside Ω , and letting $d(O, \cdot)$ be the distance from this point with respect to the metric g, we have, by (1.2.6), that

$$|\mathrm{D}u|(y) \le \mathrm{C}\frac{u(y)}{d(O,y)}$$

outside some ball containing Ω . Then,

$$|\nabla \varphi|_{\tilde{g}}(y) = \frac{|\mathrm{D}u|}{u^{\frac{n-1}{n-2}}}(y) \le C \frac{u^{-\frac{1}{n-2}}(y)}{d(O,y)} \le C C_1^{-\frac{1}{n-2}},$$

where in the last inequality we used (1.2.19). Consider now, for a given constant $\alpha > 0$, the auxiliary function

$$w_{\alpha} = |\nabla \varphi|_{\tilde{g}}^2 \mathrm{e}^{-\alpha \varphi}$$

Observe that by the just proved upper bound on $|\nabla \varphi|_{\tilde{g}}$ we have $w_{\alpha}(q) \to 0$ as $d(O,q) \to +\infty$ for any $\alpha > 0$. Moreover, a direct computation combined with (1.3.15) shows that w_{α} satisfies the identity

$$\Delta_{\tilde{g}} w_{\alpha} - (1 - 2\alpha) \left\langle \nabla w_{\alpha} \, | \, \nabla \varphi \right\rangle_{\tilde{g}} = 2 e^{-\alpha \varphi} \left[\text{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|_{\tilde{g}}^{2} \right] + \alpha (1 - \alpha) |\nabla \varphi|_{\tilde{g}}^{2} w_{\alpha}.$$

$$(1.3.32)$$

In particular, for any $\alpha \in (0,1)$ the right hand side above is nonnegative.

For the time being, we let s be a regular value. Then, applying the Maximum Principle to w_{α} , in force by (1.3.32), one gets that for every $y \in \{\varphi \ge s\}$ it holds

$$w_{\alpha}(y) \leq \sup_{\{\varphi=s\}} |\nabla \varphi|_{\tilde{g}}.$$

Letting $\alpha \to 0^+$ in the above inequality yields

$$|\nabla \varphi|_{\tilde{g}}(y) \leq \sup_{\{\varphi=s\}} |\nabla \varphi|_{\tilde{g}},$$

for every $y \in \{\varphi \geq s\}$. that in particular implies the monotonicity of Φ_{∞} . It is now easy to realise that

$$|\nabla \varphi|_{\tilde{g}}^{2} \mathbf{H}(x_{s}) = -\frac{1}{2} \frac{\partial}{\partial \nu_{s}} |\nabla \varphi|_{\tilde{g}}^{2} \ge 0, \tag{1.3.33}$$

where $x_s \in \{\varphi = s\}$ is the maximum point of $|\nabla \varphi|_{\tilde{g}}$ in $\{\varphi \geq s\}$. Indeed, the equality in (1.3.33) follows immediately from a direct computation with the second identity of (1.3.17), while the inequality immediately follows from the definition of x_s to be a maximum point.

Let us consider now the rigidity statements. Assume that $\Phi_{\infty}(S) = \Phi_{\infty}(s_0)$. Then, the maximum value of $|\nabla \varphi|_{\tilde{g}}$ in $\{\varphi \geq s_0\}$ is attained at some interior point $y \in \{\varphi \geq s_0\}$. Then, by the Strong Maximum Principle , $|\nabla \varphi|_{\tilde{g}}^2$ must be constant, since it satisfies

$$\Delta_{\tilde{g}} |\nabla \varphi|_{\tilde{g}}^2 - \langle \nabla |\nabla \varphi|_{\tilde{g}}^2 |\nabla \varphi\rangle_{\tilde{g}} \ge 0. \tag{1.3.34}$$

Lemma 1.24 then yields the desired conclusion.

On the other hand, if $|\nabla \varphi|_{\tilde{g}}^2$ is not constant in $\{\varphi \geq s\}$, the above strong maximum principle argument implies that x_s is a point of strict global maximum, and then the Hopf lemma (again in force due to (1.3.34)) immediately yields that (1.3.33) holds with a strict inequality sign.

We are left to discuss the case of singular level sets, that is, admitting a subset $Crit_s \subset \{\varphi = s\}$ (by [HS89] of Hausdorff dimension at most n-2, compare also with [CNV15]) where $|\nabla \varphi|_{\tilde{g}} = 0$. This situation is easily controlled by Sard's Theorem, ensuring that this can occur only for a set of values of measure zero. Indeed, assume that s is a critical value, that is, $\{\varphi = s\}$ is a critical level set, and let $y \in \{\varphi > s\}$. Then, by the Sard's Theorem recalled above, there exists a sequence of smooth level sets $\{\varphi = s_j\}$ with $s_j \to s$ as $j \to \infty$ satisfying $y \in \{\varphi > s_j\}$. By the proof given above for smooth level sets, we infer

$$|
abla arphi|_{\widetilde{\mathcal{S}}}^2(y) \leq \sup_{\{arphi = s_j\}} |
abla arphi|_{\widetilde{\mathcal{S}}}^2,$$

that through passing to the limit yields the monotonicity of Φ_{∞} also through singular level sets. If $\Phi_{\infty}(s_0) = \Phi_{\infty}(s_1)$ for some $s_1 > s_0$, with s_0 a possibly singular value, the monotonicity yields that $\Phi_{\infty}(s_0) = \Phi_{\infty}(s_j)$ for a sequence of $s_1 < s_j < s_0$ as above, implying that $|\nabla \varphi|_{\tilde{g}}$ is a positive constant on $\{\varphi \geq s_j\}$. In particular, by the continuity of $|\nabla \varphi|_{\tilde{g}}$, the level set $\{\varphi = s_0\}$ is actually not critical, and the splitting principle applies. Finally, observe that even for critical level sets $\{\varphi = s\}$, the point x_s lies outside Crits, and, in particular, (1.3.33) and the Hopf Lemma argument employed above for the rigidity case work substantially unchanged in this case.

A direct consequence is that the functions Φ_{β} 's are bounded.

Corollary 1.27. For every $\beta \geq 0$ the function $\Phi_{\beta} : [0, +\infty) \to \mathbb{R}$ defined in (1.3.20) is bounded as

$$\Phi_{eta}(s) \ \le \ \sup_{\partial\Omega} |
abla arphi|_{ ilde{g}}^{eta} \int\limits_{\partial\Omega} |
abla arphi|_{ ilde{g}} \, \mathrm{d}\sigma_{ ilde{g}} \,.$$

for every $s \in [0, +\infty)$.

Proof. Just observe that a simple application of the Divergence Theorem combined with the \tilde{g} -harmonicity of φ gives the constancy in s of the function

$$s \longmapsto \int_{\{\varphi=s\}} |\nabla \varphi|_{\tilde{g}} \, d\sigma_{\tilde{g}}.$$

Combining this observation with the boundon $|\nabla \varphi|_{\tilde{g}}$ following from Theorem 1.26, we have

$$\Phi_{\beta}(s) = \int\limits_{\{\varphi = s\}} |\nabla \varphi|_{\tilde{g}}^{\beta} \, |\nabla \varphi|_{\tilde{g}} \, d\sigma_{\tilde{g}} \, \leq \, \sup_{\partial \Omega} |\nabla \varphi|_{\tilde{g}}^{\beta} \int\limits_{\partial \Omega} |\nabla \varphi|_{\tilde{g}} \, d\sigma_{\tilde{g}} \, ,$$

for every $s \in [0, +\infty)$, as claimed.

A fundamental tool in the forthcoming computations, leading to the expression of the derivative of Φ_{β} in terms of a nonpositive integral (as in (1.3.23)), is the following Bochner-type identity.

Lemma 1.28 (Bochner-type identity). At every point where $|\nabla \varphi|_{\tilde{g}} \neq 0$, the following identity holds for every $\beta \geq 0$

$$\Delta_{\tilde{g}} |\nabla \varphi|_{\tilde{g}}^{\beta} - \left\langle \nabla |\nabla \varphi|_{\tilde{g}}^{\beta} \, \middle| \, \nabla \varphi \right\rangle_{\tilde{g}}^{\beta} = \beta |\nabla \varphi|_{\tilde{g}}^{\beta-2} \left[\operatorname{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|_{\tilde{g}}^{2} + (\beta - 2) |\nabla |\nabla \varphi|_{\tilde{g}}^{2} \right], \tag{1.3.35}$$

where the Ric is the Ricci tensor of the background metric g.

Proof. By a direct computation one gets

$$|\Delta_{\tilde{g}}|
ablaarphi|_{\tilde{g}}^{eta} = |
ablaarphi|_{\tilde{g}}^{eta-2} \left[\frac{eta}{2} \Delta_{\tilde{g}} |
ablaarphi|_{\tilde{g}}^2 + eta(eta-2) |
abla|
ablaarphi|_{\tilde{g}}^2 \right],$$

that, combined with (1.3.15), leads to (1.3.35).

We prove now an integral identity that will enable us to link the derivative of Φ_{β} to the Bochner-type formula above. Although, differently from the Euclidean case where u, and thus φ , is analytic, we cannot rely on the discreteness of the set of singular values of φ , the proof of such fundamental relation does not importantly differ from that of [AM20, Lemma 3.4]. We report it for the reader's sake. We also point out that a different proof was proposed in [AFM18], building on the fine estimates on the Minkowski content of the critical set provided in [CNV15]. However, that proof just recovered the monotonicity of U_{β} for $\beta \geq 1$, that was, anyway, still sufficient to provide the Willmore-type inequalities. In what follows, we let Crit $\varphi = \{x \in M \setminus \Omega \mid |\nabla \varphi| = 0\}$.

Lemma 1.29 (Fundamental Integral Identity for regular values). *Let* $0 \le s < S < +\infty$ *be regular values, that is,* Crit $\varphi \cap \{\varphi = s\} = \text{Crit } \varphi \cap \{\varphi = S\} = \emptyset$. *Then, the following identity*

holds

$$\int_{\{\varphi=S\}} \frac{\left\langle \nabla |\nabla \varphi|_{\tilde{g}}^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|_{\tilde{g}}} \right\rangle}{e^{S}} d\sigma_{\tilde{g}} - \int_{\{\varphi=s\}} \frac{\left\langle \nabla |\nabla \varphi|_{\tilde{g}}^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|_{\tilde{g}}} \right\rangle}{e^{S}} d\sigma_{\tilde{g}} =$$

$$= \beta \int_{\{g=S\}} \frac{|\nabla \varphi|_{\tilde{g}}^{\beta-2} \left[\text{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|_{\tilde{g}}^{2} + (\beta-2) |\nabla |\nabla \varphi|_{\tilde{g}}^{2} \right]}{e^{\varphi}} d\mu_{\tilde{g}},$$

$$\{s \leq \varphi \leq S\} \tag{1.3.36}$$

where the Ricci tensor is referred to the background metric g.

Proof. Since all the quantities that appear in this proof are referred to the metric \tilde{g} , except for Ric, which is the Ricci tensor of the background metric g, we shorten the notation, dropping the subscript \tilde{g} . We consider the vector field

$$X = \frac{\nabla |\nabla \varphi|^{\beta}}{e^{\varphi}},$$

that is well defined at every point where $|\nabla \varphi| \neq 0$. By (1.3.35), we have that wherever $|\nabla \varphi| \neq 0$ it holds

$$\operatorname{div} X = \frac{\Delta |\nabla \varphi|^{\beta} - \langle \nabla |\nabla \varphi|^{\beta}, \nabla \varphi \rangle}{e^{\varphi}}$$

$$= \beta \frac{|\nabla \varphi|^{\beta - 2} \left[\operatorname{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|^{2} + (\beta - 2) |\nabla |\nabla \varphi||^{2} \right]}{e^{\varphi}}.$$
(1.3.37)

Let *s* and *S* as in the statement, and consider, for $\varepsilon > 0$, the cut-off function χ_{ε} defined by

$$\begin{cases} \chi(t) = 0 & \text{in } t < \frac{1}{2}\varepsilon, \\ 0 < \dot{\chi}(t) \le 2\varepsilon^{-1} & \text{in } \frac{1}{2}\varepsilon \le t \le \frac{3}{2}\varepsilon, \\ \chi(t) = 1 & \text{in } t > \frac{3}{2}\varepsilon, \end{cases}$$
 (1.3.38)

and define accordingly the auxiliary vector field $X_{\varepsilon} = \chi_{\varepsilon}(|\nabla \varphi|^2)X$. Let also $N_{\delta} = \{|\nabla \varphi|^2 \leq \delta\}$ be the tubular neighbourhood of the critical set introduced in [AM20]. By the continuity of $|\nabla \varphi|^2$, for any ε small enough we have $\{\varphi = s\} \cap N_{\varepsilon} = \{\varphi = S\} \cap N_{3\varepsilon/2} = \emptyset$. Observing that $X_{\varepsilon} = X$ on $\{\varphi = s\}$ and $\{\varphi = S\}$ we get, using the Divergence Theorem

$$\int_{\{\varphi=S\}} \frac{\left\langle \nabla |\nabla \varphi|^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle}{e^{S}} d\sigma - \int_{\{\varphi=s\}} \frac{\left\langle \nabla |\nabla \varphi|^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle}{e^{S}} d\sigma = \int_{\{s \leq \varphi \leq S\}} \operatorname{div} X_{\varepsilon} d\mu$$

$$= \int_{\{s \leq \varphi \leq S\}} \chi_{\varepsilon} \operatorname{div} X d\mu + \int_{N_{3\varepsilon/2} \setminus N_{\varepsilon/2}} \left\langle \nabla \chi_{\varepsilon} | X \right\rangle d\mu$$

$$= \int_{\{s \leq \varphi \leq S\}} \chi_{\varepsilon} \operatorname{div} X d\mu + \beta \int_{N_{3\varepsilon/2} \setminus N_{\varepsilon/2}} \dot{\chi}_{\varepsilon} (|\nabla \varphi|^{2}) \frac{|\nabla \varphi|^{\beta-2} |\nabla |\nabla \varphi|^{2}|^{2}}{2e^{\varphi}} d\mu.$$
(1.3.39)

By (1.3.37) and (1.3.7), the integrand in the first term of the rightmost hand side is non-negative, and thus, by Monotone Convergence Theorem, its integral converges to the right hand side of (1.3.36). One is thus left to show that the second term in the rightmost hand side of (1.3.39) vanishes as $\varepsilon \to 0^+$. To see this, observe that by the coarea formula and the second property in (1.3.38) of χ_{ε}

$$\int\limits_{N_{3\varepsilon/2}\setminus N_{\varepsilon/2}}\dot{\chi}_{\varepsilon}(|\nabla\varphi|^2)\frac{|\nabla\varphi|^{\beta-2}|\nabla|\nabla\varphi|^2|^2}{2e^{\varphi}}\,\mathrm{d}\mu\leq \frac{1}{\varepsilon}\int\limits_{\varepsilon/2}^{3\varepsilon/2}s^{(\beta-2)/2}\int\limits_{\{|\nabla\varphi|^2=s\}}\frac{|\nabla|\nabla\varphi|^2|}{\mathrm{e}^{\varphi}}\,\mathrm{d}\sigma\,\mathrm{d}s.$$

By means of the Mean Value Theorem, it then suffices to show that the function $r^{(\beta-2)/2}F(r)$, with

$$F(r) = \int_{\{|\nabla \varphi|^2 = r\}} \frac{|\nabla |\nabla \varphi|^2|}{e^{\varphi}} d\sigma,$$

vanishes as $r \to 0^+$. We prove that this is actually the case if $\beta > (n-2)/(n-1)$. Indeed, observe that

$$\begin{split} F(r) &= \int\limits_{\partial N_r} \frac{\left\langle \nabla |\nabla \varphi|^2 \right\rangle}{e^{\varphi}} \, \mathrm{d}\sigma = \int\limits_{N_r} \mathrm{div} \left(\frac{\nabla |\nabla \varphi|^2}{e^{\varphi}} \right) \, \mathrm{d}\mu = 2 \int\limits_{N_r} \frac{|\nabla \nabla \varphi|^2}{e^{\varphi}} \, \mathrm{d}\mu \\ &= \int\limits_{0}^{r} \int_{\{|\nabla \varphi|^2 = s\}} \frac{|\nabla \nabla \varphi|^2}{e^{\varphi} |\nabla |\nabla \varphi|^2|} \, \mathrm{d}\sigma \, \mathrm{d}s. \end{split}$$

The above relation implies in particular that F is absolutely continuous. Then, taking derivatives, and exploiting the refined Kato's inequality provided by (1.3.7), we get

$$F'(r) = 2 \int\limits_{\{|\nabla \varphi|^2 = r\}} \frac{|\nabla \nabla \varphi|^2}{e^{\varphi} |\nabla |\nabla \varphi|^2|} \, \mathrm{d}\sigma \ge 2 \left(\frac{n}{n-1}\right) \int\limits_{\{|\nabla \varphi|^2 = r\}} \frac{|\nabla |\nabla \varphi||}{e^{\varphi} |\nabla |\nabla \varphi|^2|} \, \mathrm{d}\sigma = \frac{1}{2} \frac{n}{n-1} \frac{F(r)}{r}.$$

for almost any r > 0. Integrating, we get, for R > r,

$$\frac{F(r)}{r^{\frac{n}{2(n-1)}}} \le \frac{F(R)}{R^{\frac{n}{2(n-1)}}}.$$

Then, we deduce that for $\beta > (n-2)/(n-1)$ the function $r^{(\beta-2)/2}F(r)$ vanishes as $r \to 0^+$. To complete the proof of the present Lemma also for $\beta = (n-2)/(n-1)$, it suffices to pass to the limit as $\beta \to (n-2)/(n-1)$ from above in (1.3.36), employing the Dominated Convergence Theorem on the one side and the Monotone Convergence Theorem on the other.

Building on the Fundamental Integral Identity for regular values proved in Lemma 1.29, we are now ready to deduce the following monotonicity result, holding up also at critical level sets. Here, it becomes transparent the key role played by the density of regular values, ensured by the Sard Theorem, a property that we are going to miss in the nonlinear case treated in the sequel of this thesis.

Theorem 1.30 (Monotonicity of $e^{-s}\Phi'_{\beta}(s)$). Let $\beta \geq (n-2)/(n-1)$. The function Φ_{β} defined in (1.3.20) is differentiable for any $s \geq 0$, and its derivative satisfies

$$\Phi_{\beta}'(s) = \int_{\{\varphi = s\}} \left\langle \nabla |\nabla \varphi|_{\tilde{g}}^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|_{\tilde{g}}} \right\rangle d\sigma_{\tilde{g}} = -\beta \int_{\{\varphi = s\}} |\nabla \varphi|_{\tilde{g}}^{\beta} H_{\tilde{g}} d\sigma_{\tilde{g}}.$$
 (1.3.40)

In particular, for any $0 \le s < S < \infty$ *, we have*

$$e^{-S}\Phi_{\beta}'(S) - e^{-s}\Phi_{\beta}'(s) = \int_{\{s \leq \varphi \leq S\}} |\nabla \varphi|_{\tilde{g}}^{\beta-2} \left[\operatorname{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|_{\tilde{g}}^{2} + (\beta - 2) |\nabla |\nabla \varphi|_{\tilde{g}}^{2} \right] e^{-\varphi} d\mu_{\tilde{g}},$$

$$(1.3.41)$$

where the Ricci tensor is referred to the background metric g.

Proof. Let us drop the subscript \tilde{g} . Let $s_0 \in [0, +\infty)$, and $s \geq s_0$ be possibly singular values of φ . We first claim that

$$\Phi_{\beta}(s) - \Phi_{\beta}(s_0) = \int_{\{s_0 \le \varphi \le s\}} \operatorname{div}(|\nabla \varphi|^{\beta} \nabla \varphi) \, \mathrm{d}\mu = \int_{\{s_0 \le \varphi \le s\}} \left\langle \nabla |\nabla \varphi|^{\beta} \middle| \nabla \varphi \right\rangle \, \mathrm{d}\mu. \tag{1.3.42}$$

To see this, consider again the tubular neighbourhoods N_{ε} considered in the proof of Lemma 1.29 above. Applying the Divergence Theorem to the vector field $Y_{\beta} = |\nabla \varphi|^{\beta} \nabla \varphi$ in the set $\{s_0 \leq \varphi \leq s\} \setminus N_{\varepsilon}$ we get

$$\int_{\{s_0 \le \varphi \le s\} \setminus N_{\varepsilon}} \operatorname{div} \left(|\nabla \varphi|^{\beta} \nabla \varphi \right) d\mu = \int_{\{\varphi = s\} \setminus N_{\varepsilon}} |\nabla \varphi|^{\beta+1} d\sigma - \int_{\{\varphi = s_0\} \setminus N_{\varepsilon}} |\nabla \varphi|^{\beta+1} d\sigma + \int_{\partial N_{\varepsilon} \cap \{s_0 \le \varphi \le s\}} |\nabla \varphi|^{\beta} \left\langle \nabla \varphi \left| \frac{\nabla |\nabla \varphi|^2}{|\nabla |\nabla \varphi|^2|} \right\rangle d\mu. \right) \tag{1.3.43}$$

Observing now that

$$\operatorname{div}(|\nabla \varphi|^{\beta} \nabla \varphi) = \left\langle \nabla |\nabla \varphi|^{\beta} \middle| \nabla \varphi \right\rangle = \beta |\nabla \varphi|^{\beta} \nabla \nabla \varphi \left(\frac{\nabla \varphi}{|\nabla \varphi|}, \frac{\nabla \varphi}{|\nabla \varphi|} \right),$$

we deduce that the integrand in the left hand side of (1.3.43) is uniformly bounded, and thus by Dominated Convergence Theorem we deduce that its integral converges as $\varepsilon \to 0^+$ to the right hand side of (1.3.42). By Monotone Convergence Theorem , it is also easy to see that the first two terms in the right hand side of (1.3.43) converges to the left hand side of (1.3.42). We are thus left to show that the last term in (1.3.43) vanishes as $\varepsilon \to 0$. In fact, since $|\nabla \varphi|^2 = \varepsilon$ on ∂N_ε , we have

$$\int\limits_{\partial N_{\varepsilon}\cap \{s_{0}\leq \varphi\leq s\}} |\nabla \varphi|^{\beta} \left\langle \nabla \varphi \left| \frac{\nabla |\nabla \varphi|^{2}}{|\nabla |\nabla \varphi|^{2}|} \right\rangle \, \mathrm{d}\mu \leq \varepsilon^{(\beta+1)/2} |\partial N_{\varepsilon} \cap \{s_{0}\leq \varphi\leq s\}|,$$

that tends to 0 as $\varepsilon \to 0^+$, as desired.

We now aim at showing that the function mapping

$$s \to I(s) = e^{-s} \int_{\varphi=s} \left\langle \nabla |\nabla \varphi|^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma,$$

a priori defined just for almost any $s \in [0, \infty)$, can be extended to a continuous function defined on the whole $[0, +\infty)$. Letting D be the set of regular values of φ , it is well known that it suffices to show that I is *uniformly* continuous on $D \cap [s, S]$ for $0 \le s < S < +\infty$ to ensure the existence of such an extension in [s, S]. To see this, let $s_0 \in D \cap [s, S]$, and let $s_j \in D \cap [s, S]$ be a sequence of regular values converging to s_0 . Let us assume for simplicity that $s_j > s_0$ for any j, the general case displaying no additional difficulties. By (1.3.36), we have

$$\frac{|\nabla \varphi|^{\beta-2} \left[\mathrm{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|^2 + (\beta-2) \big| \nabla |\nabla \varphi| \big|^2 \right]}{e^{\varphi}} \in L^1(\{s \leq \varphi \leq S\}).$$

In particular, for any $\varepsilon > 0$ and $\tilde{s} \in D \cap [s, S]$, there exists $\delta = \delta(\varepsilon, \tilde{s})$ such that

$$\int\limits_{\{\tilde{s}-\delta\leq \varphi\leq \tilde{s}+\delta\}} \frac{|\nabla\varphi|^{\beta-2}\left[\mathrm{Ric}(\nabla\varphi,\nabla\varphi)+|\nabla\nabla\varphi|^2+(\beta-2)\big|\nabla|\nabla\varphi|\big|^2\right]}{e^{\varphi}}\,\mathrm{d}\mu\leq \varepsilon.$$

By the density of regular values, the union of open intervals $(\tilde{s} - \delta, \tilde{s} + \delta)$ as \tilde{s} ranges in $D \cap [s, S]$, covers the whole compact [s, S], and thus we can extract a *finite* number of intervals $(\tilde{s_i} - \delta_i, \tilde{s_i} + \delta_i)$. Let then $\delta = \min_i \delta_i > 0$, and let us come back to our sequence of regular values s_j converging to s_0 . It is now clear that, if $s_j \in (s_0, s_0 + \delta)$, by the above discussion and Lemma 1.29,

$$I(s_j) - I(s_0) = \int\limits_{\{s_0 \leq \varphi \leq s_0 + \delta\}} \frac{\left|\nabla \varphi\right|^{\beta - 2} \left[\mathrm{Ric}(\nabla \varphi, \nabla \varphi) + \left|\nabla \nabla \varphi\right|^2 + (\beta - 2) \left|\nabla \left|\nabla \varphi\right|\right|^2 \right]}{e^{\varphi}} \, \mathrm{d}\mu \leq \varepsilon.$$

The radius δ being independent of x_0 shows that I is uniformly continuous on $D \cap [s, S]$. Since this holds for any interval [s, S], we can extend I to a continuous function on $[0, +\infty)$, that we are still calling I(s). Clearly, by (1.3.36), this function satisfies

$$I(S) - I(s) = \int_{\{s \le \varphi \le S\}} \frac{|\nabla \varphi|^{\beta - 2} \left[\text{Ric}(\nabla \varphi, \nabla \varphi) + |\nabla \nabla \varphi|^2 + (\beta - 2) |\nabla |\nabla \varphi||^2 \right]}{e^{\varphi}} d\mu$$

for any $0 \le s \le S < \infty$. Moreover, with this terminology, the function $e^{(\cdot)}I(\cdot)$ is a continuous function on $[0,\infty)$ such that

$$e^{s}I(s) = \int\limits_{\{\varphi=s\}} \left\langle \nabla |\nabla \varphi|^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma \right.$$

everytime *s* is a regular value for φ .

Finally, apply the coarea formula in (1.3.42), to get

$$\Phi_{eta}(s) - \Phi_{eta}(s_0) \; = \int\limits_{s_0}^s \mathrm{d} au \int\limits_{\{arphi = au\}} \left\langle
abla |
abla arphi|^eta \, \left| rac{
abla arphi}{|
abla arphi|}
ight
angle \, \mathrm{d}\sigma \, .$$

By the Fundamental Theorem of Calculus, the continuity of (the extension of) the mapping

$$\tau \longmapsto e^{\tau} I(\tau) = \int_{\{\varphi = \tau\}} \left\langle \nabla |\nabla \varphi|^{\beta} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma \right.$$

proved above implies the differentiability of Φ_{β} , together with the first identity in (1.3.40). The second one follows from the first just by a direct computation involving (1.3.17). The proof is completed.

We are now in a position to complete the proof of Theorem 1.25.

Conclusion of the proof of Theorem 1.25. We are going to pass to the limit as $S \to +\infty$ in (1.3.41). The following argument is due to Colding and Minicozzi, [CM14b]. We first prove that the derivative of Φ_{β} has a sign. By (1.3.41) we have that, for every $\beta \geq 1$ and every $0 \leq s < S < +\infty$,

$$\Phi'_{\beta}(S) \geq e^{(S-s)}\Phi'_{\beta}(s).$$

Integrating the above differential inequality, we get

$$\Phi_{\beta}(S) \ge \left(e^{(S-s)} - 1\right) \Phi_{\beta}'(s) + \Phi_{\beta}(s).$$
(1.3.44)

for every $0 \le s < S < +\infty$. Assume now, by contradiction, that $\Phi'_{\beta}(s) > 0$ for some $s \in [0, +\infty)$. Then, passing to the limit as $S \to +\infty$ in (1.3.44), we would get $\Phi_{\beta}(S) \to +\infty$, against the boundedness of Φ_{β} provided in Corollary 1.27. Thus, $\Phi'_{\beta}(s) \ge 0$ for every $s \in [0, +\infty)$. Therefore Φ_{β} is a nonincreasing, differentiable bounded function on $[0, +\infty)$, and in particular, $\Phi'_{\beta}(S) \to 0$ as $S \to +\infty$, possibly along a subsequence. Passing to the limit as $S \to +\infty$ in (1.3.41) finally gives the monotonicity formula (1.3.23).

For the rigidity statement, assume now that $\Phi'_{\beta}(s_0)=0$ for some $s_0\in[0,+\infty)$. Then, if, $\beta>(n-2)/(n-1)$ we see from (1.3.23) that $\left|\nabla|\nabla\varphi|_{\tilde{g}}\right|_{\tilde{g}}=0$, and we can thus conclude by Lemma 1.24. If on the other hand $\beta=(n-2)/(n-1)$, (1.3.23) still shows that the terms on the right hand side of (1.3.7) vanish, and so the related rigidity statement ensures that $(\{\varphi\geq s_0\},\tilde{g})$ is a warped product. In particular, the mean curvature of $\{\varphi=s\}$ depends only on s for any $s\geq s_0$. Consequently, (1.3.22) shows us that they all have zero mean curvature. By (1.3.17), we deduce in particular that $|\nabla^{\perp}|\nabla\varphi|_{\tilde{g}}|_{\tilde{g}}=0$, that, coupled with $|\nabla^{\perp}|\nabla\varphi|_{\tilde{g}}|_{\tilde{g}}=0$ following from the vanishing of (1.3.23), implies that $|\nabla|\nabla\varphi|_{\tilde{g}}|_{\tilde{g}}=0$, and we can conclude as before appealing to Lemma 1.24.

1.4 Long time behaviour of the electrostatic potential

In this section, we are going to prove the Willmore inequality on manifolds with non-negative Ricci curvature and Euclidean volume growth, using the geometric features of the capacitary potential u of a given bounded domain with smooth boundary Ω . For the

reader's convenience we recall that u is a solution to the following problem

$$\begin{cases} \Delta u = 0 & \text{in } M \setminus \overline{\Omega} \\ u = 1 & \text{on } \partial\Omega \\ u(y) \to 0 & \text{as } d(O, y) \to +\infty, \end{cases}$$

whose existence in the present context follows from Theorem 1.14. As sketched in the Introduction, the proof makes use of the global features of the Monotonicity-Rigidity Theorem, that is, it compares the behaviour of U_{β} in the large with $U_{\beta}(1)$. The well known asymptotics of u and of its gradient in the Euclidean case were the crucial tool to compute the limits of U_{β} in [AM15], that consequently gave the sharp lower bound on the Willmore-type functional. On manifolds with nonnegative Ricci curvature and Euclidean volume growth, the work of Colding-Minicozzi [CM97b] actually implies that the asymptotic behaviour of the potential is completely analogous to that in \mathbb{R}^n . However, as we will clarify in Remark 1.39, there is no hope to get an Euclidean-like pointwise behaviour of Du in the general case. Nevertheless, using techniques developed in the celebrated [CC96], we are able to achieve asymptotic integral estimates for the gradient that in turn will let us conclude the proof of the Willmore-type inequalities (10).

We introduce the (normalised) capacity of Ω , that is

$$\operatorname{Cap}(\Omega) = \inf \left\{ \frac{1}{(n-2)|\mathbb{S}^{n-1}|} \int_{\mathbb{R}^n} |\mathrm{D}f|^2 \, \mathrm{d}\mu \, \middle| \, f \in \mathscr{C}^{\infty}_{c}(\mathbb{R}^n), f \geq 1 \text{ on } \Omega \right\}. \tag{1.4.1}$$

Such notion will arise here in the following form

$$\operatorname{Cap}(\Omega) = \frac{\int_{\partial\Omega} |\operatorname{D}u| \, \mathrm{d}\sigma}{(n-2)|S^{n-1}|}, \tag{1.4.2}$$

written in terms of the capacitary potential u of Ω . We address the reader to the proof of Theorem B.1 for the equivalence between (1.4.1) and (1.4.2).

1.4.1 Manifolds with Euclidean volume growth

A foundational result in comparison geometry, that we did not explicitly employed yet, is the Bishop-Gromov Theorem. In order to ease the reader with the conventions we adopt, we recall the classical statement in the case of nonnegative Ricci curvature.

Theorem 1.31 (Bishop-Gromov Theorem). Let (M, g) be a complete noncompact Riemannian manifold with Ric ≥ 0 . Then, for any point $x \in M$, the function $\Theta_x : (0, +\infty) \to \mathbb{R}$ defined by

$$\Theta_x(r) = \frac{|B(x,r)|}{|\mathbb{B}^n|r^n},$$

is monotone nonincreasing. Moreover, the limit as $r \to +\infty$ of Θ_x is independent of x, and it satisfies

$$\lim_{r\to\infty}\Theta_x(r)=1$$

for some $x \in M$ if and only if (M,g) is isometric to \mathbb{R}^n with the standard metric.

The above Theorem justifies the following commonly used Definition.

Definition 1.32 (Asymptotic Volume Ratio and Euclidean volume growth). *Let* (M, g) *be a complete noncompact Riemannian manifold with* Ric ≥ 0 , *and let* $O \in M$. *Then we define its*

Asymptotic Volume Ratio $AVR(g) \in [0, 1]$ as

$$AVR(g) = \lim_{r \to \infty} \frac{|B(O, r)|}{|\mathbb{R}^n| r^n}.$$

We say that (M, g) has Euclidean volume growth if AVR(g) > 0.

If follows at once that manifolds with Euclidean volume growth satisfy (1.2.3), and thus from Varopoulos' characterisation manifolds with Euclidean volume growth are non-parabolic.

Let us also recall the well known fact that in a noncompact complete Riemannian manifold (M,g) with Ric ≥ 0 also the function

$$(0,+\infty)\ni r\longmapsto \vartheta_x(r)=\frac{|\partial B(O,r)|}{r^{n-1}|\mathbb{S}^{n-1}|}$$

is monotone nonincreasing for any $x \in M$, and the asymptotic volume ratio satisfies

$$AVR(g) = \lim_{r \to +\infty} \frac{|\partial B(O, r)|}{r^{n-1}|S^{n-1}|}.$$
 (1.4.3)

The proof of the monotonicity of $\vartheta_x(r)$ is actually the main step in the classical proof of the Bishop-Gromov Theorem, see e.g. the proof given for [CLN06, Theorem 1.132] or that proposed for Theorem 3.1 in the survey [Zhu97], while the validity of (1.4.3) is easily checked through an application of de l'Hôpital rule.

Let us point out an obvious but very important feature of the Bishop-Gromov Theorem and of the Euclidean volume growth assumption, that is

$$AVR(g)|\mathbb{B}^{n}|r^{n} \le |B(x,r)| \le |\mathbb{B}^{n}|r^{n}$$
 and $AVR(g)|\mathbb{S}^{n-1}|r^{n-1} \le |\partial B(x,r)| \le |\mathbb{S}^{n-1}|r^{n-1}$. (1.4.4)

The importance of the above bounds lie also in the fact that the constants in front of r^n do not depend on the base point x. We are explicitly making use of this fact in the computation of the limit of $U_{\infty}(t)$ as $t \to 0^+$, see the proof of Proposition 1.43.

In the remainder of this section we are repeatedly going to use both the area and the volume formulations of the Bishop-Gromov Theorem without always mentioning them.

Rough estimates for the electrostatic potential

Let us now derive some rough estimates for the electrostatic potential u and its gradient Du holding on manifolds with nonnegative Ricci curvature and Euclidean volume growth, to be refined later. Here, and in the sequel, we set r(x) = d(O, x), where $O \in \Omega$.

Proposition 1.33. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth. Then, it is nonparabolic and the solution u to problem (1.2.15) for some bounded and open Ω with smooth boundary satisfies

$$C_1 r^{2-n}(x) \le u(x) \le C_2 r^{2-n}(x)$$
 (1.4.5)

on $M \setminus \Omega$ for some positive constants C_1 and C_2 depending on M and Ω . Moreover, if $\Omega \subset B(O,R)$, it holds

$$|Du|(x) \le C_3 r^{1-n}(x),$$
 (1.4.6)

on $M \setminus B(O, 2R)$ with $C_3 = C_3(M, \Omega) > 0$.

Proof. The first inequality in (1.4.5) is just (1.2.19). To obtain the second one, recall from (1.4.4) that

$$|B(O,r)| \ge \left(n|\mathbb{S}^{n-1}|\text{AVR}(g)\right)r^n.$$

Then, the second inequality in the Li-Yau estimate (1.12), combined with the second inequality in (1.2.18) completes the proof of (1.4.5). Finally, inequality (1.4.6) is achieved just by plugging the upper estimate on u given by (1.4.5) into (1.2.5).

1.4.2 Asymptotics for *u* and its gradient

The behaviour at infinity of u can be deduced along the path indicated in [CM97b]. However, we prefer to present here a simplified version of that proof, taking advantage of some of the refinements provided in [LTW97]. Let us first recall the following asymptotic behaviour of G (see [CM97b, Theorem 0.1], or [LTW97, Theorem 1.1] for a completely different proof),

$$\lim_{r(x) \to +\infty} \frac{G(O, x)}{r(x)^{2-n}} = \frac{1}{\text{AVR}(g)}.$$
 (1.4.7)

Building on this fact, we deduce precise asymptotics for the capacitary potential of Ω .

Lemma 1.34. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth, and let u be a solution to problem (1.2.15). Then

$$\lim_{r(x)\to +\infty} \frac{u(x)}{r(x)^{2-n}} = \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)}.$$
 (1.4.8)

Proof. By [LTW97, Theorem 1.2], that slightly extends [CM97b, Theorem 0.3], we have that outside some large ball B(O, R) containing Ω

$$u = -\frac{G}{(n-2)|\mathbb{S}^{n-1}|} \int_{\partial B(O,R)} \frac{\partial u}{\partial \nu} d\sigma + \nu, \qquad (1.4.9)$$

where *G* is the Green's function with pole in O, ν is the exterior unit normal to the boundary of B(O, R) and v is a harmonic function defined in $M \setminus B(O, R)$ satisfying

$$|v| \le C \frac{G}{r} \tag{1.4.10}$$

for some constant C > 0. We point out that the Green's function considered in [LTW97] is, in our notation, $G/((n-2)|S^{n-1}|)$. By the Divergence Theorem and the harmonicity of u, we infer that

$$\int_{\partial B(O,R)} \frac{\partial u}{\partial \nu} d\sigma + \int_{\partial \Omega} \frac{\partial u}{\partial \nu} d\sigma = \int_{B(O,R)\backslash \Omega} \Delta u d\mu = 0,$$

where we denote by ν the exterior unit normal to the boundary of $B(O, R) \setminus \overline{\Omega}$. Since, on $\partial \Omega$, $\nu = Du/|Du|$, we get, by the above identity, that

$$-\frac{\int_{\partial B(O,R)} \frac{\partial u}{\partial \nu} d\sigma}{(n-2)|S^{n-1}|} = \frac{\int_{\partial \Omega} |Du| d\sigma}{(n-2)|S^{n-1}|} = \operatorname{Cap}(\Omega).$$
 (1.4.11)

Dividing both sides of (1.4.9) by r^{2-n} and passing to the limit as $r \to +\infty$ taking into account (1.4.7), (1.4.11) and (1.4.10), we get the claim.

As a straightforward corollary of the above lemma, we compute the rescaled area of large geodesic spheres in M.

Corollary 1.35. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth, and let u be a solution to (1.2.15). Then

$$\lim_{R_i \to +\infty} \int_{\partial B(O,R_i)} u^{\frac{n-1}{n-2}} d\sigma = |S^{n-1}| \operatorname{AVR}(g) \left(\frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \right)^{\frac{n-1}{n-2}}$$
(1.4.12)

Proof. It follows from Lemma 1.34 and (1.4.3).

Building on the above formula (1.4.8) for the asymptotics of u, we now derive integral asymptotics for Du. They are achieved through an adaptation of the methods used in [CC96, Section 4]. Similar estimates have been widely considered in literature (see e.g. [Col12] [CM97b], [CM97a], [CM14a]). We are providing here a complete and self-contained proof.

In the computations below, we are going to consider first and second derivatives of the distance function $x \mapsto d(O, x)$. This can be readily justified by approximating the distance function by convolution, by performing the computations below for the approximating sequence and finally passing to the limit. A scheme like this, often implicit in the aforementioned literature, has been carried out in details for example in [LTW97]. We state this underlying approximation lemma, a complete proof of which can be obtained following a standard regularisation argument as in [Col97, Lemma 1.4].

Lemma 1.36 (Smooth approximation of the distance function). Let (M,g) be a complete Riemannian manifold. Let $O \in M$, and let $K \subset M$ be a compact set such that $O \in M \setminus K$. Let r(x) = d(O,x), and let $f:(0,+\infty) \to \mathbb{R}$ be a smooth function. Moreover, assume that $\Delta(f \circ r) \leq h \circ r$ in the sense of distributions for some smooth function $h:(0,+\infty) \to \mathbb{R}$. Then, there exists a sequence of smooth functions $\{f_j\}_{j\in\mathbb{N}}$ such that for any $\varepsilon > 0$ there exists a $j_0 = j_0(K,h)$ such that

(i) for any $j \geq j_0$ we have

$$|f_j(x) - f(r(x))| \le \varepsilon$$

for any $x \in K$,

(ii) for any $j \ge j_0$ we have

$$|\mathrm{D}f_i(x)| \le |\mathrm{D}f(r(x))| + \varepsilon$$

for any $x \in K$ and $Df_i(x) \to Df(r(x))$ for almost any $x \in K$,

(iii) for any $j \geq j_0$ we have

$$\Delta f_i(x) \le h(r(x)) + \varepsilon$$

for any $x \in K$.

The above lemma will be applied mostly with $f(t) = -t^{2-n}$ and h(t) = 0, for $t \in (0, +\infty)$, so that (iii) approximates the well known relation $\Delta r^{2-n} \ge 0$ in the sense of distribution, already observed in (1.2.14). Observe also that (ii) easily allows to pass to the limit under the integral sign when approximating the gradient of functions of the distance. Moreover, we are also going to repeatedly apply the Divergence Theorem to vector fields depending on (approximations of) the distance function on annuli

 $B(O, kR) \setminus B(O, R)$ with k > 1. We can do this since by the coarea formula the intersection of the cut locus with geodesic spheres is (n - 1)-negligible for almost any value of the radius. We are actually applying the Divergence Theorem to annuli bounded by such geodesic spheres (we implicitly used this observation also in the proof of Lemma 1.34). Alternatively, we can refer to the much more general weak Sard-type property of Lipschitz functions object of [ABC11].

In order to keep the presentation as transparent as possible, we are going to work directly with derivatives of the distance function, having in mind a repeated use of Lemma 1.36 as hinted above, and on geodesic annuli allowing us to apply the Divergence Theorem as just illustrated.

Proposition 1.37 (Integral Asymptotics for Du). Let (M, g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth, and let u be a solution to problem (1.2.15). Then, for every k > 1, it holds

$$\lim_{R \to +\infty} \frac{R^{2n-2}}{|A_{R,kR}|} \int_{A_{R,kR}} \left| Du - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \operatorname{D} r^{2-n} \right|^2 d\mu = 0, \qquad (1.4.13)$$

where, for R > 0, we set $A_{R,kR} = B(O,kR) \setminus \overline{B(O,R)}$.

Proof. A simple integration by parts combined with the harmonicity of u leads to the following identity

$$\int_{A_{R,kR}} \left| Du - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \operatorname{D} r^{2-n} \right|^{2} d\mu = -\int_{A_{R,kR}} \left(\frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \Delta r^{2-n} \right) \left(u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{2-n} \right) d\mu + \int_{\partial A_{R,kR}} \left(u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{2-n} \right) \left\langle \operatorname{D} u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \operatorname{D} r^{2-n} \middle| \nu \right\rangle d\sigma.$$
(1.4.14)

Let us estimate separately the integrals on the right hand side of the above identity. Let $\varepsilon > 0$. Then, by (1.4.8),

$$\left|\frac{u}{r^{2-n}} - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)}\right| < \varepsilon.$$

for r big enough. We have, for R big enough

$$\left| \int_{A_{R,kR}} \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \Delta r^{2-n} \right) \left(u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{2-n} \right) d\mu \right| \leq$$

$$\leq \int_{A_{R,kR}} \left| \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \Delta r^{2-n} \right| \left| \frac{u}{r^{2-n}} - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \right| r^{2-n} d\mu$$

$$\leq \varepsilon R^{2-n} \int_{A_{R,kR}} \left| \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \Delta r^{2-n} \right| d\mu$$

$$= \varepsilon R^{2-n} \int_{A_{R,kR}} \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \Delta r^{2-n} d\mu ,$$

$$(1.4.15)$$

where in the last identity we used $\Delta r^{2-n} \ge 0$ in the sense of distributions, already discussed above. Integrating by parts Δr^{2-n} we obtain

$$\int_{A_{R}} \Delta r^{2-n} \, \mathrm{d}\mu = (2-n) \left[(kR)^{1-n} |\{r = kR\}| - R^{1-n} |\{r = R\}| \right].$$

In particular, by the assumption on the Euclidean volume growth, the above quantity is uniformly bounded in R. We have thus proved that the first summand on the right hand side of (1.4.14), for R large enough, is bounded as follows

$$\left| \int\limits_{A_{R,kR}} \left(\frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \Delta r^{2-n} \right) \left(u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{2-n} \right) d\mu \right| \leq C_1 \varepsilon R^{2-n}. \tag{1.4.16}$$

Let us turn our attention to the second integral in the right hand side of (1.4.14). Proceeding as in (1.4.15), we have

$$\int_{\partial A_{R,kR}} \left| \left(u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{2-n} \right) \left\langle \operatorname{D} u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \operatorname{D} r^{2-n} \middle| \nu \right\rangle \right| d\sigma \leq \\
\leq \varepsilon R^{2-n} \int_{\partial A_{R,kR}} \left[|\operatorname{D} u| + (n-2) \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{1-n} \right] d\sigma. \tag{1.4.17}$$

Recall now that in Proposition 1.33 we proved that

$$|Du| \leq C_2 r^{1-n},$$

for some positive constant C_2 independent of r. Thus, by the Euclidean volume growth, the integral on the right hand side of (1.4.17) is uniformly bounded in R and so

$$\left| \int_{\partial A_{R,kR}} \left(u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{2-n} \right) \left\langle \operatorname{D} u - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} \operatorname{D} r^{2-n} \, \middle| \, \nu \right\rangle \, \mathrm{d}\sigma \right| \leq C_3 \, \varepsilon \, R^{2-n} \qquad (1.4.18)$$

for some C_3 independent of R. Finally, by (1.4.14), (1.4.16) and (1.4.18), we obtain, for R big enough, the estimate

$$\frac{1}{|A_{R,kR}|}\int\limits_{A_{R,kR}}\left|Du-\frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)}\operatorname{D}r^{2-n}\right|^{2}\mathrm{d}\mu \leq C_{4}\varepsilon\frac{R^{2-n}}{|A_{R,kR}|} \leq C_{5}\varepsilon R^{2-2n},$$

for some positive constants C_4 and C_5 independent of R. In the last inequality we used the Euclidean volume growth assumption. Our claim (1.4.13) is thus proved.

From the above Proposition we easily deduce the integral asymptotic behaviour of |Du| on geodesic spheres.

Corollary 1.38. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth, and let u be a solution to problem (1.2.15). Then, there exists a sequence

of positive real numbers R_k with $R_k \to +\infty$ such that

$$\lim_{\substack{R_k \to +\infty \\ \{r = R_k\}}} \int \left| |Du| - (n-2) \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} r^{1-n} \right| d\sigma = 0.$$
 (1.4.19)

Proof. Let us first observe that, by means of Hölder inequality, we can deduce from the L^2 asymptotics (1.4.13) an analogous L^1 behaviour. Namely, for any $\varepsilon > 0$ we have

$$\frac{\int_{A_{R,kR}} \left| \mathrm{D}u - \frac{\mathrm{Cap}(\Omega)}{\mathrm{AVR}(g)} \mathrm{D}r^{2-n} \right| \, \mathrm{d}\mu}{R^{1-n} |A_{R,kR}|} \leq \left(\frac{\int_{A_{R,kR}} \left| \mathrm{D}u - \frac{\mathrm{Cap}(\Omega)}{\mathrm{AVR}(g)} \mathrm{D}r^{2-n} \right|^2 \, \mathrm{d}\mu}{R^{2-2n} |A_{R,kR}|} \right)^{1/2} \leq \varepsilon$$

for any R large enough. By the Coarea Formula, the above L^1 estimate gives, for R large enough,

$$\frac{\int\limits_{R}^{kR}\mathrm{d}s\int\limits_{\{r=s\}}\left|\mathrm{D}u-\frac{\mathrm{Cap}(\Omega)}{\mathrm{AVR}(g)}\mathrm{D}r^{2-n}\right|\,\mathrm{d}\sigma}{R^{1-n}|A_{R,kR}|}\leq\varepsilon.$$

Thus, by the Mean Value Theorem, there exists $R_k \in (R, kR)$ such that

$$\frac{\int\limits_{\{r=R_k\}}\left|\mathrm{D}u-\frac{\mathrm{Cap}(\Omega)}{\mathrm{AVR}(g)}\mathrm{D}r^{2-n}\right|\,\mathrm{d}\sigma}{R^{-n}|A_{R,kR}|}\leq C\,\varepsilon\,,$$

for some constant C independent of R. The Euclidean volume growth of the annulus $A_{R,kR}$ as R increases then implies the existence of a sequence $R_k \to +\infty$ as in the statement.

Remark 1.39. The integral asymptotic for |Du| given by Corollary 1.38 cannot, in general, be improved to a pointwise asymptotic expansion at infinity on a manifold with nonnegative Ricci curvature and Euclidean volume growth. Indeed, the validity of such a formula would imply $|Du| \neq 0$ outside some big ball B(O,R), and, in turn, $M \setminus B(O,R)$ would be diffeomorphic to $\partial B(O,R) \times [R,+\infty)$. This would imply that M has finite topological type. However, Menguy provided in [Men00] examples of manifolds of dimension $n \geq 4$ with Ric ≥ 0 and AVR(g) > 0 with infinite topological type. On the other hand, we notice that in dimension 3 topological obstructions like these cannot occur. This is due to the structure result obtained in [Liu13], where it is proved that if a 3-dimensional complete noncompact Riemannian manifold (M,g) with Ric ≥ 0 is not diffeomorphic to \mathbb{R}^3 , then its universal cover isometrically splits a product $N \times \mathbb{R}$ where N is compact Riemannian manifold with nonnegative sectional curvature. In particular, in this case, (M,g) has nonnegative sectional curvature, and the finite topology immediately follows from the celebrated Soul Theorem proved in [CG72].

1.4.3 Limits of the monotone quantities.

Recalling that

$$U_{\beta}(t) = t^{-\beta\left(\frac{n-1}{n-2}\right)} \int_{\{u=t\}} |\mathrm{D}u|^{1+\beta} \,\mathrm{d}\sigma,$$

it is easy to realise that we have now at hand all the elements to compute the limit as $t \to 0^+$.

Proposition 1.40. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth. Then, for every $\beta \geq 0$, we have that

$$\lim_{t \to 0^+} U_{\beta}(t) = \operatorname{Cap}(\Omega)^{1 - \frac{\beta}{n-2}} \operatorname{AVR}(g)^{\frac{\beta}{n-2}} (n-2)^{\beta+1} |\mathbb{S}^{n-1}|. \tag{1.4.20}$$

Proof. We multiply and divide inside the integral in (1.4.19) by $u^{(n-1)/(n-2)}$. Using the asymptotics of u obtained in Lemma 1.34, we obtain

$$\lim_{R_i \to +\infty} \int_{\{r=R_i\}} \left| \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right| - (n-2) \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{\frac{1}{n-2}} \right| u^{\frac{n-1}{n-2}} \, \mathrm{d}\sigma = 0.$$

Let us now recall the following basic interpolation inequality

$$||f||_{L^p(X)} \leq ||f||_{L^1(X)}^{\vartheta} ||f||_{L^q(X)}^{1-\vartheta}$$

holding for any $f \in L^1(X) \cap L^q(X)$, where X is a measure space and the numbers p and q satisfy $1 and <math>1/p = \vartheta + (1 - \vartheta)/q$. We apply such an estimate with

$$f = \left| \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right| - (n-2) \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{\frac{1}{n-2}} \right|$$

 $p = 1 + \beta$, $q = 2 + \beta$, so that $\vartheta = 1/(1+\beta)^2$, and with respect to the measure $u^{(n-1)/(n-2)} d\sigma$. We get

$$\left(\int_{\{r=R_{i}\}} \left| \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right| - (n-2) \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{\frac{1}{n-2}} \right|^{1+\beta} u^{\frac{n-1}{n-2}} d\sigma \right)^{1/(1+\beta)} \leq \left(\int_{\{r=R_{i}\}} \left| \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right| - (n-2) \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{\frac{1}{n-2}} \right| u^{\frac{n-1}{n-2}} d\sigma \right)^{1/(1+\beta)^{2}} \times \left(\int_{\{r=R_{i}\}} \left| \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right| - (n-2) \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{\frac{1}{n-2}} \right|^{2+\beta} u^{\frac{n-1}{n-2}} d\sigma \right)^{\beta/(1+\beta)^{2}} . \tag{1.4.21}$$

Due to both the uniform bounds on $|Du|/u^{(n-1)/(n-2)} = |\nabla \varphi|_{\tilde{g}}$ following from Theorem 1.21 and the bound on $\int_{\{r=R_i\}} u^{\frac{n-1}{n-2}} d\sigma$, that follows from Corollary 1.35, it is easy to see that the second integral on the right hand side of (1.4.21) is bounded in R_i . Thus, by (1.4.21) and (1.4.19), we deduce that

$$\lim_{R_i\to\infty}\int\limits_{\{r=R_i\}}\left|\left|\frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}}\right|-(n-2)\left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)}\right)^{\frac{1}{n-2}}\right|^{1+\beta}u^{\frac{n-1}{n-2}}\,\mathrm{d}\sigma\ =\ 0\,.$$

The above limit in particular implies that

$$\lim_{R_{i}\to +\infty} \int_{\{r=R_{i}\}} \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right|^{1+\beta} u^{\frac{n-1}{n-2}} d\sigma = (n-2)^{1+\beta} \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{\frac{1+\beta}{n-2}} \lim_{R_{i}\to \infty} \int_{\{r=R_{i}\}} u^{\frac{n-1}{n-2}} d\sigma,$$

that, combined with (1.4.12), gives

$$\lim_{R_i \to +\infty} \int_{\{r=R_i\}} \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right|^{1+\beta} u^{\frac{n-1}{n-2}} \, \mathrm{d}\sigma \ = \ (n-2)^{1+\beta} \, \mathrm{Cap}(\Omega)^{1-\frac{\beta}{n-2}} \, \mathrm{AVR}(g)^{\frac{\beta}{n-2}} \, |\mathbb{S}^{n-1}| \, .$$

Moreover, by the asymptotic behaviour of u, we have

$$\lim_{R_i \to +\infty} \int_{\{r=R_i\}} \left| \frac{\mathrm{D} u}{u^{\frac{n-1}{n-2}}} \right|^{1+\beta} u^{\frac{n-1}{n-2}} \, \mathrm{d}\sigma = \lim_{t_i \to 0^+} \int_{\{u=t_i\}} \left| \frac{\mathrm{D} u}{u^{\frac{n-1}{n-2}}} \right|^{1+\beta} u^{\frac{n-1}{n-2}} \, \mathrm{d}\sigma = \lim_{t_i \to 0^+} U_\beta(t_i) \,,$$

and we have thus proved our claim for some sequence $t_i \to 0^+$. However, by the boundedness and monotonicity of U_β the whole limit as $t \to 0^+$ exists, and it coincides with the just computed one.

The computation of the limit of $U_{\infty}(t)$ as $t \to 0^+$ is more subtle, but it can still be returned to Proposition 1.37, thanks to a clever trick sketched by Colding in the proof of [Col12, Theorem 3.5]. The argument is based on a very nice mean value inequality for positive super-harmonic functions in manifolds with nonnegative Ricci curvature, first provided in [CCM95] and stated and proved separately in [Col12, Lemma 3.6].

Lemma 1.41 (Mean value inequality for positive super-harmonic functions). *Let* (M, g) *be a complete noncompact Riemannian manifold with nonnegative Ricci curvature and let u be a positive super-harmonic function defined in an open set* $D \subseteq M$. *Then, for any geodesic ball* $B(x,r) \subseteq D$ *we have*

$$\frac{1}{|\partial B(x,r)|} \int_{\partial B(x,r)} u \, d\sigma \le C \, u(x)$$

for some C = C(n).

Remark 1.42. The correspondent statement for positive sub-harmonic function is classical and is proved in the celebrated [LS84], see Theorem 2.1.

We can finally compute the limit of U_{∞} at infinity.

Proposition 1.43. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth. Then

$$\lim_{t \to 0^{+}} U_{\infty}(t) = (n-2) \left[\frac{\text{AVR}(g)}{\text{Cap}(\Omega)} \right]^{1/(n-2)}.$$
 (1.4.22)

Proof. Let us first observe that proving

$$\lim_{j \to \infty} \sup_{M \setminus \overline{B(O,R_i)}} \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} = (n-2)^2 \left[\frac{AVR(g)}{Cap(\Omega)} \right]^{2/(n-2)}$$
(1.4.23)

for some sequence $\{R_j\}_{j\in\mathbb{N}}$ such that R_j diverges to infinity as $j\to\infty$ suffices to get (1.4.22). Indeed, if (1.4.23) holds, it suffices to take a sequence x_{R_j} realising the supremum

of $|Du|^2/u^{2(n-1)/(n-2)}$ in $M \setminus \overline{B(O,R_j)}$. In fact, x_R is easily seen by the same argument as in the proof of Theorem 1.21 to belong to $\partial B(O,R_j)$. Up to possibly consider a subsequence of the R_j 's, let t_j such that $B(O,R_j) \subseteq M \setminus \{u \leq t_j\} \subseteq B(O,R_{j+1})$. Then, we have

$$\sup_{M\setminus \overline{B(O,R_{i+1})}} \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} \le \sup_{M\setminus \{u \le t_j\}} \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} \le \sup_{M\setminus \overline{B(O,R_i)}} \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}}.$$
 (1.4.24)

In particular, since

$$\sup_{M\setminus\{u\leq t_i\}}\frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}}=\sup_{\{u=t_i\}}\frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}}$$

as the argument in the proof of Theorem 1.21 showed, we get, passing to the limit as $j \to +\infty$ in (1.4.24) and using (1.4.23), that

$$\lim_{t_j \to 0^+} \sup_{\{u=t_j\}} \frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}} = (n-2)^2 \left[\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right]^{2/(n-2)},$$

that is obviously equivalent to (1.4.22) for the subsequence t_j . However, the monotonicity of U_{∞} provided by Theorem 1.21 yields the desired full limit.

We thus devote ourselves to prove (1.4.23). To this aim, let, for R > 0 big enough,

$$L_R = \sup_{M \setminus B(O,R)} \frac{|Du|^2}{u^{2\frac{n-1}{(n-2)}}},$$

and let $x_R \in \partial B(O, R)$, as above, to satisfy $L_R = |Du|^2/u^{2(n-1)/(n-2)}(x_R)$. Consider now on $M \setminus B(O, R)$ the *positive* function

$$f = u \left(L_R - \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} \right).$$

A direct computation shows that

$$\Delta f = -u \left[\Delta \frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}} + 2 \left\langle \mathrm{D}u \,\middle|\, \mathrm{D}\left(\frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}}\right) \right\rangle \right],$$

that amounts, by the formulas recalled in Subsection 1.3.1, to

$$\Delta f = -\mathrm{e}^{-\frac{n}{n-2}\varphi}\left(\Delta_{\widetilde{g}}|
ablaarphi|_{\widetilde{g}}^2 - \left\langle
abla|
ablaarphi|^2|
ablaarphi|^2\right) \leq 0,$$

where the nonpositivity is due to (1.3.15). In particular, the function f is a super-harmonic positive function, and so it satisfies the assumptions of Lemma 1.41. We thus have

$$\frac{1}{|\partial B(x_{4R}, \tilde{R})|} \int_{\partial B(x_{4R}, \tilde{R})} f \, d\sigma \le C f(x_{4R}) = C u(x_{4R}) (L_R - L_{4R})$$

for any $R < \tilde{R} < 3R$ and some C = C(n). In particular, we get

$$\frac{\inf_{\partial B(x_{4R},\tilde{R})} u}{u(x_{4R})} \frac{1}{|\partial B(x_{4R},\tilde{R})|} \int_{\partial B(x_{4R},\tilde{R})} \left(L_R - \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} (x_{4R}) \right) d\sigma \le C \left(L_R - L_{4R} \right). \tag{1.4.25}$$

A direct application of the Harnack's inequality (1.2.7) combined with Yau's gradient

inequality (1.2.4) shows that the ratio of on the left hand side of (1.4.25) is bounded from below from a constant C = C(n), and then we are left with

$$\frac{1}{R^{n-1}} \int_{\partial B(x_{4R}, \tilde{R})} \left(L_R - \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} (x_{4R}) \right) d\sigma \le C \left(L_R - L_{4R} \right), \tag{1.4.26}$$

for some constant C = C(AVR(g)), where we also used the assumption of Euclidean volume growth in the form of (1.4.4). Observe now that since, as observed above, L_R is nonincreasing as $R \to +\infty$, it admits a finite limit and in particular the right hand side of (1.4.26) tends to 0 as $R \to +\infty$, forcing the right hand side to do so too. To complete the proof, we are going to show that also

$$\frac{1}{R^{n-1}} \int_{\partial B(x_{4R},\tilde{R})} \left[(n-2)^2 \left(\frac{\text{AVR}(g)}{\text{Cap}(\Omega)} \right)^{2/(n-2)} - \frac{|Du|^2}{u^{2\frac{n-1}{n-2}}} (x_{4R}) \right] d\sigma \to 0$$
 (1.4.27)

as $R \to +\infty$. Indeed, since obviously

$$\begin{split} \frac{1}{R^{n-1}} \int\limits_{\partial B(x_{4R},\tilde{R})} \left(L_R - \frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}}(x_{4R}) \right) \, \mathrm{d}\sigma &= \frac{1}{R^{n-1}} \int\limits_{\partial B(x_{4R},\tilde{R})} \left[L_R - (n-2)^2 \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{2/(n-2)} \right] \, \mathrm{d}\sigma \\ &+ \frac{1}{R^{n-1}} \int\limits_{\partial B(x_{4R},\tilde{R})} \left[(n-2)^2 \left(\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right)^{2/(n-2)} - \frac{|\mathrm{D}u|^2}{u^{2\frac{n-1}{n-2}}}(x_{4R}) \right] \, \mathrm{d}\sigma, \end{split}$$

if (1.4.27) holds, we deduce by coupling it with (1.4.26) that

$$\lim_{R\to+\infty} L_R = (n-2)^2 \left[\frac{\text{AVR}(g)}{\text{Cap}(\Omega)} \right]^{2/(n-2)},$$

that is (1.4.23).

To finally prove (1.4.27), consider the annulus $A_{R,3R}(x_{4R}) = B(x_{4R},3R) \setminus \overline{B(x_{4R},R)}$ centered at x_{4R} , and observe that it is contained in the annulus $A_{R,7R}$ centered at the reference point O. By Proposition 1.37, for $\varepsilon > 0$ it holds

$$\frac{R^{2n-2}}{|A_{R,7R}|} \int_{A_{R,7R}} \left| Du - \frac{\operatorname{Cap}(\Omega)}{\operatorname{AVR}(g)} Dr^{2-n} \right|^2 d\mu \le \varepsilon$$

if *R* is big enough. Multiplying and dividing for $u^{2(n-1)/(n-2)}$, and employing Lemma 1.34, we deduce that

$$\frac{1}{R^n}\int\limits_{A_{n,n}}\left|\frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}}-(2-n)\left[\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)}\right]^{1/(n-2)}\mathrm{D}r\right|^2\mathrm{d}\mu\leq C_1\,\varepsilon,$$

for some C_1 independent of R, where we also used the Euclidean volume growth assumption to control the volume of the annulus. Since as just observed $A_{R,3R}(x_{4R}) \subset A_{R,7R}$, we also get

$$\frac{1}{R^n} \int_{A \operatorname{Dap}(Y(n))} \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} - (2-n) \left[\frac{\operatorname{AVR}(g)}{\operatorname{Cap}(\Omega)} \right]^{1/(n-2)} \operatorname{D}r \right|^2 \mathrm{d}\mu \le C_2 \, \varepsilon,$$

for some C_2 independent of R. By the coarea formula and the integral mean value theorem we actually find $\tilde{R} \in (R, 3R)$ such that

$$\frac{1}{R^{n-1}} \int\limits_{\partial B(\chi_{AR},\tilde{R})} \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} - (2-n) \left[\frac{\mathrm{AVR}(g)}{\mathrm{Cap}(\Omega)} \right]^{1/(n-2)} \mathrm{D}r \right|^2 \mathrm{d}\mu \le C_3 \, \varepsilon.$$

for any R big enough and some constant C_3 independent of R, that is, we proved (1.4.27).

Limits of the monotone quantities in the sub-Euclidean volume growth case

As happens also for the monotone quantities in [Col12], U_{β} and U_{∞} vanish as $t \to 0^+$ if (M, g) has Euclidean volume growth. In this case, we have

$$\lim_{r \to +\infty} \frac{|B(x,r)|}{r^n} = 0$$

for any $x \in M$. Then, by (1.2.18), (1.12) and (1.2.6), one easily obtains

$$\frac{|Du|}{u^{\frac{n-1}{n-2}}} \le C \left[\frac{|B(p,r)|}{r^n} \right]^{\frac{1}{n-2}}$$

outside some ball containing Ω for some $C = C(M, \Omega)$. This implies that

$$\lim_{t \to 0^{+}} U_{\beta}(t) = \lim_{t \to 0^{+}} \int_{\{u=t\}} \left| \frac{\mathrm{D}u}{u^{\frac{n-1}{n-2}}} \right|^{\beta} |\mathrm{D}u| \, \mathrm{d}\sigma = 0, \tag{1.4.28}$$

where we used the constancy of $t\mapsto \int_{\{u=t\}} |\mathrm{D}u|\,\mathrm{d}\sigma$. Similarly, one realises that $\lim_{t\to 0^+} U_\infty(t)=0$. This computation clearly shows that U_β cannot be employed to deduce a Willmore inequality on manifolds with sub-Euclidean volume growth, and it supports the perception that the infimum of the Willmore-type functional is zero on these manifolds. This actually happens for example on noncompact Riemannian manifolds with a metric which is asymptotic to a warped product metric of the following type

$$g = d\rho \otimes d\rho + C \rho^{2\alpha} g_{\Sigma}$$

where Σ is a closed hypersurface, C>0 and $0<\alpha<1$. Indeed, this is readily checked by computing the Willmore-type functional on large level sets $\{\rho=r\}$. Actually, Theorem 1.68 below, following from the arguments leading to the Isoperimetric Inequality, shows that this happens also on complete noncompact 3-manifolds with Ric ≥ 0 admitting a uniform superlinear volume growth condition.

1.4.4 The Willmore-type inequalities for nonnegative Ricci

We are finally in position to state and prove one the first main results of this work, a Willmore-type inequality on complete noncompact Riemannian manifolds with nonnegative Ricci curvature.

Theorem 1.44 (Willmore-type inequality). *Let* (M,g) *be a complete noncompact Riemannian manifold with* Ric ≥ 0 *and Euclidean volume growth. If* $\Omega \subset M$ *is a bounded and open subset*

with smooth boundary, then

$$AVR(g)|S^{n-1}| \le \int_{\partial\Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma, \qquad (1.4.29)$$

where $AVR(g) \in (0,1]$ is the asymptotic volume ratio of (M,g). Moreover, the equality holds if and only if $(M \setminus \Omega, g)$ is isometric to

$$\left(\left[r_0,+\infty\right)\times\partial\Omega,\ \mathrm{d}r\otimes\mathrm{d}r+(r/r_0)^2g_{\partial\Omega}\right),\qquad with\quad r_0=\left(\frac{\left|\partial\Omega\right|}{\mathrm{AVR}(g)|\mathbb{S}^{n-1}|}\right)^{\frac{1}{n-1}}.$$

In particular, $\partial\Omega$ is a connected totally umbilic submanifold with constant mean curvature.

Proof of Theorem 1.44. With Theorem 1.19 and Proposition 1.40 at hand, Theorem 1.44 follows exactly as in the Euclidean proof recalled in the Introduction. Precisely, let $\beta = n - 2$. Then, (1.4.20) reads

$$\lim_{t \to 0^+} U_{n-2}(t) = \text{AVR}(g) (n-2)^{n-1} |\mathbb{S}^{n-1}|.$$

Moreover, by the nonnegativity of expression (1.3.3) proved in Theorem 1.19, and the Hölder inequality,

$$\begin{split} \frac{(n-1)}{(n-2)} \int_{\partial\Omega} &|\mathrm{D} u|^{n-1} \,\mathrm{d}\sigma \ \leq \ \int_{\partial\Omega} &|\mathrm{D} u|^{n-2} \,\mathrm{H} \,\mathrm{d}\sigma \\ & \leq \ \left(\int_{\partial\Omega} &|\mathrm{D} u|^{n-1} \,\mathrm{d}\sigma \right)^{(n-2)/(n-1)} \left(\int_{\partial\Omega} &\mathrm{H}^{n-1} \,\mathrm{d}\sigma \right)^{1/(n-1)}, \end{split}$$

that gives

$$\int_{\partial \Omega} |Du|^{n-1} d\sigma \le (n-2)^{n-1} \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma.$$
 (1.4.30)

Finally, one has

$$AVR(g) (n-2)^{n-1} |S^{n-1}| = \lim_{t \to 0^+} U_{n-2}(t) \le U_{n-2}(1) = \int_{\partial \Omega} |Du|^{n-1} d\sigma \le
\le (n-2)^{n-1} \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma.$$
(1.4.31)

This completes the proof of the Willmore-type inequality. The rigidity statement when equality is attained follows straightforwardly from the rigidity part of Theorem 1.19.

Application to ALE manifolds

We can improve our Willmore-type inequality if (M, g) satisfies a *quadratic curvature decay* condition, showing that, in this case, the lower bound $AVR(g)||S^{n-1}||$ on the Willmore functional is actually an infimum. Let us recall the following well known definition.

Definition 1.45 (Quadratic curvature decay). A complete noncompact Riemannian manifold (M,g) has quadratic curvature decay if there exists a point $x \in M$ and a constant C = C(M,x) such that

$$|\text{Riem}|(y) \le C d(x, y)^2$$
,

where by Riem we denote the Riemann curvature tensor of (M,g).

When this assumption is added on a Riemannian manifold with Ric ≥ 0 and Euclidean volume growth (M,g), [CM97b, Proposition 4.1] gives the following asymptotic behaviour of the gradient and the Hessian of the minimal Green's function G.

Theorem 1.46. Let (M, g) be a complete noncompact Riemannian manifold with Ric ≥ 0 , Euclidean volume growth and quadratic curvature decay, and let G be its minimal Green's function. Then

$$\lim_{d(O,x)\to +\infty} \frac{|D_x G|(O,x)}{d^{1-n}(p,x)} = \frac{(n-2)}{AVR(g)}$$
(1.4.32)

$$\lim_{d(O,x)\to +\infty} \left| DD_x \left(G^{2/(2-n)} \right) (O,x) - 2 \left(\frac{1}{AVR(g)} \right)^{\frac{2}{2-n}} g(x) \right| = 0.$$
 (1.4.33)

Observe that arguing as in Remark 1.39, one realises that (1.4.32) actually implies that Riemannian manifolds satisfying the assumptions of the above Theorem have finite topological type.

Theorem 1.46 enables us to prove that the Willmore-type functional of large level sets of the Green's function approach $AVR(g)|S^{n-1}|$. This fact, combined with our Willmore-type inequality (1.4.29), easily yields the following refinement.

Theorem 1.47. Let (M, g) be a complete noncompact Riemannian manifold with Ric ≥ 0 , Euclidean volume growth and quadratic curvature decay. Then,

$$\inf \left\{ \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma \right| \Omega \subset M \text{ bounded and smooth} \right\} = AVR(g) |S^{n-1}|. \quad (1.4.34)$$

Moreover, the infimum is attained at some bounded and smooth $\Omega \subset M$ if and only if $(M \setminus \Omega, g)$ is isometric to

$$\left(\left[r_0, +\infty \right) \times \partial \Omega, \ dr \otimes dr + (r/r_0)^2 g_{\partial \Omega} \right), \quad \text{with} \quad r_0 = \left(\frac{|\partial \Omega|}{\text{AVR}(g)|\mathbb{S}^{n-1}|} \right)^{\frac{1}{n-1}}.$$

Proof. Let $p \in M$ be fixed, and let G be the minimal Green's function of (M,g). Let us denote again by G the function $x \mapsto G(O,x)$. In light of Theorem 1.44, it suffices to prove that

$$\lim_{t \to +\infty} \int_{\{G^{2/(2-n)} = t\}} \left| \frac{H}{n-1} \right|^{n-1} d\sigma = AVR(g) |\mathbb{S}^{n-1}|.$$
 (1.4.35)

To see this, consider, at a point x, orthonormal vectors $\{e_1, \dots, e_{n-1}\}$ tangent to the level set $\{G = G(x)\}$. Then, letting r(x) = d(O, x), we have, by (1.4.33),

$$\lim_{r(x)\to\infty} \sum_{i=1}^{n-1} DD\left(G^{\frac{2}{2-n}}\right)(e_i, e_i)(x) = 2(n-1)\left(\frac{1}{AVR(g)}\right)^{\frac{1}{2-n}}.$$
 (1.4.36)

The mean curvature of the level sets of $G^{2/(2-n)}$ is easily computed as

$$H_{G^{2/(2-n)}} = \frac{\sum_{i=1}^{n-1} DD\left(G^{\frac{2}{2-n}}\right)(e_i, e_i)}{|DG^{2/(2-n)}|},$$

that, combined with (1.4.36), (1.4.32) and (1.4.7) gives

$$\lim_{r(x)\to\infty} r(x) \, \mathcal{H}_{G^{2/(2-n)}}(x) = (n-1).$$

Combining it again with (1.4.7) and Euclidean volume growth gives (1.4.35), in fact completing the proof.

We now particularise Theorem 1.47 to *Asymptotically Locally Euclidean* (ALE) manifolds, proving Corollary 1.49. We adopt the following definition, that is a sort of extension of the one considered in the celebrated [BKN89] – where striking relations between curvature decay conditions and behaviour at infinity of manifolds are drawn – and sensibly weaker than the one used by Joyce in the classical reference [Joy00].

Definition 1.48 (ALE manifolds). We say that a complete noncompact Riemannian manifold (M,g) is ALE (of order τ) if there exist a compact set $K \subset M$, a ball $B \subset \mathbb{R}^n$, a diffemorphism $F: M \setminus K \to \mathbb{R}^n \setminus B$, a subgroup $\Gamma < SO(n)$ acting freely on $\mathbb{R}^n \setminus B$ and a number $\tau > 0$ such that

$$(F^{-1} \circ \pi)^* g(z) = g_{\mathbb{R}^n} + O(|z|)^{-\tau}$$
(1.4.37)

$$\left|\partial_i[(F^{-1}\circ\pi)^*g]\right|(z) = O(|z|)^{-\tau-1}$$
 (1.4.38)

$$\left|\partial_i \partial_j [(F^{-1} \circ \pi)^* g]\right|(z) = O(|z|)^{-\tau - 2},$$
 (1.4.39)

where π is the natural projection $\mathbb{R}^n \to \mathbb{R}^n / \Gamma$, $z \in \mathbb{R}^n \setminus B$ and i, j = 1, ..., n.

The following is the version of Theorem 1.47 sharpened for ALE manifolds.

Corollary 1.49. *Let* (M,g) *be an ALE Riemannian manifold with* Ric ≥ 0 . Then,

$$\inf \left\{ \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma \, \middle| \, \Omega \subset M \text{ bounded and smooth} \right\} = \frac{|S^{n-1}|}{\operatorname{card} \Gamma}. \tag{1.4.40}$$

Moreover, if the infimum is attained by some Ω *, then* $M \setminus \Omega$ *is isometric to*

$$\left(\left[r_0,+\infty\right)\times\left(\mathbb{S}^{n-1}/\Gamma\right),\ \mathrm{d} r\otimes\mathrm{d} r+r^2g_{\mathbb{S}^{n-1}/\Gamma}\right),\qquad with\quad r_0=\left(\frac{\mathrm{card}\,\Gamma|\partial\Omega|}{|\mathbb{S}^{n-1}|}\right)^{\frac{1}{n-1}},\tag{1.4.41}$$

for some $r_0 > 0$ and some finite subgroup Γ of SO(n). In particular, $(\partial\Omega, g_{\partial\Omega})$ is homothetic to $(S^{n-1}/\Gamma, g_{S^{n-1}/\Gamma})$.

Proof. Conditions (1.4.37), (1.4.38), (1.4.39) readily imply that ALE manifolds have Euclidean volume growth and quadratic curvature decay. Moreover, condition (1.4.37) and a direct computation give that

$$AVR(g) = \frac{|\mathbb{S}^{n-1}/\Gamma|}{|\mathbb{S}^{n-1}|} = \frac{1}{\operatorname{card}\Gamma}.$$
 (1.4.42)

The characterisation (1.4.40) then follows immediately from (1.4.34).

Assume now that the infimum of the Willmore functional is attained at some $\Omega \subset M$. Then, by the rigidity part in Theorem 1.47, $M \setminus \Omega$ is isometric to a truncated cone over $\partial \Omega$. However, by (1.4.37), (M,g) is also C^0 -close at infinity to a metric cone with link S^{n-1}/Γ . Since the cross sections of a cone are all homothetic to each other, $\partial \Omega$ is homothetic to

 S^{n-1}/Γ , that is, they are diffeomorphic and $g_{\partial\Omega} = \lambda^2 g_{S^{n-1}/\Gamma}$ for some positive constant λ . This fact, together with (1.4.42) in the rigidity part of Theorem 1.44 imply that

$$\left(\left[r_0,+\infty\right)\times\partial\Omega,\ \mathrm{d} r\otimes\mathrm{d} r+(\lambda r/r_0)^2g_{\mathbf{S}^{n-1}/\Gamma}\right),\qquad \mathrm{with}\quad r_0=\left(\frac{|\partial\Omega|}{|\mathbf{S}^{n-1}/\Gamma|}\right)^{\frac{1}{n-1}}.$$

In particular, one has

$$|\partial B(O,R)| = \left(\frac{R\lambda}{r_0}\right)^{n-1} |\mathbb{S}^{n-1}/\Gamma|.$$

Combining this with (1.4.42) we conclude that $\lambda = r_0$, proving the isometry with (1.4.41) and completing the proof.

1.5 Other consequences of the Monotonicity-Rigidity Theorem for nonparabolic manifolds

In this short section we draw some other consequences of Theorem 1.19, in particular of the monotonicity of U_{∞} , that we did not apply yet.

1.5.1 Capacity estimates

The following capacity estimate hold on any nonparabolic manifold with Ric ≥ 0 , and, at this point not surprisingly, holds as equality only on truncated cones.

Proposition 1.50 (Capacity estimates for nonparabolic manifolds). Let (M, g) be a complete noncompact Riemannian manifold with nonnegative Ricci curvature, and let $\Omega \subset M$ be a bounded set with smooth boundary. Then, for any $\beta \geq (n-2)/(n-1)$, we have

$$\operatorname{Cap}(\Omega) \le \frac{|\partial \Omega|}{|\mathbb{S}^{n-1}|} \left(\oint_{\partial \Omega} \left| \frac{H}{n-1} \right|^{1+\beta} d\sigma \right)^{1/(1+\beta)}. \tag{1.5.1}$$

Moreover, equality is achieved if and only if $(M \setminus \Omega, g)$ *is isometric to*

$$\left(\left[r_0, +\infty \right) \times \partial \Omega, \ dr \otimes dr + (r/r_0)^2 g_{\partial \Omega} \right), \quad \text{with} \quad r_0 = \left(\frac{|\partial \Omega|}{\text{AVR}(g)|\mathbb{S}^{n-1}|} \right)^{\frac{1}{n-1}}.$$

Proof. By the nonnegativity of expression (1.3.3) for t = 1, we get, as for (1.4.30), that

$$\int_{\partial \Omega} |Du|^{\beta+1} d\sigma \le (n-2)^{1+\beta} \int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{1+\beta} d\sigma. \tag{1.5.2}$$

Recalling that

$$\operatorname{Cap}(\Omega) = \frac{\int_{\partial \Omega} |\operatorname{D} u| \, \mathrm{d}\sigma}{(n-2)|S^{n-1}|}$$

we easily obtain (1.5.1) by applying the Hölder inequality in the above definition of capacity and combining with (1.5.2). See the analogous proof of the nonlinear version Theorem 2.5.1 in \mathbb{R}^n for additional details. The rigidity statement follow immediately from the equality case of Theorem 1.19.

We would like to highlight, among the various inequalities provided in (1.5.1), the one obtained setting $\beta = n - 2$, reading

$$\mathsf{Cap}(\Omega) \leq \frac{|\partial \Omega|^{\frac{(n-2)}{(n-1)}}}{|\mathsf{S}^{n-1}|} \left(\int_{\partial \Omega} \left| \frac{\mathsf{H}}{n-1} \right|^{n-1} \, \mathrm{d}\sigma \right)^{1/(n-1)},$$

since it combines the Willmore-type functional and the capacity of Ω . In the next chapter we will show how in \mathbb{R}^n the nonlinear version of it, that is (2.5.4), involving the p-capacity of Ω , actually leads to the Willmore inequality in the limit as $p \to n$.

In the the above inequality we just used $U_{\beta}'(0) \geq 0$, and this is why it is meaningful on any nonparabolic manifold with nonnegative Ricci curvature. The following one, on the other hand, will employ $\lim_{t\to 0^+} U_{\beta}(t) \leq U_{\beta}(1)$, and this is where, by Proposition 1.40, the Asymptotic Volume Ratio and thus the Euclidean volume growth assumption enters.

Proposition 1.51 (Capacity estimates for manifolds with Euclidean volume growth). Let (M,g) be a complete noncompact Riemannian manifold with nonnegative Ricci curvature and Euclidean volume growth, and let $\Omega \subset M$ be a bounded set with smooth boundary. Then, for every $\beta \geq (n-2)/(n-1)$ we have

$$\operatorname{AVR}(g)^{\beta/(n-2)}\frac{|S^{n-1}|}{|\partial\Omega|} \leq \left[\operatorname{Cap}(\Omega)\right]^{\frac{\beta-(n-2)}{n-2}} \oint_{\partial\Omega} \left|\frac{\operatorname{H}}{n-1}\right|^{1+\beta} \, \mathrm{d}\sigma.$$

Moreover, equality is achieved if and only if $(M \setminus \Omega, g)$ is isometric to

$$\left([r_0, +\infty) \times \partial \Omega, dr \otimes dr + (r/r_0)^2 g_{\partial \Omega} \right), \quad with \quad r_0 = \left(\frac{|\partial \Omega|}{AVR(g)|\mathbb{S}^{n-1}|} \right)^{\frac{1}{n-1}}.$$

Proof. This is substantially (1.4.31) with a generic $\beta \ge (n-2)/(n-1)$, obtained combining (1.4.20) with (1.5.2) in the proof above, and suitably rearranging the terms. The rigidity statement follows as usual from the rigidity statement of Theorem 1.19.

As before, we would like to point out a particularly relevant inequality given by the above result, that is, the inequality obtained through setting $\beta = 1$, reading

$$\left(\operatorname{AVR}(g)\operatorname{Cap}^{n-3}(\Omega)\right)^{1/(n-2)} \leq \frac{1}{|\mathbb{S}^{n-1}|} \int_{\partial\Omega} \left|\frac{\operatorname{H}}{n-1}\right|^2 \, \mathrm{d}\sigma.$$

This is, with the terminology of Theorem 2.24 in the next chapter, the L^2 -Minkowski inequality in the general setting of manifolds with nonnegative Ricci curvature and Euclidean volume growth. The L^p version of it, indeed, will be seen to lead, in \mathbb{R}^n , to the (extended) Minkowski inequality.

1.5.2 Rigidity Theorems under pinching conditions

Observe that, passing to the limit as $\beta \to \infty$ in (1.5.1) after taking the $1 + \beta$ -power on both sides, one gets

$$\left\lceil \frac{\text{AVR}(g)}{\text{Cap}(\Omega)} \right\rceil^{\frac{1}{n-2}} \le \sup_{\partial \Omega} \left| \frac{\text{H}}{n-1} \right|.$$

However, with this procedure one cannot apply the rigidity statement of Theorem 1.19, that regards finite values of β , in order to prove that equality in the above inequality characterises truncated cones. On the other hand, this can be deduced from the monotonicity property of U_{∞} and the value of its limit (1.4.22).

Theorem 1.52. Let (M,g) be a complete noncompact Riemannian manifold with nonnegative Ricci curvature and Euclidean volume growth. If there exists a bounded set $\Omega \subset M$ with smooth boundary satisfying

$$-\left[\frac{\text{AVR}(g)}{\text{Cap}(\Omega)}\right]^{\frac{1}{n-2}} \le \frac{H}{n-1} \le \left[\frac{\text{AVR}(g)}{\text{Cap}(\Omega)}\right]^{\frac{1}{n-2}} \tag{1.5.3}$$

on every point of $\partial\Omega$, then $(M \setminus \Omega, g)$ is isometric to

$$\left(\left[r_0,+\infty\right)\times\partial\Omega,\ \mathrm{d} r\otimes\mathrm{d} r+(r/r_0)^2g_{\partial\Omega}\right),\qquad with\quad r_0=\left(rac{\left|\partial\Omega\right|}{\mathrm{AVR}(g)|\mathbb{S}^{n-1}|}
ight)^{rac{1}{n-1}}.$$

Proof. By Theorem 1.21 and Proposition 1.43, we immediately get

$$\lim_{t\to 0^+} U_{\infty}(t) = (n-2) \left[\frac{\operatorname{AVR}(g)}{\operatorname{Cap}(\Omega)} \right]^{1/(n-2)} \leq \sup_{\partial \Omega} |\mathrm{D}u| \leq (n-2) \sup_{\partial \Omega} \left| \frac{\mathrm{H}}{n-1} \right|,$$

where the equality is achieved only if $(M \setminus \Omega, g)$ splits as in the statement. In particular, this happens if (1.5.3) holds.

The above result can be interpreted as a rigidity theorem under a pinching condition on the mean curvature of $\partial\Omega$ with respect to the capacity of Ω , first observed for \mathbb{R}^n in [BMM19], where it can be interpreted as a sphere theorem. An analogous statement is immediately deduced from the proof above for |Du|, stating that

$$\left[\frac{\text{AVR}(g)}{\text{Cap}(\Omega)}\right]^{1/(n-2)} \le \sup_{\partial \Omega} \left|\frac{\text{D}u}{n-2}\right|,\tag{1.5.4}$$

with equality achieved only on truncated metric cones. An application of the above sharp inequality yields the following analogue of [BMM19, Corollary 1.4], that, in our case, gives a sufficient condition to infer that the whole manifold is actually \mathbb{R}^n with Euclidean metric and Ω is a ball.

Theorem 1.53. Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and Euclidean volume growth, and let $\Omega \subset M$ be bounded set with smooth boundary. Let u be the solution to (1.2.15), and assume that

$$\sup_{\partial\Omega} \left| \frac{\mathrm{D}u}{n-2} \right| \le \mathrm{AVR}(g) \left(\frac{|\mathbb{S}^{n-1}|}{|\partial\Omega|} \right)^{\frac{1}{n-1}}. \tag{1.5.5}$$

Then (M,g) is isometric to \mathbb{R}^n with the Euclidean metric and Ω is a ball.

Proof. Under the validity of (1.5.5), we get

$$\operatorname{Cap}(\Omega) = \frac{\int_{\partial\Omega} |\operatorname{D} u| \, \mathrm{d}\sigma}{(n-2)|S^{n-1}|} \le \operatorname{AVR}(g) \left(\frac{|S^{n-1}|}{|\partial\Omega|}\right)^{-\frac{n-2}{n-1}},$$

that in particular yields

$$\left(\frac{|\mathbb{S}^{n-1}|}{|\partial\Omega|}\right)^{\frac{n-2}{n-1}} \le \frac{\operatorname{AVR}(g)}{\operatorname{Cap}(\Omega)} \le \sup_{\partial\Omega} \left|\frac{\operatorname{D}u}{n-2}\right|^{n-2} \le \operatorname{AVR}(g)^{n-2} \left(\frac{|\mathbb{S}^{n-1}|}{|\partial\Omega|}\right)^{\frac{n-2}{n-1}},\tag{1.5.6}$$

where we used (1.5.4) and again (1.5.5) respectively in the second and third inequality. In particular, the chain above implies that AVR(g) = 1, that implies, by the Bishop-Gromov

Theorem, that (M, g) is actually isometric to $(\mathbb{R}^n, g_{\mathbb{R}^n})$. Moreover, since the second inequality in (1.5.6) becomes an equality, we infer from Theorem 1.21 that $\partial\Omega$ is a compact connected totally umbilical submanifold, that is, Ω is a ball.

The strong rigidity statement proved above is the same that happens in the equality case of the Isoperimetric Inequality for manifolds with Ric ≥ 0 , that is Theorem 1.60.

1.6 Monotonicity-Rigidity Theorems for parabolic manifolds and the Enhanced Kasue's Theorem

For $\beta \geq 0$, we recall the definition of the function $\Psi_{\beta} : [0, +\infty) \to \mathbb{R}$ satisfying

$$\Psi_{\beta}(s) = \int_{\{\psi=s\}} |\mathrm{D}\psi|^{\beta+1} \,\mathrm{d}\sigma,$$

and that of $\Psi_{\infty}:[0,+\infty)\to\mathbb{R}$

$$\Psi_{\infty}(s) = \sup_{\{\psi=s\}} |\mathrm{D}\psi|$$
,

where ψ is a solution to problem (1.2.22) for some bounded $\Omega \subset M$ with smooth boundary and $\beta \geq 0$. The following is the statement of the monotonicity-rigidity properties of Ψ_{β} .

Theorem 1.54 (Monotonicity-Rigidity Theorem for parabolic manifolds). Let (M,g) be a parabolic manifold with Ric ≥ 0 . Let $\Omega \subset M$ be a bounded and open subset with smooth boundary, and let ψ be a solution to problem (1.2.22). Then, for every $\beta \geq (n-2)/(n-1)$, the function Ψ_{β} is differentiable with derivative

$$\frac{d\Psi_{\beta}}{ds}(s) = -\beta \int_{\{\psi=s\}} |D\psi|^{\beta} H d\sigma,$$

where H is the mean curvature of the level set $\{\psi = s\}$ computed with respect to the unit normal vector field $\nu = D\psi/|D\psi|$. Moreover, for every $s \ge 0$, the derivative fulfils

$$\frac{d\Psi_{\beta}}{ds}(s) = -\beta \int_{\{\psi \ge s\}} |D\psi|^{\beta-2} \left\{ \text{Ric}(D\psi, D\psi) + \left(\beta - \frac{n-2}{n-1}\right) |D|D\psi|^{2} + |D\psi|^{2} \left| h - \frac{H}{n-1} \right|^{2} + \left(\frac{n-2}{n-1}\right) |D^{\top}|D\psi|^{2} \right\} d\mu.$$
(1.6.1)

In particular, $d\Psi_{\beta}/ds$ is always nonpositive. Moreover, $(d\Psi_{\beta}/ds)(s_0) = 0$ for some $s_0 \ge 1$ and some $\beta \ge (n-2)/(n-1)$ if and only if $(\{\psi \ge s_0\}, g)$ is isometric to the Riemannian product $([s_0, +\infty) \times \{\psi = s_0\}, d\rho \otimes d\rho + g_{\{\psi = s_0\}})$. In this case, in particular, $\partial\Omega$ is a connected totally geodesic submanifold.

Analogously, these properties are shared by Ψ_{∞} .

Theorem 1.55 (Monotonicity-Rigidity Theorem for Ψ_{∞}). Let (M, g) be a complete parabolic manifold with Ric ≥ 0 . Then, the function $\Psi_{\infty} : [0, \infty) \to \mathbb{R}$ defined by (13) is monotone

nonincreasing. Moreover, if $x_s \in \{\psi = s\}$ is such that

$$|\mathrm{D}\psi|(x_s) = \sup_{\{\psi=s\}} |\mathrm{D}\psi|,$$

then

$$H(x_s) = -\frac{\partial}{\partial \nu_s} \log |D\psi| \ge 0, \tag{1.6.2}$$

where v_s is the unit normal to $\{\psi = s\}$ given by $v = D\psi/|D\psi|$. Moreover, equality holds in the inequality of (1.6.2) for some $s = s_0 \in [0, \infty)$ or $\Psi_\infty(s_1) = \Psi_\infty(s_0)$ for some $s_1 > s_0$ if and only if $\{\psi \geq s_0\}, g\}$ is isometric to the Riemannian product $([s_0, +\infty) \times \{\psi = s_0\}, d\rho \otimes d\rho + g_{\{\psi = s_0\}})$. In this case, in particular, $\partial\Omega$ is a connected totally geodesic submanifold.

As for the proofs of Theorems 1.25 and 1.26, the triggering factor implying the monotonicity of the above functions is a Bochner-type identity. In the parabolic case, it reads, for $\beta \ge 0$,

$$\Delta |\mathrm{D}\psi|^{\beta} = \beta |\mathrm{D}\psi|^{\beta-2} \left[|\mathrm{D}\mathrm{D}\psi|^2 + (\beta-2) \left| \mathrm{D}|\mathrm{D}\psi| \right|^2 + \mathrm{Ric}(\mathrm{D}\psi,\mathrm{D}\psi) \right]. \tag{1.6.3}$$

The details of the above basic computation are analogous to those giving the Bochner-type identity (1.3.35).

As done in Section 1.3, we first show the monotonicity of Ψ_{∞} , together with the related rigidity statements.

Proof of Theorem 1.55. Define now the auxiliary function w_{α}

$$w_{\alpha} = |\mathrm{D}\psi|^2 (\psi + 1)^{-\alpha}.$$

Applying (1.6.3) with $\beta = 2$, a straightforward computation gives

$$\Delta w_{\alpha} + 2\alpha \left\langle Dw_{\alpha} \mid D\log(\psi + 1) \right\rangle = 2 \left[|DD\psi|^{2} + Ric(D\psi, D\psi) \right] + \alpha(1 - \alpha)|D\psi^{2}|w_{\alpha}.$$

If $\alpha \in (0,1)$, we see that the right hand side of the equation above is nonnegative, and in particular, by the Maximum Principle applied in $\{s \leq \psi \leq S\}$ we find that that the maximum of w_{α} in such a set is achieved on $\{\psi = s\} \cup \{\psi = S\}$. Since by Theorem 1.17 the function $|D\psi|^2$ is uniformly bounded, and then w_{α} vanishes at infinity, we infer that $\sup_{\{\varphi \geq s\}} w_{\alpha} = \sup_{\{\varphi = s\}} w_{\alpha}$ for any $\alpha \in (0,1)$. Letting $\alpha \to 0$, we get the desired upper bound for $|D\psi|^2$, giving as a consequence the monotonicity of Ψ_{∞} . The inequality (1.6.2) follows as in the proof of Theorem 1.26 by the property of x_s to be a maximum point.

Once the monotonicity and (1.6.2) are established, the rigidity statements are obtained exactly as in the proof of Theorem 1.26, using that, by (1.6.3)

$$\Delta |D\psi|^2 \ge 0$$

in place of (1.3.34). The case of singular level sets is treated as there too.

Now, we prove the Monotonicity-Rigidity Theorem for parabolic manifolds with non-negative Ricci curvature.

Proof of Theorem 1.54. Observe that the right hand side of (1.6.3) is nonnegative if $\beta \ge (n-2)/(n-1)$, by means of the refined Kato's inequality (1.3.7) for harmonic functions. Analogously to (1.3.42), we first prove, applying the Divergence Theorem to the vector

field $Z = |D\psi|^{\beta}D\psi$ in the set $\{s_0 \le \psi \le s\}$, that

$$\Psi_{\beta}(s) - \Psi_{\beta}(s_0) = \int_{\{s_0 \le \varphi \le s\}} \operatorname{div} \left(|D\psi|^{\beta} D\psi \right) d\mu = \int_{\{\sigma_0 \le \varphi \le s\}} \left\langle D|D\psi|^{\beta} \left| \frac{D\psi}{|D\psi|} \right\rangle d\mu.$$
(1.6.4)

Applying the Divergence Theorem on to the vector field $W_{\varepsilon} = \chi_{\varepsilon}(|D\psi|^2)W$, with $W = D|D\psi|^{\beta}$ and χ_{ε} the same cut-off function as in the proof of Lemma 1.29, we can show that

$$\int_{\{\psi=S\}} \left\langle \mathbf{D} | \mathbf{D} \psi |^{\beta} \left| \frac{\mathbf{D} \psi}{|\mathbf{D} \psi|} \right\rangle d\sigma - \int_{\{\psi=s\}} \left\langle \mathbf{D} | \mathbf{D} \psi |^{\beta} \left| \frac{\mathbf{D} \psi}{|\mathbf{D} \psi|} \right\rangle d\sigma \right| =$$

$$= \beta \int_{\{s \leq \psi \leq S\}} |\mathbf{D} \psi|^{\beta-2} \left[|\mathbf{D} \mathbf{D} \psi|^2 + (\beta-2) |\mathbf{D} | \mathbf{D} \psi| \right]^2 + \text{Ric}(\mathbf{D} \psi, \mathbf{D} \psi) \right] d\mu$$
(1.6.5)

everytime s < S are regular values. Sard's Theorem then allows to extend the function

$$s \to \int\limits_{\{\psi=s\}} \left\langle \mathrm{D}|\mathrm{D}\psi|^{\beta} \left| \frac{\mathrm{D}\psi}{|\mathrm{D}\psi|} \right\rangle \,\mathrm{d}\sigma \right|$$

to a continuous function on $[0, \infty)$, and finally, by applying the coarea formula to (1.6.4), this permits to apply the Fundamental Theorem of Calculus in order to get

$$\Psi'_{\beta}(s) = -\beta \int_{\{\psi=s\}} \left\langle \mathrm{D} |\mathrm{D}\psi|^{\beta} \left| \frac{\mathrm{D}\psi}{|\mathrm{D}\psi|} \right\rangle \, \mathrm{d}\sigma, \right.$$

where the above right hand side is defined by density also on singular values. In particular, with this convention, (1.6.5) yields

$$\Psi'_{\beta}(S) - \Psi'_{\beta}(s) = \beta \int_{\{s \le \psi \le S\}} |D\psi|^{\beta - 2} \Big(\operatorname{Ric}(D\psi, D\psi) + |DD\psi|^2 + (\beta - 2) |D|D\psi|^2 \Big) d\mu \ge 0.$$
(1.6.6)

Combining the uniform bound on $|D\psi|$ given in Theorem 1.17 with the constancy of

$$s \to \int_{\{\psi=s\}} |\mathsf{D}\psi| \,\mathrm{d}\sigma,$$

we obtain, as in Corollary 1.27, that Ψ_{β} is uniformly bounded in s for any $\beta \geq 0$. We can thus argue as in the conclusion of the proof of Theorem 1.25 to pass to the limit as $S \to +\infty$ in (1.6.6) and obtain, plugging also the Kato-type identity (1.22), the monotonicity formula (1.6.1). The rigidity part of the statement is obtained exactly as for that of Theorem 1.25.

1.6.1 The Enhanced Kasue's Theorem

The Enhanced Kasue's Theorem 1.56 now follows at once.

Theorem 1.56 (Enhanced Kasue's Theorem). *Let* (M, g) *be a complete noncompact Riemannian manifold with* Ric ≥ 0 , and let $\Omega \subset M$ be a bounded and open subset with smooth boundary. Then, the following assertions hold true.

(i) If (M, g) is nonparabolic, then, for every $\beta \ge (n-2)/(n-1)$,

$$\sup_{\partial\Omega} \mathbf{H} \geq \frac{1}{\int_{\partial\Omega} |\mathrm{D}u|^{\beta} \, \mathrm{d}\sigma} \left[\left(\frac{n-1}{n-2} \right) U_{\beta}(0) + \frac{1}{\beta} \frac{\mathrm{d}U_{\beta}}{\mathrm{d}t}(0) \right] > 0, \tag{1.6.7}$$

and

$$\sup_{\partial\Omega} \mathbf{H} \geq \left(\frac{n-1}{n-2}\right) |\mathbf{D}u|(x_1) - \frac{\partial}{\partial \nu} \log \frac{|\mathbf{D}u|}{u^{\frac{n-1}{n-2}}}(x_1) \geq \left(\frac{n-1}{n-2}\right) \sup_{\partial\Omega} |\mathbf{D}u| > 0, \quad (1.6.8)$$

where H is the mean curvature of $\partial\Omega$, $\nu = -Du/|Du|$ and x_1 is defined in Theorem 1.21.

(ii) If (M,g) is parabolic, then, for every $\beta \ge (n-2)/(n-1)$,

$$\sup_{\partial \Omega} H \geq -\frac{1}{\int_{\partial \Omega} |D\psi|^{\beta} d\sigma} \left[\frac{1}{\beta} \frac{d\Psi_{\beta}}{ds}(0) \right] \geq 0, \tag{1.6.9}$$

and

$$\sup_{\partial \Omega} H \ge -\frac{\partial}{\partial \nu} \log |D\psi|(x_0) \ge 0 \tag{1.6.10}$$

where H is the mean curvature of $\partial\Omega$, $v = D\psi/|D\psi|$ and x_0 is defined in Theorem 1.55. Moreover, the second inequality in (1.6.9) and the second inequality in (1.6.10) hold with the equality sign if and only if $(M \setminus \Omega, g)$ is isometric to the Riemannian product $([0, +\infty) \times \partial\Omega, d\rho \otimes d\rho + g_{\partial\Omega})$ and $\partial\Omega$ is a totally geodesic connected submanifold of (M, g).

Kasue's Theorem follows immediately as a corollary.

Corollary 1.57 (Kasue's Theorem). Let (M,g) be a complete noncompact Riemannian manifold with Ric ≥ 0 and let $\Omega \subset M$ be a bounded and open subset with smooth boundary such that $H_{\partial\Omega} \leq 0$ on $\partial\Omega$. Then $(M \setminus \Omega, g)$ is isometric to the Riemannian product $([0, +\infty) \times \partial\Omega, dr \otimes dr + g_{\partial\Omega})$ and $\partial\Omega$ is a totally geodesic connected submanifold of (M,g).

Proof of Theorem 1.56 and Corollary 1.57. Assume first that (M, g) is nonparabolic. Then, it is sufficient to use (1.3.3) and (1.3.4) at t = 1 to get

$$\left(\frac{n-1}{n-2}\right)U_{\beta}(0) + \frac{1}{\beta}\frac{\mathrm{d}U_{\beta}}{\mathrm{d}t}(0) = \int\limits_{\partial\Omega} \mathrm{H}\,|\mathrm{D}u|^{\beta}\,\mathrm{d}\sigma \leq \sup\limits_{\partial\Omega} \mathrm{H}_{\partial\Omega}\int\limits_{\partial\Omega} |\mathrm{D}u|^{\beta}\,\mathrm{d}\sigma,$$

that is (1.6.7). If (M,g) is parabolic, one can prove inequality (1.6.9) in a completely analogous fashion. The inequalities (1.6.8) and (1.6.10) on the other hand follow immediately from (1.3.6) and (1.6.2). Since $U_{\beta} > 0$, it is easy to deduce from (1.6.7) and (1.6.9) that $H_{\partial\Omega} \leq 0$ on $\partial\Omega$ if and only if (M,g) is parabolic and $d\Psi/ds_{\beta}(0) = 0$. The rigidity statements of Theorems 1.54 and 1.55 imply at once the rigidity statement in Theorem 1.56, that in particular implies Corollary 1.57.

Corollary 1.57 can also be interpreted as a rigidity statement when the equality is attained in (1.4.29) if AVR(g) = 0. The following is then a direct consequence of Theorem 1.44 and Corollary 1.57.

Corollary 1.58. *Let* (M, g) *be a complete noncompact Riemannian manifold with* Ric ≥ 0 . *If* $\Omega \subset M$ *is a bounded subset with smooth boundary, then*

$$AVR(g)|S^{n-1}| \leq \int_{\partial\Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma.$$
 (1.6.11)

If AVR(g) > 0, then equality in (1.6.11) holds if and only if $(M \setminus \Omega, g)$ is isometric to

$$\left(\left[r_0,+\infty\right)\times\partial\Omega,\ \mathrm{d} r\otimes\mathrm{d} r+(r/r_0)^2g_{\partial\Omega}\right),\qquad with\quad r_0=\left(\frac{|\partial\Omega|}{\mathrm{AVR}(g)\,|\mathbb{S}^{n-1}|}\right)^{\frac{1}{n-1}}.$$

In particular, $\partial\Omega$ is a connected submanifold with constant mean curvature. If AVR(g) = 0, equality holds in (1.6.11) if and only if $(M \setminus \Omega, g)$ is isometric to a Riemannian product $([0, +\infty) \times \partial\Omega, dr \otimes dr + g_{\partial\Omega})$. In particular, $\partial\Omega$ is a connected totally geodesic submanifold of (M, g).

1.6.2 A Maximum Principle/Splitting Theorem for manifolds with two boundary components

The Splitting Theorem by Kasue, that we rediscovered in Corollary 1.57, is known to be true also in a compact versions. Namely, it is proved in [Kas83, Theorem B 1] or in [CK92, Theorem 1] that if a compact manifold (M,g) has nonnegative Ricci curvature and boundary $\partial M = N \sqcup N_1$, with N a compact connected submanifold, such that ∂M has nonpositive mean curvature, then it splits a compact cylinder $([0,T]\times N, \,\mathrm{d} t\otimes \mathrm{d} t + g_N)$. We show that we can recover also this result by means of a suitable boundary value problem. As in Theorem 1.56, this new approach actually bounds from below the supremum of the mean curvature of ∂M in terms of the nonnegative normal derivative on the boundary of a suitable sub-harmonic function, that vanishes exactly when the splitting occurs. In spite of this enhancement, we point out that our assumptions are more restrictive than those of [Kas83, Theorem B 1], in that we also ask for N_1 to be a compact smooth submanifold. The reason for this extra condition is in the definition of a boundary value problem with Dirichlet data on ∂M . This seems to be a delicate issue in case of noncompact boundary, and we are not intentioned to address it here.

Theorem 1.59. Let (M,g) be a compact Riemannian manifold with Ric ≥ 0 and nonempty boundary $\partial M = N \sqcup N_1$, where N and N_1 are complete connected smooth manifolds of dimension n-1. Let v the solution to the boundary value problem

$$\begin{cases}
\Delta v = 0 & \text{in } M \\
v = 0 & \text{on } N \\
v = 1 & \text{on } N_1.
\end{cases}$$
(1.6.12)

Let $x \in \partial M$ realise $|Dv|(x) = \sup_{\partial M} |Dv|$. Then, we have

$$\sup_{\partial M} \mathbf{H} \ge -\frac{\partial}{\partial \nu} \log |\mathbf{D}v|(x) \ge 0, \tag{1.6.13}$$

where H is the mean curvature of ∂M with respect to the inward pointing unit normal v. Moreover, equality holds in the second inequality above if and only if (M,g) splits as $([0,T] \times N, dt \otimes dt + g_N)$. In particular, this happens if and only H = 0 on the whole ∂M .

Proof. Since v satisfies (1.6.12), the Bochner inequality yields

$$\Delta |Dv|^2 = 2 [|DDv|^2 + Ric(Dv, Dv)].$$
 (1.6.14)

In particular, by the Maximum Principle

$$\sup_{M} |Dv|^2 = \sup_{\partial M} |Dv|^2$$

Let then x as in the statement be the point on the boundary of N where the maximum of $|Dv|^2$ is achieved. Letting v the inward normal, it is immediately checked, as in the proof of Theorems 1.3 and 1.55 that

$$\frac{\partial}{\partial v}|Dv|^2(x) = -2H|Dv|^2(x) \le 0, \tag{1.6.15}$$

where the sign is due to the property of x to be a maximum point. This already proves (1.6.13). Assume now that equality holds in the second inequality of (1.6.13), or equivalently in the inequality of (1.6.15). By the Hopf Lemma, this can happen only if $|Dv|^2$ is constant in M. By (1.6.14) and the assumption of nonnegative Ricci curvature, this implies that the Hessian of v vanishes, and this yields, by Corollary 1.23, the claimed splitting.

1.7 The isoperimetric inequality for 3-manifolds

As already discussed in the Introduction, we show here how to use our Willmore inequality (1.4.29) to improve a result stated by Huisken in [Hui], in which the infimum of the Willmore energy is characterised in terms of the infimum of the isoperimetric ratio on 3-manifolds with nonnegative Ricci curvature.

Theorem 1.60 (AVR(g) & Isoperimetric Constant). Let (M,g) be a complete noncompact 3-manifold with Ric ≥ 0 and Euclidean Volume Growth. Then,

$$\inf \frac{|\partial \Omega|^3}{36\pi |\Omega|^2} = \inf \frac{\int_{\partial \Omega} H^2 d\sigma}{16\pi} = AVR(g), \tag{1.7.1}$$

where the infima are taken over bounded and open subsets $\Omega \subset M$ with smooth boundary. In particular, the following isoperimetric inequality holds for any bounded and open $\Omega \subset M$ with smooth boundary

$$\frac{|\partial\Omega|^3}{|\Omega|^2} \ge 36\pi \,\text{AVR}(g). \tag{1.7.2}$$

Moreover, equality is attained in (1.7.2) if and only if $M = \mathbb{R}^3$ and Ω is a ball.

We point out that in the 3-dimensional case Theorem 1.60 extends Theorem 1.47 to any complete noncompact manifold with $\text{Ric} \geq 0$ and Euclidean volume growth, with no curvature assumptions at infinity. Actually, we can even relax the volume growth assumption, see Theorem 1.68.

1.7.1 Huisken's argument.

Let us briefly present Huisken's heuristic argument to deduce an isoperimetric inequality from Willmore's through the *mean curvature flow*. We first recall that a sequence of orientable hypersurfaces $F_t(x): \Sigma \to M$ immersed in a Riemann manifold (M,g), evolves through the Mean Curvature Flow if

$$\frac{\mathrm{d}}{\mathrm{d}t}F_t(x) = -\mathrm{H}(t,x)\,\nu(t,x)\,,$$

where H is the mean curvature of $\Sigma_t = F_t(\Sigma)$ and ν is its (exterior, in the case where Σ_t is the boundary of a domain) unit normal. Accordingly, we say that $\{\Omega_t\}$ is a mean curvature flow if the boundaries are evolving through mean curvature flow in the sense

explained above. Let then Ω be an open bounded set with smooth boundary and let $\{\Omega_t\}$, with $t \in [0,T)$, be a smooth mean curvature flow starting from Ω . Suppose, in addition, that

$$\lim_{t \to T^{-}} |\Omega_t| = 0. \tag{1.7.3}$$

Consider, for some constant C > 0 to be defined later, the *isoperimetric difference*

$$D(t) = |\partial \Omega_t|^{3/2} - C|\Omega_t|. \tag{1.7.4}$$

Taking derivatives in t, and using standard formulas (see for example [HP99, Theorem 3.2]), one finds

$$\frac{\mathrm{d}}{\mathrm{d}t}D(t) = -\frac{3}{2} |\partial\Omega_t|^{1/2} \int_{\partial\Omega_t} \mathrm{H}^2 \,\mathrm{d}\sigma + C \int_{\partial\Omega_t} \mathrm{H} \,\mathrm{d}\sigma,$$

that, through Hölder inequality, gives

$$\frac{\mathrm{d}}{\mathrm{d}t}D(t) \, \leq \, \left(|\partial\Omega_t| \int\limits_{\partial\Omega_t} H^2 \, \mathrm{d}\sigma \right)^{1/2} \, \left[C \, - \, \frac{3}{2} \left(\int\limits_{\partial\Omega_t} H^2 \, \mathrm{d}\sigma \right)^{1/2} \right].$$

Thus, if we choose C such that

$$C \leq \frac{3}{2} \left(\int_{\partial \Omega} H^2 d\sigma \right)^{1/2} \tag{1.7.5}$$

for any bounded and smooth $\Omega \subset M$, $t \mapsto D(t)$ is nonincreasing. This implies that

$$D(0) = |\partial \Omega|^{3/2} - C|\Omega| \ge \lim_{t \to T^{-}} D(t) \ge 0,$$

where we have also used (1.7.3). The above comparison in particular gives the (possibly non sharp) isoperimetric inequality

$$\frac{|\partial\Omega|^{3/2}}{|\Omega|} \geq C.$$

In [Hui], the constant C is chosen to be the infimum of the right hand side of (1.7.5), when Ω varies in the class of outward minimising sets.

1.7.2 Tools from the Mean Curvature Flow of mean-convex domains.

We are first concerned with the accurate justification of the above computations. This will be accomplished with the help of a couple of important results due to Schulze and White, respectively. In the first part of our treatment we assume that the boundary $\partial\Omega$ of the bounded set Ω is smooth and mean-convex, that we understand as H>0. We will see later how to deal with the general cases.

Since the Mean Curvature Flow (MCF for short) is likely to develop singularities, one needs to consider an appropriate weak notion in order to state the following useful results. In particular, we consider the Weak Mean Curvature Flow in the sense defined in [ES91]. A special case of the regularity theorem [Whi00, Theorem 1.1] gives

Theorem 1.61 (White's Regularity Theorem). *Let* (M, g) *be a complete noncompact* 3-*dimensional Riemannian manifold, let* $\Omega \subset M$ *be a bounded set with smooth mean-convex boundary and let*

 $\{\Omega_t\}_{t\in[0,T)}$ be its Weak Mean Curvature Flow. Then, the boundary of Ω_t is smooth for almost every $t\in[0,T)$.

We point out that the maximal time *T* might *a priori* be infinite on a general Riemannian manifold. We are going to combine the above regularity result with the following special case of [Sch08, Proposition 7.2], that is a weak version of the monotonicity of the isoperimetric ratio. It can be checked, indeed, that the computations performed to obtain such a result do not involve the geometry of the underlying manifold.

Theorem 1.62 (Schulze). Let (M,g) be a 3-dimensional Riemannian manifold, let $\Omega \subset M$ be a bounded set with smooth mean-convex boundary and let $\{\Omega_t\}_{t\in[0,T)}$ be its Weak Mean Curvature Flow. Assume there exists a universal constant $C \geq 0$ such that

$$C \leq \frac{3}{2} \left(\int\limits_{\partial\Omega_t} \mathrm{H}^2 \,\mathrm{d}\sigma \right)^{1/2}$$

for almost every $t \in [0,T)$. Then, the isoperimetric difference $t \mapsto D(t)$ defined as in (1.7.4) using the constant C, is nonincreasing for every $t \in [0,T)$.

Remark 1.63. A different tool that might be used to deal with the singularities would be the theory of the Mean Curvature Flow with surgery, recently developed by Brendle and Huisken in [BH16] and [BH18]. On this regard, one should first make clear whether the monotonicity of the isoperimetric difference survives the surgeries.

The following theorem provides a complete description of the long time behaviour of the Weak MCF of a surface moving inside a 3-dimensional Riemannian manifold, and it can be readily deduced from [Whi00, Theorem 11.1].

Theorem 1.64 (Long time behaviour of MCF). Let (M,g) be a 3-dimensional Riemannian manifold, let $\Omega \subset M$ be a bounded set with smooth mean-convex boundary and let $\{\Omega_t\}_{t\in[0,T)}$ be its Weak Mean Curvature Flow. If $|\Omega_t|$ and $|\partial\Omega_t|$ do not vanish at finite time as $t\to T^-$, then Ω_t converges smoothly to a subset K, and the boundary of any connected component of K is a stable minimal submanifold.

As a consequence, if (M, g) contains no bounded subsets with minimal boundary, the Weak MCF of a bounded set with mean-convex boundary is going to vanish. In particular, combining Corollary 1.57 with Theorem 1.64, one gets the following corollary.

Corollary 1.65. Let (M,g) be a complete, noncompact, 3-dimensional Riemannian manifold with Ric ≥ 0 and no cylindrical ends, let $\Omega \subset M$ be a bounded set with smooth mean-convex boundary and let $\{\Omega_t\}_{t\in[0,T)}$ be its Weak Mean Curvature Flow. Then, T is finite and $|\Omega_t|$ and $|\partial\Omega_t|$ tend to 0 as $t\to T^-$.

1.7.3 Some properties of the strictly outward minimising hull

In order to prove the isoperimetric inequality for any set Ω with smooth and possibly not mean-convex boundary, we are going to consider the strictly outward minimising hull Ω^* . This notion, with roots in [BT84], manifested its importance in the celebrated construction of weak solutions the Inverse Mean Curvature Flow of [HI01], that led to the proof of the Riemannian Penrose Inequality. We are going to extensively discuss this notion in Chapter 3. For the time being, we limit ourselves to define briefly the relevant notions and state the fundamental facts needed in the proof. We first recall the notion of *strictly outward minising sets*, a variational property of bounded sets with finite

perimeter that played a key role in the foundation of the Weak Inverse Mean Curvature Flow worked out in [HI01]. Simply, we say that a bounded set with finite perimeter $E \subset \mathbb{R}^n$ is *outward minimising* if for any bounded set with finite perimeter $F \supset E$ with finite perimeter and $|F \setminus E| > 0$ it holds $P(F) \ge P(E)$. We say that it is *strictly* outward minimising if for any such set F we have in fact P(F) > P(E). We can thus define the *strictly outward minimising hull* Ω^* of an open bounded set Ω as the intersection of all the strictly outward minimising sets containing Ω . What mostly matters to us now, is that on any complete noncompact Riemannian manifold with nonnegative Ricci curvature and Euclidean volume growth Ω^* is an open bounded strictly outward minimising set with finite perimeter containing Ω and satisfying

$$P(\Omega^*) = \inf\{P(E) \mid \Omega \subseteq E\}.$$

In particular, if Ω is outward minimising then Ω^* satisfies $P(\Omega) = P(\Omega^*)$. This is (part of) the content of Theorem 3.1, that is the main result of Chapter 3. Let us just briefly remark that complete noncompact Riemannian manifolds with nonnegative Ricci curvature satisfy (i) in that statement, compare with the discussion in the Introduction. By [SZW91], $\partial\Omega^*$ enjoys $\mathscr{C}^{1,1}$ -regularity where it touches the obstacle Ω , and it is a locally area minimising hypersurface where it does not. See Theorem 3.14 for the complete statement. In particular, if Ω has smooth boundary, the strictly outward minimising hull also enjoys $P(\Omega^*) = |\partial\Omega^*|$, see Remark 3.16. By minimal surfaces regularity, up to ambient dimension 7, $\partial\Omega$ is then a $\mathscr{C}^{1,1}$ -hypersurface. These last deep and celebrated results on the regularity of locally area minimising hypersurfaces were pioneered in [DG61] and later completed in [Alm66] and [Sim67]. See also the comprehensive [HI01, Regularity Theorem 1.3] for an account on the regularity of solutions to more general obstacle problems. We complete this overview on the properties of $\partial\Omega$ pointing out that, by the outward minimising property, the weak mean curvature $H_{\partial\Omega^*}$ is nonnegative. This readily follows from the standard first variation argument.

We will actually flow Ω^* by mean curvature. To this aim, we will invoke [HI01, Lemma 5.6], where the authors show that bounded set E with $\mathscr{C}^{1,1}$ -boundary having nonnegative variational mean curvature can be approximated in \mathscr{C}^1 by smooth hypersurfaces with strictly positive mean-curvature. These approximating hypersurfaces happen be boundaries of sets E_{ε} that are strictly outward minimising if E was. Moreover, the Willmore energy of the approximators converges to the Willmore energy (in terms of the variational mean curvature) of ∂E . Interestingly, the approximating sequence is built through an appropriate notion of mean curvature flow starting from such a $\mathscr{C}^{1,1}$ -hypersurface. Such a result has found many other applications in literature, see for example [LW17, Lemma 4.2] in the ambient setting of a Kottler-Schwarzschild manifold, and the proof of [Sch08, Corollary 1.2]. Moreover, it has been generalised to \mathscr{C}^1 boundaries in [HI08, Lemma 2.6] admitting nonnegative weak mean curvature, and the approximation is also shown to take actually place in $W^{2,q}$. We include here a general statement that combines [HI01, Lemma 5.6] and [HI08, Lemma 2.6].

Lemma 1.66 (Huisken-Ilmanen's approximation lemma). Let (M,g) be a complete Riemannian manifold, let $E \subset M$ be a bounded set with \mathscr{C}^1 -boundary and let $F_0: \partial E \hookrightarrow M$ be its immersion . Assume that ∂E has nonnegative weak mean curvature, that is, it admits a nonnegative function $H \in L^1_{loc}(\Sigma)$ such that

$$\int_{\partial E} \operatorname{div}_{\Sigma} X \, d\sigma = \int_{\partial E} H \, \langle X, \nu \rangle \, d\sigma \tag{1.7.6}$$

for any compactly supported vector field X of M. Assume also that ∂E is not minimal, that is, there exist a subset $K \subset \Sigma$ of positive measure such that H > 0 on K. Then, there exists a family of bounded open sets E_{ε} with smooth boundaries ∂E_{ε} given by the immersions $F(\varepsilon, \cdot) : \partial E \hookrightarrow M$, with $\varepsilon \in (0, \varepsilon_0]$, such that

$$\frac{\mathrm{d}}{\mathrm{d}\varepsilon}F(\varepsilon,x) = -\mathrm{H}(\varepsilon,x)\nu(\varepsilon,x)$$

for any $\varepsilon \in (0, \varepsilon_0]$, where $H(\varepsilon, x)$ is the mean curvature of the immersion $F(\varepsilon, \cdot)$ at point x, $v(\varepsilon, x)$ its outer unit normal at point x, and we have

$$\lim_{\varepsilon \to 0^+} F(\varepsilon, \cdot) = F_0(\cdot)$$

in $\mathscr{C}^{1,\alpha}\cap W^{2,q}$ for any $0<\alpha<1$ and $q\geq 1$. Moreover, $H_{\partial E_\epsilon}>0$ for any $\epsilon\in(0,\epsilon_0]$, and if E is strictly outward minimising then so is E_ϵ .

We point out that the weak mean curvature of a closed hypersurface Σ is easily verified to exist if Σ is of class $\mathscr{C}^{1,1}$. In the following remark we show how through a parabolic maximum principle the mean curvature flow naturally approximates mean-convex hypersurface with strictly mean-convex ones. This behaviour lies at the core of the proof of Lemma 1.66 given in [HI01], together with fine higher derivative estimates of [EH91].

Remark 1.67. Observe that if Σ is a (non minimal) closed \mathscr{C}^2 hypersurface with $H_{\Sigma} \geq 0$, then the approximation of Σ by means of a family of smooth mean-convex hypersurfaces $\{\Sigma_{\varepsilon}\}_{\varepsilon>0}$ is a straightforward procedure. Indeed, it is sufficient to run the MCF starting at Σ for short time (see [Man11] for an account about the classical existence theory), say until some time $\varepsilon_0 > 0$. This provides a family of hypersurfaces $\{\Sigma_{\varepsilon}\}_{\varepsilon \in (0,\varepsilon_0]}$, whose mean curvatures satisfy (see e.g. [HP99, Theorem 3.2]) the following reaction-diffusion equation,

$$\frac{\partial}{\partial \varepsilon} H = \Delta H + H(|h|^2 + Ric(\nu, \nu)),$$

where h is the second fundamental form of the evolving hypersurface and Ric is the Ricci tensor of the ambient manifold. Then, a standard maximum principle for parabolic equations (see e.g. Theorem 7 and subsequent remarks in [PW84]) shows that $H_{\Sigma_{\epsilon}} > 0$ for every $\epsilon \in (0, \epsilon_0]$, unless H_{Σ} is constantly null. The latter case is excluded by the non minimality of Σ .

Finally, we have at hand all the ingredients that allow to completely justify the computations of Subsection 1.7.1 and in turn to prove Theorem 1.60.

Proof of Theorem 1.60. The following argument follows the lines of the proof of [Sch08, Corollary 1.2]. Let us first suppose that the boundary of $\partial\Omega$ is strictly mean-convex, that is, $H_{\partial\Omega} > 0$. Let $\{\Omega_t\}_{t\in[0,T)}$ be a mean curvature flow starting from Ω . Then, by Theorem 1.61, for almost any $t\in[0,T)$ the boundary $\partial\Omega_t$ is a smooth submanifold. Let C be defined as

$$C = \inf \left\{ \frac{3}{2} \left(\int_{\partial \Omega} H^2 d\sigma \right)^{1/2} \middle| \Omega \subset M \text{ bounded set with smooth boundary} \right\}. \quad (1.7.7)$$

Then, Theorem 1.62 guarantees that, with the above choice of C, the isoperimetric difference $t \mapsto D(t)$ defined in (1.7.4) is nonincreasing for $t \in [0, T)$. Moreover, by Corollary 1.65 D(t) tends to 0 as $t \to T^-$. This implies the inequality

$$\frac{|\partial\Omega|^{3/2}}{|\Omega|}\,\geq\,C$$

for any Ω with smooth mean-convex boundary. If this is not the case, take the minimising hull Ω^* of Ω (see [HI01, Section 1] for details). As discussed above $\partial\Omega^*$ is a $\mathscr{C}^{1,1}$ hypersurface. Observe that, by the minimising property, $|\partial\Omega^*| \leq |\partial\Omega|$, while trivially $|\Omega^*| \geq |\Omega|$. Hence, proving a lower bound on the isoperimetric ratio for Ω^* readily implies that the same lower bound holds for Ω . Moreover, again by the minimising property, we have that $H_{\partial\Omega^*} \geq 0$ (see also [HI01, (1.15)]). Also notice that $\partial\Omega^*$ cannot be minimal, for otherwise (M,g) will have a cylindrical end, in virtue of Corollary 1.57, against the Euclidean volume growth assumption. By Lemma 1.66, we find a sequence bounded sets E_ε with smooth and strictly mean-convex boundary approximating $\partial\Omega^*$ in \mathscr{C}^1 . Arguing as above, we thus obtain the isoperimetric inequality

$$\frac{|\partial E_{\varepsilon}|^{3/2}}{|E_{\varepsilon}|} \geq C,$$

that, through letting $\varepsilon \to 0^+$, gives the isoperimetric inequality for Ω^* , and, in turn, for any bounded Ω with smooth boundary. Combining it with our Willmore inequality (1.4.29), we get

$$\inf \frac{|\partial \Omega|^3}{36\pi |\Omega|^2} \ge \inf \frac{\left(\int_{\partial \Omega} H^2 d\sigma\right)}{16\pi} \ge AVR(g), \tag{1.7.8}$$

where the infima are taken over any bounded Ω with smooth boundary. We now want to prove that the equality sign hold in both the above inequalities, as stated in (1.7.1). To do so, we fix a point $O \in M$ and we observe that by the Bishop-Gromov Theorem, we can find, for every $\delta > 0$, a radius R_{δ} such that

$$\frac{|\partial B(O, R_{\delta})|^3}{36\pi |B(O, R_{\delta})|^2} \leq \text{AVR}(g) + \delta.$$

Observe that we can suppose $\partial B(O, R_{\delta})$ to be smooth. Otherwise, it suffices to consider in place of $B(O, R_{\delta})$ a smooth set whose perimeter and volume approximate $|\partial B(O, R_{\delta})|$ and $|B(O, R_{\delta})|$, respectively (this can be done by standard tools, see e.g. [Mag12, Remark 13.2]). This proves that

$$\inf \left\{ \frac{|\partial \Omega|^3}{36\pi |\Omega|^2} \ \middle| \ \Omega \subset M \text{ bounded and smooth} \right\} \leq \text{AVR}(g).$$

Combining the above inequality with (1.7.8), gives (1.7.1).

To prove the rigidity statement, we assume now that (1.7.2) holds with the equality sign for a smooth and bounded $\Omega \subset M$. In virtue of (1.7.7) and of (1.7.1), we have that

$$C = \sqrt{36\pi AVR(g)},$$

By the minimising property, Ω^* satisfies the same equality (recall that we actually proved the isoperimetric inequality for minimising hulls). We claim that it also holds for any

approximating E_{ε} as above. Indeed, by Lemma 1.66, this family is a smooth mean curvature flow, and then, by the monotonicity of the isoperimetric difference, for any fixed $\varepsilon_1 \in (0, \varepsilon_0]$ we have

$$D(\varepsilon) \geq D(\varepsilon_1) \geq 0$$

for every $\varepsilon \in (0, \varepsilon_1)$. Since ∂E_{ε} converges strongly enough to $\partial \Omega^*$ as $\varepsilon \to 0^+$, and on Ω^* the isoperimetric difference is 0, $D(\varepsilon) \to 0^+$ as $\varepsilon \to 0^+$, and thus $D(\varepsilon_1) = 0$ as well. Since ε_1 was arbitrarily chosen, E_{ε} satisfies the equality in the isoperimetric inequality for $\varepsilon \in (0, \varepsilon_0]$, as claimed. In particular, from the same computations as in Subsection 1.7.1, for any fixed $\varepsilon \in (0, \varepsilon_0]$, we have that

$$0 = \frac{\mathrm{d}D}{\mathrm{d}\varepsilon}(\varepsilon) \leq \left(|\partial E_{\varepsilon}| \int_{\partial E_{\varepsilon}} \mathrm{H}^2 \, \mathrm{d}\sigma \right)^{1/2} \left[\sqrt{36\pi \mathrm{AVR}(g)} - \frac{3}{2} \left(\int_{\partial E_{\varepsilon}} \mathrm{H}^2 \, \mathrm{d}\sigma \right)^{1/2} \right] \leq 0.$$

This implies that the equality sign holds in the Willmore inequality for Σ_{ε} , and thus, by the rigidity statement in Theorem 1.44, $(M \setminus E_{\varepsilon}, g)$ is isometric to the truncated cone

$$\Big(\big[r_{\varepsilon}, +\infty \big) imes \partial E_{\varepsilon}, \ \mathrm{d} r \otimes \mathrm{d} r + (r/r_{\varepsilon})^2 g_{\partial E_{\varepsilon}} \Big), \qquad \mathrm{where} \quad r_{\varepsilon} = \Big(\frac{|\partial E_{\varepsilon}|}{4\pi \mathrm{AVR}(g)} \Big)^{1/2}.$$

Hence, it is easily seen that the MCF $\{\partial E_{\varepsilon}\}_{\varepsilon \in (0,\varepsilon_0]}$ is given by totally umbilic hypersurfaces coinciding with the cross sections of the cone

$$(M \setminus E_{\varepsilon}, g) \cong \left(\left[r_{\varepsilon}, +\infty \right) \times \partial \Omega^*, \ dr \otimes dr + \frac{4\pi \text{AVR}(g)}{|\partial \Omega^*|} r^2 g_{\partial \Omega^*} \right).$$
 (1.7.9)

We now claim that the MCF $\{\partial E_{\varepsilon}\}_{\varepsilon>0}$ does not develop singularities before the extinction time ε^* . Letting $(0, \varepsilon_*)$ be the maximal interval of existence of the smooth MCF, we claim that $\varepsilon_* = \varepsilon^*$. In fact, from (1.7.9) one can easily see that the mean curvature of Σ_{ε} is given by $(n-1)/r_{\varepsilon}$, and in turn the squared norm of its second fundamental form is equal to $(n-1)/r_{\varepsilon}^2$. It follows then by [Hui86, Theorem 7.1] that ε_* is such that $r_{\varepsilon_*} = 0$, and thus coincides with the extinction time of the flow, i.e., $\varepsilon^* = \varepsilon_*$. We have hence deduced the isometry

$$(M\setminus\{O\},g)\cong\Big(\big(0,+\infty)\times\partial\Omega^*,\ \mathrm{d} r\otimes\mathrm{d} r+\frac{4\pi\mathrm{AVR}(g)}{|\partial\Omega^*|}r^2g_{\partial\Omega^*}\Big),$$

for some $O \in M$. In particular, the surface area of the geodesic balls centered at O decays as $4\pi r^2 \text{AVR}(g)$, and, since g is smooth at O, this implies that AVR(g) = 1. By Bishop-Gromov, we infer that (M,g) is isometric to $(\mathbb{R}^3, g_{\mathbb{R}^3})$ and $\partial \Omega^*$ is isometric to a sphere. This implies that $\Omega = \Omega^*$, since, otherwise, the mean curvature of $\partial \Omega^*$ would be null on the points not belonging to $\partial \Omega$ (compare with Theorem 3.14), leading to a contradiction. We have thus shown that Ω is a ball, completing the proof.

Observe that the proof illustrated above works with no modification also without the Euclidean volume growth assumption, that is with AVR(g) = 0, if one were able to ensure that a bounded solution to the least area problem with a bounded open set with smooth boundary exists anyway. Even more weakly, it would suffice to show that there exists an exhausting sequence of sets with smooth and mean-convex boundaries. The existence of such a sequence does not seem evident. However, in Chapter 3, Theorem 3.1, we prove that if the volume growth is uniformly superlinear in the sense of (1.7.10) below, Ω^* is a well defined set with the properties needed above. Thus, we have actually proved also the following result.

Theorem 1.68. Let (M,g) be a complete noncompact Riemannian manifold with nonnegative Ricci curvature. Assume also that there exists $O \in M$ and 1 < b < n and a constant C such that

$$C^{-1}r^b \le |B(O,r)| \le Cr^b$$
 (1.7.10)

for any $r \ge r_0$ for some fixed $r_0 > 0$. Then,

$$\inf \frac{|\partial \Omega|^3}{|\Omega|^2} = \inf \int_{\partial \Omega} H^2 d\sigma = 0.$$
 (1.7.11)

Observe that the interest in the above result lies in the second equality of (1.7.11), that generalises, in the 3-dimensional case, the examples provided at the end of Subsection 1.4.3.

1.7.4 Perspectives and applications of the Isoperimetric Inequalities

We discuss here some topics related to the Isoperimetric Inequality for 3-manifolds with nonnegative Ricci curvature. Namely, we state it in the equivalent form of Sobolev Inequality, we show how it yields the sharp constant in the Faber-Krahn Inequality, we discuss its natural extension to higher dimensions and we compare it with the existence of isoperimetric sets.

The Sobolev Inequality

Let us first observe how it translates in terms of sharp Sobolev Inequality.

Theorem 1.69 (AVR(g) & Sobolev Constant). Let (M, g) be a complete noncompact 3-manifold with Ric > 0 and Euclidean Volume Growth. Then

$$\inf_{f \in W_0^{1,1}(M)} \frac{\int_M |Df| \, d\sigma}{\left(\int_M |f|^{3/2} \, d\sigma\right)^{2/3}} = \sqrt[3]{36\pi \, \text{AVR}(g)}.$$

The equivalence between the Sobolev and the Isoperimetric Inequality is very well known, and seemingly it was first displayed along [FF60, Remark 6.6], where the argument given actually does not depend on the ambient manifold. We refer also to [Cha84, Chapter IV, Theorem 4] for a presentation.

Faber-Krahn Inequality

A very famous application of the Euclidean Isoperimetric Inequality is the celebrated Faber-Krahn inequality, stating that for any bounded $\Omega \subset \mathbb{R}^n$ with smooth boundary there holds

$$\lambda_1(\Omega) \ge \lambda_1(B_{\Omega}),\tag{1.7.12}$$

where B_{Ω} is the ball with $|\Omega| = |B_{\Omega}|$, and $\lambda_1(\Omega)$ is the first Dirichlet eigenvalue of Ω , that is the lowest and strictly greater than zero number λ such that

$$\begin{cases}
-\Delta u = \lambda u & \text{in } \Omega \\
u = 0 & \text{on } \partial\Omega
\end{cases}$$
(1.7.13)

.

admits a solution. Moreover, a rigidity statement characterises the equality case in (1.7.12), stating that equality occurs only if $B_{\Omega} = \Omega$.

It is also well known that the Faber-Krahn inequality (1.7.12) actually *just* depends on the Euclidean Isoperimetric Inequality, and in particular it holds on any Riemannian manifold supporting this structural property. This is carried out in details in the book [Cha84], see Theorem 2 in Chapter 4. As a consequence, (1.7.12) holds on Cartan-Hadamard manifolds of dimension 3 and 4, since with this assumptions the Euclidean Isoperimetric Inequality is known to be true, as proved in [Kle92] and [Cro84]. A simple generalisation of the argument in the proof of the Faber-Krahn Inequality actually allowed to provide it in closed Riemannian manifolds with Ricci curvature bounded below by a positive number, by using, in place of the Euclidean Isoperimetric Inequality, the classical Levy-Gromov Isoperimetric Inequality (see for example [GHL04, Theorem 4.6]). The details of this version of the Faber-Krahn Inequality in closed manifolds are worked out in [BM82].

Here, we point out that the Isoperimetric Inequality for 3-manifolds with nonnegative Ricci curvature (1.7.2) yields a Faber-Krahn Inequality on these manifolds. Since the isoperimetric constant is not the Euclidean one, but it depends on the Asymptotic Volume Ratio, the same is happening for the constant in the eigenvalue comparison.

Theorem 1.70 (Faber-Krahn inequality for 3-manifolds with nonnegative Ricci curvature and Euclidean Volume Growth). Let (M, g) be a complete noncompact Riemannian 3-manifold with nonnegative Ricci curvature and Euclidean Volume Growth. Let $\Omega \subset M$ be a bounded set with smooth boundary, and let $B_{\Omega} \subset \mathbb{R}^n$ satisfy $|B_{\Omega}|_{\mathbb{R}^n} = |\Omega|$. Then, we have

$$\lambda_1(\Omega) \ge \text{AVR}(g)^{2/3} \lambda_1(B_{\Omega}),$$
 (1.7.14)

where $\lambda_1(\Omega)$ is the first Dirichlet eigenvalue of Ω in M with respect to the metric g and $\lambda_1(B_\Omega)$ is the first Dirichlet eigenvalue of B_Ω in \mathbb{R}^n . Moreover, equality holds in (1.7.14) if and only if (M,g) is isometric to $(\mathbb{R}^n,g_{\mathbb{R}^n})$ and Ω is isometric to B_Ω .

It is not surprising, in light of the discussion above, that the above result follows from the Isoperimetric Inequality Theorem 1.60 by adapting the classical argument. We provide here the details, mainly to clarify how the constant AVR(g) arises in (1.7.14). It follows from the following fact, that is a slight generalisation of what is known in literature as Pólia-Szegö Principle. The argument is exactly that of [Cha84]. We sketch it, in order to clarify the role played by the isoperimetric constant.

Proposition 1.71 (Generalised Pólia-Szegö Principle). *Let* (M, g) *be a Riemannian manifold, and assume that for any bounded* $\Omega \subset M$ *with smooth boundary there holds*

$$\frac{|\partial\Omega|^n}{|\Omega|^{n-1}} \ge n^n |\mathbb{B}^n| C_{\text{iso}}$$
 (1.7.15)

for some C_{iso} independent of Ω .

Let, for $\Omega \subset M$ a bounded set with smooth boundary, $f \in \mathscr{C}^{\infty}(\overline{\Omega})$ be a positive function with f = 0 on $\partial\Omega$. Then, there exists a function $F \in W_0^{1,2}(B_{\Omega})$, where $B_{\Omega} \subset \mathbb{R}^n$ is a ball with $|\Omega| = |B_{\Omega}|_{\mathbb{R}^n}$ such that

$$\int_{\Omega} |Df|^2 d\mu \ge C_{\rm iso}^{\frac{2}{n}} \int_{B_{\Omega}} |DF|_{\mathbb{R}^n}^2 d\mu_{\mathbb{R}^n} \qquad \int_{\Omega} f^2 d\mu = \int_{B_{\Omega}} F^2 d\mu_{\mathbb{R}^n}.$$
 (1.7.16)

Moreover, equality holds in the inequality above if and only Ω satisfies equality in (1.7.15).

Proof. Let $V:[0,T] \to \mathbb{R}$, where $T = \max_{\overline{\Omega}} f$, the function defined by $V(t) = |\{f \ge t\}|$. This function is easily seen to be continuous and actually \mathscr{C}^{∞} on regular values of f.

Consider $B_t = B_{\{f \geq t\}}$, that is, the ball in \mathbb{R}^n with (Euclidean) volume equal to V(t). This defines a bijective function $\rho: [0,T] \to [0,\rho(0)]$ such that $V(t) = |B(\rho(t))|_{\mathbb{R}^n}$, where $B(\rho(t))$ is any ball of radius $\rho(t)$. Define finally $F: \overline{B_\Omega} \to \mathbb{R}$ by $F = \rho^{-1} \circ r$, where r is the distance from the center of B_Ω . It is shown exactly as in [Cha84] that with this definition the identity in (1.7.16) is satisfied. By coarea formula, we have

$$\int_{\Omega} |Df|^2 d\mu = \int_{0}^{T} \int_{\{f=t\}} |Df| d\sigma dt \ge \int_{0}^{T} |\{f=t\}|^2 \int_{\{f=t\}} \frac{1}{|Df|} d\sigma dt$$
 (1.7.17)

On the other hand, by (1.7.15),

$$|\{f=t\}|^2 \ge |\{f\ge t\}|^{2\frac{n-1}{n}}n^2|\mathbb{B}^n|^{\frac{2}{n}}C_{\text{iso}}^{\frac{2}{n}} = |B_t|_{\mathbb{R}^n}^{2\frac{n-1}{n}}n^2|\mathbb{B}^n|^{\frac{2}{n}}C_{\text{iso}}^{\frac{2}{n}} = |\partial B_t|_{\mathbb{R}^n}^2C_{\text{iso}}^{2/n}, \quad (1.7.18)$$

where in the last equality we used the fact that balls in \mathbb{R}^n satisfy equality in the Isoperimetric Inequality. Using that

$$\rho'(t) = -\frac{1}{|\partial B_t|_{\mathbb{R}^n}} \int_{\{f=t\}} \frac{1}{|\mathrm{D}f|} \, \mathrm{d}\sigma \qquad \left(\rho^{-1}\right)'(\rho(t)) = \frac{1}{\rho'(t)},$$

we get, plugging the outcome of (1.7.18) in (1.7.17) and using again the coarea formula we obtain

$$\int_{\Omega} |Df|^2 d\mu \ge C_{\text{iso}}^{2/n} \int_{0}^{T} \left[\left(\rho^{-1} \right)' \right]^2 |\partial B_t|_{\mathbb{R}^n} \rho'(t) dt = C_{\text{iso}}^{2/n} \int_{B_{\Omega}} |DF|_{\mathbb{R}^n}^2 d\mu_{\mathbb{R}^n}, \quad (1.7.19)$$

as desired.

Assume now that equality holds in the inequality of (1.7.16). Then, we deduce from (1.7.19) and (1.7.18) that

$$\int_{0}^{T} |\{f=t\}|^{2} - |\partial B_{t}|_{\mathbb{R}^{n}}^{2} C_{\text{iso}}^{\frac{2}{n}} dt = 0,$$

that, by the nonnegativity of the integrand, implies that the sets $\{f \geq t\}$ satisfy equality in the Isoperimetric Inequality of (M,g) for any regular value t. Letting $t \to 0^+$, that is possible by Sard's Theorem, we conclude that Ω satisfy equality in (1.7.15).

The Faber-Krahn inequality for 3-manifolds with nonnegative Ricci curvature of Theorem 1.70 now follows as a direct consequence of the Isoperimetric Inequality 1.60 and the above general principle

Proof of Theorem 1.70. Just recalling that the first Dirichlet eigenvalue minimises the Rayleigh quotient, we get, applying (1.7.16) to the solution of (1.7.13) with $\lambda = \lambda_1(\Omega)$, and observing that by the sharp Isoperimetric inequality (1.7.2) we have $AVR(g) = C_{iso}$, we get

$$\lambda_1(\Omega) = \frac{\int_{\Omega} |Df|^2 d\mu}{\int_{\Omega} f^2 d\mu} \ge AVR(g)^{2/3} \frac{\int_{B_{\Omega}} |DF|_{\mathbb{R}^3}^2 d\mu_{\mathbb{R}^3}}{\int_{B_{\Omega}} F^2 d\mu_{\mathbb{R}^3}} \ge AVR(g)^{2/3} \lambda_1(B_{\Omega}), \quad (1.7.20)$$

that is (1.7.14). If equality holds in (1.7.14), then, equalities hold in (1.7.20), that implies equality holds in the inequality of (1.7.16). The rigidity statement of Proposition 1.71 then shows that Ω satisfies equality in the Isoperimetric Inequality of (M, g), and then the

rigidity statement of Theorem 1.60 applies, yielding the isometry of (M, g) with $(\mathbb{R}^3, g_{\mathbb{R}^3})$ and the isometry of Ω with B_{Ω} .

Expected Isoperimetric Inequality in higher dimensions

The Isoperimetric Inequality for complete noncompact 3-manifolds with nonnegative Ricci curvature (1.7.2) asserts that the sharp isoperimetric constant is the limit of the isoperimetric quotient of geodesic balls as the radius tends to infinity. It becomes then absolutely natural to conjecture that on manifolds with nonnegative Ricci curvature there holds

$$\frac{|\partial\Omega|^n}{|\Omega|^{n-1}} \geq n^n |\mathbb{B}^n| \text{AVR}(g).$$

It should also happen that equality is achieved only on metric balls of flat \mathbb{R}^n . Needless to say that with such an equality at hand, the Faber-Krahn equality

$$\lambda_1(\Omega) \ge \text{AVR}(g)^{2/n} \lambda_1(B_{\Omega})$$

would follow by Proposition 1.71 exactly as shown for its 3-dimensional version.

We find interesting to observe that this topic is somewhat parallel to the Cartan-Hadamard conjecture, stating that the isoperimetric constant of a complete noncompact simply connected Riemannian manifolds with nonpostive sectional curvature is that of flat \mathbb{R}^n . As already remarked, this conjecture is proved only in dimensions 3 and 4. Let us mention however that [GS19] should give the crucial insights for the resolution of the general case. It is worth pointing out that a Mean Curvature Flow proof of the 3-dimensional version of the Cartan-Hadamard conjecture is carried out in [Sch08].

The Isoperimetric Inequality and isoperimetric sets

The very strong rigidity statement of Theorem 1.60 in particular implies that in any 3-manifold with nonnegative Ricci curvature different from flat \mathbb{R}^3 there are no bounded sets with smooth boundary attaining the equality in the Isoperimetric Inequality. One could then wonder whether there exists isoperimetric sets, that is, sets minimising the perimeter under a volume constraint. Actually, the existence of isoperimetric sets in manifolds with nonnegative Ricci curvature is proved in [MN16] if the manifold considered is also \mathscr{C}^0 -asymptotically locally Euclidean, that is, satisfying (1.4.37). In particular, these isoperimetric sets, in dimension 3, do not satisfy equality in our sharp Isoperimetric Inequality (1.7.2). We bring to the reader's attention also [CEV17], where a more detailed description of isoperimetric sets is obtained under the additional assumption of $\mathscr{C}^{2,\alpha}$ -asymptotic conicality. To the author's knowledge, existence result for isoperimetric sets on manifolds with nonnegative Ricci curvature without similar additional assumptions on the structure at infinity are not yet available in literature.

Chapter 2

Minkowski Inequalities and related results via Nonlinear Potential Theory

2.1 Structure of the chapter

In the following Section 2.2, we collect some preliminary results, define the relevant quantities and carry out the computations picturing the conformal setting where the core analysis is realised. Here, we do prove the sharp Kato-type identity for p-harmonic functions in Proposition 2.5, yielding as a corollary the one for harmonic functions employed in the previous chapter. In Section 2.3 we prove the effective monotonicity of (the conformal version of) U^p_β and U^p_∞ discussed in the Introduction, also showing how it improves to a full monotonicity when no critical points are developed. In Section 2.4 we derive the Extended Minkowski Inequality and discuss the application to Volumetric Minkowski Inequality and to the nearly umbilical estimates. In the last Section 2.5, we draw some additional consequences of the effective monotonicity formulas, such as capacity estimates and sphere theorems, yielding the nonlinear version of results from [AM20] and [BMM19] and improving inequalities from [Xia17]. In the end of this last section we also show how to relate with some techniques in overdetermined boundary value problems.

2.2 Effective Monotonicity. Statements and preparatory material.

Before giving the precise statements of the Effective Monotonicity Theorems along the level set flow of the *p*-capacitary potential, let us give some precise background.

2.2.1 Preliminaries on p-capacitary potentials

We recall the well known notion of *p*-capacity, introducing at the same time a normalised version of it that is suitable for our applications.

Definition 2.1 (p-capacity & normalised p-capacity). Let Ω be a bounded open subset of \mathbb{R}^n with smooth boundary.

• The p-capacity of Ω is defined as

$$\operatorname{Cap}_p(\Omega) = \inf \left\{ \int_{\mathbb{R}^n} |\mathrm{D}f|^p \, \mathrm{d}\mu \, \middle| \, f \in \mathscr{C}^\infty_c(\mathbb{R}^n), \, f \geq 1 \, \text{on} \, \Omega \right\}.$$

• The normalised p-capacity of Ω is defined as

$$C_p(\Omega) = \inf \left\{ \left(\frac{p-1}{n-p} \right)^{p-1} \frac{1}{|\mathbb{S}^{n-1}|} \int_{\mathbb{R}^n} |\mathrm{D}f|^p \, \mathrm{d}\mu \, \left| \, f \in \mathscr{C}_c^{\infty}(\mathbb{R}^n), \, f \ge 1 \text{ on } \Omega \right\}. \right. \tag{2.2.1}$$

Through this chapter, we are anyway mostly referring to $C_{(\Omega)}$ simply as *p*-capacity of Ω .

The variational structure of the above definition leads naturally to the formulation of the following problem

$$\begin{cases}
\Delta_p u = 0 & \text{in } \mathbb{R}^n \setminus \overline{\Omega}, \\
u = 1 & \text{on } \partial\Omega, \\
u(x) \to 0 & \text{as } |x| \to \infty.
\end{cases}$$
(2.2.2)

It is well known that, for every bounded open set Ω with smooth boundary and every 1 , problem (2.2.2) admits a unique weak solution. Such a solution is called the <math>p-capacitary potential associated with Ω . For the reader's convenience, we recall that a function v is a weak solution of $\Delta_p v = 0$ in an open set V if $v \in W^{1,p}_{loc}(V)$ and

$$\int_{V} \left\langle |\mathrm{D}v|^{p-2} \mathrm{D}v \, \middle| \, \mathrm{D}\psi \right\rangle \, \mathrm{d}\mu = 0$$

for any test function $\psi \in \mathscr{C}^{\infty}_{c}(V)$. By the important contributions [DiB83; Eva82; Lew83] and [Ura68], we know that weakly *p*-harmonic functions are $\mathscr{C}^{1,\alpha}_{loc}$.

Note that the uniqueness of the solution to problem (2.2.2) can be easily proved by suitably applying the Comparison Theorem for weakly p-harmonic functionsfirst provided in [Tol83] on large balls of radius R, and letting then $R \to +\infty$. With the same argument one can also show that the solution u to problem (2.2.2) is such that 0 < u(x) < 1 for every $x \in \mathbb{R}^n \setminus \overline{\Omega}$. Finally, we recall that such a solution realises the infimum in (2.2.1). This can be proved using a standard exhaustion scheme (for example the one proposed in [CS03a]) and invoking the $\mathscr{C}^{1,\alpha}_{loc}$ regularity to guarantee the convergence of the scheme itself. Let us summarise some of these facts in the following statement, a generalised version of which is proved in Appendix B. For most of the basics on p-harmonic functions we employ in this chapter, we refer the reader to the nice lecture notes [Lin17].

Theorem 2.2 (Existence and regularity of *p*-capacitary potentials). Let Ω be a bounded open subset of \mathbb{R}^n with smooth boundary, and let $1 . Then, there exists a unique weak solution <math>u \in \mathscr{C}^{1,\alpha}_{loc}(\mathbb{R}^n \setminus \overline{\Omega}) \cap \mathscr{C}(\mathbb{R}^n \setminus \Omega)$ to problem (2.2.2), and it fulfils

$$C_p(\Omega) = \left(\frac{p-1}{n-p}\right)^{p-1} \frac{1}{|\mathbb{S}^{n-1}|} \int_{\mathbb{R}^n \setminus \overline{\Omega}} |\mathrm{D}u|^p \,\mathrm{d}\mu,$$

where $C_p(\Omega)$ is the normalised p-capacity of Ω defined in (2.2.1).

The following important result yields the precise asymptotic behaviour of u and |Du| at infinity.

Lemma 2.3 (Asymptotic expansions of u and |Du|). Let Ω be a bounded open subset of \mathbb{R}^n with smooth boundary, and let 1 . Then, the solution <math>u to (2.2.2) satisfies

(i)
$$\lim_{|x|\to+\infty} u(x) |x|^{\frac{n-p}{p-1}} = C_p(\Omega)^{\frac{1}{p-1}}$$
,

(ii)
$$\lim_{|x|\to+\infty} |Du(x)| |x|^{\frac{n-1}{p-1}} = (\frac{n-p}{p-1}) C_p(\Omega)^{\frac{1}{p-1}}$$

where $C_p(\Omega)$ is the normalised p-capacity of Ω defined in (2.2.1). In particular, Crit u is a compact subset of $\mathbb{R}^n \setminus \overline{\Omega}$, possibly with full measure.

For the proof of this lemma we refer the reader to [KV86] (see also the more recent [Pog18, Lemma 2.3 and (2.2)] for a precise statement). It is also worth mentioning [GS99], where similar expansions are employed to infer rotational symmetry of star-shaped domains supporting a solution to problem (2.2.2) with constant normal derivative on the boundary.

On the other hand, the classical regularity theory for quasilinear nondegenerate elliptic equations (see e.g. [LU68]) ensures that they are analytic around the points where the gradient does not vanish. Note that since $\partial\Omega$ is assumed to be smooth, by the Hopf Lemma for p-harmonic functions (see [Tol83, Proposition 3.2.1]), we have that $|Du| \neq 0$ in a neighbourhood of this hypersurface. In particular, u is analytic in such a neighbourhood, and it smoothly extends to $\partial\Omega$. Coupling this with the asymptotic expansion of the gradient given in Lemma 2.3 implies that $\operatorname{Crit} u = \left\{x \in \mathbb{R}^n \setminus \overline{\Omega} \mid \operatorname{D} u(x) = 0\right\}$ is a compact subset of $\mathbb{R}^n \setminus \overline{\Omega}$ (generically depending on p), and in turn that u is analytic outside this set. Finally, it is worth recalling that for $p \neq 2$, the set $\operatorname{Crit} u$ is a priori allowed to have full measure. The following characterisation of the p-capacity of Ω is widely used in the literature and it is also very useful for our purposes. We include a proof, well suited in the general framework of p-nonparabolic Riemannian manifolds in Appendix B.

Lemma 2.4. Let Ω be a bounded open subset of \mathbb{R}^n with smooth boundary, and let 1 . Then, the solution <math>u to (2.2.2) satisfies

$$C_{p}(\Omega) = \left(\frac{p-1}{n-p}\right)^{p-1} \frac{1}{|S^{n-1}|} \int_{\partial \Omega} |Du|^{p-1} d\sigma = \left(\frac{p-1}{n-p}\right)^{p-1} \frac{1}{|S^{n-1}|} \int_{\{u=t\}} |Du|^{p-1} d\sigma,$$
(2.2.3)

where $C_p(\Omega)$ is the normalised p-capacity of Ω defined in (2.2.1) for almost any $t \in (0,1]$ including any regular value t.

In the remaining part of the chapter we will always assume that 1 , unless otherwise stated.

2.2.2 A Kato-type identity for *p*-harmonic functions

We provide the Kato-type identity for *p*-harmonic functions on a general Riemannian manifolds, that in particular implies the one stated in Proposition 2.5 for harmonic functions. Similar identities, but without any refinement coming from being solution of an equation, were important also in [FMV13], see Proposition 1.8 there.

Proposition 2.5 (Kato-type identity for *p*-harmonic functions). Let (M,g) be a Riemannian manifold, and let f be p-harmonic function defined on some subset of M, for p>1. Then, in a neighbourhood of any point where f is such that $|\nabla f|(x)>0$ there holds

$$|\nabla \nabla f|^{2} - \left(1 + \frac{(p-1)^{2}}{n-1}\right) \left|\nabla |\nabla f|\right|^{2} = |Df|^{2} \left|h - \frac{H}{n-1}g^{\top}\right|^{2} + \left(1 - \frac{(p-1)^{2}}{n-1}\right) \left|\nabla^{\top} |\nabla f|\right|^{2},$$

$$(2.2.4)$$

where the tangential elements are referred to the level sets of f. Moreover, if

$$\left| h - \frac{H}{n-1} g^{\top} \right|^2 (x) = 0$$
 (2.2.5)

$$\left| \mathbf{D}^{\top} | \mathbf{D} f | \right|^2(x) = 0 \tag{2.2.6}$$

for any x in $\{\rho_0 \le f \le \rho_1\}$ and |Df| > 0 in this set, then the Riemannian manifold $(\{\rho_0 \le f \le \rho_1\}, g)$ is isometric to the warped product $([\rho_0, +\rho_1] \times \{f = \rho_0\}, d\rho \otimes d\rho + \eta^2(\rho) g_{|\{f=\rho_0\}})$, where η , f and ρ are related as

$$\eta(\rho) = \left(\frac{f'(\rho_0)}{f'(\rho)}\right)^{\frac{p-1}{n-1}}.$$

Proof. We consider on x an orthonormal frame $\{e_1, \ldots, e_{n-1}, e_n = Df/|Df|\}$. We introduce the notation $DD^{\top}f$ to denote the tensor

$$\left(DD^{\top}f\right)_{ij} = DDf(e_i, e_j)$$
 $i, j = 1, \dots, n-1.$

Observe that $DD^{T}f = |Df|h$. We can write the norm of DDf as follows

$$|DDf|^2 = |DD^{\top}f|^2 + 2\sum_{i=1}^{n-1}|DDf(e_n, e_i)|^2 + |DDf(e_n, e_n)|^2.$$
 (2.2.7)

Moreover, we can write the first term in the right hand side of (2.2.7) as follows

$$|DD^{\top}f|^2 = \frac{(\Delta^{\top}f)^2}{n-1} + \left|DD^{\top}f - \frac{\Delta^{\top}f}{n-1}g^{\top}\right|^2,$$
 (2.2.8)

where we denote by $\Delta^{\top} f$ the trace of DD $^{\top}$. We now exploit the *p*-harmonicity of *f*. By

$$\Delta_p f = |\mathbf{D}f|^{p-2} \left(\Delta f + (p-2) \nabla^2 f \left(\frac{\mathbf{D}f}{|\mathbf{D}f|}, \frac{\mathbf{D}f}{|\mathbf{D}f|} \right) \right) = 0$$

and $|Df|(x) \neq 0$, we have

$$\Delta f = -(p-2)DDf\left(\frac{Df}{|Df|}, \frac{Df}{|Df|}\right).$$

The above identity implies

$$\Delta^{\top} f = \Delta f - \text{DD} f(e_n, e_n) = -(p-1) \text{DD} f\left(\frac{Df}{|Df|}, \frac{Df}{|Df|}\right), \tag{2.2.9}$$

that, plugged into (2.2.8), gives

$$|\mathrm{DD}^{\top} f|^2 = \frac{(p-1)^2}{n-1} \left| \mathrm{DD} f \left(\frac{\mathrm{D} f}{|\mathrm{D} f|} \left| \frac{\mathrm{D} f}{|\mathrm{D} f|} \right) \right|^2 + \left| \mathrm{DD}^{\top} f - \frac{\Delta^{\top} f}{n-1} g^{\top} \right|^2. \tag{2.2.10}$$

We now turn our attention to the second and the third term in (2.2.7). An easy computation shows

$$DDf(e_n, e_j) = \langle D|Df||e_j\rangle$$

for any j = 1, ..., n - 1, and thus we have

$$\sum_{j=1}^{n-1} |DDf(e_n, e_j)|^2 = |D^{\top}|Df||^2$$
 (2.2.11)

and

$$\left| DDf(e_n, e_n) \right|^2 = \left\langle D|Df| \left| \frac{Df}{|Df|} \right\rangle^2.$$
 (2.2.12)

Finally, plugging (2.2.10), (2.2.11) and (2.2.12) into (2.2.7) we obtain (2.2.4).

Let us now assume $|\mathrm{D}f|>0$ on $\{\rho_0\leq f\leq \rho_1\}$ and conditions (2.2.5)-(2.2.6) hold on this set. Then, by the first assumption, we deduce by standard results in differential geometry (see e.g. [Hir76, Theorem 2.2]) that $\{\rho_0\leq f\leq \rho_1\}$ is diffeomorphic to $[\rho_0,\rho_1)\times \{f=\rho_0\}$, and in particular there exist new coordinates $\{f,\vartheta^1,\ldots,\vartheta^{n-1}\}$ on $\{f\geq \rho_0\}$ such that

$$g = \frac{\mathrm{d}f \otimes \mathrm{d}f}{|\mathrm{D}f|^2} + g_{ij}(f,x) \,\mathrm{d}\vartheta^i \otimes \vartheta^j x^j,$$

where, i, j range in $1, \ldots, n-1$ and $\{\vartheta^i\}_{i=1}^{n-1}$ are coordinates on $\{f = \rho_0\}$. Observe now that by (2.2.5), the function |Df| is constant on each level set of f. In other words, it is a function of f alone. We can then define a new coordinate by $d\rho = df/|Df|$ so that the metric becomes

$$g = d\rho \otimes d\rho + g_{ij}(\rho, x) d\vartheta^i \otimes d\vartheta^j,$$

with some abuse of notation. In this coordinates, standard computations show that the Hessian is computed as

$$DDf = f'' d\rho \otimes d\rho + f'DD\rho = f'' d\rho \otimes d\rho + \frac{1}{2} f' \partial_{\rho} g_{ij} d\theta^{i} \otimes d\theta^{j}, \qquad (2.2.13)$$

where by f' and f'' we denote the derivatives of f with respect to ρ .

Let us now consider, for any fixed point $x \in \{\rho_0 \le \rho \le \rho_1\}$ the orthonormal frame $\{e_1, \ldots, e_{n-1}, e_n = \mathrm{D}f/|\mathrm{D}f| = \mathrm{D}\rho\}$, already used in the first part of this proof. Then we have

$$DDf(e_n, e_j) = \langle D|Df| | e_j \rangle = 0$$

by (2.2.6), and

$$\mathrm{DD}^{\top} f = -\frac{p-1}{n-1} \mathrm{DD} f \left(\frac{\mathrm{D} f}{|\mathrm{D} f|}, \frac{\mathrm{D} f}{|\mathrm{D} f|} \right) g^{\top}$$

by (2.2.5) combined with (2.2.9). In particular, the Hessian of f can also be computed as

$$DDf = DDf\left(\frac{Df}{|Df|}, \frac{Df}{|Df|}\right) d\rho \otimes d\rho - \frac{p-1}{n-1}DDf\left(\frac{Df}{|Df|}, \frac{Df}{|Df|}\right) g_{ij} d\theta^{i} \otimes d\theta^{j}.$$

A comparison with (2.2.13) then gives the system of ordinary differential equations

$$\partial_{\rho} \log g_{ij}(\rho, \vartheta) = -2 \frac{p-1}{n-1} \partial_{\rho} \log f'(\rho), \tag{2.2.14}$$

that, integrated, yields

$$g_{ij}(\rho,\vartheta) = g_{ij}(\rho_0,\vartheta) \left(\frac{f'(\rho_0)}{f'(\rho)}\right)^{2\frac{p-1}{n-1}},$$

that is, *g* has the warped product structure claimed.

As a corollary of the above Proposition, we record the following refined Kato's inequalities for p-harmonic functions, together with a characterisation of the equality case. We will not need this corollary in the sequel, but it actually is of some independent interest. Let us also point out that setting p = 2 we recover the well known, and widely used in the last chapter, refined Kato's inequality for harmonic functions, whose associated rigidity statement is discussed in [BC12]. For general p, it is obtained also in [CCW12], but without discussion of related rigidity statements.

Corollary 2.6 (Refined Kato's inequalities for *p*-harmonic functions). Let (M, g) be a Riemannian manifold, and let f be a p-harmonic function on some subset of M, with $p \in (1, n)$.

(i) If $(p-1)^2 < n-1$, then , in a neighbourhood of any x such that $|\mathrm{D} f|(x) > 0$,

$$|\mathrm{DD}f|^2 \ge \left(1 + \frac{(p-1)^2}{n-1}\right) \left|\mathrm{D}|\mathrm{D}f|\right|^2.$$

Moreover, if equality is achieved on $\{\rho_0 \leq f \leq \rho_1\}$, and |Df| > 0 in this region, then $(\{\rho_0 \leq f \leq \rho_1\}, g)$ has the same warped product structure as in the rigidity case of Proposition 2.5.

(ii) If $(p-1)^2 > n-1$, then, in a neighbourhood of any x such that |Df|(x) > 0,

$$|\nabla \nabla f|^2 \ge 2 |\nabla |\nabla f||^2. \tag{2.2.15}$$

Moreover, if equality is achieved on $\{\rho_0 \leq f \leq \rho_1\}$ and $|\nabla f| > 0$ in this region, then $(\{\rho_0 \leq f \leq \rho_1\}, g)$ splits as a Riemannian product $([\rho_0, \rho_1] \times g_{\{f=\rho_0\}}, d\rho \otimes d\rho + g_{\{f=\rho_0\}})$ and f is an affine function of ρ .

(iii) If $(p-1)^2 = n-1$, then, in a neighbourhood of any x such that |Df|(x) > 0 it holds (2.2.15). If equality holds on (2.2.15) at some x with |Df| > 0, then x is an umbilical point of $\{f = f(x)\}$, that in a neighbourhood of x is a smooth hypersurface.

Proof. The assertions in (i) follow straightforwardly from Proposition 2.5. Let now $(p-1)^2 \ge n-1$. Plugging

$$\left|\mathbf{D}^{\top}|\mathbf{D}f|\right|^{2} = \left|\mathbf{D}|\mathbf{D}f|\right|^{2} - \left\langle\nabla|\mathbf{D}\frac{\mathbf{D}f}{|\mathbf{D}f|}|\right\rangle^{2},$$

into (2.2.4) we obtain (2.2.15), Assume now that equality holds in (2.2.15) for x with $|Df|(x) \neq 0$. Then, by (2.2.4) and the above identity we obtain

$$|\mathrm{D}f|^2 \Big| \mathbf{h} - \frac{\mathbf{H}}{n-1} \, g^\top \Big|^2 + \left(\frac{(p-1)^2}{n-1} - 1 \right) \left\langle \mathrm{D}|\mathrm{D}f|, \frac{\mathrm{D}f}{|\mathrm{D}f|} \right\rangle^2 = 0.$$

If $(p-1)^2 = n-1$, we can only deduce

$$\left| \mathbf{h} - \frac{\mathbf{H}}{n-1} \, \mathbf{g}^{\top} \right|^2 = 0,$$

that is, x is an umbilical point of $\{f = f(x)\}$. This shows (iii). If $(p-1)^2 > n-1$, and equality holds in(2.2.15) on a region $\{\rho_0 \le f \le \rho_1\}$ where $|Df| \ne 0$, then we also get

$$\left\langle \mathbf{D}|\mathbf{D}f| \left| \frac{\mathbf{D}f}{|\mathbf{D}f|} \right\rangle^2 = 0.$$

In particular, by the computations performed in the proof of Proposition 2.5, we deduce that |DDf| = 0. The isometry claimed follows by Corollary 1.23.

Observe that in order to deduce Proposition 2.5 from Proposition 2.5 one should remove, in the case p=2 the assumption of nonvanishing gradient. Such an assumption is added in Proposition 2.5 just to ensure the possibility of taking second derivatives, as we pointed out in the discussion of the regularity of p-harmonic functions above. The smoothness of harmonic functions then allows to remove this assumption.

2.2.3 The effectively monotone quantities

We now rigorously state our Effective Monotonicity-Rigidity Theorems. Let us first recall the definitions of the relevant quantities we just hinted in the Introduction. For $\Omega \subset \mathbb{R}^n$ and 1 , consider <math>u the solution of (2.2.2). Let, for $\beta \geq 0$ the function U_{β}^p : $(0,1] \setminus u(\operatorname{Crit}(u)) \to \mathbb{R}$ be defined by

$$U_{\beta}^{p}(t) = t^{-\beta(p-1)\frac{(n-1)}{(n-p)}} \int_{\{u=t\}} |Du|^{(\beta+1)(p-1)} d\sigma, \qquad (2.2.16)$$

and the function $U_{\infty}^{p}:(0,1]\setminus u(\operatorname{Crit} u)\to \mathbb{R}$ be defined by

$$U_{\infty}^{p}(t) = \sup_{\{u=t\}} \frac{|Du|}{u^{\frac{n-1}{n-p}}}.$$
 (2.2.17)

We stress that the above functions are defined just on regular values of u. The following is the Effective Monotonicity Theorem for U_{β}^{p} , when $\beta \geq (n-p)/[(p-1)(n-1)]$.

Theorem 2.7 (Effective Monotonicity of U_{β}^p). Let $\Omega \subset M$ be a bounded set with smooth boundary, and let $U_{\beta}^p:(0,1]\setminus u(\mathrm{Crit}(u))\to \mathbb{R}$ be the function defined by (2.2.16). Then, if $\beta\geq (n-p)/[(p-1)(n-1)]$ on any regular value $0< t\leq 1$, we have

$$(U_{\beta}^{p})'(t) = \beta t^{-\beta(p-1)\left(\frac{n-1}{n-p}\right)} \int_{\{u=t\}} |Du|^{(\beta+1)(p-1)-1} \left[H - \frac{(n-1)(p-1)}{(n-p)} |D\log u| \right] d\sigma \ge 0$$
(2.2.18)

and

$$\lim_{t\to 0^+} U_{\beta}^p(t) \le U_{\beta}^p(1).$$

Moreover, equality holds in (2.2.18) for some regular $0 < t_0 \le 1$ if and only if $\{u = t_0\}$ is isometric to a sphere.

On the other hand, the following is the Effective Monotonicity Theorem for U_{∞}^p .

Theorem 2.8 (Effective Monotonicity of U_{∞}^p). Let $\Omega \subset M$ be a bounded set with smooth boundary, and let $U_{\infty}^p: (0,1] \setminus u(\operatorname{Crit} u) \to (0,\infty)$ be the function defined by (2.2.17). Then, on any regular value $0 < t \le 1$ and any regular T < t we have

$$U_{\infty}^{p}(T) \le U_{\infty}^{p}(t). \tag{2.2.19}$$

Moreover, on any regular $0 < t \le 1$ we have

$$\[H - \frac{(n-1)}{(p-1)(n-p)} |D\log u|\] (x_t) = -(p-1) \frac{\partial}{\partial \nu_t} \log \frac{|Du|}{u^{\frac{n-1}{n-p}}} (x_t) \ge 0, \tag{2.2.20}\]$$

where $x_t \in \{u = t\}$ is the point where $\sup_{\{u = t\}} |Du|/u^{(n-1)/(n-p)}$ is achieved and $v_t = -Du/|Du|$ is the unit normal to $\{u = t\}$.

Equality holds in (2.2.19) for regular $T < t_0$ or in (2.2.20) for regular t_0 if and only if $\{u = t_0\}$ is isometric to a sphere.

Observe that, differently from U^p_β with finite β , the function U^p_∞ is easily defined also on singular values of u, and, as it will be clear from the proof, the value T in (2.2.19) could be taken also singular.

Analogously to what we did Chapter 1 in proving the Willmore-type inequalities, the L^p -Minkowski Inequality as well as some other inequalities will follow combining $(U^p_\beta)'(1) \geq 0$ with $\lim_{t\to 0^+} U^p_\beta(t) \leq U^p_\beta(1)$. The computation of the latter limit is a straightforward consequence of Lemma 2.3, that in fact yields

$$\lim_{t \to 0^+} U_{\beta}^p(t) = C_p^{1-\beta \frac{p-1}{n-p}} \left(\frac{n-p}{p-1} \right)^{(\beta+1)(p-1)} |S^{n-1}|. \tag{2.2.21}$$

Similarly, in the applications of the effective monotonicity of U_{∞}^{p} we are going to use

$$\lim_{t\to 0^+} U_{\infty}^p(t) = \left(\frac{n-p}{p-1}\right) C_p(\Omega), \tag{2.2.22}$$

again following from the asymptotic expansion of *u* and its gradient.

2.2.4 The conformal setting

The heuristics leading to cylindrical conformal change of metric introduced in [AM15] and [AM20] are formally sustainable also starting from problem (2.2.2). Indeed, when Ω is a ball, the solution to (2.2.2) is (proportional to) $r^{-(n-p)/(p-1)}$ where r is the distance from the center of the ball, and we can define a metric g satisfying

$$g = u^{2\frac{p-1}{n-p}} g_{\mathbb{R}^n} = (Cr)^{-2} \left(dr \otimes dr + r^2 g_{\mathbb{S}^{n-1}} \right) = d\rho \otimes d\rho + g_{\mathbb{S}^{n-1}},$$

where C is a constant depending on the radius of the ball, and ρ is defined by $d\rho = (Cr)^{-1} dr$. In this case, g is the cylindrical metric on $M \setminus \Omega$, and ρ is harmonic with respect to g. Encouraged by this basic observation we devote this section to fully describe the conformal background where we are going to work.

A conformal reformulation of the boundary value problem

Define, for u a solution to (2.2.2) with 1 , the metric

$$g = u^{2\frac{p-1}{n-p}} g_{\mathbb{R}^n}. \tag{2.2.23}$$

Let

$$\varphi = -\frac{n-2}{n-p}(p-1)\log u. \tag{2.2.24}$$

Observe that, in light of the optimal $\mathscr{C}^{1,\alpha}$ -regularity of u, the metric g is not a smooth Riemannian metric, but just \mathscr{C}^1 . For this reason, in the computations that follow those involving just first derivative of the metric (and equivalently, of φ) make sense in the whole of $\mathbb{R}^n \setminus \Omega$, while those involving higher derivatives have to be thought as carried out in the complement of the critical set of u, that is, on $\{|Du| \neq 0\}$.

Fixing local coordinates $\{x^{\alpha}\}_{\alpha=1}^{n}$ in M and using standard formulas (that can be found for example in [Bes08; HE73]) we get

$$\Gamma_{\alpha\beta}^{\gamma} = -\frac{1}{n-2} \left(\delta_{\alpha}^{\gamma} \partial_{\beta} \varphi + \delta_{\beta}^{\gamma} \partial_{\alpha} \varphi - g_{\alpha\beta}^{\mathbb{R}^{n}} g_{\mathbb{R}^{n}}^{\gamma\eta} \partial_{\eta} \varphi \right) \tag{2.2.25}$$

$$R_{\alpha\beta}^{g} = D_{\alpha}D_{\beta}\varphi + \frac{\partial_{\alpha}\varphi\partial_{\beta}\varphi}{n-2} - \frac{|D\varphi|^{2} - \Delta\varphi}{n-2}g_{\alpha\beta}^{\mathbb{R}^{n}},$$
(2.2.26)

where $\Gamma_{\alpha\beta}^{\gamma}$ are the Christoffel symbols associated to the metric g, $R_{\alpha\beta}^{g}$ represents the components of the Ricci tensor and ∇_{α} the covariant derivative. Moreover,

$$\nabla_{\alpha}\nabla_{\beta}w = D_{\alpha}D_{\beta}w + \frac{1}{n-2}\left(\partial_{\alpha}w\partial_{\beta}\varphi + \partial_{\alpha}\varphi\partial_{\beta}w - \langle Dw|D\varphi\rangle\,g_{\alpha\beta}^{\mathbb{R}^{n}}\right),\tag{2.2.27}$$

$$\Delta_{g}w = e^{\frac{2\varphi}{n-2}} \left(\Delta w - \langle Dw | D\varphi \rangle \right) \tag{2.2.28}$$

for any $w \in \mathscr{C}^2$, where Δ_g and Δ denote the Laplacian with respect to g and $g_{\mathbb{R}^n}$ respectively. Let T be a vector field. Then,

$$\operatorname{div}_g T = g^{ik} \left(\frac{\partial T_k}{\partial x_i} - \Gamma_{ik}^l T_l \right) = u^{-2\frac{p-1}{n-p}} g_{\mathbb{R}^n}^{ik} \left(\frac{\partial T_k}{\partial x_i} - \Gamma_{ik}^l T_l \right).$$

Using (2.2.25), we get

$$\Gamma_{ik}^{l} = \frac{p-1}{n-p} \left(\delta_{i}^{l} \frac{\partial_{k} u}{u} + \delta_{k}^{l} \frac{\partial_{i} u}{u} - g_{ik}^{\mathbb{R}^{n}} \frac{D^{l} u}{u} \right)$$

and

$$\begin{aligned} \operatorname{div}_{g}(T) \\ &= u^{-2\frac{p-1}{n-p}} g_{\mathbb{R}^{n}}^{ik} \left(\frac{\partial T_{k}}{\partial x_{i}} - \frac{p-1}{n-p} T_{i} \frac{\partial_{k} u}{u} - \frac{p-1}{n-p} T_{k} \frac{\partial_{i} u}{u} + g_{ik}^{\mathbb{R}^{n}} \frac{p-1}{n-p} \left\langle \frac{\mathrm{D} u}{u} \middle| T \right\rangle_{\mathbb{R}^{n}} \right) \\ &= u^{-2\frac{p-1}{n-p}} \operatorname{div}_{g_{\mathbb{R}^{n}}} T + \frac{(n-2)(p-1)}{(n-p)} \left\langle \frac{\mathrm{D} u}{u} \middle| T \right\rangle_{g}. \end{aligned}$$

where $\langle \cdot | \cdot \rangle_{\mathbb{R}^n}$ and $\langle \cdot | \cdot \rangle_g$ are the scalar products associated to $g_{\mathbb{R}^n}$ and g respectively. Setting $T = |\mathrm{D} u|^{p-2}\mathrm{D} u$ and recalling that $\Delta_p u = 0$ we obtain

$$\operatorname{div}_{g}(|\operatorname{D} u|^{p-2}\operatorname{D} u) = \frac{(n-2)(p-1)}{n-p} \left\langle \frac{\operatorname{D} u}{u} \middle| |\operatorname{D} u|^{p-2}\operatorname{D} u \right\rangle_{g}$$

$$= \frac{(n-2)(p-1)}{n-p} \frac{|\operatorname{D} u|^{p-2}}{u} \left\langle \operatorname{D} u \middle| \operatorname{D} u \right\rangle_{g}.$$
(2.2.29)

By the definition of *g* we immediately get

$$|Du|^{p-2} = u^{\frac{(p-1)(p-2)}{n-p}} |\nabla u|_g^{p-2}.$$
 (2.2.30)

Using (2.2.30) in (2.2.29) we get

$$\operatorname{div}_{g}(u^{\frac{(p-1)(p-2)}{n-p}}|\nabla u|_{g}^{p-2}\nabla u) = \frac{(n-2)(p-1)}{n-p}u^{\frac{(p-1)(p-2)}{n-p}-1}|\nabla u|_{g}^{p},$$

and so

$$\Delta_p^g u = \text{div}_g(|\nabla u|_g^{p-2} \nabla u) = (p-1) \frac{|\nabla u|_g^p}{u}.$$
 (2.2.31)

Lemma 2.9. Let u be the solution to (2.2.2), let φ be defined in (2.2.24) and g the metric defined by (2.2.23). Then

$$\Delta_n^g \varphi = 0$$

on $(\mathbb{R}^n \setminus \overline{\Omega}) \setminus \text{Crit } u$.

Proof. On $(\mathbb{R}^n \setminus \overline{\Omega}) \setminus \text{Crit } u$ the computations performed above are actual. In light of the definition (2.2.24), it obviously suffices to show that $\log u$ is p-harmonic with respect to the metric g. Letting $f = \log u$, we have $\nabla f = \frac{\nabla u}{u}$ and $|\nabla f|_g^{p-2} = \frac{|\nabla u|_g^{p-2}}{u^{p-2}}$. Thus,

$$\Delta_p^g f = \operatorname{div}_g(|\nabla f|_g^{p-2} \nabla f) = \operatorname{div}_g\left(\frac{|\nabla u|_g^{p-2}}{u^{p-2}} \frac{\nabla u}{u}\right).$$

Therefore,

$$\Delta_p^g f = u^{1-p} \Delta_p^g u + \left\langle \nabla u^{1-p} \middle| |\nabla u|_g^{p-2} \nabla u \right\rangle_g$$
$$= u^{1-p} \Delta_p^g u + (1-p) u^{-p} |\nabla u|_g^p$$

and recalling (2.2.31) we get the thesis.

We now compute the Ricci curvature of g, in order to complete the reformulation of (2.2.2).

Lemma 2.10. Let u be the solution to (2.2.2), let φ be defined by (2.2.24) and g be the metric defined by (2.2.23). Then,

$$\operatorname{Ric}_{g} - \nabla \nabla \varphi + \frac{\operatorname{d} \varphi \otimes \operatorname{d} \varphi}{n-2} = \left(\frac{|\nabla \varphi|_{g}^{2}}{n-2} - \frac{p-2}{n-2} \frac{\nabla \nabla \varphi (\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|_{g}^{2}} \right) g \tag{2.2.32}$$

on $(\mathbb{R}^n \setminus \overline{\Omega}) \setminus \text{Crit } u$.

Proof. As in the proof of Lemma 2.9 above, let $f = \log u$. Keeping in mind formulas (2.2.26) and (2.2.27), recalling that $\varphi = -\frac{(n-2)(p-1)}{n-p}f$ and $R_{\alpha\beta}^{\mathbb{R}^n} = 0$ we obtain

$$R_{\alpha\beta}^{g} = -\frac{(n-2)(p-1)}{n-p} D_{\alpha} D_{\beta} f + \frac{(n-2)(p-1)^{2}}{(n-p)^{2}} \partial_{\alpha} f \partial_{\beta} f$$

$$-\frac{p-1}{n-p} \left(\Delta f + \frac{(n-2)(p-1)}{n-p} |Df|^{2} \right) e^{-\frac{2(p-1)f}{n-p}} g_{\alpha\beta}$$
(2.2.33)

and

$$D_{\alpha}D_{\beta}f = \nabla_{\alpha}\nabla_{\beta}f + \frac{p-1}{n-p}\Big(2\partial_{\alpha}f\partial_{\beta}f - |\nabla f|_{g}^{2}g_{\alpha\beta}\Big). \tag{2.2.34}$$

Using (2.2.34) in (2.2.33) we get

$$R_{\alpha\beta}^{g} = -\frac{(n-2)(p-1)}{n-p} \nabla_{\alpha} \nabla_{\beta} f - \frac{(n-2)(p-1)^{2}}{(n-p)^{2}} \partial_{\alpha} f \partial_{\beta} f +$$

$$+ \frac{(n-2)(p-1)^{2}}{(n-p)^{2}} |\nabla f|_{g}^{2} g_{\alpha\beta}$$

$$- \frac{p-1}{n-p} \Delta f e^{-\frac{2(p-1)f}{n-p}} g_{\alpha\beta} - \frac{(n-2)(p-1)^{2}}{(n-p)^{2}} |Df|^{2} e^{-\frac{2(p-1)f}{n-p}} g_{\alpha\beta}$$

$$= \nabla_{\alpha} \nabla_{\beta} \varphi - \frac{\partial_{\alpha} \varphi \partial_{\beta} \varphi}{n-2} + \frac{|\nabla \varphi|_{g}^{2}}{n-2} g_{\alpha\beta} + \frac{1}{n-2} e^{-\frac{2(p-1)f}{n-p}} \left(\Delta \varphi - |D\varphi|^{2}\right) g_{\alpha\beta}$$

$$= \nabla_{\alpha} \nabla_{\beta} \varphi - \frac{\partial_{\alpha} \varphi \partial_{\beta} \varphi}{n-2} + \frac{|\nabla \varphi|_{g}^{2}}{n-2} g_{\alpha\beta} + \frac{1}{n-2} \Delta_{g} \varphi g_{\alpha\beta}$$

where in the last equality we used (2.2.28). Since

$$0 = \Delta_p^g \varphi = |\nabla \varphi|_g^{p-2} \Delta_g \varphi + \left\langle \nabla |\nabla \varphi|_g^{p-2} \, \middle| \, \nabla \varphi \right\rangle_g$$

we obtain

$$\Delta_{g}\varphi = -(p-2)\frac{\nabla\nabla\varphi(\nabla\varphi,\nabla\varphi)}{|\nabla\varphi|_{g}^{2}}$$

$$= -\frac{p-2}{2}\frac{\left\langle\nabla|\nabla\varphi|_{g}^{2}\mid\nabla\varphi\right\rangle_{g}}{|\nabla\varphi|_{g}^{2}}.$$
(2.2.36)

Using (2.2.36) in (2.2.35) we can write

$$R_{\alpha\beta}^{g} = \nabla_{\alpha}\nabla_{\beta}\varphi - \frac{\partial_{\alpha}\varphi\partial_{\beta}\varphi}{n-2} + \left(\frac{|\nabla\varphi|_{g}^{2}}{n-2} - \frac{p-2}{n-2}\frac{\nabla\nabla\varphi(\nabla\varphi,\nabla\varphi)}{|\nabla\varphi|_{g}^{2}}\right)g_{\alpha\beta},$$

that is
$$(2.2.32)$$
.

We are finally in position to reformulate problem (2.2.2) as

$$\begin{cases}
\Delta_{p}^{g}\varphi = 0 & \text{in} \quad (\mathbb{R}^{n} \setminus \overline{\Omega}) \setminus \text{Crit } \varphi \\
\operatorname{Ric}_{g} - \nabla \nabla \varphi + \frac{\operatorname{d}\varphi \otimes \operatorname{d}\varphi}{n - 2} = \left(\frac{|\nabla \varphi|_{g}^{2}}{n - 2} - \frac{p - 2}{n - 2} \frac{\nabla \nabla \varphi(\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|_{g}^{2}}\right) g & \text{in} \quad (\mathbb{R}^{n} \setminus \overline{\Omega}) \setminus \operatorname{Crit} \varphi \\
\varphi = 0 & \text{on} \quad \partial \Omega \\
\varphi(x) \to +\infty & \text{as} \quad |x| \to \infty.
\end{cases}$$
(2.2.37)

We explicitly observe that if p=2 then (2.2.37) coincides with the problem studied in [AM15]. We conclude this part by isolating from the computations above the useful relation between $|\nabla \varphi|_g$ and |Du|:

$$|\nabla \varphi|_{g} = \frac{(n-2)(p-1)}{n-p} \frac{|\mathrm{D}u|}{u^{\frac{n-1}{n-p}}}$$
(2.2.38)

Observe that involving just u and its gradient, the above is a continuous function of $\mathbb{R}^n \setminus \Omega$.

The geometry of the level sets of u and φ

Let us consider the $g^{\mathbb{R}^n}$ –unit vector field

$$\nu := -\mathrm{D}u/|\mathrm{D}u| = \mathrm{D}\varphi/|\mathrm{D}\varphi|$$

and the g-unit vector field

$$\nu_{g} := -\nabla u/|\nabla u|_{g} = \nabla \varphi/|\nabla \varphi|_{g}.$$

Accordingly, we consider the second fundamental forms h and h_g of the level sets of u and φ with respect to the Euclidean metric $g^{\mathbb{R}^n}$ and the conformally related ambient metric g are respectively given by

$$\mathbf{h}_{ij} = -\frac{\mathbf{D}_i \mathbf{D}_j u}{|\mathbf{D} u|} = \frac{\mathbf{D}_i \mathbf{D}_j \varphi}{|\mathbf{D} \varphi|}, \qquad \mathbf{h}_{ij}^{g} = -\frac{\nabla_i \nabla_j u}{|\nabla u|_g} = \frac{\nabla_{ij}^2 \varphi}{|\nabla \varphi|_g} \quad \text{for } i, j = 1, \dots, n-1.$$

Taking the trace of the above expressions with respect to the induced metric we obtain the following expressions for the mean curvatures in the two settings

$$\mathbf{H} = -\frac{\Delta u}{|\mathrm{D}u|} + \frac{\mathrm{D}\mathrm{D}u(\mathrm{D}u,\mathrm{D}u)}{|\mathrm{D}u|^3}, \qquad \mathbf{H}_g = \frac{\Delta_g \varphi}{|\nabla \varphi|_g} - \frac{\nabla \nabla \varphi(\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|_g^3}.$$

Recalling that $\Delta_p u = 0$ and $\Delta_p^g \varphi = 0$ we have

$$H = \frac{p-1}{p} \frac{\langle D|Du|^p|Du\rangle}{|Du|^{p+1}} = (p-1) \frac{DDu(Du, Du)}{|Du|^3},$$

and

$$H_{g} = -\frac{p-1}{p} \frac{\left\langle \nabla |\nabla \varphi|_{g}^{p} |\nabla \varphi\rangle_{p}}{|\nabla \varphi|_{g}^{p+1}} = -(p-1) \frac{\nabla \nabla \varphi(\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|_{g}^{3}}.$$
 (2.2.39)

The second fundamental forms h and h_g are related by the following formula:

$$h_g(X,Y) = u^{\frac{p-1}{n-p}} \Big(h(X,Y) - \frac{p-1}{n-p} \frac{|Du|}{u} \langle X \mid Y \rangle \Big),$$
 (2.2.40)

for any X, Y tangent vectors to the level sets of u. Tracing the above identity with respect to g we obtain the useful relation between the mean curvatures H and H $_g$

$$H_g = u^{-\frac{p-1}{n-p}} \left(H - \frac{(n-1)(p-1)}{(n-p)} \frac{|Du|}{u} \right).$$
 (2.2.41)

Relation between the metric-induced measures We recall the relation between the Lebesgue measure $d\mu$ and the volume measure $d\mu_g$ induced by g on M

$$d\mu_g = u^{(p-1)\frac{n}{n-p}} d\mu (2.2.42)$$

and the relation between the are elements $d\sigma_g$ and $d\sigma$ induced respectively by g and the flat metric on smooth hypersurfaces

$$d\sigma_g = u^{(p-1)\frac{n-1}{n-p}} d\sigma. \tag{2.2.43}$$

Bochner formula for solutions of (2.2.37)

As in the last chapter, it is natural to apply the Bochner formula to a solution of (2.2.2). To do so, we start from the version of the Bochner formula suited for the *p*-Laplacian functions that has been worked out and applied in [Val13], see Proposition 3.1.2 there. Combining it with the first two equations in (2.2.37), we directly get the following relation. We do not provide the details, being straightforward.

Proposition 2.11. Let $\Omega \subset \mathbb{R}^n$ be an open bounded subset with smooth boundary. Let φ be a solution to (2.2.37). Then, in a neighbourhood of any $x \in \mathbb{R}^n \setminus \overline{\Omega}$ such that $|\nabla \varphi|(x) > 0$, we have

$$\Delta |\nabla \varphi|^{p} + (p-2) \frac{\nabla \nabla |\nabla \varphi|^{p} (\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|^{2}} - \frac{n-p}{n-2} \langle \nabla |\nabla \varphi|^{p} |\nabla \varphi\rangle =$$

$$= p|\nabla \varphi|^{p-2} \left(|\nabla \nabla \varphi|^{2} + p(p-2) \left\langle \nabla |\nabla \varphi| \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle^{2} \right).$$
(2.2.44)

As we are going to see soon, (2.2.44) will be the key ingredient to show the monotonicity of (the conformal version of) U_{∞}^{p} . An involved processing of it will instead lead to the vector field with nonnegative divergence encoding the monotonicity of U_{β}^{p} . Here, we just point out the enhancement of Corollary 1.23 for solutions of (2.2.37)

Lemma 2.12. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Let φ be a solution to (2.2.37). Assume that $|\nabla \varphi|_g > 0$ on $\{s_0 \leq \varphi \leq s_1\}$ for some $s_0 \in [0, \infty)$, and that $\nabla |\nabla \varphi|_g = 0$ on this region . Then, the Riemannian manifold ($\{s_0 \leq \varphi \leq s_1\}, g$) is isometric to the Riemannian product ($[s_0, s_1] \times \{\varphi = s_0\}$, $d\rho \otimes d\rho + g_{\{\varphi = s_0\}}$), and φ is an affine function of ρ .

Proof. If $\nabla |\nabla \varphi|_g = 0$, then, plugging this information in the relation (2.2.44), we immediately deduce that $|\nabla \nabla \varphi| = 0$ on $\{s_0 \leq \varphi \leq s_1\}$. The result claimed now follows from Corollary 1.23.

The conformal version of the Effective Monotonicity Theorems

We start introducing the conformal version of the functions U^p_β and U^p_∞ introduced before. Let as always $1 , and let <math>\beta \ge 0$. For φ defined in (2.2.24), let Φ^p_β : $[0, +\infty) \setminus \varphi[\operatorname{Crit} \varphi] \to \mathbb{R}$ and Φ^p_∞ : $[0, +\infty) \setminus \varphi[\operatorname{Crit} \varphi] \to \mathbb{R}$ be defined by

$$\Phi_{\beta}^{p}(s) = \int_{\{\varphi=s\}} |\nabla \varphi|_{g}^{(\beta+1)(p-1)} d\sigma_{g}, \qquad (2.2.45)$$

and

$$\Phi^p_{\infty}(s) = \sup_{\{\varphi = s\}} |\nabla \varphi|_g. \tag{2.2.46}$$

The above functions are defined on regular values of φ .

We are now going to state the conformal version of Theorem 2.7. Observe first that straightforward computations involving (2.2.38) and (2.2.43) show the following relation between the functions U_{β}^{p} and Φ_{β}^{p} .

$$U_{\beta}^{p}(t) = \Phi_{\beta}^{p} \left(-\frac{(n-2)(p-1)}{(n-p)} \log t \right)$$
$$-t \frac{dU_{\beta}^{p}}{dt}(t) = \frac{d\Phi_{\beta}^{p}}{dt} \left(-\frac{(n-2)(p-1)}{(n-p)} \log t \right)$$
 (2.2.47)

and between U^p_{∞} and Φ^p_{∞}

$$U_{\infty}^{p}(t) = \Phi_{\infty}^{p} \left(-\frac{(n-2)(p-1)}{(n-p)} \log t \right).$$

With the above relations, it becomes evident that the effective monotonicity of U^p_β and U^p_∞ is proved if we prove the substantial monotonicity of Φ^p_β and Φ^p_∞ . The following statements are the conformal versions of Theorems 2.7 and 2.8.

Theorem 2.13 (Effective monotonicity of Φ_{β}^{p}). Let $\Omega \subset M$ be an open bounded set with smooth boundary and let g, φ and Φ_{β}^{p} be defined as in (2.2.23), (2.2.24) and (2.2.45). Then, for any regular value $s \geq 0$, we have

$$(\Phi_{\beta}^{p})'(s) = -\beta \int_{\{\varphi = s\}} |\nabla \varphi|_{g}^{(\beta+1)(p-1)-1} H_{g} d\sigma_{g} \le 0$$
 (2.2.48)

and

$$\Phi^p_{\beta}(s) \le \Phi^p_{\beta}(0). \tag{2.2.49}$$

Moreover, equality holds in (2.2.48) for some regular s_0 if and only if $(\{\varphi \geq s_0\}, g)$ splits as $([s_0, +\infty) \times \{\varphi = s_0\}, d\rho \otimes d\rho + g_{\{\varphi = s_0\}})$ and φ , when restricted to $\{\varphi \geq s_0\}$, is an affine function of ρ .

On the other hand, the following is the conformal version of Theorem 2.14.

Theorem 2.14 (Effective monotonicity of Φ^p_∞). Let $\Omega \subset M$ be an open bounded set with smooth boundary and let g, φ and Φ^p_∞ be defined as in (2.2.23), (2.2.24) and (2.2.46). Then, for any regular $s \geq 0$ and any S > s we have

$$\Phi_{\infty}^{p}(S) \le \Phi_{\infty}^{p}(s). \tag{2.2.50}$$

Moreover, for any regular $s \geq 0$, we have

$$H_{g}(x_{s}) = -(p-1)\frac{\partial}{\partial \nu_{s}}\log|\nabla \varphi|_{g} \ge 0, \qquad (2.2.51)$$

where $x_s \in \{\varphi = s\}$ is the point where $\sup_{\{\varphi = s\}} |\nabla \varphi|_g$ is achieved and ν_s is the unit normal $\nabla \varphi / |\nabla \varphi|_g$ to $\{\varphi = s\}$. Moreover, equality holds in (2.2.50) for $s_0 < S$ regular values or in (2.2.51) at s_0 with s_0 a regular value if and only if $(\{\varphi \geq s_0\}, g)$ splits as $([s_0, +\infty) \times \{\varphi = s_0\}, d\rho \otimes d\rho + g_{\{\varphi = s_0\}})$ and φ , when restricted to $\{\varphi \geq s_0\}$, is affine function of ρ .

It is easy to see from formulas (2.2.47), (2.2.38), (2.2.41) and (2.2.42) that the effective monotonicity of Φ^p_β and Φ^p_∞ implies that of U^p_β and U^p_∞ , as well as (2.2.48) and (2.2.51) imply (2.2.18) and (2.2.20). It remains to check that the cylindrical splitting of ($\{\varphi \geq$

 s_0 , g) implies that the level set of u corresponding to { $\varphi = s_0$ } is isometric to a sphere in \mathbb{R}^n .

The cylindrical splitting of g implies rotational symmetry of u. As anticipated above, we just show that the rigidity statements of Theorems 2.13 and 2.14 imply those related to the effective monotonicity of U^p_β and U^p_∞ . It suffices to notice that, if φ is an affine function of ρ , as claimed in the conformal versions of the monotonicity theorems, then $|\nabla\nabla\varphi|^2_g=0$. By means of (2.2.4), this implies that $h_g=H_g/(n-1)g^\top$. Since, by (2.2.27) together with (2.2.40) and (2.2.41), we have, after a lengthy computation,

$$|\nabla \varphi|_g^2 \left| \mathbf{h}_g - \frac{\mathbf{H}_g}{n-1} g^\top \right|_{g^\top}^2 = \left[\frac{(p-1)(n-2)}{(n-p)} \right]^2 u^{-2\frac{n+p-2}{n-p}} |\mathbf{D}u|^2 \left| \mathbf{h} - \frac{\mathbf{H}}{n-1} g_{\mathbb{R}^n}^\top \right|_{g_{\mathbb{R}^n}}^2,$$

where the tangential elements of the left hand side are referred to the level set $\{\varphi = s_0\}$, and the tangential elements in the right hand side are referred to correspondent level set of u, we deduce that $\{u = t_0\} = \{\varphi = s_0\}$ is totally umbilical, and thus a sphere.

2.3 Proof of the Effective Monotonicity Theorems

The aim of this section is to give a complete proof of Theorems 2.13 and 2.14. Being much easier, we will establish first the effective monotonicity of Φ^p_∞ .

Since all the computations of this section will be performed in the conformally related setting, the subscript *g* will be dropped from the notations.

2.3.1 The Effective Monotonicity of Φ^p_{∞} .

In proving the maximum principle enjoyed by $|\nabla \varphi|$ that as in the last chapter roughly corresponds to the monotonicity of Φ^p_∞ , we are not using the same argument used for harmonic functions. Indeed, what we propose is an adaptation to *p*-capacitary potentials of the argument used for the sharp estimate on the harmonic Green's function on manifolds with nonnegative Ricci curvature provided by Colding in [Col12, Theorem 3.1].

We start by showing that $|\nabla \varphi|^p$ is a subsolution of the operator $\mathscr L$ acting on a smooth function f as

$$\mathscr{L}f = \Delta f + (p-2)\nabla\nabla f\left(\frac{\nabla\varphi}{|\nabla\varphi|}, \frac{\nabla\varphi}{|\nabla\varphi|}\right) - \frac{n-p}{n-2}\langle\nabla f \mid \nabla\varphi\rangle, \tag{2.3.1}$$

in a neighbourhood of any point $x \in \mathbb{R}^n \setminus \overline{\Omega}$ such that $|\nabla \varphi| > 0$, with φ a solution to (2.2.37). Observe that it appears on the right hand side of (2.2.44) applied to the function $|\nabla \varphi|^p$.

Lemma 2.15. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Let \mathscr{L} be the differential operator defined in (2.3.1). Then, a solution φ to (2.2.37) satisfies

$$\mathcal{L}(|\nabla \psi|^p) \ge 0 \tag{2.3.2}$$

in a neighbourhood of any point $x \in \mathbb{R}^n \setminus \overline{\Omega}$ such that $|\nabla \varphi|(x) > 0$.

Proof. By (2.2.44), we have

$$\mathscr{L}(|\nabla \varphi|^p) = p|\nabla \varphi|^{p-2} \left(|\nabla \nabla \varphi| + p(p-2) \left\langle \nabla |\nabla \varphi| \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle^2 \right).$$

By the standard Kato inequality $|\nabla\nabla\varphi|^2 \ge |\nabla|\nabla\varphi||^2$, we obtain

$$\mathscr{L}(|\nabla \varphi|^p) \geq p|\nabla \varphi|^{p-2} \left(|\nabla |\nabla \varphi||^2 + p(p-2) \left\langle \nabla |\nabla \varphi| \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle^2 \right),$$

and, since

$$|\nabla|\nabla\varphi||^2 + p(p-2)\left\langle\nabla|\nabla\varphi|\left|\frac{\nabla\varphi}{|\nabla\varphi|}\right\rangle^2 = \left|\nabla^\top|\nabla\varphi|\right|^2 + (p-1)^2\left\langle\nabla|\nabla\varphi|\left|\frac{\nabla\varphi}{|\nabla\varphi|}\right\rangle^2 \ge 0,$$
 we get (2.3.2).

The following is just a computational lemma yielding a function lying in the kernel of (2.3.1).

Lemma 2.16. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Let \mathscr{L} be the differential operator defined in (2.3.1). Let φ be a solution to (2.2.37). Then

$$\mathscr{L}\left(\mathrm{e}^{\frac{n-p}{(n-2)(p-1)}\varphi}\right)=0,$$

in a neighbourhood of any $x \in \mathbb{R}^n \setminus \overline{\Omega}$ with $|\nabla \varphi|(x) > 0$.

Proof. We compute separately the three terms forming $\mathscr{L}(\mathrm{e}^{\frac{n-p}{(n-2)(p-1)}\phi})$. First, we have

$$\Delta\left(e^{\frac{n-p}{(n-2)(p-1)}\varphi}\right) = \frac{n-p}{(n-2)(p-1)}e^{\frac{n-p}{(n-2)(p-1)}\varphi}\left(\frac{n-p}{(n-2)(p-1)}|\nabla\varphi|^2 + \Delta\varphi\right)$$
(2.3.3)

Using

$$\Delta_p \varphi = |\nabla \varphi|^{p-2} \left(\Delta \varphi + (p-2) \nabla \nabla \varphi \left(\frac{\nabla \varphi}{|\nabla \varphi|}, \frac{\nabla \varphi}{|\nabla \varphi|} \right) \right) = 0,$$

where the last equality follows from the *p*-harmonicity of φ , we obtain, from (2.3.3) and the fact that $|\nabla \varphi| \neq 0$

$$\Delta\left(e^{\frac{n-p}{(n-2)(p-1)}\varphi}\right) = \frac{n-p}{(n-2)(p-1)}e^{\frac{n-p}{(n-2)(p-1)}\varphi}\left[\frac{n-p}{(n-2)(p-1)}|\nabla\varphi|^2 - (p-2)\nabla\nabla\varphi\left(\frac{\nabla\varphi}{|\nabla\varphi|},\frac{\nabla\varphi}{|\nabla\varphi|}\right)\right].$$

The second term to be computed in $\mathscr{L}(\mathrm{e}^{\frac{n-p}{(n-2)(p-1)}\varphi})$ is

$$\nabla\nabla\left(e^{\frac{n-p}{(n-2)(p-1)}\varphi}\right)\left(\frac{\nabla\varphi}{|\nabla\varphi|},\frac{\nabla\varphi}{|\nabla\varphi|}\right) = \frac{n-p}{(n-2)(p-1)}e^{\frac{n-p}{(n-2)(p-1)}\varphi}\left[\frac{n-p}{(n-2)(p-1)}|\nabla\varphi|^{2} + \nabla\nabla\varphi\left(\frac{\nabla\varphi}{|\nabla\varphi|},\frac{\nabla\varphi}{|\nabla\varphi|}\right)\right],$$
(2.3.5)

and the last one is

$$\left\langle \nabla e^{\frac{n-p}{(n-2)(p-1)}\varphi} \,\middle| \, |\nabla \varphi \right\rangle = \frac{n-p}{(n-2)(p-1)} e^{\frac{n-p}{(n-2)(p-1)}\varphi} |\nabla \varphi|^2. \tag{2.3.6}$$

By (2.3.4), (2.3.5) and (2.3.6) we get
$$\mathcal{L}(e^{\frac{n-p}{(n-2)(p-1)}\varphi}) = 0$$
, as claimed.

We now apply the above lemmas to trigger the barrier argument in turn leading to Theorem 2.14.

Proof of Theorem 2.14. We claim that

$$|\nabla \varphi|(x) \le \sup_{\{\varphi = s\}} |\nabla \varphi| \tag{2.3.7}$$

for any $x \in \{\varphi \ge s\}$ and any noncritical s. This clearly suffices to prove (2.2.50). Indeed, for any $S \ge s$, (2.3.7) implies

$$\sup_{\{\varphi=s\}} |\nabla \varphi| \geq \sup_{\{\varphi \geq s\}} |\nabla \varphi| \geq \sup_{\{\varphi \geq S\}} |\nabla \varphi| \geq \sup_{\{\varphi=S\}} |\nabla \varphi|.$$

Let then s be a noncritical value of φ . By the asymptotic expansions recalled in Proposition 2.3, the relation (2.2.38) and the continuity of $|\nabla \varphi|$ following from the $\mathscr{C}^{1,\alpha}$ regularity of u, it is immediate to deduce that

$$|\nabla \varphi| \leq C$$

uniformly on $\mathbb{R}^n \setminus \Omega$. Consider then, for such a constant C and for an auxiliary S > s that we suppose to be regular (as usual, this is possible by the compactness of $Crit\varphi$), the function

$$w = |\nabla \varphi|^p - \sup_{\{\varphi = s\}} |\nabla \varphi|^p - C^p e^{\frac{n-p}{(n-2)(p-1)}(\varphi - S)}$$

defined on $\{s \le \varphi \le S\}$. Let then $\delta > 0$ such that $N_{\delta} = \{|\nabla \varphi| \le \delta\} \subset \{s < \varphi < S\}$. By the smoothness of φ , and in turn of $|\nabla \varphi|$, around the boundary of N_{δ} , we can suppose, by Sard Theorem, that ∂N_{δ} is a smooth hypersurface. By definition, we have

$$\sup_{\{\varphi=s\}\cup\{\varphi=S\}\cup\partial N_\delta}w\leq 0.$$

Moreover, by Lemma 2.15 and 2.16, w satisfies $\mathscr{L}w \geq 0$ on $\{s < \varphi < s\} \setminus \overline{N_\delta}$, and since this operator is uniformly elliptic in this bounded set with smooth boundary, the Maximum Principle applies and yields

$$|\nabla \varphi|^p \le \sup_{\{\varphi = s\}} |\nabla \varphi|^p + C^p e^{\frac{n-p}{(n-2)(p-1)}(\varphi - S)}.$$
 (2.3.8)

on $\{s \leq \varphi \leq S\} \setminus N_{\delta}$. On the other hand, upon choosing a smaller δ , we can clearly suppose that (2.3.8) is satisfied also on N_{δ} , and thus in the whole $\{s \leq \varphi \leq S\}$. By passing to the limit $S \to +\infty$, this proves (2.3.7).

We now turn to prove (2.2.51). Let then, for s nonsingular, $x_s \in \{\varphi = s\}$ be the point where the maximum of $|\nabla \varphi|$ in $\{\varphi = s\}$ is achieved. By (2.3.7), it is also the maximum point of $\{\varphi \geq s\}$, and in particular

$$\frac{\partial}{\partial \nu} |\nabla \varphi|^p(x_s) \le 0. \tag{2.3.9}$$

Using (2.2.39) we get

$$\left\langle \nabla |\nabla \varphi|^p, \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle (x_s) = -\frac{p}{p-1} (|\nabla \varphi|^p \mathbf{H}) (x_s),$$

where H is the mean curvature of the set $\{\varphi = x_s\}$ and inequality (2.2.51) follows.

Assume now that (2.2.50) holds with equality sign for $s_0 < S$ with s_0 regular. Then, by the effective monotonicity just proved we also have $\Phi^p_\infty(s_0) = \Phi^p_\infty(\tilde{S})$ with $\tilde{S} > s_0$ close enough to s_0 so that $|\nabla \varphi| > 0$ on $\{s_0 \le \varphi \le \tilde{S}\}$. Letting $x_{\tilde{S}}$ the maximum point of $|\nabla \varphi|^p$ on $\{\varphi = \tilde{S}\}$, we get in particular that

$$\sup_{\{\varphi=s_0\}\cup\{\varphi=\tilde{S}+\delta\}}|\nabla\varphi|^p=|\nabla\varphi|^p(x_{\tilde{S}}),$$

for some $\delta>0$ small enough so that $|\nabla\varphi|^p>0$ on $\{s_0\leq\varphi\leq\tilde{S}+\delta\}$. Since the point $x_{\tilde{S}}$ lies in the interior of $\{s_0\leq\varphi\leq\tilde{S}+\delta\}$, the Strong maximum principle, in force because of (2.3.2) ensures that $|\nabla\varphi|$ is a positive constant in $\{s_0\leq\varphi\leq\tilde{S}+\delta\}$. This fact, combined with the continuity of $|\nabla\varphi|$, coming from the $\mathscr{C}^{1,\alpha}$ -regularity of u, also easily shows that no singular values bigger than s_0 can occur. In particular, δ could be taken arbitrarily big, showing that $|\nabla\varphi|$ is constant on the whole $\{\varphi\geq s_0\}$, and we deduce the claimed splitting principle from Lemma 2.12.

We are left to show that the same splitting happens if (2.2.51) holds with equality sign. In this case, in particular, the normal derivative in (2.3.9) vanishes. But since x_s is a global maximum value for $|\nabla \varphi|^p$ on $\{s \le \varphi \le S\}$ for any S > s such that $|\nabla \varphi| > 0$ on $\{s \le \varphi \le S\}$, and $|\nabla \varphi|^p$ is subsolution of the elliptic equation $\mathcal{L}f = 0$ by (2.3.2), Hopf's lemma implies that $|\nabla \varphi|^p$ is constant on this region. We can deduce arguing as above that there are no singular values bigger than s_0 , that then $|\nabla \varphi|$ is constant on the whole $\{\varphi \ge s_0\}$ and conclude with Lemma 2.12.

2.3.2 The Effective Monotonicity of Φ^p_{β} .

For a given 1 , let us consider the vector field

$$X = e^{-\frac{(n-p)}{(n-2)(p-1)}\varphi} |\nabla \varphi|^{p-2} \left(\nabla |\nabla \varphi|^{\beta(p-1)} + (p-2)\nabla^{\perp} |\nabla \varphi|^{\beta(p-1)} \right)$$
(2.3.10)

defined around points where $|\nabla \varphi|$ does not vanish. Let us first check the relation between the vector field X and the derivative of Φ_{β}^{p} at a regular value s.

Proposition 2.17. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Let Φ_{β}^p be the function defined in (2.2.45) and X the vector field defined in (2.3.10). Then, for any regular value $s \geq 0$, we have

$$e^{-\frac{(n-p)}{(n-2)(p-1)}s} (\Phi_{\beta}^{p})'(s) = \frac{1}{p-1} \int_{\{\varphi=s\}} \left\langle X \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma.$$
 (2.3.11)

Proof. It suffices to show that

$$\Phi_{\beta}'(s) = \int_{\{\varphi = s\}} \left\langle |\nabla \varphi|^{p-2} \nabla |\nabla \varphi|^{\beta(p-1)} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma.$$
 (2.3.12)

Indeed, it is immediately checked that

$$e^{-\frac{(n-p)}{(n-2)(p-1)}s}\left\langle |\nabla \varphi|^{p-2}\nabla |\nabla \varphi|^{\beta(p-1)}\left|\frac{\nabla \varphi}{|\nabla \varphi|}\right\rangle = \frac{1}{p-1}\left\langle X\left|\frac{\nabla \varphi}{|\nabla \varphi|}\right\rangle.$$

Let us then show (2.3.12). Let $\delta > 0$ small enough so that $s + \delta$ is still a regular value. We can write

$$\begin{split} \Phi_{\beta}^{p}(s+\delta) - \Phi_{\beta}^{p}(s) &= \int\limits_{\{\varphi = s + \delta\}} \left\langle \nabla \varphi | \nabla \varphi |^{(\beta+1)(p-1)-1} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle \, \mathrm{d}\sigma \right. \\ &- \int\limits_{\{\varphi = s\}} \left\langle \nabla \varphi | \nabla \varphi |^{(\beta+1)(p-1)-1} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle \, \mathrm{d}\sigma. \end{split}$$

Using the Divergence Theorem, we can write the above quantity as

$$\Phi^p_{\beta}(s+\delta) - \Phi^p_{\beta}(s) = \int\limits_{\{s \leq \varphi \leq s+\delta\}} \operatorname{div} \left(\nabla \varphi |\nabla \varphi|^{p-2} |\nabla \varphi|^{\beta(p-1)} \right) \, \mathrm{d}\mu.$$

Since $\Delta_p \varphi = 0$, we get

$$\operatorname{div}\left(\nabla\varphi|\nabla\varphi|^{p-2}|\nabla\varphi|^{\beta(p-1)}\right) = \left\langle\nabla|\nabla\varphi|^{\beta(p-1)}\,\Big|\,\nabla\varphi\right\rangle|\nabla\varphi|^{p-2},$$

and by the coarea formula we obtain

$$\Phi_{\beta}^{p}(s+\delta) - \Phi_{\beta}^{p}(s) = \int_{s}^{s+\delta} \int_{\{\varphi=\tau\}} \left\langle \left(\nabla |\nabla \varphi|^{\beta(p-1)}\right) |\nabla \varphi|^{p-2} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma d\tau.$$

By dividing both sides of the above equality by δ , and letting $\delta \to 0^+$, we get (2.3.11) by the Fundamental Theorem of Calculus provided that the function I mapping

$$\tau \mapsto \int\limits_{\{\varphi = \tau\}} \left\langle \left(\nabla |\nabla \varphi|^{\beta(p-1)} \right) |\nabla \varphi|^{p-2} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle \, \mathrm{d}\sigma \right.$$

is continuous. In fact, fixed $\tau_0 \geq 0$ we have for any $\tau > \tau_0$ close enough to τ_0 , by Divergence Theorem

$$\left|I(\tau)-I(\tau_0)\right| \leq \int\limits_{\{\tau_0 \leq \varphi \leq \tau\}} \left|\operatorname{div}\left(\nabla |\nabla \varphi|^{\beta(p-1)}\right)|\nabla \varphi|^{p-2}\right)\right| \mathrm{d}\mu,$$

and the right hand side vanishes in the limit as $\tau \to \tau_0^+$ by Dominated Convergence Theorem.

In the next fundamental lemma, we compute the divergence of *X*.

Lemma 2.18 (Divergence of X). Let X be the vector field defined by (2.3.10). Then, the following identity holds in a neighbourhood of any point $x \in \mathbb{R}^n \setminus \overline{\Omega}$ such that $|\nabla \varphi|(x) \neq 0$

$$\operatorname{div} X = e^{-\frac{(n-p)}{(n-2)(p-1)}\varphi} Q > 0, \qquad (2.3.13)$$

where

$$Q = \beta (p-1) |\nabla \varphi|^{\beta(p-1)+p-4} \left\{ |\nabla \varphi|^2 \left| \mathbf{h} - \frac{\mathbf{H}}{n-1} g^\top \right|^2 + (p-1) \left[\beta + \left(\frac{p-2}{p-1} \right) \right] \left| \nabla^\top |\nabla \varphi| \right|^2 + (p-1)^2 \left[\beta - \frac{(n-p)}{(p-1)(n-1)} \right] \left| \nabla^\perp |\nabla \varphi| \right|^2 \right\},$$

$$(2.3.14)$$

where h and H are respectively the second fundamental form and the mean curvature of the level sets of φ with respect to the unit normal $\nabla \varphi / |\nabla \varphi|$.

Proof. For the sake of clearness, we write

$$X = e^{-\frac{(n-p)}{(n-2)(p-1)}}(W+Z),$$

where

$$W = \left| \nabla \varphi \right|^{p-2} \nabla \left| \nabla \varphi \right|^{\beta(p-1)}$$
,

and

$$Z = (p-2)|\nabla \varphi|^{p-2} \left\langle \nabla |\nabla \varphi|^{\beta(p-1)} \; \left| \; \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle \frac{\nabla \varphi}{|\nabla \varphi|}.$$

We now compute separately the divergence of W and Z.

The divergence of W. Simple computations give

$$\nabla |\nabla \varphi|^p = p|\nabla \varphi|^{p-1}\nabla |\nabla \varphi| \tag{2.3.15}$$

and

$$\nabla \nabla |\nabla \varphi|^{p}(\nabla \varphi, \nabla \varphi) = p(p-1)|\nabla \varphi|^{p-2}\langle \nabla |\nabla \varphi| |\nabla \varphi\rangle^{2} + p|\nabla \varphi|^{p-1}\nabla \nabla |\nabla \varphi|(\nabla \varphi, \nabla \varphi).$$
(2.3.16)

Moreover, we can write

$$W = \frac{\beta(p-1)}{p} |\nabla \varphi|^{\beta(p-1)-2} \nabla |\nabla \varphi|^{p}. \tag{2.3.17}$$

Using (2.3.15) and (2.3.17) we get

$$\begin{split} \operatorname{div} W &= \frac{\beta(p-1)}{p} |\nabla \varphi|^{\beta(p-1)-2} \Delta |\nabla \varphi|^p \\ &+ \big(\beta(p-1)\big) \big(\beta(p-1)-2\big) |\nabla \varphi|^{\beta(p-1)+p-4} \Big|\nabla |\nabla \varphi|\Big|^2. \end{split}$$

Plugging (2.2.44) into the above relation, and using identity (2.3.17), we obtain

$$\begin{split} \operatorname{div} W - \frac{n-p}{n-2} \langle W \, | \, \nabla \varphi \rangle &= \Big(\beta(p-1)\Big) |\nabla \varphi|^{\beta(p-1)+p-4} \\ &\times \left[|\nabla \nabla \varphi|^2 + \Big(\beta(p-1)-2\Big) \Big|\nabla |\nabla \varphi|\Big|^2 \\ &\quad - \frac{(p-2)}{p} \frac{\nabla \nabla |\nabla \varphi|^p (\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|^p} \\ &\quad + p(p-2) \bigg\langle \nabla |\nabla \varphi| \, \Big| \frac{\nabla \varphi}{|\nabla \varphi|} \bigg\rangle^2 \bigg], \end{split}$$

that, through (2.3.16), becomes

$$\operatorname{div} W = \frac{n-p}{n-2} \langle W, \nabla \varphi \rangle$$

$$+ \beta (p-1) |\nabla \varphi|^{\beta(p-1)+p-4} \left[|\nabla \nabla \varphi|^2 + \left[\beta(p-1) - 2 \right] |\nabla |\nabla \varphi| \right]^2$$

$$+ (p-2) \left(\left\langle \nabla |\nabla \varphi| \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle^2 - \frac{\nabla \nabla |\nabla \varphi| (\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|} \right) \right].$$

$$(2.3.18)$$

It is time to insert the Kato-type identity (2.2.4) in (2.3.18). By that and the frequently used decomposition

$$\left|\nabla|\nabla\varphi|\right|^2 = \left|\nabla^{\top}|\nabla\varphi|\right|^2 + \left|\nabla^{\perp}|\nabla\varphi|\right|^2, \tag{2.3.19}$$

we immediately get

$$\operatorname{div} W = \frac{n-p}{n-2} \left\langle W \mid \nabla \varphi \right\rangle + \beta \left(p-1 \right) |\nabla \varphi|^{\beta(p-1)+p-4} \left\{ \beta(p-1) \left| \nabla^{\top} |\nabla \varphi| \right|^{2} + \left[\beta(p-1) + \frac{(p-1)^{2}}{n-1} - 1 + (p-2) \right] \left| \nabla^{\perp} |\nabla \varphi| \right|^{2} - (p-2) \frac{\nabla \nabla |\nabla \varphi| (\nabla \varphi, \nabla \varphi)}{|\nabla \varphi|} \right\}.$$

$$(2.3.20)$$

The divergence of Z. Clearly, by the *p*-harmonicity of φ , we have

$$\operatorname{div} Z = (p-2)|\nabla \varphi|^{p-2} \left\langle \nabla \left[\left\langle \nabla |\nabla \varphi|^{\beta(p-1)} \left| \frac{\nabla \varphi}{|\nabla \varphi|^2} \right\rangle \right] \right| \nabla \varphi \right\rangle.$$

We then write

$$\left\langle \nabla \left| \nabla \varphi \right|^{\beta(p-1)} \, \left| \, \frac{\nabla \varphi}{\left| \nabla \varphi \right|^2} \right\rangle = \beta(p-1) |\nabla \varphi|^{\beta(p-1)-3} \left\langle \nabla \left| \nabla \varphi \right| \, \left| \, \nabla \varphi \right\rangle,$$

and by a straightforward computation

$$\begin{aligned} \operatorname{div} Z &= (p-2)\,\beta\,(p-1)|\nabla\varphi|^{\beta(p-1)+p-4} \Bigg\{ \frac{\nabla\nabla|\nabla\varphi|(\nabla\varphi,\nabla\varphi)}{|\nabla\varphi|} \\ &\quad + \left[\beta(p-1)^2 - \beta(p-1) - 2(p-2)\right] \left\langle \nabla|\nabla\varphi|, \frac{\nabla\varphi}{|\nabla\varphi|} \right\rangle^2 \\ &\quad + \left(p-2\right) \left|\nabla^\top|\nabla\varphi|\right|^2 \Bigg\}, \end{aligned} \tag{2.3.21}$$

where we just used the decomposition (2.3.19) and the general fact

$$\frac{\nabla \nabla \varphi (\nabla |\nabla \varphi|, \nabla \varphi)}{|\nabla \varphi|} = |\nabla |\nabla \varphi||^2.$$

Completion of the computation. Summing up the expressions (2.3.20) and (2.3.21), and observing that

$$\langle W \, | \, \nabla \varphi \rangle = \frac{1}{p-1} \, \langle X \, | \, \nabla \varphi
angle$$
 ,

we finally get

$$\operatorname{div}(W+Z) - \frac{(n-p)}{(n-1)(p-1)} \langle (W+Z) \mid \nabla \varphi \rangle = \beta (p-1) |\nabla \varphi|^{\beta(p-1)+p-4} \left\{ |\nabla \varphi|^2 \left| \mathbf{h} - \frac{\mathbf{H}}{n-1} g^\top \right|^2 + (p-1) \left[\beta + \left(\frac{p-2}{p-1} \right) \right] \left| \nabla^\top |\nabla \varphi| \right|^2 + (p-1)^2 \left[\beta - \frac{(n-p)}{(p-1)(n-1)} \right] \left| \nabla^\perp |\nabla \varphi| \right|^2 \right\},$$

that is clearly equivalent to (2.3.13).

In absence of critical points, the Divergence Theorem applied to the vector field X on the open region $\{s < \varphi < S\}$, with 0 < s < S, easily yields the inequality

$$\int\limits_{\{\varphi=s\}}\!\!\left\langle X\left|\,\frac{\nabla\varphi}{|\nabla\varphi|}\right\rangle\,\mathrm{d}\sigma\right| \leq \int\limits_{\{\varphi=S\}}\!\!\left\langle X\left|\,\frac{\nabla\varphi}{|\nabla\varphi|}\right\rangle\,\mathrm{d}\sigma\,,$$

and in turns, thanks to (2.3.11), (2.3.22) below. In presence of a possibly wild critical set, this direct argument is no longer working. Fortunately, some of the new ideas introduced in [AM20] to treat the same issues in the case of harmonic functions are exportable to the case of p-harmonic functions, where one does not know a priori that the critical set is (n-1)-negligible. As a consequence, we are still able to provide an effective version of the considered monotonicity, showing that (2.3.22) is actually in force, provided s is small enough and s is large enough. The desired effective inequality $(\Phi_{\beta}^p)'(s) \leq 0$ for nonsingular values s, that is (2.2.48), will follow at once. It will also be clear that $(\Phi_{\beta}^p)'(s_0) = 0$ only if $(\{\varphi \geq s_0\}, g)$ splits a Riemannian product.

Remark 2.19 (Asymptotic boundedness of Φ_{β}^{p}). Before proving the above facts, let us briefly observe that the function Φ_{β}^{p} are asymptotically bounded for any $\beta \geq 0$. Indeed,

this immediately follows from the smoothness of such function at values s big enough and the finite limit it achieves at infinity, coinciding with that of U_6^p computed in (2.2.21).

Theorem 2.20 (First substantial inequality). Let $\Omega \subset \mathbb{R}^n$ be a bounded set with smooth boundary, and let Φ_{β}^p be the function defined in (2.2.45). Then, for all regular values s < S and any $\beta \ge (n-p)/[(n-1)(p-1)]$, the inequality

$$\frac{(\Phi_{\beta}^{p})'(s)}{e^{\frac{(n-p)}{(n-1)(p-1)}s}} \leq \frac{(\Phi_{\beta}^{p})'(s)}{e^{\frac{(n-p)}{(n-1)(p-1)}s}}$$
(2.3.22)

holds true. In particular, one has that $(\Phi_{\beta}^p)'(s) \leq 0$ for any nonsingular $s \geq 0$, that is (2.2.48). Moreover, if $(\Phi_{\beta}^p)'(s_0) = 0$ for some nonsingular $s_0 \geq 0$, then $(\{\varphi \geq s_0\}, g)$ is isometric to the truncated Riemannian cylinder $([s_0, \infty), d\rho \otimes d\rho + g_{\{\varphi = s_0\}}, and \varphi$ is an affine function of ρ in $\{\varphi \geq 0\}$.

Proof. Let $\varepsilon > 0$, we consider a smooth nonnegative cut-off-function $\chi : [0, +\infty) \to \mathbb{R}$, such that

$$\begin{cases} \chi(t) = 0 & \text{in } t < \frac{1}{2}\varepsilon, \\ \dot{\chi}(t) \ge 0 & \text{in } \frac{1}{2}\varepsilon \le t \le \frac{3}{2}\varepsilon, \\ \chi(t) = 1 & \text{in } t > \frac{3}{2}\varepsilon. \end{cases}$$
 (2.3.23)

Since $\chi(|\nabla \varphi|^{\beta(p-1)}) = 0$ on Crit $\varphi \cap \{s < \varphi < S\}$ we can apply the Divergence Theorem to the smooth vector field

$$\widetilde{X} = \chi(|\nabla \varphi|^{\beta(p-1)}) X$$

in the domain $\{s < \varphi < S\}$. Observe that, choosing ε small enough, we can make sure that $\chi(|\nabla \varphi|^{\beta(p-1)}) = 1$ on $\{\varphi = s\}$ and $\{\varphi = S\}$. Having this in mind, we compute

$$\int_{\{\varphi=S\}} \left\langle X \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma - \int_{\{\varphi=s\}} \left\langle X \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma \right| = \int_{\{s < \varphi < S\}} \operatorname{div} \widetilde{X} d\mu$$

$$= \int_{\{s < \varphi < S\} \setminus N_{\varepsilon/2}} \chi(|\nabla \varphi|^{\beta(p-1)}) \operatorname{div} X d\mu + \int_{N_{3\varepsilon/2} \setminus N_{\varepsilon/2}} \dot{\chi}(|\nabla \varphi|^{\beta(p-1)}) \left\langle X \left| \nabla |\nabla \varphi|^{\beta(p-1)} \right\rangle d\mu,$$

$$= \int_{\{s < \varphi < S\} \setminus N_{\varepsilon/2}} \chi(|\nabla \varphi|^{\beta(p-1)}) \operatorname{div} X d\mu + \int_{N_{3\varepsilon/2} \setminus N_{\varepsilon/2}} \dot{\chi}(|\nabla \varphi|^{\beta(p-1)}) \left\langle X \left| \nabla |\nabla \varphi|^{\beta(p-1)} \right\rangle d\mu,$$
(2.3.24)

where in the last identity we have used the tubular neighbourhood of Crit $\varphi \cap \{s < \varphi < S\}$ defined for every $\delta > 0$ as $N_{\delta} = \{|\nabla \varphi|^{\beta(p-1)} \leq \delta\}$. In view of (2.3.11), (2.3.13), (2.3.23) and (2.3.24), the inequality (2.3.22) is proved if we show that $\langle X | \nabla | \nabla \varphi |^{\beta(p-1)} \rangle \geq 0$ on $N_{3\varepsilon/2} \setminus N_{\varepsilon/2}$. On the other hand, a direct computation gives

$$\begin{split} \left\langle X \, \big| \, \nabla |\nabla \varphi|^{\beta(p-1)} \right\rangle \; &= \; \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} \, |\nabla \varphi|^{p-2} \, \left[\; \big| \nabla |\nabla \varphi|^{\beta(p-1)} \big|^2 + (p-2) \big| \nabla^{\perp} |\nabla \varphi|^{\beta(p-1)} \big|^2 \; \right] \\ &= \; \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} \, |\nabla \varphi|^{p-2} \, \left[\; \big| \nabla^{\top} |\nabla \varphi|^{\beta(p-1)} \big|^2 + (p-1) \big| \nabla^{\perp} |\nabla \varphi|^{\beta(p-1)} \big|^2 \; \right] \\ &\geq \; 0. \end{split}$$

This completes the proof of (2.3.22). We show that it implies $(\Phi_{\beta}^{p})'(s) \leq 0$. Recalling that by the asymptotic expansions in Lemma 2.3 any level set $\varphi(S)$ is smooth for S big

enough, say for $S \ge S_p$, it follows at once from (2.3.22) that, for every $S \ge S_p$, it holds

$$e^{\frac{n-p}{(n-1)(p-1)\varphi}S}\left(\Phi_{\beta}^{p}\right)'(s) \leq e^{\frac{n-p}{(n-1)(p-1)\varphi}S}\left(\Phi_{\beta}^{p}\right)'(S).$$

Integrating both sides of the above inequality in the variable S on an interval of the form (S_p, S) , we obtain

$$e^{\frac{(n-p)}{(n-2)(p-1)}S}\,(\Phi_{\beta}^p)'(s)\,+\,e^{\frac{n-p}{(n-1)(p-1)\varphi}s}\,\Phi_{\beta}^p(S_p)\,-\,e^{\frac{n-p}{(n-2)(p-1)}S_p}\,(\Phi_{\beta}^p)'(s)\,\,\leq\,\,e^{\frac{n-p}{(n-1)(p-1)\varphi}s}\Phi_{\beta}^p(S)\,.$$

If by contradiction, $(\Phi_{\beta}^p)'(s) > 0$, then, letting $S \to +\infty$ in the above identity, we would deduce that $\Phi_{\beta}^p(S) \to +\infty$, against the asymptotic boundedness of Φ_{β}^p discussed in Remark 2.19.

We are left to show the rigidity statement occurring when $(\Phi_{\beta}^p)'(s_0) = 0$ for some $s_0 \ge 0$. Let $S > s_0$ be a regular value close enough to s_0 so that $|\nabla \varphi| > 0$ on $\{s_0 \le \varphi \le S\}$. Then, the Divergence Theorem and (2.3.11) show that

$$0 \geq (p-1)e^{-\frac{(n-p)}{(n-2)(p-1)}S} \left(\Phi_{\beta}^{p}\right)'(S) = \int_{\{\varphi=S\}} \left\langle X \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma - \int_{\{\varphi=s_{0}\}} \left\langle X \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma \right.$$

$$= \int_{\{s < \varphi < S\}} \operatorname{div} X d\mu \geq 0,$$

$$\{s < \varphi < S\}$$

$$(2.3.25)$$

and thus div X vanishes on $\{s_0 \le \varphi \le S\}$. If $\beta > (n-p)/[(n-1)(p-1)]$, then the expression (2.3.13) implies that $|\nabla|\nabla\varphi||=0$, and thus by Lemma 2.12 ($\{s_0 \le \varphi \le S\}, g\}$) splits a compact Riemannian product. Arguing as in the proof of the rigidity part of Theorem 2.14, this also implies by the continuity of $|\nabla\varphi|$ that no singular values bigger than s_0 can exist, and the isometry with the Riemannian product extends to the whole $(\{\varphi \ge s_0\}, g)$. If on the other hand $\beta = (n-p)/[(n-1)(p-1)]$, then (2.3.25) coupled with (2.3.13) just implies that (2.2.5) and (2.2.6) holds for φ . The rigidity statement in the Kato-type identity Proposition 2.5 implies that $(\{s_0 \le \varphi \le S\}, g)$ splits a warped product, whose cross section are given by level sets of φ . By the vanishing of $(\Phi_\beta^p)'(S)$ deduced from (2.3.25) together with the second expression in (2.3.11), these level sets are actually minimal. By the relation (2.2.39), this implies that also $\nabla^\top|\nabla\varphi|=0$, and this fact coupled with (2.2.6) for φ implies that $|\nabla|\nabla\varphi||=0$ on $(\{s_0, \le \varphi \le S\}, g)$, and we can conclude as before.

As already observed several times, the presence of critical points and critical values possibly arranged in sets with full measure makes the full monotonicity not expectable in general. In fact, the lack of a sufficiently strong Sard-type property for the p-capacitary potentials prevents any kind of straightforward adaptation of the arguments presented in [AM20]. In other words, there is no hope for deducing the global inequality $\Phi^p_\beta(+\infty) \le \Phi^p_\beta(0)$ from the pointwise inequality $(\Phi^p_\beta)'(s) \le 0$ through integration, since the latter inequality may fail to be true – or even well defined – for too many values of $s \in [0, +\infty)$. To face this main difficulty, we craft a new family of auxiliary monotonicity formulas. For a given $1 and a given <math>0 < \lambda < 1$, we consider the vector field

$$Y_{\lambda} = \left(e^{\frac{(n-p)}{(n-2)(p-1)}\varphi} - \lambda \right) X - \left(\frac{n-p}{n-2} \right) |\nabla \varphi|^{\beta(p-1)+p-2} \nabla \varphi, \qquad (2.3.26)$$

where *X* has been defined in (2.3.10). It is convenient to observe that at a regular value of φ it holds

$$\left(\frac{e^{\frac{(n-p)}{(n-2)(p-1)}s} - \lambda}{e^{\frac{(n-p)}{(n-2)(p-1)}s}}\right) (\Phi_{\beta}^{p})'(s) - \frac{(n-p)}{(n-2)(p-1)} \Phi_{\beta}^{p}(s) = \frac{1}{(p-1)} \int_{\{\varphi=s\}} \left\langle Y_{\lambda} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma,$$
(2.3.27)

by the definitions of the vector fields and the functions involved and Proposition 2.17. In the next lemma, we compute the divergence of Y_{λ} , that happens to be nonnegative when the so is the divergence of X.

Lemma 2.21 (Divergence of Y_{λ}). Let Y_{λ} be the vector field defined in (2.3.26). Then, the following identity holds in a neighbourhood of any point $x \in \mathbb{R}^n \setminus \overline{\Omega}$ such that $|\nabla \varphi|(x) \neq 0$

$$\operatorname{div}Y_{\lambda} = \left(\frac{e^{\frac{(n-p)}{(n-2)(p-1)}\varphi} - \lambda}{\frac{(n-p)}{Q(n-2)(p-1)}\varphi}\right)Q \geq 0,$$

where Q is the quantity defined in (2.3.14).

Proof. By the very definition of Y_{λ} , we have that

$$\begin{array}{ll} \mathrm{div} Y_{\lambda} \; = \; \left(\mathrm{e}^{\frac{(n-p)}{(n-2)(p-1)} \varphi} - \lambda \right) \, \mathrm{div} X \; + \; \frac{(n-p)}{(n-2)(p-1)} \, \mathrm{e}^{\frac{(n-p)}{(n-2)(p-1)} \varphi} \, \langle X \, | \, \nabla \varphi \rangle \\ & - \left(\frac{n-p}{n-2} \right) \, \mathrm{div} \big(|\nabla \varphi|^{\beta(p-1)+p-2} \nabla \varphi \big) \, . \end{array}$$

Using the definition (2.3.10) of the vector field X, we compute

$$\langle X \,|\, \nabla \varphi \rangle = (p-1) \, \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} \, |\nabla \varphi|^{p-2} \, \langle \nabla |\nabla \varphi| \, |\, \nabla \varphi \rangle \,.$$

Exploiting the *p*-harmonicity of φ , we get

$$\operatorname{div} \! \left(|\nabla \varphi|^{\beta(p-1)+p-2} \nabla \varphi \right) \; = \; |\nabla \varphi|^{p-2} \left\langle \nabla |\nabla \varphi|^{\beta(p-1)} \, \big| \, \nabla \varphi \right\rangle \, .$$

We conclude that

$$\operatorname{div}Y_{\lambda} = \left(e^{\frac{(n-p)}{(n-2)(p-1)}\varphi} - \lambda\right)\operatorname{div}X = \left(\frac{e^{\frac{(n-p)}{(n-2)(p-1)}\varphi} - \lambda}{e^{\frac{(n-p)}{(n-2)(p-1)}\varphi}}\right)Q,$$

where in the last equality we made use of the identity (2.3.13).

Again, in absence of critical points, the Divergence Theorem applied to the vector field Y_{λ} on the open region $\{s < \varphi < S\}$ easily yields the inequality

$$\int\limits_{\{\varphi=s\}} \left\langle Y_{\lambda} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma \right| \leq \int\limits_{\{\varphi=S\}} \left\langle Y_{\lambda} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma,$$

and in turns, thanks to (2.3.27), the inequality (2.3.28) below. As usual, the difficult part is the treatment of the critical points. However, a quite surprising computation in the spirit of Theorem 2.20 shows that it is always possible to deduce the second effective inequality, that is (2.2.49). This fact, together with Theorem 2.22, completes the proof of the Effective Monotonicity-Rigidity Theorem for Φ^p_β .

Theorem 2.22 (Second effective inequality). Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Then, for every $0 < \lambda < 1$ and any couple nonsingular values s < S, the inequality

$$\left(\frac{e^{\frac{(n-p)s}{(n-2)(p-1)}} - \lambda}{e^{\frac{(n-p)s}{(n-2)(p-1)}}}\right) (\Phi_{\beta}^{p})'(s) - \frac{(n-p)\Phi_{\beta}^{p}(s)}{(n-2)(p-1)} \leq \left(\frac{e^{\frac{(n-p)S}{(n-2)(p-1)}} - \lambda}{e^{\frac{(n-p)S}{(n-2)(p-1)}}}\right) (\Phi_{\beta}^{p})'(S) - \frac{(n-p)\Phi_{\beta}^{p}(S)}{(n-2)(p-1)} \tag{2.3.28}$$

holds true, where Φ_{β}^{p} is the function defined in (2.2.24). In particular, one has that $\Phi_{\beta}^{\hat{p}}(S) \leq \Phi_{\beta}^{p}(0)$, that is (2.2.49).

Proof. Let $\chi:[0,+\infty)\to\mathbb{R}$ be the same smooth nonnegative cut-off function as in the proof of Theorem 2.20, so that the properties (2.3.23) are in force. To simplify the notation, let us also set

$$\eta_{\lambda}(\varphi) = \frac{1}{e^{\frac{(n-p)}{(n-2)(p-1)}\varphi} - \lambda}.$$

Finally, let us consider the smooth vector field

$$\widetilde{Y}_{\lambda} = \chi \left(\eta_{\lambda}(\varphi) |\nabla \varphi|^{\beta(p-1)} \right) Y_{\lambda},$$

where Y_{λ} has been defined in (2.3.26). Again, choosing ε small enough, we can suppose $\widetilde{Y}_{\lambda} = Y_{\lambda}$ on $\{\varphi = s\}$ and $\{\varphi = S\}$, with for nonsingular s and S and apply the Divergence Theorem to the smooth vector field \widetilde{Y}_{λ} on the region $\{s < \varphi < S\}$. It yields

$$\int_{\{\varphi=S\}} \left\langle Y_{\lambda} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma - \int_{\{\varphi=s\}} \left\langle Y_{\lambda} \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle d\sigma \right| = \int_{\{s < \varphi < S\}} \operatorname{div} \widetilde{Y}_{\lambda} d\mu \right| =$$

$$= \int_{\{s < \varphi < S\} \setminus N_{\varepsilon/2}} \chi(\eta_{\lambda}(\varphi) |\nabla \varphi|) \operatorname{div} Y_{\lambda} d\mu + \int_{N_{3\varepsilon/2} \setminus N_{\varepsilon/2}} \dot{\chi}(\eta_{\lambda}(\varphi) |\nabla \varphi|) \left\langle Y_{\lambda} \left| \nabla \left(\eta_{\lambda}(\varphi) |\nabla \varphi|^{\beta(p-1)} \right) \right\rangle d\mu,$$

$$= \int_{\{s < \varphi < S\} \setminus N_{\varepsilon/2}} \chi(\eta_{\lambda}(\varphi) |\nabla \varphi|) \operatorname{div} Y_{\lambda} d\mu + \int_{N_{3\varepsilon/2} \setminus N_{\varepsilon/2}} \dot{\chi}(\eta_{\lambda}(\varphi) |\nabla \varphi|) \left\langle Y_{\lambda} \left| \nabla \left(\eta_{\lambda}(\varphi) |\nabla \varphi|^{\beta(p-1)} \right) \right\rangle d\mu,$$

where this time the tubular neighbourhoods of Crit are defined, for every $\delta > 0$, as $N_{\delta} = \{ \eta_{\lambda}(\varphi) | \nabla \varphi|^{\beta(p-1)} \leq \delta \}$. Since, as observed in Lemma 2.21, the divergence of Y_{λ} is nonnegative on $\{ s \leq \varphi \leq S \} \setminus N_{\epsilon/2}$, where clearly $|\nabla \varphi| \neq 0$, in light of (2.3.27) the inequality (2.3.28) is proved if we can show that

$$\left\langle Y_{\lambda} \, \middle| \, \nabla \Big(\eta_{\lambda}(\varphi) |\nabla \varphi|^{\beta(p-1)} \Big) \right\rangle \, \geq \, 0$$

on $N_{3\varepsilon/2} \setminus N_{\varepsilon/2}$. A direct – though not immediately evident – computation, combined with the definition (2.3.26) of Y_{λ} yields

$$\begin{split} \left\langle Y_{\lambda} \left| \left. \nabla \left(\frac{|\nabla \varphi|}{\mathrm{e}^{\frac{(n-p)}{(n-2)(p-1)} \varphi}} \right) \right\rangle \right. &= \left. \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} \left| \nabla \varphi \right|^{p-2} \right. \left| \nabla^{\top} |\nabla \varphi|^{\beta(p-1)} \right|^{2} \\ &+ \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} \left| \nabla \varphi \right|^{p-2} (p-1) \left| \nabla^{\perp} |\nabla \varphi|^{\beta(p-1)} \right|^{2} \\ &- 2 \left(\frac{n-p}{n-2} \right) \eta_{\lambda}(\varphi) \left. \left| \nabla \varphi \right|^{(\beta+1)(p-1)} \left\langle \nabla |\nabla \varphi|^{\beta(p-1)} \right| \left| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle \right. \\ &+ \left(\frac{n-p}{n-2} \right)^{2} \eta_{\lambda}^{2}(\varphi) \left. \frac{\mathrm{e}^{\frac{(n-p)}{(n-2)(p-1)} \varphi} |\nabla \varphi|^{2\beta(p-1)+p}}{p-1} \\ &= \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} \left| \nabla \varphi \right|^{p-2} \left| \nabla^{\top} |\nabla \varphi|^{\beta(p-1)} \right|^{2} \\ &+ \left[\left(\frac{n-p}{n-2} \right) \eta_{\lambda}(\varphi) \left(\frac{\mathrm{e}^{\frac{(n-p)}{(n-2)(p-1)} \varphi} |\nabla \varphi|^{2\beta(p-1)+p}}{(p-1)} \right)^{1/2} \\ &- \left. \left\langle \nabla |\nabla \varphi|^{\beta(p-1)} \right| \frac{\nabla \varphi}{|\nabla \varphi|} \right\rangle \left((p-1) \, \mathrm{e}^{-\frac{(n-p)}{(n-2)(p-1)} \varphi} |\nabla \varphi|^{p-2} \right)^{1/2} \right]^{2} \end{split}$$

This completes the proof of the first part of the statement, since the rightmost hand side is manifestly nonnegative. It remains to show that $\Phi_{\beta}^{p}(S) \leq \Phi_{\beta}^{p}(0)$ for any regular S > 0. Applying the just proved inequality (2.3.28) with s = 0, we get

$$\frac{(n-p)}{(n-2)(p-1)} \left(\Phi_{\beta}^{p}(S) - \Phi_{\beta}^{p}(0) \right) \leq -(1-\lambda) \left(\Phi_{\beta}^{p} \right)'(0) + \left(\frac{e^{\frac{(n-p)}{(n-2)(p-1)}S} - \lambda}{e^{\frac{(n-p)}{(n-2)(p-1)}S}} \right) (\Phi_{\beta}^{p})'(S) \\
\leq -(1-\lambda) \left(\Phi_{\beta}^{p} \right)'(0) ,$$

where in the last inequality we used $(\Phi_{\beta}^{p})'(S)$ from Theorem 2.20. Letting $\lambda \to 1^{-}$, we get the claimed inequality.

2.3.3 Full monotonicity in absence of critical points

It directly follows Theorem 2.20 that if $|\nabla \varphi|_g > 0$ on the whole $\mathbb{R}^n \setminus \Omega$, then Φ_β^p is fully monotone nonincreasing. In particular, in this case, we can infer the full monotonicity of U_β^p . Since, as already recalled, standard elliptic regularity theory ensures in this case that u is analytic, from the vanishing of $(U_\beta^p)'$ at some $0 < t_0 \le 1$ we can deduce that Ω is a ball. We just write the full statement for U_β^p , leaving the details of its conformal version to the interested reader.

Theorem 2.23 (Full Monotonicity of U_{β}^{p} in absence of critical points). Let $\Omega \subset \mathbb{R}^{n}$ be an open bounded set with smooth boundary, and let u be the solution to (2.2.2). Assume that |Du| > 0. Then, the function $U_{\beta}^{p} : (0,1] \to \mathbb{R}$ defined by

$$U_{\beta}^{p}(t) = t^{-\beta(p-1)\frac{(n-1)}{n-p}} \int_{\{u=t\}} |Du|^{(\beta+1)(p-1)}$$

is differentiable and satisfies

$$\frac{\mathrm{d}U_{\beta}^{p}}{\mathrm{d}t}(t) = \beta t^{-\beta(p-1)\left(\frac{n-1}{n-p}\right)} \int_{\{u=t\}} |\mathrm{D}u|^{(\beta+1)(p-1)-1} \left[H - \frac{(n-1)(p-1)}{(n-p)} |\mathrm{D}\log u| \right] \mathrm{d}\sigma$$

$$= \frac{\beta}{t^{2}} \int_{\{u \le t\}} u^{2-\beta(p-1)\left(\frac{n-1}{n-p}\right)} |\mathrm{D}u|^{(\beta+1)(p-1)-3} \left\{ |\mathrm{D}u|^{2} \left| h - \frac{H}{n-1} g^{\top} \right|^{2} + \left[\beta(p-1) + p - 2\right] |\mathrm{D}^{\top}|\mathrm{D}u|^{2} + (p-1)^{2} \left[\beta - \frac{(n-p)}{(n-1)(p-1)} \right] |\mathrm{D}u|^{2} \left[H - \left[\frac{(n-1)(p-1)}{n-p}\right] |\mathrm{D}\log u| \right]^{2} \right\} \mathrm{d}\mu.$$
(2.3.29)

In particular, the derivative of U_{β}^{p} is always nonnegative. Moreover $(U_{\beta}^{p})'(t_{0})=0$ for some $t_{0}\in(0,1]$ if and only if Ω is a ball and u is rotationally symmetric.

Proof. From the assumptions we have Crit $u = \emptyset$, and thus Theorem 2.2.48 and a direct perusal of the proof of Theorem, 2.20, precisely, by (2.3.24), we get

$$\begin{split} (\Phi_{\beta}^{p})'(s) &= -\beta \int_{\{\varphi = s\}} |\nabla \varphi|_{g}^{(\beta+1)(p-1)-1} H_{g} d\sigma_{g} \\ &= -\beta e^{\frac{n-p}{(n-2)(p-1)}s} \int_{\{\varphi \geq s\}} e^{-\frac{n-p}{(n-2)(p-1)}\varphi} |\nabla \varphi|_{g}^{(\beta+1)(p-1)-3} \left\{ |\nabla \varphi|_{g}^{2} \left| h_{g} - \frac{H_{g}}{n-1} g^{\top} \right|_{g^{\top}}^{2} + \left[\beta(p-1) + p - 2 \right] |\nabla^{\top}|\nabla \varphi|_{g} \right|_{g}^{2} \\ &+ (p-1)^{2} \left[\beta - \frac{(n-p)}{(n-1)(p-1)} \right] |\nabla^{\perp}|\nabla \varphi|_{g} \right|_{g}^{2} \right\} d\mu_{g}. \end{split}$$

$$(2.3.30)$$

The main identities to be used in order to recover (2.3.29) from (2.3.30) are the following.

$$|\nabla \varphi|_{g}^{2} \left| \mathbf{h}_{g} - \frac{\mathbf{H}_{g}}{n-1} g^{\top} \right|_{g^{\top}}^{2} = \left[\frac{(p-1)(n-2)}{(n-p)} \right]^{2} u^{-2\frac{n+p-2}{n-p}} |\mathbf{D}u|^{2} \left| \mathbf{h} - \frac{\mathbf{H}}{n-1} g_{\mathbb{R}^{n}}^{\top} \right|_{g_{\mathbb{R}^{n}}^{\top}}^{2},$$

where in the left hand side the second fundamental form and the mean curvature are those of the level sets of φ , in the right hand one they are referred to the corresponding level sets of u. Moreover

$$\left|\nabla^{\top}|\nabla\varphi|_{\mathcal{S}}\right|_{\mathcal{S}}^{2} = \left[\frac{(p-1)(n-2)}{(n-p)}\right]^{2} u^{-2\frac{n+p-2}{n-p}} \left|D^{\top}|Du|_{\mathcal{S}^{\mathbb{R}^{n}}}\right|_{\mathcal{S}^{\mathbb{R}^{n}}}^{2},$$

and

$$\left|\nabla^{\perp}|\nabla\varphi|_{g}\right|_{g}^{2}=\left(\frac{n-2}{n-p}\right)^{2}u^{-2\frac{n+p-2}{n-p}}|\mathrm{D}u|^{2}\left[\mathrm{H}-\frac{(n-1)(p-1)}{(n-p)}\frac{|\mathrm{D}u|}{u}\right]^{2}.$$

If $(U_{\beta}^p)'(t_0) = 0$ for some $t_0 \in (0,1]$, then by (2.3.29) all the level sets $\{u = t\}$ for $t \le t_0$ are spheres. Let R_0 such that $\{u = t_0\} = \partial B(O, R_0)$ for some origin $O \in \mathbb{R}^n$. Let then

$$v(x) = t_0 \left(\frac{R_0}{|x|}\right)^{\frac{n-p}{p-1}},$$

that solves

$$\begin{cases} \Delta_p v = 0 & \text{in} \quad \mathbb{R}^n \setminus \overline{B}_{R_0} \\ v = t_0 & \text{on} \quad \partial B_{R_0} \\ v(x) \to 0 & \text{as} \quad |x| \to \infty. \end{cases}$$

Since u solves the same problem then by the uniqueness of solutions we get u = v in $\mathbb{R}^n \setminus B_{R_0}$, and in turn u is rotationally symmetric in this region. Finally, since v can be extended to $\mathbb{R}^n \setminus \{0\}$, we have that u and this extension of v are both analytic function coinciding on an open subset of M, and thus they must coincide on the whole $\mathbb{R}^n \setminus \Omega$. \square

It is natural to wonder whether there are geometric conditions on Ω ensuring that the p-capacitary potential u does not develop critical points. To the author's knowledge, the only complete result in this direction available in literature is contained in the main theorem of [Lew77], implying that if Ω is strictly convex then $|Du| \neq 0$ on $\mathbb{R}^n \setminus \Omega$. Actually, [Lew77] considers the p-capacitary function in convex rings, that in particular entails the case of the p-harmonic function u_R on $B(O,R) \setminus \overline{\Omega}$ with $\Omega \in B(O,R)$ such that u=1 on $\partial\Omega$ and u=0 on $\partial B(O,R)$, and proves that $|Du_R|>0$. However, it can be checked that the positive lower bound on $|Du_R|$ does not depend on R (see also the more general [BC18, Lemma B.1] for an explicit computation), and then the limit of u_R as $R \to +\infty$, that produces the p-capacitary function u of Ω , still satisfies |Du|>0. We refer the reader to the proof of Theorem B.1 in Appendix B for details on the local \mathscr{C}^1 -convergence of u_R to u on compact sets.

It is worth pointing out that the strict convexity assumption can be relaxed to strict starshapedness in the linear case p=2. This is a well known result, and it can be found in [Eva10, Theorem 1, Section 9.5]. It is based on the harmonicity of the support function $\langle x \mid Du \rangle$ when u is harmonic, and a straightforward maximum principle argument yields the result. Finding a suitable elliptic equation solved by the support function when u is p-harmonic functions does not seem easy, and the validity of this extension to p-capacitary potentials remains to the author's knowledge an open problem. It is worth pointing out, on the other hand, that starshapedness of the level sets of the p-capacitary function of a starshaped set Ω is valid, and it is proved in [Sal05], despite the techniques used do not seem sufficient to infer the strict starshapedness, that would imply the absence of critical points.

2.4 Minkowski Inequalities and applications

In the following Subsections, we re going to easily deduce from the Effective Monotonicity Theorem for U^p_β a L^p -version of the Minkowski inequality, that will lead in the limit as $p\to 1^+$ to the Extended Minkowski inequality, and analyse some consequences of our new generalised version of this classical result.

2.4.1 From the L^p -Minkowski inequality to the Extended Minkowski Inequality

The following is the statement of the L^p -Minkowski inequality, together with the related rigidity statement.

Theorem 2.24 (L^p -Minkowski Inequality). Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Then, for every 1 , the following inequality holds

$$C_p(\Omega)^{\frac{n-p-1}{n-p}} \le \frac{1}{|\mathbb{S}^{n-1}|} \int\limits_{\partial \Omega} \left| \frac{H}{n-1} \right|^p \mathrm{d}\sigma, \qquad (2.4.1)$$

where $C_p(\Omega)$ is the normalised p-capacity of Ω introduced in Definition 2.1. Moreover, equality holds in (2.4.1) if and only if Ω is a ball.

Proof. We let $\beta=1/(p-1)$, and call for simplicity $U_p=U_p^{1/(p-1)}$. The substantial monotonicity of U_p proved in Theorem 2.13 yields $U_p'(1) \geq 0$ and $U_p(0^+) \leq U_p(1)$, respectively. The first inequality, combined with the expression for the derivative of U_p at regular values of u, given in (2.2.18) implies that

$$\int\limits_{\partial\Omega} \left(\frac{p-1}{n-p}\right) |\mathrm{D}\log u|^p \,\mathrm{d}\sigma \,\, \leq \,\, \int\limits_{\partial\Omega} |\mathrm{D}u|^{p-1} \frac{\mathrm{H}}{n-1} \,\mathrm{d}\sigma \,.$$

Applying the Hölder inequality to the above right hand side, with conjugate exponents a = p/(p-1) and b = p, one is left with

$$\int_{\partial\Omega} |Du|^p d\sigma \leq \left(\frac{n-p}{p-1}\right)^p \int_{\partial\Omega} \left|\frac{H}{n-1}\right|^p d\sigma. \tag{2.4.2}$$

Using the inequality $U_p(0^+) \leq U_p(1)$ in combination with (2.2.21) we get

$$\left(\frac{n-p}{p-1}\right)^p |\mathbb{S}^{n-1}| |C_p(\Omega)^{\frac{n-p-1}{n-p}}| = \lim_{\tau \to 0^+} U_p(\tau) | \leq |U_p(1)| = \int_{\partial \Omega} |Du|^p d\sigma,$$

that coupled with (2.4.2) gives the desired

$$C_p(\Omega)^{\frac{n-p-1}{n-p}} \leq \frac{1}{|\mathbb{S}^{n-1}|} \int\limits_{\partial\Omega} \left| \frac{\mathsf{H}}{n-1} \right|^p \mathrm{d}\sigma.$$

Assume now that equality holds in (2.24). Then, equality holds in (2.4.2), and consequently $U_p'(1) = 0$. The isometry of $\partial\Omega$ with a sphere then follows from the rigidity statement in Theorem 2.7.

We are now interested in passing to the limit as $p \to 1^+$ in the L^p -Minkowski Inequality (2.4.1). The main task here is to compute – and characterise geometrically – the limit of the variational p-capacity of a bounded set with smooth boundary. This topic will be discussed in full details and great generality in the next Chapter. The notion of Ω^* was anyway outlined in Chapter 1 Subsection 1.7.3. Importantly, the following approximation of $|\partial \Omega^*|$ holds true for an open bounded set Ω with smooth boundary

$$\lim_{p \to 1^+} C_p(\Omega) = \frac{|\partial \Omega^*|}{|S^{n-1}|}.$$
(2.4.3)

We refer the reader to the next Chapter, and in particular to Theorem 3.1, for a rigorous proof of (2.4.3) and of various other aspects of Ω^* .

We can now prove the Extended Minkowski Inequality, together with its Corollary 2.26.

Theorem 2.25 (Extended Minkowski Inequality). *If* $\Omega \subset \mathbb{R}^n$ *is a bounded open subset with smooth boundary, then*

$$\left(\frac{\left|\partial\Omega^*\right|}{\left|\mathbb{S}^{n-1}\right|}\right)^{\frac{n-2}{n-1}} \le \frac{1}{\left|\mathbb{S}^{n-1}\right|} \int\limits_{\partial\Omega} \left|\frac{\mathbf{H}}{n-1}\right| \, \mathrm{d}\sigma, \tag{2.4.4}$$

where Ω^* is the strictly outward minimising hull of Ω defined as in (3.2.17). Moreover, the dimensional constants appearing here are optimal, in the sense that

$$\min \left\{ \left| \partial \Omega^* \right|^{-\frac{n-2}{n-1}} \int |\mathsf{H}| \, \mathrm{d}\sigma \, \middle| \, \Omega \Subset \mathbb{R}^n \,, \, \text{with } \partial \Omega \, \text{smooth} \right\} \, = \, (n-1) \, |\mathbb{S}^{n-1}|^{\frac{1}{n-1}} \,,$$

and the minimum is achieved on spheres.

Proof. It suffices to pass to the limit as $p \to 1^+$ in (2.4.1), that is

$$C_p(\Omega)^{\frac{n-p-1}{n-p}} \le \frac{1}{|\mathbb{S}^{n-1}|} \int\limits_{\partial \Omega} \left| \frac{\mathsf{H}}{n-1} \right|^p \mathrm{d}\sigma. \tag{2.4.5}$$

Indeed, recalling the relation between *p*-capacity and normalised *p*-capacity given in Definition 2.1, the discussion above, or precisely Theorem 3.1, shows that the left hand side of the above inequality satisfies

$$\lim_{p\to 1^+} \mathsf{C}_p(\Omega)^{\frac{n-p-1}{n-p}} = \left(\frac{|\partial\Omega^*|}{|\mathsf{S}^{n-1}|}\right)^{\frac{n-2}{n-1}},$$

while the right-hand side of (2.4.5) is immediately seen to converge to the right hand side of (2.4.4). Spheres show the optimality of the estimate since their mean curvature is given by (n-1)/R, where R is the radius of the ball they enclose.

The Extended Minkowski inequality readily yields the known version for outward minimising sets. We are spending some lines to prove the rigidity statement when the set considered is actually strictly outward minimising and its boundary is strictly mean-convex. For this last observation we are going to appeal to the Inverse Mean Curvature Flow, since we still do not know how to use our (effective) monotonicity formulas to recover the rigidity statement. The difficulty obviously arises by the need of passing to $p \to 1^+$, that makes impossible to use the effective monotonicity result Theorem 2.7.

Corollary 2.26 (Minkowski Inequality for Outward Minimising Sets). *If* $\Omega \subset \mathbb{R}^n$ *is a bounded outward minimising open subset with smooth boundary, then*

$$\left(\frac{\left|\partial\Omega\right|}{\left|\mathbb{S}^{n-1}\right|}\right)^{\frac{n-2}{n-1}} \le \frac{1}{\left|\mathbb{S}^{n-1}\right|} \int_{\partial\Omega} \frac{H}{n-1} \, d\sigma. \tag{2.4.6}$$

Moreover, the dimensional constants appearing here are optimal, in the sense that

$$\min \left\{ \left| \partial \Omega \right|^{-\frac{n-2}{n-1}} \int\limits_{\partial \Omega} H \ d\sigma \ \middle| \ \Omega \Subset \mathbb{R}^n outward \ minimising \ , \ with \ \partial \Omega \ smooth \right\} \ = \ (n-1) \ |\mathbb{S}^{n-1}|^{\frac{1}{n-1}} \ ,$$

and the minimum is achieved on spheres. Viceversa, if the equality holds in (2.4.6) for some bounded strictly outward minimising open subset with smooth and strictly mean convex boundary, then Ω is isometric to a round ball.

Proof. Inequality (2.4.6) immediately follows from the fact that outward minimising sets with smooth boundary satisfy $|\partial \Omega^*| = |\partial \Omega|$ and the mean curvature of their boundaries is nonnegative (see Remark 3.16).

We are left to consider the equality case in (2.4.6) for some strictly outward minimising sets with smooth and strictly mean-convex boundary. To this aim, let $\{\partial\Omega_t\}_{t\in[0,T)}$ be the evolution of $\partial\Omega$ under smooth IMCF, up to some T>0. By [HI01, Lemma 2.4], the sets E_t evolving by weak IMCF starting at $\partial\Omega$, whose definition is recalled in Subsection 3.3.2 in the next chapter, coincide with Ω_t in some time interval $[0,T^*)$, with T^* possibly smaller than T. In particular, by [HI01, Lemma 1.4], Ω_t is strictly outward minimising and strictly mean-convex for every $t\in[0,T^*)$, and then (2.4.6) holds for every $\partial\Omega_t$ with $t\in[0,T^*)$. We can then define, for $t\in[0,T^*)$, the monotonic quantity already discussed in the Introduction

$$\mathscr{Q}(t) = |\partial \Omega_t|^{-\frac{n-2}{n-1}} \int_{\partial \Omega_t} H d\sigma.$$

Observe that the inequality (2.4.6) is equivalent to $\mathcal{Q}(0) \geq (n-1)|\mathbb{S}^{n-1}|^{1/(n-1)}$, and assuming equality in (2.4.6) is equivalent to $\mathcal{Q}(0) = |\mathbb{S}^{n-1}|^{1/(n-1)}$. By the smoothness of the flow, the function $\mathcal{Q}(t)$ is differentiable for $t \in [0,T)$, and then a straightforward computation involving the standard evolution equations provided e.g. in [HP99, Theorem 3.2] show that

$$\mathscr{Q}'(0) = -|\partial\Omega|^{-\frac{n-2}{n-1}} \int_{\partial\Omega} \frac{|\mathring{\mathbf{h}}|^2}{H} \, \mathrm{d}\sigma \le 0.$$
 (2.4.7)

However, since we assumed $\mathcal{Q}(0) = 0$, $\mathcal{Q}'(0) < 0$ would imply $\mathcal{Q}(t) < 0$ for some $t \in (0, T^*)$ that is equivalent to falsify (2.4.6) for some outward minimising Ω_t with strictly mean-convex boundary. Then $\mathcal{Q}'(0) = 0$, and by formula (2.4.7) this means that $\partial\Omega$ is totally umbilical, and thus a sphere.

We remark once again that (2.4.6) in particular holds for strictly starshaped sets with mean-convex boundary, see Proposition 3.25.

It is not quite clear, at least at first glance, whether one can in fact deduce the Extended Minkowski Inequality (2.4.4) from its corollary for outward minimising sets. We are able to show that this is the case if $n \le 7$.

Corollary 2.26 implies Theorem 2.25 if $n \le 7$. Let $\Omega \subset \mathbb{R}^n$ be a bounded set with smooth boundary, and $n \le 7$. Then, as recalled in Subsection 1.7.3 in the previous chapter, Ω^* is of class $\mathscr{C}^{1,1}$. We can then apply, as there, Lemma 1.66 to Ω^* to obtain a sequence of strictly outward minimising sets E_{ε} with smooth and strictly mean-convex boundary approximating Ω^* in $\mathscr{C}^1 \cap W^{1,1}$ -topology as $\varepsilon \to 0^+$. In particular, for these sets, (2.4.6) is in force, and then we get, letting $\varepsilon \to 0^+$

$$\left(\frac{\left|\partial\Omega^*\right|}{\left|S^{n-1}\right|}\right)^{\frac{n-2}{n-1}} \leq \frac{1}{\left|S^{n-1}\right|} \int\limits_{\partial\Omega^*} \frac{H}{n-1} \, \mathrm{d}\sigma \leq \int\limits_{\partial\Omega} \frac{H}{n-1} \, \mathrm{d}\sigma,$$

where the last inequality is due to the fact that any region of $\partial\Omega$ with negative mean curvature is replaced by regions of $\partial\Omega^*$ that are minimal, as immediately deduced from Theorem 3.14, anticipated by the discussion in Subsection 1.7.3.

The above argument evidently breaks down when $n \geq 8$, due to area minimising hypersurface regularity issues. Indeed, in this case, nonempty singular sets can appear in $\partial\Omega^*$, preventing this boundary to be \mathscr{C}^1 and thus making the application of Lemma 1.66 no more possible. We are not aware of other approximation results, strong enough

to let the L^1 -norm of the mean-curvature of the approximator converge, holding true for boundaries with (small) singular sets.

2.4.2 Applications of the Extended Minkowski Inequality

The two applications we give consist in a Volumetric Minkowski Inequality, known in literature also as Higher Order Isoperimetric Inequality, and in the De Lellis-Müller nearly umbilical estimates with sharp constant for outward minimising sets.

Volumetric Minkowski Inequality

An application of the Euclidean Isoperimetric Inequality (see e.g. [Mag12, Theorem 14.1]) immediately yields the following.

Theorem 2.27 (Volumetric Minkowski Inequality). *Let* $\Omega \subset \mathbb{R}^n$ *be an open bounded set with smooth boundary. Then*

$$\left(\frac{|\Omega|}{|\mathbb{B}^n|}\right)^{\frac{n-2}{n}} \le \frac{1}{|\mathbb{S}^{n-1}|} \int_{\partial \Omega} \left|\frac{H}{n-1}\right| d\sigma. \tag{2.4.8}$$

Moreover, equality holds in (2.4.8) if and only if Ω is a ball.

Proof. Applying the Euclidean Isoperimetric Inequality to the left hand side of (2.4.4), we get

$$\left(\frac{|\Omega|}{|\mathbb{B}^n|}\right)^{\frac{n-2}{n}} \leq \left(\frac{|\Omega^*|}{|\mathbb{B}^n|}\right)^{\frac{n-2}{n}} \leq \left(\frac{|\partial\Omega^*|}{|\mathbb{S}^{n-1}|}\right)^{\frac{n-2}{n-1}} \leq \frac{1}{|\mathbb{S}^{n-1}|} \int_{\partial\Omega} \left|\frac{H}{n-1}\right| d\sigma, \qquad (2.4.9)$$

where the first inequality is just due to $\Omega \subseteq \Omega^*$. If equality holds in (2.4.8), then by (2.4.9) Ω^* satisfies equality in the Euclidean Isoperimetric Inequality, and then Ω^* is a ball up to negligible sets. Arguing as in the proof of the Isoperimetric Inequality for complete noncompact 3-manifolds with nonnegative Ricci curvature Theorem 1.7.2, we conclude that Ω is thus a ball.

To our knowledge, the above inequality was previously known to hold for domains with a strictly mean-convex boundary of positive scalar curvature (for short $\partial\Omega\in\Gamma_2^+$). On this regard, we refer the reader to the paper [CW13] and the subsequent [Qiu15], where the inequality was proved with methods based on Optimal Transport.

Nearly umbilical estimates with optimal constant for outward minimising sets.

This subsection is devoted to the proof of an optimal version of the celebrated De Lellis-Müller nearly umbilical estimates for outward minimising domains.

Theorem 2.28 (Optimal Nearly Umbilical Estimate). *If* $\Omega \subset \mathbb{R}^3$ *is a bounded outward minimising open domain with smooth boundary, then*

$$\int_{\partial\Omega} \left| \mathbf{h} - \frac{\overline{\mathbf{H}}}{2} g_{\partial\Omega} \right|^2 d\sigma \leq 2 \int_{\partial\Omega} \left| \mathring{\mathbf{h}} \right|^2 d\sigma, \qquad (2.4.10)$$

where $g_{\partial\Omega}$ is the metric induced on $\partial\Omega$ by the Euclidean metric of \mathbb{R}^3 , and

$$\overline{\mathrm{H}} = \int_{\partial\Omega} \mathrm{H} \,\mathrm{d}\sigma, \qquad \qquad \mathring{\mathrm{h}} = \mathrm{h} - \frac{\mathrm{H}}{2} g_{\partial\Omega}.$$

Moreover, the equality is achieved in (2.4.10) by some strictly mean-convex and strictly outward minimising domain Ω if and only if Ω is isometric to a round ball.

A first main tool we are going to use in order to deduce Theorem 2.28 from Theorem 2.25 is the classical Gauss' equation for surfaces in \mathbb{R}^3 , yielding

$$R_{\partial\Omega} = H^2 - |h|^2,$$
 (2.4.11)

where $R_{\partial\Omega}$ is the scalar curvature of $\partial\Omega$ computed with respect to the metric $g_{\partial\Omega}$ induced on it by the Euclidean metric of \mathbb{R}^3 . A second main tool we need to recall is the famous Gauss-Bonnet formula, stating that

$$\int_{\partial\Omega} R_{\partial\Omega} d\sigma = 4\pi \chi(\partial\Omega), \qquad (2.4.12)$$

for any domain with smooth boundary Ω , where $\chi(\partial\Omega)$ is the Euler characteristic of the surface $\partial\Omega$. The need of a connected boundary in order to apply (2.4.12) is the reason behind the assumption of Ω to be a domain in Theorem 2.28.

We are finally going to show how the Minkowski inequality (2.4.6) for outward minimising sets combined with these basic identities in differential geometry give the optimal nearly umbilical estimate (2.4.10).

Proof of Theorem 2.28. Expanding the squares, it is straightforwardly seen that (2.4.10) is equivalent to

$$\int\limits_{\partial\Omega} \left(|H|^2 - |h|^2\right) d\sigma \leq \frac{\overline{H}^2}{2} |\partial\Omega|.$$

Invoking Gauss' equation (2.4.11) and Gauss-Bonnet formula (2.4.12), we then obtain that (2.4.10) is equivalent to

$$8\pi\chi(\partial\Omega) = 2\int\limits_{\partial\Omega} R_{\partial\Omega} d\sigma = 2\int\limits_{\partial\Omega} \left(|H|^2 - |h|^2\right) d\sigma \leq \overline{H}^2 |\partial\Omega|,$$

that is, to

$$\sqrt{2\pi\chi(\partial\Omega)|\partial\Omega|} \le \int_{\partial\Omega} \frac{H}{2} d\sigma. \tag{2.4.13}$$

Since obviously $\chi(\partial\Omega) \le 2$, the inequality (2.4.13) follows from the Minkowski inequality (2.4.6).

Assume now that equality holds for some outward minimising set Ω with smooth and strictly mean-convex boundary. Let $\{\partial\Omega_t\}_{t\in[0,T)}$ be evolving by IMCF with initial datum $\partial\Omega$. By [HI01, Lemma 2.4], the sets E_t evolving by weak IMCF starting at $\partial\Omega$ coincide with Ω_t for $t\in[0,T^*)$, for some T^* possibly smaller than T. In particular, Ω_t is outward minimising and strictly mean-convex for any $t\in[0,T^*)$, for some $T^*>0$, and then (2.4.10) holds for $\partial\Omega_t$ for any $t\in[0,T^*)$. We can then define, for $T\in[0,T^*)$, the quantity

$$\mathscr{P}(t) = \int_{\partial \Omega_t} |\mathring{\mathbf{h}}|^2 d\sigma - \frac{1}{2} \int_{\partial \Omega_t} \left(\mathbf{H} - \frac{1}{|\partial \Omega_t|} \int_{\partial \Omega_t} \mathbf{H} \right)^2 d\sigma,$$

introduced in [Per11, Chapter 3]. Observe that inequality (2.4.10) is equivalent to $\mathcal{P}(0) \ge 0$, and assuming equality in (2.4.10) is equivalent to $\mathcal{P}(0) = 0$. By the smoothness of the

flow, the function $\mathcal{P}(t)$ is differentiable for $t \in [0, T)$, and then [Per11, Lemma 3.4] yields

$$\mathscr{P}'(0) = -\overline{H} \int_{\partial\Omega} \frac{|\mathring{\mathbf{h}}|^2}{H} \, \mathrm{d}\sigma \le 0 \tag{2.4.14}$$

However, since we assumed $\mathcal{P}(0) = 0$, $\mathcal{P}'(0) < 0$ would imply $\mathcal{P}(t) < 0$ for some $t \in (0, T^*)$ that is equivalent to falsify (2.4.10) for some outward minimising Ω_t with strictly mean-convex boundary. Then $\mathcal{P}'(0) = 0$, and by formula (2.4.14) this means that $\partial\Omega$ is totally umbilical, thus a sphere.

Inequality (2.4.13) in the above proof, that we just showed to be equivalent to the nearly umbilical estimate (2.4.10), is actually equivalent to the Minkowski inequality for mean-convex hypersurfaces (2.4.6) if $\chi(\partial\Omega)=2$, that is, if $\partial\Omega$ is diffeomorphic to a sphere. We want to show, with the following easy proposition, that such a diffeomorphism exists each time the right-hand side of (2.4.10) is smaller than 16π .

Proposition 2.29. *Let* $\Omega \subset \mathbb{R}^3$ *be a bounded open subset with smooth and* mean-convex *boundary. If*

$$\int_{\partial\Omega} \left| \mathbf{h} - \frac{\mathbf{H}}{2} g_{\partial\Omega} \right|^2 d\sigma \le 8\pi \tag{2.4.15}$$

then $\partial\Omega$ is diffeomorphic to a sphere. In particular, if (2.4.15) holds, then $\chi(\Omega)=2$, and in particular the Minkowski inequality (2.4.6) for outward minimising domains is equivalent to the optimal nearly umbilical estimate (2.4.10) for outward minimising domains.

Proof. If (2.4.15) holds, we have, by the classical Willmore inequality [Wil68]

$$\int_{\partial \Omega} \left| h - \frac{H}{2} g_{\partial \Omega} \right|^2 d\sigma \le 8\pi \le \frac{1}{2} \int H^2 d\sigma,$$

that implies

$$\int\limits_{\partial\Omega} \left(|\mathbf{H}|^2 - |\mathbf{h}|^2 \right) \mathrm{d}\sigma \ge 0. \tag{2.4.16}$$

Moreover, since equality is attained in the Willmore inequality if and only if $\partial\Omega$ is isometric to a sphere with the round metric, the same rigidity statement holds if equality is attained in (2.4.16). On the other hand, applying the Gauss' equation (2.4.11) and the Gauss-Bonnet formula (2.4.12), (2.4.16) is equivalent to

$$\chi(\partial\Omega) \geq 0.$$

If $\chi(\partial\Omega)=0$ then by the characterisation of the equality case in the Willmore inequality $\partial\Omega$ would be even isometric to a sphere, and this is a contradiction. Then $\chi(\partial\Omega)=2$, as claimed.

2.5 Other consequences of the Effective Monotonicity Theorems

Here, we illustrate some others consequences and applications of the (effective) monotonicity of U^p_β for various values of β and of U^p_∞ . The following two subsections can be compared to Subsections 1.5.1 and Subsection 1.5.2 in the previous chapter.

2.5.1 *p*-capacitary inequalities and the relations with Willmore-type functionals

The following sharp inequality for the *p*-capacity of Ω follows just from $(U_{\beta}^{p})'(1) \geq 0$.

Theorem 2.30. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Then

$$C_p(\Omega) \leq \frac{|\partial \Omega|}{|S^{n-1}|} \left(\int_{\partial \Omega} \left| \frac{H}{n-1} \right|^{(p-1)(1+\beta)} d\sigma \right)^{1/(1+\beta)}. \tag{2.5.1}$$

Moreover, the above inequality holds with equality sign if and only if Ω is a ball.

Proof. By $(U_6^p)'(1) \ge 0$, and more precisely by (2.2.18) for t = 1, we get

$$\int\limits_{\partial\Omega} \left(\frac{p-1}{n-p}\right) |\mathrm{D}\log u|^{(p-1)(\beta+1)} \,\mathrm{d}\sigma \ \leq \int\limits_{\partial\Omega} |\mathrm{D}u|^{(p-1)(\beta+1)-1} \frac{\mathrm{H}}{n-1} \,\mathrm{d}\sigma.$$

Applying the Hölder inequality to the right hand side, with conjugate exponents $a = (\beta + 1)(p - 1)/[(\beta + 1)(p - 1) - 1]$ and $b = (\beta + 1)(p - 1)$, we get

$$\int_{\partial\Omega} |Du|^{(\beta+1)(p-1)} d\sigma \leq \left(\frac{n-p}{p-1}\right)^{(\beta+1)(p-1)} \int_{\partial\Omega} \left|\frac{H}{n-1}\right|^{(\beta+1)(p-1)} d\sigma. \tag{2.5.2}$$

Applying now the Hölder inequality with conjugate exponents $\beta + 1$ and $(\beta + 1)/\beta$ to $\int_{\partial\Omega} |Du|^{p-1} d\sigma$, we get

$$\left(\frac{n-p}{p-1}\right)^{p-1} |\mathbb{S}^{n-1}| C_p(\Omega) = \int_{\partial\Omega} |\mathrm{D}u|^{p-1} \,\mathrm{d}\sigma \le \left(\int_{\partial\Omega} |\mathrm{D}u|^{(\beta+1)(p-1)} \,\mathrm{d}\sigma\right)^{\frac{1}{\beta+1}} |\partial\Omega|^{\frac{\beta}{\beta+1}},$$
(2.5.3)

where the first identity follows from (2.2.3) for t = 1. Coupling (2.5.3) with (2.5.2) we are left with (2.5.1). It is immediately seen that if equality holds in (2.5.1) then $(U_{\beta}^{p})'(1) = 0$, and the rigidity statement follows from that of Theorem 2.7.

We find particularly interesting in the family of inequalities (2.5.1) the one estimating the Willmore-type functional, largely discussed in a general Riemannian context in the last chapter. It is deduced just by choosing $\beta = (n - p)/(p - 1)$.

Corollary 2.31 (*p*-capacitary estimates of the Willmore-type functional). *Let* $\Omega \subset \mathbb{R}^n$ *be an open bounded set with smooth boundary. Then,*

$$C_{p}(\Omega) |S^{n-1}|^{\frac{p-1}{n-1}} \leq \frac{|\partial\Omega|}{|S^{n-1}|}^{\frac{n-p}{n-1}} \left(\int_{\partial\Omega} \left| \frac{H}{n-1} \right|^{n-1} d\sigma \right)^{\frac{p-1}{n-1}}.$$
 (2.5.4)

Moreover equality holds if and only if Ω *is a ball.*

To our knowledge, the above inequality, as well as those in (2.5.1), were known only for strictly mean-convex strictly starshaped domains. It has been provided, together with other inequalities, by J. Xiao in [Xia17, Theorem 3.1] via the Inverse Mean Curvature Flow and relying on Gerhardt's long time existence theory for strictly mean-convex strictly starshaped initial data developed [Ger90].

The Willmore-type inequality through $p \rightarrow n^-$

In his just mentioned paper, Xiao also observes that the classical Euclidean Willmoretype inequality can be deduced by letting $p \to n^-$ in (2.5.4). Indeed, letting $B(x_0, r) \subset \Omega \subset B(x_0, R)$ one deduces from the obvious monotonicity property of the p-capacity with respect to set-inclusion and its explicitly computed value on balls that

$$r^{n-p} \le C_p(\Omega) \le R^{n-p},\tag{2.5.5}$$

yielding

$$C_v(\Omega) \to 1$$

as $p \to n^-$. Plugging this information in (2.5.4), we are left with the classical Willmore-type inequality. This alternative proof is interesting in relation of our monotonicity formulas, since it just uses $(U^p_\beta)'(1) \ge 0$, that is just a local information. On the other hand, it is heavily bound to the very special geometry of \mathbb{R}^n , encoded in the explicit formula for the p-capacity of balls used in (2.5.5).

The Willmore-type inequality from any value of p.

We observe that, algebraically, there was nothing special about the value p=2 for achieving the Willmore-type inequality. Indeed, let as for Corollary 2.31 $\beta=(n-p)/(p-1)$ and observe that from $\lim_{t\to 0^+} U^p_\beta(t) \leq U^p_\beta(1)$ we get

$$\left(\frac{n-p}{p-1}\right)^{n-1}|\mathbb{S}^{n-1}| \le \int_{\partial\Omega} |\mathrm{D}u|^{n-1} \,\mathrm{d}\sigma,$$

that, coupled with $(U_{\beta}^p)'(1) \geq 0$ through Hölder inequality as in (2.5.3) leaves us with the classical Willmore-type inequality. In spite of this, p=2 is anyway the most convenient value to use if one is interested in the Willmore-type inequality, both for the easier algebraic computations and primarily for the wonderful regularity properties of harmonic functions, that we employed successfully in the more general geometric framework treated in the last Chapter.

2.5.2 Spherical symmetry under pinching conditions

Now, we finally give some applications of the substantial monotonicity of U_{∞}^p . They constitute the nonlinear versions of the main results obtained in [BMM19], where a different geometric construction named *spherical ansatz* was employed.

Theorem 2.32. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Then, if the mean curvature of $\partial\Omega$ satisfies

$$-\left(\frac{1}{C_p(\Omega)}\right)^{\frac{1}{n-p}} \le \frac{H}{n-1} \le \left(\frac{1}{C_p(\Omega)}\right)^{\frac{1}{n-p}} \tag{2.5.6}$$

on every point of $\partial\Omega$, then Ω is a ball.

Proof. By means of the global aspect of the substantial monotonicity of U_{∞}^{p} (2.2.19), the computation of its limit in (2.2.22) and the local aspect of the substantial monotonicity

(2.2.20) we deduce

$$\lim_{t\to 0^+} U^p_\infty(t) = \left(\frac{n-p}{p-1}\right) C_p(\Omega)^{-\frac{1}{n-p}} \le \sup_{\partial \Omega} |Du| \le \left(\frac{n-p}{p-1}\right) \sup_{\partial \Omega} \left|\frac{H}{n-1}\right|,$$

with equality achieved in any of the above inequalities if and only if Ω is a ball, by the rigidity statement in Theorem 2.8. In particular, this happens if (2.5.6) is satisfied.

As a consequence of the above argument we get

$$\left(\frac{1}{C_p(\Omega)}\right)^{\frac{1}{n-p}} \le \sup_{\partial\Omega} \left| \frac{H}{n-1} \right|. \tag{2.5.7}$$

Letting $p \to 1^+$, we get, by (2.4.3), the following completely geometric consequence.

Corollary 2.33. Let $\Omega \subset \mathbb{R}^n$ be an open bounded set with smooth boundary. Then, the following inequality holds

$$\left(\frac{|S^{n-1}|}{|\partial\Omega^*|}\right)^{\frac{1}{n-1}} \le \sup_{\partial\Omega} \left|\frac{H}{n-1}\right|. \tag{2.5.8}$$

In particular, it holds

$$\left(\frac{|S^{n-1}|}{|\partial\Omega|}\right)^{\frac{1}{n-1}} \le \sup_{\partial\Omega} \left|\frac{H}{n-1}\right|. \tag{2.5.9}$$

If equality in (2.5.8) is achieved on a strictly outward minimising set with strictly mean-convex boundary, then Ω is a ball.

Proof. Letting $p \to 1^+$ in (2.5.7) and using (2.4.3), we get, coupling it with $|\partial \Omega^*| \le |\partial \Omega|$, we get

$$\left(\frac{|S^{n-1}|}{|\partial\Omega|}\right)^{\frac{1}{n-1}} \le \left(\frac{|S^{n-1}|}{|\partial\Omega^*|}\right)^{\frac{1}{n-1}} \le \sup_{\partial\Omega} \frac{H}{n-1}.$$

We now have to justify the rigidity statement. Actually, in the case of outward minimising sets, it is immediately seen that inequality (2.5.8) also follows directly from the Minkowski Inequality (2.4.6), and equality is achieved only if it is achieved in the Minkowski inequality. In particular, if Ω is *strictly* outward minimising with strictly mean-convex boundary, the claimed rigidity statement follows from the rigidity statement of Theorem 2.26.

We observe that if Ω is not outward minimising, (2.5.8) is strictly stronger than (2.5.9). Moreover, in general (2.5.8) does not seem to be implied by the Extended Minkowski Inequality, if Ω is not outward minimising.

2.5.3 Relations with a classical overdetermined problem

Here, we discuss how the monotonicity of U^p_∞ constitutes a new interpretation of a maximum principle for for $|\mathrm{D}u|/u^{(n-1)/(n-p)}$, that is, up to a multiplicative constant, $|\nabla \varphi|_g$, widely used in literature to establish symmetry results for overdetermined boundary value problems. Such a function, known in literature as P-function, together with related maximum principles, was first conceived by Payne-Philippin in [PP79] for harmonic functions. Its generalisation to p-harmonic functions on rings first, and then on exterior domains was treated respectively in [Sar98], that generalised the results and the techniques of [Phi90] for the harmonic case, and [GS99].

The overdetermined boundary value problem considered is the following

$$\begin{cases}
\Delta_{p}u = 0 & \text{in } \mathbb{R}^{n} \setminus \overline{\Omega}, \\
u = 1 & \text{on } \partial\Omega, \\
u(x) \to 0 & \text{as } |x| \to \infty \\
|Du| = c & \text{on } \partial\Omega,
\end{cases}$$
(2.5.10)

for some positive constant c>0. It is well known that the above problem admits a solution if and only if Ω is a ball. This was first provided by Reichel in [Rei96] for bounded sets Ω with \mathscr{C}^2 -boundary using the moving planes method, and later showed by Garofalo-Sartori in [GS99] through the P-function approach for starshaped sets with no a priori smoothness assumptions. Reichel's result was recently recovered in [Pog18] building on Garofalo-Sartori maximum principle for the P-function but concluding the proof with a very interesting integral formula involving the torsion function of Ω provided in [MP19]. We are coming back to this point in a while.

Our Substantial Monotonicity Theorem for U^p_∞ actually yields a new interpretation, with a new proof, of the main maximum principle for the *P*-function. Indeed, (2.2.20) yields for the *p*-capacitary potential u of Ω

$$H(x) \ge \frac{(n-1)(p-1)}{(n-p)}|Du|(x),$$
 (2.5.11)

where $x \in \partial\Omega$ is the point where |Du| achieves its maximum on $\partial\Omega$, and actually where the P-function $|Du|/u^{(n-1)/(n-p)}$ achieves its maximum on $\mathbb{R}^n \setminus \Omega$. If u is also assumed to satisfy the additional condition of |Du| = c on $\partial\Omega$, then in particular (2.5.11) holds for any $x \in \partial\Omega$. In particular, since by an easy computation (see [Pog18, Lemma 2.4])

$$c = \frac{(n-p)}{n(p-1)} \frac{|\partial \Omega|}{|\Omega|},$$

we deduce that

$$H \ge \frac{(n-1)}{n} \frac{|\partial\Omega|}{|\Omega|} \tag{2.5.12}$$

on $\partial\Omega$. The above estimate was the main outcome of the maximum principle for the P-function worked out in [GS99]. Moreover, in light of the rigidity statement for U^p_∞ in Theorem 2.8, equality holds in (2.5.12) at some point of $\partial\Omega$ if and only if Ω is a ball. The symmetry of Ω under the existence of a solution to (2.5.10) is then established if we can prove that equality holds somewhere on $\partial\Omega$. This was accomplished in [GS99] using integral identities valid under the starhapedness of Ω . In [Pog18], this was obtained without the starshapedness assumption by means of the sharp inequality, valid for any bounded Ω with smooth boundary,

$$\int_{\partial\Omega} \left(H - \frac{(n-1)}{n} \frac{|\partial\Omega|}{|\Omega|} \right) |Dv|^2 d\sigma \le 0, \qquad (2.5.13)$$

where v is the torsional function of Ω , that is solving

$$\begin{cases} \Delta v = n & \text{in } \Omega, \\ v = 0 & \text{on } \partial\Omega. \end{cases}$$

The inequality (2.5.13) is actually a corollary of the remarkable integral identity [MP19, Theorem 2.2]. In particular, the combination of (2.5.12) and (2.5.13) implies that equality

is achieved in (2.5.12), and we can thus conclude. We find this interplay between the exterior problem for the p-capacitary potential and the interior problem for the torsion quite intriguing, and somehow reminiscent of the proof of the Isoperimetric Inequality worked out in the previous Chapter, where an exterior boundary value problem was exploited in combination with a shrinking flow.

Chapter 3

On strictly outward minimising hulls and p-capacity in Riemannian manifolds

3.1 Main result and structure of the chapter

The main goal of this chapter is to prove the following result, already vastly discussed in the Introduction.

Theorem 3.1. Let (M, g) be a complete noncompact Riemannian manifold satisfying at least one of the following two conditions.

(i) (M,g) satisfies an Euclidean-like Isoperimetric Inequality, namely, there exists $C_{\rm iso}>0$ such that

$$\frac{|\partial\Omega|^n}{|\Omega|^{n-1}} \ge C_{\text{iso}} \tag{3.1.1}$$

for any bounded set Ω with smooth boundary.

(ii) (M, g) has nonnegative Ricci curvature and the superlinear uniform volume growth condition holds

$$C_{\text{vol}}^{-1} r^b \le |B(O, r)| \le C_{\text{vol}} r^b$$
 (3.1.2)

for some b > 1 and $C_{vol} > 0$, for any $r \ge R$ for some R > 0.

Let $\Omega \subset M$ be an open bounded set with finite perimeter. Then, its strictly outward minimising hull Ω^* is an open bounded maximal volume solution to the least area problem with obstacle Ω . Moreover, if Ω has smooth boundary, then

$$\lim_{p \to 1^+} \operatorname{Cap}_p(\Omega) = |\partial \Omega^*|. \tag{3.1.3}$$

It is well known that the validity of the Euclidean-like Isoperimetric Inequality (3.1.1) is equivalent to that of the L^1 -Sobolev Inequality, that is, (3.1.1) is equivalent to the validity of

$$\left(\int_{M} f^{n/(n-1)} d\mu\right)^{(n-1)/n} \leq C_{\text{Sob}} \int_{M} |\nabla f| d\mu$$

for any $f \in C_0^{\infty}(M)$ and some positive constant C_{Sob} . We briefly point out that condition (3.1.2) can actually be weakened or replaced. We are discussing it in Remark 3.28, after having discussed the difficulties in removing it. Still, we preferred the above presented form for the sake of geometric clarity.

This chapter is structured as follows. Section 3.2, after proving basic properties of (strictly) outward minimising sets, is mostly committed to show the equivalence, in a

general complete noncompact Riemannian manifold, among the existence of a maximal volume solution to the least area problem with obstacle, the property of Ω^* to be such a solution and the existence of an exhausting sequence of strictly outward minimising sets. This is summarised in Theorem 3.13. Examples of Riemannian manifolds where these equivalent properties are not satisfied are worked out in Examples 3.8 and 3.9. The last part of the section precisely recalls the regularity properties of Ω^* , that were important also in the previous chapters. Here, these properties are important in order to apply an exterior approximation result due to [Sch15], see Theorem 3.17 below. In Section 3.3, we prove the first part of Theorem 3.1, that is the well-posedness of the strictly outward minimising hull Ω^* under assumptions (i) or (ii). When dealing with (ii) in Subsection 3.3.2, we recall the notion of Weak Inverse Mean Curvature Flow, yielding, in relation with that some insights including a proof of the outward minimising property of starshaped sets with smooth mean-convex boundary, not necessarily strictly mean-convex. Finally, in Section 3.4 we infer (3.1.3) under the assumptions of Theorem 3.1.

3.2 The strictly outward minising hull of a set with finite perimeter

3.2.1 Preliminaries

The main sources are Maggi's book [Mag12]. Miranda's [Mir64] and Caraballo's comprehensive paper [Car11]. Clearly, although these references deal only with the case $M = \mathbb{R}^n$, the notions and results that we recall in this Subsection are adapted with no effort to a Riemannian ambient (M, g).

Let $A \subseteq M$ be an open set. Then, we let P(E, A) denote the De Giorgi's *relative* perimeter of Lebesgue-measurable set E in A. It is defined as

$$P(E,A) = \sup \left\{ \int_E \operatorname{div} T \, \mathrm{d}\mu \mid T \in \Gamma_c(A), \sup_A |T| \le 1 \right\},$$

where by $\Gamma_c(A)$ we denote the class of vector fields with compact support in A.

When A = M, we just talk about the perimeter of E and we denote it by P(E). Given a set E with finite perimeter it is well known [Mag12, Proposition 12.1] the existence of a vector-valued Radon measure μ_E satisfying (and actually defined by)

$$\int_{E} \operatorname{div} T = \int_{M} \langle T | \, \mathrm{d}\mu_{E} \rangle$$

for any $T \in \Gamma_c(M)$. Denoting by $|\mu_E|$ the total variation measure of μ_E , one actually has $P(E,A) = |\mu_E|(A)$ for any open set $A \subseteq \mathbb{R}^n$. With the notion of *perimeter measure* μ_E at hand, De Giorgi's *reduced boundary* $\partial^* E$ of a set with finite perimeter E can be defined as

$$\partial^* E = \left\{ x \in \operatorname{supp} \mu_E \quad \Big| \quad \lim_{r \to 0^+} \frac{\mu_E(B(x,r))}{|\mu_E|(B(x,r))} \text{ exists and belong to } \mathbb{S}^{n-1} \right\}.$$

In fact, one has (see [Mag12, Remark 15.3]) supp $\mu_E = \overline{\partial^* E}$.

We define now the *measure theoretic interior* of a measurable set E as the points of density one for E, namely

$$Int(E) = \left\{ x \in M \mid \lim_{r \to 0^+} \frac{|E \cap B(x,r)|}{|B(x,r)|} = 1 \right\}.$$
 (3.2.1)

It follows from Lebesgue Differentiation Theorem (see [Mag12, Theorem 5.16] that

$$|E\Delta \operatorname{Int}(E)| = 0. \tag{3.2.2}$$

Importantly, a set with finite perimeter *E* satisfies,

$$\partial \operatorname{Int}(E) = \overline{\partial^* E},$$
 (3.2.3)

that is, the topological boundary of the measure theoretic interior of a set with finite perimeter coincides with the closure of its reduced boundary. We address the reader to [Car11, Theorem 10] for a proof of this nice property.

3.2.2 Basic properties of outward minimising sets

We recall, according to [HI01], the notion of *outward* and *strictly outward minimising sets*.

Definition 3.2 (Outward minimising and strictly outward minimising sets). Let (M,g) be a complete Riemannian manifold. Let $E \subset M$ be a bounded set with finite perimeter. We say that E is outward minimising if for any $F \subset M$ with $E \subseteq F$ we have $P(E) \leq P(F)$. We say that E is strictly outward minimising if it is outward minimising and any time P(E) = P(F) for some $F \subset M$ with $E \subseteq F$ we have $|F \setminus E| = 0$.

We observe at once that the notions of outward minimising and strictly outward minimising are stable under zero-measure modification. We give a proof of this basic yet fundamental fact.

Lemma 3.3. Let (M,g) be a complete Riemannian manifold. Let $N \subset M$ be a null set, that is |N| = 0. Then, a bounded set $E \subset M$ with finite perimeter is (strictly) outward minimising if and only if $E \cup N$ is (strictly) outward minimising.

Proof. We first show that if E is (strictly) outward minimising then $E \cup N$ is strictly outward minimising. Let F be a set with finite perimeter such that $E \cup N \subseteq F$. In particular, $E \subseteq F$. Then one has, by the invariance of the perimeter under measure zero modifications and since E is outward minimising

$$P(E \cup N) = P(E) \le P(F). \tag{3.2.4}$$

that is, $E \cup N$ is outward minimising. Assume now that E is strictly outward minimising, and assume $P(F) = P(E \cup N)$. Then equalities hold in (3.2.4), and thus $|F \setminus E| = 0$, that obviously implies $|F \setminus (E \cup N)| = 0$, that is, $E \cup N$ is strictly outward minimising.

Assume now that $E \cup N$ is outward minimising, and let F be a set with finite perimeter containing E. Then, since $E \cup N$ is contained in $F \cup N$ we have

$$P(E) = P(E \cup N) \le P(F \cup N) = P(F),$$
 (3.2.5)

that is, E is outward minimising. Assume now that $E \cup N$ is strictly outward minimising, and assume that P(F) = P(E). Then, (3.2.5) and $E \cup N$ being strictly outward minimising imply that $|(F \cup N) \setminus (E \cup N)| = 0$, that immediately implies $|F \setminus E| = 0$.

It is well known that locally area minimising sets satisfy upper and lower density estimates. It is then not surprising to realise that outward minimising sets, that can be thought as one sided minimisers, satisfy upper density estimates. We include a proof, inspired by that of [Mag12, Theorem 16.14].

Lemma 3.4. Let (M,g) be a complete Riemannian manifold, and let $E \subset M$ be an outward minimising set. Then, there exists $r_0 > 0$ such that for any $0 < r < r_0$ and $x \in \overline{\partial^* E}$ there holds

$$\frac{|E \cap B(x,r)|}{|B(x,r)|} \le C$$

for some constant 0 < C < 1 independent of x

Proof. Consider $E^c = M \setminus E$. Then, for any $F \subset M$ such that $E \subseteq F$ it is immediately seen from the outward minimising property of E that

$$P(E^c) = P(E) \le P(F) \le P(F^c).$$
 (3.2.6)

Let $x \in \overline{\partial^* E}$ and r > 0, and choose $F = (E^c \setminus B(x,r))^c \supset E$. Then, for s > r, we have, applying (3.2.6)

$$P(E^{c}, B(x,s)) + P(E^{c}, M \setminus B(x,s)) = P(E^{c}) \le P(F^{c}) = P(F^{c}, B(x,s)) + P(F^{c}, M \setminus B(x,s)).$$
(3.2.7)

Since $E^c \setminus F^c \in B(x,s)$, we actually have $P(F^c, \mathbb{R}^n \setminus B(x,s)) = P(E^c, \mathbb{R}^n \setminus B(x,s))$, so we deduce from (3.2.7), the definition of F and basic set operations for the perimeter (see e.g. [Mag12, Theorem 16.3])

$$P(E^{c}, B(x,s)) \leq P(F^{c}, B(x,s)) = P(E^{c} \setminus B(x,r), B(x,s))$$
$$= |Int(E^{c}) \cap B(x,r)| + P(E^{c}, B(x,s) \setminus \overline{B(x,r)}),$$

that, upon letting $s \to r^+$, yields

$$P(E^c, B(x,r)) \le |\operatorname{Int}(E^c) \cap B(x,r)|. \tag{3.2.8}$$

Recalling again from basic set operations for the perimeter that

$$P(E^c, B(x,r)) + |\operatorname{Int}(E^c) \cap B(x,r)| = P(E^c \cap B(x,r)),$$

we deduce from (3.2.8) the estimate

$$P(E^c \cap B(x,r)) \le 2|\text{Int}(E^c) \cap B(x,r)|.$$

For r small enough, but uniform as x ranges in the compact $\overline{\partial^* E}$, we can apply an Euclidean-like type inequality to left hand side of the above inequality, leading to

$$C_1(n,r_0) \le |E^c \cap B(x,r)|^{(n-1)/n} \le 2|\text{Int}(E^c) \cap B(x,r)|$$
 (3.2.9)

for any $r \in (0, r_0)$. The constant $C_1(n, r_0)$ can be seen to approach $n|\mathbb{B}^n|$ as $r_0 \to 0^+$, see e.g. [BM82, Appendice C]. Define, for $0 < r < r_0$ the function $m(r) = |E^c \cap B(x, r)|$. Applying the coarea formula (see [Mag12, Theorem 13.1 and Remark 13.4]), one gets that m is an absolutely continuous function satisfying $m'(r) = |\operatorname{Int}(E^c) \cap B(x, r)|$ for almost any $r \in (0, r_0)$, and thus (3.2.9) gives

$$C_1(n,r_0)m(r)^{(n-1)/n} \le 2m'(r)$$
 (3.2.10)

for almost any $r \in (0, r_0)$. Moreover, since $x \in \overline{\partial^* E} = \text{supp } \mu_E$, we have, by [Mag12, Proposition 12.19], that m > 0 and tends to 0 as $r \to 0^+$. Integrating (3.2.10), and taking

into account that for small r we can approximate |B(x,r)| with $|\mathbb{B}^n|r^n$, we then easily get

$$\frac{|E^c \cap B(x,r)|}{|B(x,r)|} \ge C_2(n,r_0),$$

for some constant $0 < C_2(n, r_0) < 1$ that approaches 2^{-n} as $r_0 \to 0^+$. This immediately gives

$$\frac{|E \cap B(x,r)|}{|B(x,r)|} \le C(n,r_0),$$

with $0 < C(n, r_0) < 1$ for any $0 < r < r_0$, as claimed.

In [Car11, Theorem 6, (i)], the author showed that if a set with finite perimeter satisfies relaxed density estimates then its measure theoretic interior and exterior are open. We report (a part of) his argument, in order to clarify that an upper density estimate suffices to get the openness of the measure theoretic interior.

Lemma 3.5. Let (M,g) be a complete Riemannian manifold. Let $E \subset \mathbb{R}^n$ be a set with finite perimeter, and suppose that for any $x \in \overline{\partial^* E}$

$$\lim_{r \to 0^+} \frac{|E \cap B(x,r)|}{|B(x,r)|} < \delta < 1 \tag{3.2.11}$$

uniformly on $\overline{\partial^* E}$. Then Int (E) is open.

Proof. Let $y \in \text{Int}(E)$. First observe that $y \notin \overline{\partial^* E}$, since otherwise (3.2.11) would contradict the condition (3.2.1) defining Int(E). We now construct a ball centered at y fully contained in Int(E). Let $d = \text{dist}(y, \overline{\partial^* E})$. By the definition (3.2.1) of Int(E),

$$\frac{|E \cap B(y,r)|}{|B(y,r)|} > 0$$

for some $r' \in (0, d)$. Then we can deduce

$$\frac{|(M \setminus E \cap B(y, r'))|}{|B(y, r')|} = 0, \tag{3.2.12}$$

since, otherwise, the relative isoperimetric inequality [Mag12, Proposition 12.37] would yield $|\partial^* E \cap B(y,r')| > 0$, in turn resulting in the contradiction $\operatorname{dist}(y,\overline{\partial^* E}) < d$. Observe that such inequality clearly holds for small balls also in Riemannian manifolds, and it can be shown just by performing the computations in a local chart. The above argument in particular works up to eventually choose a suitable, uniform d' smaller than d. By (3.2.12), for any $y \in B(y,r')$, we have

$$\frac{|E \cap B(z, r'')|}{|B(z, r'')|} = 1$$

for any $r'' \in (0, r' - \operatorname{dist}(z, y))$. This clearly implies that $B(y, r') \subset \operatorname{Int}(E)$.

As a direct consequence of (3.2.2), Lemma 3.4 and Lemma 3.5, we obtain the remarkable property of outward minimising sets to have an open representative.

Proposition 3.6. *Let* (M,g) *be a complete Riemannian manifold. Let* $E \subset M$ *be outward minimising. Then* Int(E) *is open.*

3.2.3 Maximal volume solutions to the least area problem with obstacle in Riemannian manifolds

Let (M,g) be a complete noncompact Riemannian manifold of dimension $n \geq 2$. We are interested in the existence of a *bounded* open set with finite perimeter E solving the least area problem with obstacle Ω , where $\Omega \subset M$ is a bounded open set with finite perimeter. We say that E solves the *least area problem with obstacle* Ω if it is a bounded set containing Ω satisfying

$$P(E) = \inf\{P(F) \mid \Omega \subset F, F \text{ bounded set with finite perimeter}\}$$
 (3.2.13)

A solution to (3.2.13) could could clearly be non-unique. We can try to select among these solutions a set E'

$$|E'| = \sup\{|E| \mid E \text{ solves the least area problem with obstacle } \Omega\}.$$
 (3.2.14)

We refer to a *bounded* set E' with finite perimeter containing Ω , that solves *both* (3.2.13) and (3.2.14) as *maximal volume solutions of the least area problem with obstacle* Ω

Remark 3.7. Observe that bounded solutions to the least area problem with obstacle Ω are outward minimising, as well maximal volume solutions are strictly outward minimising. In particular, outward minimising sets E are characterised by being solutions to the least area problem with obstacle E itself. Interestingly, it is immediately seen that E is strictly outward minimising if and only if it is also a maximal volume solution to the least area problem with obstacle E.

In the Euclidean case, the existence of a maximal volume solution to the least area problem with obstacle Ω is an easy consequence of the Direct Method (compare with [BT84] and [SWZ93]). In a general noncompact Riemannian manifold, minimising sequences could fatally be unbounded, as the following examples show.

Example 3.8 (Cuspidal manifolds). Let (M,g) be a complete noncompact Riemannian manifold whose metric splits as $g = d\rho \otimes d\rho + e^{-2\rho}g_{\mathbb{S}^{n-1}}$ on $M \setminus B = [1, +\infty) \times \mathbb{S}^{n-1}$ for some precompact set $B \subset M$. Then, the area of $\partial \{\rho < r\}$ decays as e^{-r} , and in particular enveloping a bounded set Ω with $\{\rho < r\}$, for bigger and bigger r, we see that

$$P(E) = \inf\{P(F) \mid \Omega \subset F, F \text{ bounded set with finite perimeter}\} = 0.$$

In particular, there cannot exist a solution to the least area problem with obstacle Ω . Observe also that (M,g) has finite volume, and thus it does not support a L^1 -Euclidean-like Sobolev inequality by [PST14, Proposition 3.1] nor the Ricci curvature is nonnegative, and thus coherently the assumptions of Theorem 3.1 are not satisfied.

Example 3.9 (Manifolds with a cylindrical end). Let (M,g) be a complete noncompact Riemannian manifold whose metric splits as $g = d\rho \otimes d\rho + g_{\mathbb{S}^{n-1}}$ on $M \setminus B = [1, +\infty) \times \mathbb{S}^{n-1}$ for some precompact set $B \subset M$. Let Ω be any bounded set with finite perimeter containing B. Then, any set of the form $B \cup \{\rho < r\}$ containing Ω solves the least area problem with obstacle Ω , since the sets $\{\rho = r\}$ are even area minimising, but since $|\{\rho = r_2\}| = |\{\rho = r_1\}| \text{ and } |B \cup \{\rho < r_2\}| = |B \cup \{\rho < r_1\}| + (r_2 - r_1) \text{ for any } r_2 \ge r_1 \ge 1$, we see at once that

$$\sup\{|E| \mid E \text{ solves the least area problem with obstacle } \Omega\} = +\infty.$$

In particular, it cannot exist a *maximal* volume solution to the least area problem with obstacle Ω . Observe that (M,g) has linear volume growth, and therefore by [PST14, Proposition 3.1] the L^1 -Euclidean-like Sobolev inequality cannot hold nor (3.1.2) holds.

In the following basic result, we isolate as a necessary and sufficient condition for obtaining a maximal volume solution to the least area problem with obstacle in a complete noncompact Riemannian manifold the existence of an exhausting sequence of bounded strictly outward minimising sets. Observe that, in \mathbb{R}^n , such a sequence is trivially given by balls.

Theorem 3.10 (Existence of maximal volume solutions to the least area problem in Riemannian manifolds-abstract criterion). Let (M,g) be a noncompact Riemannian manifold. Let $O \in M$. Then, for any R > 0 there exists a (strictly) outward minimising set S_R with $B(O,R) \in S_R$ if and only if there exists a maximal volume solution to the least area problem with obstacle Ω for any bounded $\Omega \subset M$ with finite perimeter.

Proof. One direction is almost trivial. Indeed if for any bounded $\Omega \subset M$ with finite perimeter there exists a maximal volume solution to the least area problem with obstacle Ω , then, taking a sequence of geodesic balls $B(O,R_j)$ with $R_j \to \infty$ and considering the bounded maximal volume solution to the least area problem with obstacle $B(O,R_j)$ yields the desired sequence of strictly outward minimising sets (compare with Remark 3.7). Observe that $B(O,R_j)$ can be assumed to be of finite perimeter substantially by coarea formula, and that maximal volume solutions are obviously strictly outward minimising. Let $\Omega \subset M$ be a bounded set with finite perimeter. Let $\{F_j\}_{j\in\mathbb{N}}$ be a minimising sequence for the least area problem with obstacle Ω . Define

$$m = \inf\{P(F) \mid \Omega \subset F, F \text{ bounded set with finite perimeter}\}.$$

Then, for any $\varepsilon > 0$, there exists $j_{\varepsilon} \in \mathbb{N}$ such that

$$m \leq P(F_j) \leq m + \varepsilon$$

for any $j \ge j_{\varepsilon}$. Let S be an outward minimising set containing Ω , that exists by assumption. By a standard set-theoretical property of the perimeter (see e.g. [Mag12, Lemma 12.22]), we have

$$P(F_j \cap S) \le P(F_j) + P(S) - P(F_j \cup S).$$

Since *S* is outward minimising, we have $P(S) \leq P(F_i)$, and then we deduce that

$$P(F_i \cap S) \le P(F_i). \tag{3.2.15}$$

In particular, since $\Omega \subset (F_i \cap S)$ we have

$$m \leq P(F_i \cap S) \leq m + \varepsilon$$

for any $j \geq j_{\varepsilon}$, that is, the sets $F_j \cap S$ form an equibounded minimising sequence. Then, the Compactness Theorem [Mag12, Theorem 12.26] yields a bounded set of finite perimeter F such that $\chi_{F_j \cap S} \to \chi_F$ in L^1 , possibly along a subsequence. Moreover, by the lower semicontinuity of the perimeter [Mag12, Proposition 12.15], the set F is a minimiser for the minimisation problem (3.2.13). We are left to show that we can modify F in order to obtain a set E containing Ω with P(E) = P(F). Actually, it suffices to define $E = F \cup (\Omega \setminus F)$. Clearly $\Omega \subseteq E$. Moreover, since $\Omega \subset F_j \cap S$ for any j, and $\chi_{F_j \cap S} \to \chi_F$ in L^1 , we have $|\Omega \setminus F| = 0$. Since De Giorgi's perimeter is defined up to null sets, we have P(E) = P(F).

Let now $\{E_j\}_{j\in\mathbb{N}}$ be a sequence of bounded sets with finite perimeter containing Ω with

$$|E_i| \to \sup\{|E| \mid E \text{ solves the least area problem with obstacle } \Omega\}$$
 (3.2.16)

as $j \to +\infty$. In particular, $P(E_j) = m$ for any j. Let S be a bounded *strictly* outward minimising set containing Ω . Then, as in (3.2.15), we have $P(E_j \cap S) \leq P(E_j)$. We thus have

$$m \leq P(E_j \cap S) \leq P(E_j) = m.$$

Since S is strictly outward minimising, we deduce that $|E_j \setminus (E_j \cap S)| = 0$ for any j. This implies $|E_j| = |E_j \cap S|$ for any j, and then we can consider $E_j \cap S$ in place of E_j in (3.2.16). The Compactness Theorem for sets with finite perimeter then yields a bounded set E' realising the supremum in (3.2.16). Up to a zero-measure modification as before, we can also suppose $\Omega \subseteq E'$. By the lower semicontinuity of the perimeter we also have

$$m = \liminf P(E_i \cap S) \ge P(E') \ge m$$
.

We showed that P(E') = m and its volume realises the supremum in (3.2.16), that is, E' is a maximal volume solution to the least area problem with obstacle Ω .

3.2.4 The strictly outward minimising hull of a set with finite perimeter

Once we know that maximal volume solutions to the least area problem exist, we are interested in a specific representative, that will naturally arise as level set of the Weak Inverse Mean Curvature Flow. We thus define the *strictly outward minimising hull* of a set with finite perimeter.

Definition 3.11 (Strictly outward minimising hull). *Let* $\Omega \subset M$ *be a bounded open set with finite perimeter. We define the* strictly outward minimising hull *of* Ω *as*

$$\Omega^* = \operatorname{Int}\left(\bigcap_{E \in \operatorname{SOM}(\Omega)} \operatorname{Int}(E)\right), \tag{3.2.17}$$

where

SOM
$$(\Omega) = \{ E \subset M \mid \Omega \subseteq E \text{ and } E \text{ is strictly outward minimising} \}.$$

The following is the main result of this section.

Theorem 3.12. Let (M,g) be a complete noncompact Riemannian manifold admitting an exhausting sequence of bounded strictly outward minimising sets. Let $\Omega \subset M$ be a bounded set with finite perimeter. Then, Ω^* is an open, bounded, maximal volume solution to the least area problem with obstacle Ω . In particular Ω^* is strictly outward minimising.

Proof. We split the proof in several steps.

Step 1. We prove that that finite intersections of strictly outward minimising sets are strictly outward minimising. It clearly suffices to prove that the intersection of two strictly outward minimising sets E_1 and E_2 is outward minimising. We do it by first showing that it is outward minimising. This preliminary passage is carried out in the proof of [BT84, Proposition 1.3], but we include the argument, since it will be important also in the sequel. Let $F \subset M$ such that $E_1 \cap E_2 \subset F$, and define $L = F \setminus (E_1 \cap E_2)$. Then, applying the already recalled well known property holding for sets with finite perimeter

$$P(F \cap G) + P(F \cup G) \le P(F) + P(G)$$

to the couples of sets $F = (E_1 \cap E_2) \cup L$ and $G = E_1$, $F = (E_1 \cap E_2) \cup (L \cap E_1)$ and $G = E_2$, give respectively

$$P((E_1 \cap E_2) \cup (L \cap E_1)) + P(E_1 \cup L) \le P((E_1 \cap E_2) \cup L) + P(E_1)$$
 (3.2.18)

and

$$P(E_1 \cap E_2) + P((E_1 \cap L) \cup E_2) \le P((E_1 \cap E_2) \cup (L \cap E_1)) + P(E_2).$$
 (3.2.19)

Combining (3.2.19) with (3.2.18) we obtain at once the following chain of inequalities

$$P(E_{1} \cap E_{2}) \leq P\left(\left(E_{1} \cap E_{2}\right) \cup \left(L \cap E_{1}\right)\right) + P(E_{2}) - P\left(\left(E_{1} \cap L\right) \cup E_{2}\right)$$

$$\leq P\left(\left(E_{1} \cap E_{2}\right) \cup \left(L \cap E_{1}\right)\right)$$

$$\leq P\left(\left(E_{1} \cap E_{2}\right) \cup L\right) + P(E_{1}) - P(E_{1} \cup L)$$

$$\leq P\left(\left(E_{1} \cap E_{2}\right) \cup L\right) = P(F)$$

$$(3.2.20)$$

where the second and fourth inequality are due to the property of E_1 and E_2 to be outward minimising. Assume now that that $P(E_1 \cap E_2) = P(F)$ for some $F \supseteq E_1 \cap E_2$. Then, the second and the last inequality in (3.2.20) become equalities, yielding respectively

$$P(E_2) = P((E_1 \cap L) \cup E_2)$$
 (3.2.21)

$$P(E_1) = P(E_1 \cup L). \tag{3.2.22}$$

Since E_1 is strictly outward minimising, we can deduce from (3.2.22) that

$$|(E_1 \cup L) \setminus E_1| = |L \setminus E_1| = 0.$$
 (3.2.23)

On the other hand, by (3.2.21) and by E_2 being strictly outward minimising we obtain

$$\left| \left(E_1 \cap L \right) \cup E_2 \right) \setminus E_2 \right| = 0. \tag{3.2.24}$$

By the definition of L it is easily seen that the sets $(E_1 \cap L) \cup E_2$ and E_2 are disjoint, and then (3.2.24) actually reads

$$|E_1 \cap L| = 0. (3.2.25)$$

Combining (3.2.25) with (3.2.23) we obtain

$$|F \setminus (E_1 \cap E_2)| = |L| = |L \setminus E_1| + |L \cap E_1| = 0$$
,

that is, $E_1 \cap E_2$ is strictly outward minimising, as claimed.

Step 2. We prove that Ω^* is strictly outward minimising. Let $\{E_i\}_{i\in\mathbb{N}}$ be a sequence of set realising Ω^* as a countable intersection, that is

$$\Omega^* = \operatorname{Int}\left(\bigcap_{i=1}^{\infty} \operatorname{Int}\left(E_i\right)\right)$$

By (3.2.2) and Lemma 3.3, it suffices to show that $\widetilde{\Omega} = (\bigcap_{i=1}^{\infty} \operatorname{Int}(E_i))$ is strictly outward minimising. First, [BT84, Proposition 1.3] ensures that Ω^* , is outward minimising. Assume then that $P(\widetilde{\Omega}) = P(F)$ for some $F \subset M$ containing $\widetilde{\Omega}$. Clearly, F is outward minimising. Let F' be a maximal volume solution to the least area problem with obstacle F, that exists by Theorem 3.10. Then F' is strictly outward minimising (compare with Remark 3.7) and P(F') = P(F), being F outward minimising. For ease of notation, assume $E_i = \operatorname{Int}(E_i)$ for any $i \in \mathbb{N}$. By (3.2.2) and Lemma 3.3, this will result in no loss of generality. Let $A_j = \bigcap_{k=1}^j E_k$. Since F' is strictly outward minimising, by Step 1 $A_j \cap F'$

is strictly outward minimising. In particular, since $A_{j+1} \cap F' \subset A_j \cap F' \subset F'$, we have $P(A_{j+1} \cap F') \leq P(A_j \cap F') \leq P(F')$. Then, by the lower semicontinuity of the perimeter, the definition of $\widetilde{\Omega}$ and the just observed monotonicity of $P(A_j \cap F)$ as j increases we get

$$P(F') = P(\widetilde{\Omega}) \le \liminf_{j \to \infty} P(A_j \cap F') \le P(A_k \cap F') \le P(F')$$

for any $k \in \mathbb{N}$, where the last inequality is again due to the fact that A_k is outward minimising. In particular $P(A_k \cap F') = P(F')$ for any k. Since $A_k \cap F'$ is in fact strictly outward minimising, we deduce that $|F' \setminus (A_k \cap F')| = |F' \setminus A_k| = 0$ for any k. Since we have $|A_k \setminus \widetilde{\Omega}| \to 0$ as $k \to \infty$, we conclude that $|F' \setminus \Omega^*| = \lim_{k \to \infty} |F' \setminus A_k| = 0$. Since $F \subseteq F'$ this trivially implies $|F \setminus \widetilde{\Omega}| = 0$. We have shown that $\widetilde{\Omega}$, and consequently Ω^* , is strictly outward minimising, as claimed.

Step 3. We show that Ω^* is a maximal volume solution to the least area problem with obstacle Ω . Let E' be any solution to (3.2.14). We want to show that $P(\Omega^*) = P(E)$ and $|\Omega^*| = |E'|$. Since Ω^* is an admissible competitor for (3.2.13), we have

$$P(E') \le P(\Omega^*). \tag{3.2.26}$$

Clearly E' is strictly outward minimising. Letting Ω be defined as in Step 2, we thus get by definition $\widetilde{\Omega} \subseteq \operatorname{Int}(E')$, and thus, since $\widetilde{\Omega}$ is outward minimising and differs from Ω^* by a set of measure zero, we get also

$$P(\Omega^*) = P(\widetilde{\Omega}) \le P(E'),$$

that, combined with (3.2.26), implies $P(\Omega^*) = P(E')$. The set Ω^* becomes in particular a valid competitor in (3.2.14), giving immediately

$$|\widetilde{\Omega}| = |\Omega^*| \le |E'| = |\operatorname{Int}(E')|. \tag{3.2.27}$$

However, since $P(\widetilde{\Omega}) = P(\operatorname{Int}(E'))$, and $\widetilde{\Omega} \subseteq \operatorname{Int}(E')$, the strict inequality sign in (3.2.27) would contradict the fact that Ω^* is strictly outward minimising. It follows that $|\Omega^*| = |E'|$ proving the claim.

Step 4. We are left to observe that Ω^* is open. Since Ω^* is outward minimising by Step 2, and by definition $Int(\Omega^*) = \Omega^*$, this follows at once from Proposition 3.6.

Let us summarise Theorem 3.10 and Theorem 3.12 in the following general statement.

Theorem 3.13. Let (M, g) be a complete noncompact Riemannian manifold. Then, the following are equivalent.

- (i) (M,g) admits an exhausting sequence of bounded strictly outward minimising sets.
- (ii) For any bounded $\Omega \subset M$ with finite perimeter, there exists a bounded maximal volume solution to the least area problem with obstacle Ω .
- (iii) Let $\Omega \subset M$ be a bounded set with finite perimeter. Then, Ω^* is an open, bounded, maximal volume solution to the least area problem with obstacle Ω . In particular Ω^* is strictly outward minimising.

Regularity and approximations of the strictly outward minimising hull

We recall a fundamental regularity result for solutions of the least area problem with obstacle, under an additional \mathscr{C}^2 assumption on the regularity on $\partial\Omega$. This is the main

result of [SZW91]. According to the comprehensive [HI01, Theorem 1.3], it applies also to general Riemannian ambient settings. By Theorem 3.13, we can apply it directly to Ω^* .

Theorem 3.14 (Regularity of the strictly outward minimising hull). Let (M,g) be a complete noncompact Riemannian manifold admitting an exhausting sequence of bounded strictly outward minimising sets. Let $\Omega \subset M$ be a bounded set with \mathcal{C}^2 boundary. Then

- (i) $\partial \Omega^*$ is a $\mathscr{C}^{1,1}$ hypersurface in a neighbourhood of any point in $\partial \Omega^* \cap \partial \Omega$.
- (ii) $\partial \Omega^*$ is area minimising in $\partial \Omega^* \setminus \partial \Omega$. In particular there exists a singular set Sing $\subset \partial \Omega^* \setminus \partial \Omega$ with Hausdorff dimension at most n-8 such that $\partial \Omega^* \setminus \partial \Omega$ is a real analytic hypersurface in a neighbourhood of any point in $(\partial \Omega^* \setminus \partial \Omega) \setminus \text{Sing}$.

Remark 3.15. In [SZW91], the authors in fact consider a representative E satisfying $\partial E = \overline{\partial^* E}$. However, since, by our definition $Int(\Omega^*) = \Omega^*$, this condition is automatically satisfied due to (3.2.3).

From now on we are now going to deal just with set with sets with \mathscr{C}^2 boundary. In fact, for simplicity, we will always assume $\partial\Omega$ to be a smooth hypersurface.

Let us record, for future references, two direct and easy consequences of the above regularity result.

Remark 3.16. Theorem 3.14 implies $|\partial^* \Omega^*| = |\partial \Omega^*|$. In particular, taking into account Remark 3.7, we can characterise outward minimising sets as those satisfying

$$|\partial\Omega| = |\partial\Omega^*|. \tag{3.2.28}$$

Observe that, by Theorem 3.13, we always have $|\partial \Omega^*| \leq |\partial \Omega|$. Checking condition (3.2.28) then amounts to check that $|\partial \Omega| \leq |\partial \Omega^*|$.

Moreover, let us also point out that, by a very standard variational argument, the weak mean curvature (see (1.7.6)) of $\partial\Omega^*$ is nonnegative.

We finally state the following nice one sided approximation result due to Schmidt [Sch15], that we are going to use in the *p*-capacitary approximation of $|\partial\Omega^*|$. It asserts that a bounded set with finite perimeter admit a one sided approximation in perimeter by bounded sets with smooth boundary if $P(\Omega) = |\partial\Omega|$. This is clearly the case for Ω^* , due to Theorem 3.14. The arguments being purely local, Schmidt's result applies straightforwardly in Riemannian manifolds. Observe also that although in [Sch15] just interior approximation is worked out, an analogous exterior approximation obviously follows as well.

Theorem 3.17 (Exterior approximation of Ω^*). Let (M,g) be a complete noncompact Riemannian manifold admitting an exhausting sequence of bounded strictly outward minimising sets. Let $\Omega \subset M$ be an open bounded set with smooth boundary, and Ω^* is strictly outward minimising hull. Then, there exists a sequence of bounded sets $\{\Omega_k\}_{k\in\mathbb{N}}$ with smooth boundary such that

$$\Omega^* \subset \Omega_k$$
, $|\partial \Omega_k| \to |\partial \Omega^*|$.

3.3 Families of manifolds admitting the strictly outward minimising hull

Here, we show that in the assumptions of Theorem 3.1 the notion of strictly outward minimising hull recalled in Definition 3.11 gives a well posed open bounded maximal volume solution to the least area problem with obstacle.

3.3.1 Euclidean-like isoperimetric inequality and strictly outward minimising hull

The following result affirms that a maximal volume solution to the least area problem with obstacle exists on any Riemannian manifold where an Euclidean-like Isoperimetric Inequality is available, that is, satisfying (3.1.1) for any bounded $\Omega \subset M$ with smooth boundary. In particular, by Theorem 3.10 any bounded $\Omega \subset M$ with finite perimeter admits a strictly outward minimising hull. We express our gratitude to Prof. G.P. Leonardi for having outlined the core of the following argument.

Proposition 3.18. Let (M,g) be a complete noncompact Riemannian manifold satisfying admitting a constant $C_{iso} > 0$ such that

$$\frac{|\partial\Omega|^{n/(n-1)}}{|\Omega|} \ge C_{\text{iso}} \tag{3.3.1}$$

for any bounded set Ω with smooth boundary. Then, for any bounded $\Omega \subset M$ with finite perimeter there exists a maximal volume solution to the least area problem with obstacle Ω .

Proof. The main goal is showing that the Isoperimetric Inequality (3.3.1) forces a minimising sequence for the least area problem to stay uniformly inside a ball, and then to conclude by compactness and lower semicontinuity as in the proof of Theorem 3.10. Let then F_j be a minimising sequence for the least area problem with obstacle made of bounded sets with finite perimeter. Clearly, we can suppose $P(F_j) \leq P(\Omega)$, and thus the well known *local* compactness-lower semicontinuity properties of the perimeter (namely, the combination of [Mag12, Corollary 12.17] with [Mag12, Proposition 12.15]) yield a set F such that a subsequence (relabeled as usual) of the F_j 's locally converges in L^1 to F and

$$P(F) \leq \liminf_{j \to +\infty} P(F_j),$$

so that F realises the infimum in the least area problem with obstacle Ω . We claim that, up to null sets, $F \subset B(O,r)$ for some r>0. Assume then by contradiction $|F\setminus B(O,r_j)|>0$ for some sequence $r_j\to +\infty$ as $r_j\to +\infty$. In particular, this implies that $|F\setminus B(O,r)|>0$ for any r big enough. The assumed Isoperimetric Inequality (clearly in force, by approximation, for any set with finite perimeter) yields

$$C_{\text{iso}}^{\frac{n-1}{n}} |F \setminus B(O,R)|^{\frac{n-1}{n}} \le P(F \setminus B(O,r)) = P(F, \overline{B(O,r)}^c) + |\partial B(O,r) \cap F|$$
 (3.3.2)

for any r such that $\partial B(O,r)$ has a (n-1)-negligible singular set. In particular, (3.3.2) holds for almost any r>0. In the above inequality, as in the following arguments, we are possibly considering a representative for F. Observe that by coarea formula the function $m(r) = |F \setminus B(O,r)|$ is absolutely continuous, with derivative $m'(r) = -|\partial B(O,r) \cap F|$ for almost any r>0. We claim that

$$P(F, \overline{B(O,r)}^c) \le |\partial B(O,r) \cap F|, \tag{3.3.3}$$

in order to deduce from (3.3.2) a differential inequality leading in turn to a contradiction. This is accomplished through an argument similar to the classic one used in the proof of Lemma 3.4, relying on the minimising property of F. Let $L = F \setminus \overline{B(O,r)}^c$, and observe that for R big enough we have $\Omega \subset L$. By the minimising property of F, we thus get, for s < r

$$P(F, \overline{B(O,s)}^c) + P(F, B(O,s)) = P(F) \le P(L) = P(L, \overline{B(O,s)}^c) + P(L, B(O,s)).$$

Since clearly P(L, B(O, s)) = P(F, B(O, s)), we deduce

$$P(F, \overline{B(O,s)}^c) \le P(L, \overline{B(O,s)}^c). \tag{3.3.4}$$

On the other hand, observe that, as usual up to representatives

$$P(L, \overline{B(O,s)}^c) = P(F, B(O,r) \setminus \overline{B(O,s)}) + |\partial B(O,R) \cap F|.$$

Letting $s \to r^-$, the first term in the right hand side above vanishes and thus plugging this information into (3.3.4) leaves us with the claimed (3.3.3). Inserting (3.3.3) into (3.3.2) yields, as explained above, the differential inequality

$$C_{\mathrm{iso}}^{\frac{n-1}{n}}m(r)^{\frac{n-1}{n}}\leq -2m'(r),$$

holding true for almost any r big enough, with $m(r) = |E \setminus B(O, r)|$. Integrating it, we get

$$\frac{1}{2}C_{\text{iso}}^{(n-1)/n}(r_2 - r_1) \le n \left[m(r_1)^{\frac{1}{n}} - m(r_2)^{\frac{1}{n}} \right]$$

for any $r_2 > r_1$ big enough. Letting $r_2 \to +\infty$ the right hand side converges to $nm(r_1)^{1/n}$, since the other summand vanishes by the property of E to have finite volume, while the left hand side cleary diverges. This contradiction arose from $|E \setminus B(O, r_j)| > 0$ for some diverging sequence of r_j 's, and thus we proved that there exists a ball B(O, R) containing, up to a null set, the solution F to the least area problem with obstacle Ω .

It remains to show that that we can find a bounded maximal volume solution to the least area problem. To see this, let E_j be a maximising (for the volume) sequence of bounded solutions to the least area problem with obstacle Ω . Then, by the compactness property used in the first part of this proof, such a sequence converges locally in L^1 to a set with finite perimeter E, that by lower semicontinuity is still a possibly unbounded solution to the least area problem with obstacle Ω . The same argument as above involving the Isoperimetric Inequality then shows that, up a to null sets, E is contained in some ball, completing the proof.

For future reference, let us state the following immediate Corollary of Proposition 3.18 and Theorem 3.13.

Corollary 3.19. Let (M,g) be a complete noncompact Riemannian manifold satisfying the assumption (i) in the statement of Theorem 3.1. Then, for any bounded $\Omega \subset M$ with finite perimeter, the strictly outward minimising hull Ω^* is an open bounded maximal volume solution to the least area problem with obstacle Ω .

3.3.2 Weak Inverse Mean Curvature Flow and strictly outward minimising hulls

In this Subsection we show that the existence of a *proper* solution to the Weak Inverse Mean Curvature Flow naturally yields an exhausting sequence of strictly outward minimising sets, and that the strictly outward minimising hull of a bounded set Ω with smooth boundary is simply given by the interior of the the zero-level set of such a solution. This is substantially the content of [HI01, Lemma 1.4], but here we want to emphasise and fully detail the relation between the *existence* of the strictly outward minimising hull and that of the Weak Inverse Mean Curvature Flow.

Let us recall the notion of Inverse Mean Curvature Flow, as well as its weak formulation introduced by Huisken-Ilmanen in [HI01]. Let (M,g) be a complete noncompact Riemannian manifold, and let $\Omega \subset M$ be a bounded subset with smooth boundary given by the immersion $F_0: \partial\Omega \to \mathbb{R}^n$. Assume in addition that the mean curvature of $\partial\Omega$ is strictly positive. Then, we say that the hypersurfaces $\{\partial\Omega_t\}_{t\in[0,T)}$, for some T>0 are evolving by IMCF with initial datum $\partial\Omega$ if they are given by immersions $F(t,\cdot):\partial\Omega\to\mathbb{R}^n$ satisfying

$$\frac{\partial}{\partial t}F(t,x) = \frac{1}{H}(t,x)\nu(t,x), \qquad F(0,x) = F_0(x) \tag{3.3.5}$$

where ν is the exterior unit normal to the hypersurface $\partial\Omega_t$ and H is its related mean curvature. It is well known that the IMCF of a strictly mean-convex hypersurface enjoys existence in some time interval [0, T), see e.g. the comprehensive [HP99, Theorem 3.1].

The weak formulation of the IMCF actually regards the degenerate elliptic problem solved by the function whose level sets evolve by inverse mean curvature. Namely, looking at the evolving hypersurfaces $\partial\Omega_t$ as level sets $\{w=t\}$ of a smooth function, it is easily seen that w must satisfy the equation

$$\operatorname{div}\left(\frac{\mathrm{D}w}{|\mathrm{D}w|}\right) = |\mathrm{D}w|\tag{3.3.6}$$

in the region foliated by the evolving hypersurfaces. In particular, if there exists a smooth solution to (3.3.5) made of embedded closed hypersurfaces, then it is well defined the smooth function w with nonvanishing gradient solving (3.3.6).

A weak solution to (3.3.6) starting from Ω , that we will call Weak Inverse Mean Curvature Flow starting from Ω , will be a function $w \in \operatorname{Lip}_{\operatorname{loc}}(M)$ satisfying the following conditions.

(i) For every $v \in \operatorname{Lip}_{\operatorname{loc}}(M)$ with $\{w \neq v\} \in M \setminus \overline{\Omega}$ and any compact set $K \subset M \setminus \Omega$ containing $\{w \neq v\}$,

$$J_w^K(w) \le J_w^K(v)$$

where

$$J_w^K(v) = \int_K |\mathrm{D}v| + v |\mathrm{D}w| \,\mathrm{d}\mu.$$

(ii) The set Ω is the 0-sublevel set of w, that is

$$\Omega = \{ w < 0 \}.$$

Remark 3.20 (Properness of the Weak IMCF). We say that the Weak Inverse Mean Curvature Flow is *proper* if w is a proper function. In the rest of this note, we will always consider Weak Inverse Mean Curvature Flows that are proper. Observe that if $w(x) \to +\infty$ as $d(O, x) \to +\infty$, then w is proper. The validity of this condition is assumed as definition of properness in [KN09]. However, there may exist a proper Weak IMCF w such that $w \not\to +\infty$ at infinity (see Example 3.29 below).

It is definitively convenient to rephrase (i) in the definition of Weak Inverse Mean Curvature Flow in terms of the level sets of the solution.

(i-bis) For any $t \ge 0$, the sets $\{w \le t\}$ satisfies

$$J_w^K(\{w \le t\}) \le J_w^K(F)$$

for any $F \subset M$ with locally finite perimeter satisfying $\{w \leq t\}\Delta F \Subset M \setminus \overline{\Omega}$ and any compact $K \subset M \setminus \overline{\Omega}$ containing $\{w \leq t\}\Delta F$, where

$$J_w^K(F) = |\partial^* F \cap K| + \int_{F \cap K} |\mathrm{D}w| \,\mathrm{d}\mu.$$

We refer the reader to [HI01, Lemma 1.1 and Lemma 2.2] and the discussion thereafter for the proof of the equivalence between conditions (i) and (i-bis). One can then deduce from the fundamental Minimizing Hull Property Lemma 1.4 in [HI01] that the sets $\mathrm{Int}\{w \leq t\}$ are strictly outward minimising. In particular, the condition of having an exhausting sequence of strictly outward minimising sets is fulfilled every time there exists the Weak IMCF, and the analysis of the preceding section allows then to define the strictly outward minimising hull of Ω . It finally turns out that $\Omega^* = \mathrm{Int}\{w \leq 0\}$. Let us carefully prove this fact, for the reader's sake.

Proposition 3.21. Let (M,g) be a complete noncompact Riemannian manifold. Assume that for any bounded $\Omega \subset M$ with smooth boundary there exists a proper weak IMCF w starting from Ω , then the strictly outward minimising hull Ω^* is an open bounded maximal volume solution to the least area problem with obstacle Ω . Moreover, $\operatorname{Int}\{w \leq 0\} = \Omega^*$.

Proof. Let w be the Weak IMCF emanating from the bounded open set with smooth boundary Ω . By [HI01, Lemma 1, 4, (ii)], the set $\mathrm{Int}\{w \leq 0\}$ is bounded and strictly outward minimising. In particular, if w exists for any such Ω , taking an exhausting sequence of bounded open sets with smooth boundary yields an exhausting sequence of bounded strictly outward minimising sets, that allows us to use Theorem 3.13 to show that Ω^* defined in Definition 3.11 is an open bounded maximal volume solution to the least area problem with obstacle Ω . By Proposition 3.6, the set $\mathrm{Int}\{w \leq 0\}$ is also open. By definition, we have $\Omega^* \subseteq \mathrm{Int}\{w \leq 0\}$. Then, by properties (i-bis) and (ii) in the definition of the Weak IMCF recalled above, we have, up to a suitable choice of K,

$$P(\{w \leq 0\}) + \int\limits_{\operatorname{Int}\{w \leq 0\} \setminus \Omega} |\mathsf{D}w| \, \mathrm{d}\mu \, \leq \, P(\Omega^*).$$

In particular the set $\operatorname{Int}\{w \leq 0\}$ contains Ω and satisfies $P(\operatorname{Int}\{w \leq 0\}) = P(\{w \leq 0\}) \leq P(\Omega^*)$, where we used (3.2.2). This implies in fact that $P(\operatorname{Int}\{w \leq 0\}) = P(\Omega^*)$, since by Theorem 3.13 the set Ω^* solves the least area problem with obstacle Ω . Since by the same result Ω^* is also strictly outward minimising, we deduce in turn $|(\operatorname{Int}\{w \leq 0\}) \setminus \Omega^*| = 0$. Since, by Theorem 3.14, the volume measure of $\partial \Omega^*$ is zero, we also have that the *open* set $(\operatorname{Int}\{w \leq 0\}) \setminus \overline{\Omega^*}$ is null, but by openness this is possible only if $\operatorname{Int}\{w \leq 0\} = \Omega^*$. \square

In [MRS19], the authors showed that under the assumptions of (ii) in Theorem 3.1, any bounded set with smooth boundary can be evolved through a proper weak solution to the Inverse Mean Curvature Flow. In particular from this fact and Proposition 3.21 we get that under these assumptions Ω^* satisfies the desired properties.

Corollary 3.22. Let (M, g) be a complete noncompact Riemannian manifold satisfying the assumptions of (ii) in Theorem 3.1. Then, for any open bounded Ω with smooth boundary, the set Ω^* is an open bounded maximal volume solution to least area problem with obstacle Ω .

Other aspects of strictly outward minimising sets and IMCF

Here, we observe that a *smooth* solution to the Inverse Mean Curvature Flow of a bounded open set Ω contained in a complete Riemannian manifolds never completely leaves Ω^* , if the latter is an open bounded maximal volume solution to the least area problem with obstacle Ω . In particular, this holds under the assumption of Theorem 3.1. This phenomenon is substantially a direct corollary of the outward minimising property of the hypersurfaces evolving by IMCF, that was observed in [HI01, Smooth Flow Lemma 2.3]. We add a proof of this fact, because it implies, together with the long time existence theory for strictly starshaped sets with smooth and strictly mean-convex boundary, that the latters are strictly outward minimising. A simple approximation argument involving the Mean Curvature Flow will also yield that any mean-convex strictly starshaped set is outward minimising.

Proposition 3.23 (No Escape from Ω^*). Let (M,g) be a complete noncompact Riemannian manifold satisfying one of the equivalent conditions in Theorem 3.13. Then, for $\Omega \subset M$ a bounded open set with smooth boundary, let $\{\partial \Omega_t\}_{t \in [0,T]}$ be evolving by IMCF with initial datum $\partial \Omega$. Then, if $\Omega^* \in \Omega_t$ for some $t \in (0,T]$, then Ω is strictly outward minimising and we have $\Omega = \Omega^*$.

Proof. Let $w: \Omega_T \to \mathbb{R}$ such that $\{w = t\} = \partial \Omega_t$, so that w classically satisfies the level set equation (3.3.6). Then

$$0 \leq \int_{\Omega^* \setminus \overline{\Omega}} |Dw| \, d\mu = \int_{\Omega^* \setminus \overline{\Omega}} \operatorname{div} \left(\frac{Dw}{|Dw|} \right) \, d\mu$$

$$= \int_{\partial^* \Omega^*} \left\langle \frac{Dw}{|Dw|} \middle| \nu_{\partial^* \Omega^*} \right\rangle \, d\sigma - \int_{\partial \Omega} \left\langle \frac{Dw}{|Dw|} \middle| \frac{Dw}{|Dw|} \right\rangle \, d\sigma$$

$$\leq |\partial \Omega^*| - |\partial \Omega| \leq 0,$$
(3.3.7)

where in the second equality $\nu_{\partial^*\Omega^*}$ is the measure theoretic unit normal to the reduced boundary $\partial^*\Omega^*$ and in the last inequality we used the Divergence Theorem for sets of finite perimeter coupled with $|\partial\Omega^*|=|\partial^*\Omega^*|$. In particular, (3.3.7) implies that $|\Omega^*\setminus\overline{\Omega}|=0$. By openness, that follows from (3.6), this implies that $\Omega=\Omega^*$ and in particular it is strictly outward minimising.

The celebrated result of Gerhardt [Ger90] and Urbas [Urb90], asserting that strictly starshaped shaped sets of \mathbb{R}^n with smooth and strictly mean-convex boundary admits an immortal solution to their Inverse Mean Curvature Flow immediately combines with Proposition 3.23 to show that these sets are in fact strictly outward minimising. Let us recall that in \mathbb{R}^n a bounded set Ω with smooth boundary is strictly starshaped with respect to some point $x_0 \in \Omega$ if and only

$$\langle x - x_0 | \nu \rangle > 0$$

for any $x \in \partial \Omega$, where ν is the outward unit normal to the boundary of Ω .

Corollary 3.24. Let $\Omega \subset \mathbb{R}^n$ be a bounded strictly starshaped set with smooth strictly mean-convex boundary. Then, Ω is strictly outward minimising.

A simple approximation argument based on the Mean Curvature Flow, that we actually already used in the proof of the Isoperimetric Inequality in Chapter 1, yields that the strict mean-convexity can be relaxed to mean-convexity, that is, the mean curvature H of

 $\partial\Omega$ is allowed to satisfy H=0 on some point of $\partial\Omega$. However, in this case, we just show that Ω is just outward minimising.

Proposition 3.25 (Starshaped mean-convex sets are outward minimising). *Let* $\Omega \subset \mathbb{R}^n$ *be a strictly* starshaped *bounded open set with smooth* mean-convex *boundary. Then* Ω *is outward minimising.*

Proof. Let $F_0: \partial\Omega \to \mathbb{R}^n$ be the immersion of $\partial\Omega$ in \mathbb{R}^n . Evolve this hypersurface by Mean Curvature Flow defining time dependent immersions $F: [0, \delta) \times \partial\Omega \to \mathbb{R}$ satisfying

$$\frac{\partial F}{\partial s}(s,x) = -\mathbf{H}(F(s,x))\,\nu(s,x), \qquad F(0,x) = F_0(x), \tag{3.3.8}$$

where ν is the exterior unit normal to the hypersurface $\partial\Omega_s$ given by the immersion $F(s,\cdot):\partial\Omega\to\mathbb{R}^n$, and H is its related mean curvature. The standard short-time existence theory for geometric evolution equations (see e.g. [HP99, Theorem 3.1]) ensures the existence of a $\delta>0$ such that a solution F to (3.3.8) is well-defined. In other words, we have defined a sequence of bounded open sets $\{\Omega_s\}_{s\in[0,\delta)}$ with smooth boundary evolving by Mean Curvature Flow (3.3.8). It is well known (see e.g. [HP99, Theorem 3.2]) that the mean curvature of these boundaries evolves by

$$\frac{\partial}{\partial s}H = \Delta H + H|h|^2,$$

where by h we denote the second fundamental form that, as the other quantities appearing in the equation above, is to be understood with respect to the evolving metric on $\partial\Omega_s$. In particular, the standard Maximum Principle for parabolic equations implies that the mean curvature of $\partial\Omega_s$ for $s\in(0,\delta)$ is strictly positive. Since, by the smoothness of the flow, the sets Ω_s are still strictly starshaped for small s, we can conclude by Corollary 3.24 that these approximating sets Ω_s are strictly outward minimising. Observe now that since the mean curvature H of the initial datum $\partial\Omega$ is nonnegative, the flow (3.3.8) is actually a shrinking flow, and thus $\Omega_s\subseteq\Omega\subseteq\Omega^*$. Then, by the minimising property of Ω_s we have $|\partial\Omega_s|\leq |\partial\Omega^*|$, that, upon letting $s\to0^+$, implies $|\partial\Omega|\leq |\partial\Omega^*|$. This means, by Remark 3.16, that Ω is outward minimising.

We conclude by pointing out that in literature there are many generalisations of Gerhardt-Urbas results in non-flat ambient manifolds, for a suitable notion of starshapedness. We mention in particular the work of Brendle-Hung-Wang [BHW16] that covers a considerable variety of warped product ambient manifolds and, outside rotationally symmetric ambients, the work of Pipoli [Pip16] in the Complex Hyperbolic Space. It is easy to see that, thanks to these results, the above argument shows that the theses of Corollary 3.24 and Proposition 3.25 hold in these ambient manifolds too.

3.4 Convergence of *p*-capacities to $|\partial \Omega^*|$

Aim of this section is to show that in the relevant classes of manifolds satisfying the assumptions of Theorem 3.1 we can recover the value of $|\partial\Omega^*|$ as limit for $p \to 1^+$ of the p-capacity of $\partial\Omega$, thus completing the proof of the latter Theorem. Before doing so, we recall some notation and the basic existence result of p-capacitary potentials.

3.4.1 p-nonparabolicity and p-capacitary potentials

Let (M, g) be a complete noncompact Riemannian manifold, and let $p \ge 1$. We define the variational p-capacity of a bounded set with smooth boundary $\Omega \subset M$ as

$$\operatorname{Cap}_p(\Omega) \, = \, \inf \left\{ \int_{\mathbb{R}^n} |\mathrm{D} f|^p \, \mathrm{d} \mu \, \left| \, f \geq \chi_\Omega, \, f \in \mathscr{C}_0^\infty(\mathbb{R}^n) \right. \right\}. \tag{3.4.1}$$

Let now p > 1. Then (M, g) is said to be p-nonparabolic if there exists a *positive* p-Green's function $G : (M \times M) \setminus \text{Diag}(M) \to \mathbb{R}$, that is, satisfying

$$\int_{M} \left\langle |D G_{p}(O, \cdot)|^{p-2} D G_{p}(O, \cdot) | D \varphi \right\rangle d\mu = \varphi(O)$$
(3.4.2)

for any $\varphi \in C_c^\infty(M)$. Relation (3.4.2) is the weak formulation of the equation $-\Delta_p G(O, \cdot) = \delta_O$, where δ_O is the Dirac delta centered at O. Moreover, when referring to the p-Green's function of a p-nonparabolic manifold we mean the minimal one.

One can show that if M is p-nonparabolic and $G_p \to 0$ at the infinity of any end then, for any open bounded $\Omega \subset M$ with smooth boundary there exists a unique weak solution $u_p \in W^{1,p}(M \setminus \overline{\Omega})$ to

$$\begin{cases}
\Delta_p u = 0 & \text{in } M \setminus \overline{\Omega}, \\
u = 1 & \text{on } \partial\Omega, \\
u(x) \to 0 & \text{as } d(O, x) \to \infty,
\end{cases}$$
(3.4.3)

where we can suppose $O \in \Omega$, generalising the correspondent existence theorem for harmonic functions on 2-nonparabolic manifolds Theorem 1.14 that we proved in Chapter 1. Moreover, the integral of $|\mathrm{D}u_p|^p$ on $M\setminus\overline{\Omega}$ realises the p-capacity. Since a complete and self-contained proof of this general result does not seem be easy to find in literature, we included a proof in Appendix B. Here, we just note that the self-contained argument used in Theorem 1.14, cannot evidently be directly exported to our case. Instead, an adaptation of the argument used for [CS03b] together with the deep $\mathscr{C}^{1,\alpha}$ -estimates holding true up to the boundary of [Lie88] work also in the general p-nonparabolic case.

Theorem 3.26. Let (M,g) be a complete noncompact p-nonparabolic Riemannian manifold. Let $\Omega \subset M$ be an open bounded set with smooth boundary, and let $O \in M$. Assume also that the p-Green's function G_p satisfies $G_p(O,x) \to 0$ as $d(O,x) \to \infty$. Then, there exists a weak solution u_p to (3.4.3). Moreover, it holds

$$\int_{M\setminus\overline{\Omega}} |\mathrm{D}u|^p \,\mathrm{d}\mu = \mathrm{Cap}_p(\Omega). \tag{3.4.4}$$

Remark 3.27. It is worth pointing out that, by [Hol99], the above general result fully describes the nonnegative Ricci curvature case in terms of growth of the volume of geodesic balls. Indeed, we know that if (M, g) has nonnegative Ricci curvature and

$$\int_{1}^{+\infty} \left(\frac{t}{|B(O,t)|}\right)^{1/(p-1)} dt < +\infty \tag{3.4.5}$$

for any $O \in M$, then [Hol99, Proposition 5.10] gives the p-nonparabolicity of (M, g), and the decay estimate for the positive p-Green's function G_p allows to conclude that $G_p \to 0$ at infinity. This is observed in [Bia+18, Corollary 2.6].

On the other hand, if the integral in (3.4.5) diverges, by [Hol99, Proposition 1.7] (M,g) is p-parabolic (actually regardless of curvature conditions), this in particular implies that $\operatorname{Cap}_p(\Omega) = 0$ (see e.g. [Hol99, (1.5)]) for any bounded $\Omega \subset M$ with smooth boundary and in particular the thesis of Theorem 3.26 cannot hold true.

We are finally in position to complete the proof of Theorem 3.1. Namely, we are going to show that under the assumptions of this theorem the variational p-capacity of Ω approximates the area of Ω^* . In the most important and original step in the proof (namely, Step~3 below), the arguments we use are different depending on the validity of assumption (i) or (ii) in Theorem 3.1. In presence of an Euclidean-like Isoperimetric Inequality, an argument inspired by [Xu96] allows to show that

$$\operatorname{Cap}_1(\Omega) \leq \operatorname{C}_{n,p} \operatorname{Cap}_p(\Omega)$$
,

for some constant $C_{n,p}$ fulfilling $C_{n,p} \to 1$ as $p \to 1^+$.

On the other hand, if (M,g) is a manifold with nonnegative Ricci curvature satisfying the superlinear uniform volume growth condition, we are still able to prove (3.1.3) by exploiting a decay estimate of the p-Green's function of (M,g) with an explicit dependence on p. This estimate originated in [Hol99, Proposition 5.7], where it was proved for any point on the boundary of an end, and it has been applied in [MRS19] together with the assumed (3.1.2) to obtain an analogous inequality holding true on any point outside some compact set (see [MRS19, Theorem 3.8]). In this regard, we observe that being able to work out Holopainen's inequality without the restriction of lying in the boundary of an end would allow to relax the assumption (3.1.2) in [MRS19], and consequently in the present theory, to

$$\int_{1}^{+\infty} \frac{t}{|B(O,t)|} \, \mathrm{d}t < +\infty,\tag{3.4.6}$$

that is, roughly speaking, a strictly superlinear volume growth assumption.

Remark 3.28. We point out that in [MRS19] the authors actually assumed

$$\frac{|B(O,t)|}{|B(O,s)|} \ge C\left(\frac{t}{s}\right)^b \tag{3.4.7}$$

for any $t \ge s > 0$ and some constant C > 0 to provide the inequality described above. In turn, (3.4.7), that is implied by (3.1.2), is used to invoke the technical [Min09b, Proposition 2.8], allowing to control that the size of the bounded components of the complement of $M \setminus B(O, R)$ for big R > 0 are not too big. In fact, the desired decay estimate on the p-Green's function, and in turn, the validity of Theorem 3.1 is ensured on any Riemannian manifold with nonnegative Ricci curvature satisfying (3.4.6) such that for any R big enough $M \setminus B(O, R)$ does not have bounded components.

Proof of Theorem 3.1. Let (M,g) be a complete noncompact Riemannian manifold satisfying the lower bound (1.2.19) on the Ricci curvature. Then, if (i) or (ii) hold, then by Corollary 3.19 and Corollary 3.22 respectively, for any bounded Ω with finite perimeter the strictly outward minimising hull Ω^* is an open bounded maximal volume solution to the least area problem with obstacle Ω .

We aim at proving that

$$|\partial\Omega^*| \leq \operatorname{Cap}_1(\Omega) \leq \liminf_{p \to 1^+} \operatorname{Cap}_p(\Omega) \leq \limsup_{p \to 1^+} \operatorname{Cap}_p(\Omega) \leq |\partial\Omega^*|. \tag{3.4.8}$$

We divide the proofs in several steps.

Step 1. The first and easiest inequality to show in (3.4.8) is

$$|\partial \Omega^*| \le \operatorname{Cap}_1(\Omega). \tag{3.4.9}$$

It suffices to observe that for any $f \in \mathscr{C}_c^{\infty}(M)$ with $f \geq \chi_{\Omega}$ we have, by co-area formula,

$$\int_{M} |Df| d\mu \ge \int_{0}^{1} |\{f = t\}| dt \ge \inf \{|\partial E| \mid \Omega \subset E, \partial E \operatorname{smooth}\} \ge |\partial \Omega^{*}|.$$

In particular, taking the infimum over any such f, we get (3.4.9).

Step 2. Now, we prove

$$\limsup_{p\to 1^+} \operatorname{Cap}_p(\Omega) \le |\partial\Omega^*|. \tag{3.4.10}$$

Let *E* be any open and bounded set in *M* with smooth boundary such that $\Omega \subset E$. Define, for $x \in M$, the function $d_E(x) = \operatorname{dist}(x, E)$. Moreover, let us introduce a smooth cut-off function χ_{ε} fulfilling

$$\begin{cases} \chi_{\varepsilon}(t) = 1 & \text{in } t < \varepsilon, \\ -\frac{1}{\varepsilon} < \dot{\chi}(t) < 0 & \text{in } \varepsilon \le t \le 2\varepsilon \\ \chi_{\varepsilon}(t) = 0 & \text{in } t > 2\varepsilon, \end{cases}$$
 (3.4.11)

and let us set $\eta_{\varepsilon}(x) = \chi_{\varepsilon}(d_{\varepsilon}(x))$. Choosing ε small enough, it is easily seen, by the regularity of d_{ε} in a neighbourhood of ε (first observed for \mathbb{R}^n in [GT01, Lemma 14.6], see [Man, Proposition 5.17] for a self contained proof in a general ambient manifold), that the function η_{ε} is an admissible competitor in (3.4.1). Then,

$$\operatorname{Cap}_p(\Omega) \leq \int\limits_M |\mathrm{D}\eta_{\varepsilon}|^p \,\mathrm{d}\mu$$

for any $p \ge 1$. Letting $p \to 1^+$, we get

$$\limsup_{p\to 1^+}\operatorname{Cap}_p(\Omega) \leq \int\limits_M |\mathrm{D}\eta_\varepsilon|\,\mathrm{d}\mu = \int\limits_\varepsilon^{2\varepsilon} |\dot{\chi_\varepsilon}(t)|\,|\{d_E=t\}|\,\,\mathrm{d}t,$$

where in the last equality we applied the coarea formula combined with the fact that $|Dd_E| = 1$ in a neighbourhood of E. By the Mean Value Theorem, there exist $r_{\varepsilon} \in (\varepsilon, 2\varepsilon)$ such that the above right hand side satisfies

$$\int_{\varepsilon}^{2\varepsilon} |\dot{\chi}_{\varepsilon}(t)| \left| \left\{ d_{E} = t \right\} \right| \, \mathrm{d}t = \varepsilon |\dot{\chi}_{\varepsilon}(r_{\varepsilon})| \left| \left\{ d_{E} = r_{\varepsilon} \right\} \right| < \left| \left\{ d_{E} = r_{\varepsilon} \right\} \right|,$$

where the last inequality is due to the second condition in (3.4.11). Since, as $r_{\varepsilon} \to 0^+$, we clearly have

$$|\{d_E=r_{\varepsilon}\}| \to |\partial E|,$$

we conclude that

$$\limsup_{p\to 1^+}\operatorname{Cap}_p(\Omega)\leq |\partial E|$$

for any bounded open set E with smooth boundary containing Ω . In particular, considering a sequence of bounded sets Ω_k with smooth boundary containing Ω^* and with

 $|\partial\Omega_k| \to |\partial\Omega^*|$ as $k \to \infty$, provided in Lemma 3.17, we get (3.4.10).

Step 3. The most involved step in the proof of (3.4.8) is inequality

$$\operatorname{Cap}_1(\Omega) \leq \liminf_{p \to 1^+} \operatorname{Cap}_p(\Omega). \tag{3.4.12}$$

To do this, we treat separately the cases when (M, g) satisfies an Euclidean-like Isoperimetric Inequality and the case where (M, g) has nonnegative Ricci curvature and satisfies (3.1.2).

Case 1. (M,g) satisfies an Euclidean-like Isoperimetric Inequality. Let C_{Sob} be the L^1 -Sobolev constant, that is

$$\left(\int_{M} f^{n/(n-1)} \,\mathrm{d}\mu\right)^{(n-1)/n} \le C_{\text{Sob}} \int_{M} |\nabla f| \,\mathrm{d}\mu \tag{3.4.13}$$

for any nonnegative $f \in \mathscr{C}^\infty_0(M)$. It is well known and easy to check that applying (3.4.13) to $f^{\frac{n-1}{n-p}p}$ yields the L^p -Sobolev inequality

$$\left(\int_{M} f^{p^*} d\mu\right)^{(n-1)/n} \le C_p \int_{M} |Df|^p d\mu \tag{3.4.14}$$

with

$$C_{n,p} = \left[\frac{C_{\text{Sob}}(n-1)p}{(n-p)}\right],\tag{3.4.15}$$

and

$$p^* = \frac{np}{n-p}$$

for any 1 . In particular, by [PST14, Theorem 3.2], <math>(M, g) is p-nonparabolic for p in this range The argument yielding the key estimate (3.4.18) below is inspired by the proof of [Xu96, Theorem 3.2]. By the definition of 1-capacity and Hölder inequality we get, for any q > 0,

$$\operatorname{Cap}_{1}(\Omega) \leq \int_{M} |Df^{q}| \, \mathrm{d}\mu = q \int_{M} f^{q-1} |Df| \, \mathrm{d}\mu \leq q \left(\int_{M} f^{(q-1)\frac{p}{p-1}} \, \mathrm{d}\mu \right)^{(p-1)/p} \left(\int_{M} |Df|^{p} \, \mathrm{d}\mu \right)^{1/p}. \tag{3.4.16}$$

Choose then

$$q_p = 1 + p^* \frac{(p-1)}{p} \tag{3.4.17}$$

and observe that $q_p > 1$. Then, applying the L^p -Sobolev inequality (3.4.14) we obtain

$$\operatorname{Cap}_{1}(\Omega) \leq q_{p} \operatorname{C}_{n,p}^{(p-1)/p} \left(\int_{M} |\operatorname{D} f|^{p} d\mu \right)^{p^{*}(p-1)/p^{2}+1/p} = q_{p} \operatorname{C}_{n,p}^{(p-1)/p} \left(\int_{M} |\operatorname{D} f|^{p} d\mu \right)^{(n-1)/(n-p)}.$$

Taking the infimum in the rightmost hand side of the inequality above over any $f \in \mathscr{C}^\infty_c(M)$ we are left with

$$\operatorname{Cap}_{1}(\Omega) \leq q_{p} \operatorname{C}_{n,p}^{(p-1)/p} \operatorname{Cap}_{p}(\Omega), \tag{3.4.18}$$

Letting $p \to 1^+$ in the above inequality, and observing that from the expressions (3.4.15)

and (3.4.17) both q_p and $C_{n,p}$ tend to 1, we get (3.4.12) under the validity of an Euclidean-like Isoperimetric Inequality.

Case 2. (M, g) has nonnegative Ricci curvature and satisfies (3.1.2). The superlinear volume growth following from (3.1.2) guarantees that Holopainen's condition

$$\int_{1}^{+\infty} \frac{t}{|B(O,t)|} \, \mathrm{d}t < +\infty$$

is satisfied, and in particular his [Hol99, Theorem 5.10] guarantees that (M, g) is p-nonparabolic for $1 . Moreover, [MRS19, Theorem 3.8] gives the following decay estimate for <math>G_p$

$$G_p(O,x) \le \frac{C^{1/(p-1)}}{(p-1)^2} \int_{d(O,x)}^{\infty} \left[\frac{t}{|B(O,t)|} \right]^{1/(p-1)} dt.$$

for any $x \in M \setminus B(O,R_1)$ for some $R_1 > 0$. This estimate in particular allows to apply Theorem 3.26 that yields a p-harmonic function $u_p \in W^{1,p}(\mathbb{R}^n \setminus \overline{\Omega})$ realising (3.4.4). We claim that the same type of estimate holds for u_p . Indeed, choosing $O \in \Omega$, a trivial comparison argument immediately yields $u_p(x) \leq G_p(O,x)/(\inf_{\partial\Omega}G_p)$. Moreover, it follows from the convergence results in [MRS19] that the function $-(p-1)\log G_p(O,\cdot)$ converges locally uniformly to a continuous function (more precisely to a solution of the IMCF with initial condition O) and in particular $-(p-1)\log G_p(O,\cdot)$ is uniformly bounded in the compact set $\partial\Omega$ uniformly in p small enough. This implies that $(\inf_{\partial\Omega}G_p) \geq C_2^{1/(p-1)}$ for some C_2 independent on p. Combining it with the above comparison we thus obtain

$$u_p(x) \le \frac{C_3^{1/(p-1)}}{(p-1)^2} \int_{d(O,x)}^{\infty} \left[\frac{t}{|B(O,t)|} \right]^{1/(p-1)} dt.$$

outside some ball. The volume growth condition $|B(O,t)| \ge C_{\text{vol}} t^b$, following from (3.1.2), improves the above estimate to

$$u_p(O,x) \le \frac{1}{(b-p)} \frac{C_4^{1/(p-1)}}{(p-1)} r(x)^{-(b-p)/(p-1)},$$
 (3.4.19)

where we denoted r(x) = d(O, x), outside some big ball and for a positive constants C_4 uniform in p as $p \to 1^+$.

Extend now u_p to be equal to u on $M \setminus \Omega$, and equal to 1 on Ω , for simplicity with the same name. We claim that $u_p^{q_p}$ is in $W^{1,1}(M)$, for p close enough to 1, where q_p is defined in (3.4.17). Clearly, (3.4.19) implies that $u_p^{q_p} \in L^1(M)$ for p close enough to 1, and also so is u_p . Applying Hölder inequality as in (3.4.16) we also get

$$\int_{M} |\mathrm{D}u_{p}^{q_{p}}| \,\mathrm{d}\mu \leq q_{p} \left(\int_{M} u_{p}^{p^{*}} \,\mathrm{d}\mu \right)^{(p-1)/p} \left(\int_{M \setminus \overline{\Omega}} |\mathrm{D}u_{p}|^{p} \,\mathrm{d}\mu \right)^{1/p} \\
= q_{p} \left(\int_{M} u_{p}^{p^{*}} \,\mathrm{d}\mu \right)^{(p-1)/p} \operatorname{Cap}_{p}(\Omega)^{1/p} , \qquad (3.4.20)$$

$$\leq q_{p} \left(\int_{M} u_{p} \,\mathrm{d}\mu \right)^{(p-1)/p} \operatorname{Cap}_{p}(\Omega)^{1/p} , \qquad (3.4.20)$$

where the equality is (3.4.4), and the last inequality is due to $u_p \le u_p^{p*}$ following from $0 < u \le 1$ and $p^* > 1$. Since $u_p \in L^1(M)$ uniformly in p close to 1, and so does the p-capacity of Ω as $p \to 1^+$ by (3.4.10) in $Step\ 2$, (3.4.20) implies that $u_p^{q_p} \in W^{1,1}(M)$

uniformly as $p \to 1^+$. Since the class of competitors for $\operatorname{Cap}_1(\Omega)$ can be easily relaxed to the class of $W^{1,1}(M)$ without changing the infimum (see e.g. [HKM06, Chapter 2]), we can estimate $\operatorname{Cap}_1(\Omega)$ from above with $u_p^{q_p}$ and get, from (3.4.20), that

$$\operatorname{Cap}_{1}(\Omega) \leq q_{p} \left(\int_{M} u_{p} \, \mathrm{d}\mu \right)^{(p-1)/p} \operatorname{Cap}_{p}(\Omega)^{1/p}. \tag{3.4.21}$$

We focus on the first round bracket in the rightmost hand side of (3.4.21). Let us decompose M in $\overline{B(O,R)} \cup (M \setminus \overline{B(O,R)})$. We obtain, with the aid of (3.4.19),

$$\int_{M} u_{p} \, \mathrm{d}\mu \leq \left| \overline{B(O,R)} \right| + \frac{1}{(b-p)} \frac{C_{4}^{1/(p-1)}}{(p-1)} \int_{M \setminus \overline{B(O,R)}} r^{-(b-p)/(p-1)} \, \mathrm{d}\mu. \tag{3.4.22}$$

We estimate the integral in the right had side above as follows. We have

$$\int_{M\backslash \overline{B(O,R)}} r^{-(b-p)/(p-1)} d\mu = \sum_{j=1}^{+\infty} \int_{B(O,2jR)\backslash \overline{B(O,jR)}} r^{-(b-p)/(p-1)} d\mu$$

$$\leq \sum_{j=1}^{+\infty} (jR)^{-[(b-p)/(p-1)]} \Big[|B(O,2jR)| - |B(O,jR)| \Big]$$

$$\leq |\mathbb{B}^{n}| 2^{n} R^{n-(b-p)/(p-1)} \sum_{j=1}^{+\infty} j^{n-[(b-p)/(p-1)]} \leq C_{5} R^{n-[(b-p)/(p-1)]},$$
(3.4.23)

where we used Bishop-Gromov to estimate $|B(O,2jR)| \le |\mathbb{B}^n|(2jR)^n$, and, in the last step, the convergence of the series, that holds true for p close to 1 with a value that is uniform in p. Resuming, we have, plugging the outcome of (3.4.23) into (3.4.22),

$$\int_{M} u_{p} \, \mathrm{d}\mu \leq \left| \overline{B(O,R)} \right| + R^{n} \frac{1}{(b-p)(p-1)} \left[\frac{C_{6}}{R^{(b-p)}} \right]^{1/(p-1)}$$

for any R big enough. Choose then $R^{b-p} > C_6$ for any p close enough to 1 to see that, with this choice, the second summand in the above right hand side vanishes in the limit as $p \to 1^+$. In particular, letting $p \to 1^+$ in (3.4.21) we infer

$$\operatorname{Cap}_1(\Omega) \leq \liminf_{p \to 1^+} \operatorname{Cap}_p(\Omega),$$

also under the assumption of nonnegative Ricci curvature with uniform volume growth (3.1.2).

Arranging (3.4.9), (3.4.10) and (3.4.12) into (3.4.8) completes the computation of the limit of $\operatorname{Cap}_n(\Omega)$ as $p \to 1^+$.

In light of Theorem 3.1, one could wonder whether the existence of a well posed (in the sense of Theorem 3.13) strictly outward minimising hull implies a p-capacitary approximation of $|\partial\Omega^*|$ in the sense of (3.1.3). The following example, inspired in part by [KN09, Section 4], provides a manifold admitting an exhausting sequence of strictly outward minimising sets but such that $\operatorname{Cap}_p(\Omega)=0$ for any $\Omega\subset M$ with smooth boundary. *Example* 3.29. Consider, for $n\geq 2$ the complete noncompact Riemannian manifold (M,g) whose metric splits as $g=\operatorname{d}\rho\otimes\operatorname{d}\rho+\operatorname{tanh}^2(\rho)g_{\mathbb{S}^{n-1}}$ on $[0,+\infty)\times\mathbb{S}^{n-1}$. For n=2, this is

the celebrated Hamilton's cigar soliton [Ham88]. The Riemannian manifold (M, g) has linear volume growth and nonnegative Ricci curvature. In particular,

$$\int_{1}^{+\infty} \frac{t}{|B(O,t)|} \, \mathrm{d}t = +\infty,$$

and by [Hol99, Proposition 1.7] (M,g) is p-parabolic for any p>1. In particular, it is well known (see e.g. [Hol99, (1.5)]) that $\operatorname{Cap}_p(\Omega)=0$ for any open bounded $\Omega\subset M$ with smooth boundary. However, the level sets of ρ are strictly outward minimising, and so they provide the exhausting sequence (in fact a foliation) required in Theorem 3.13 for the well-posedness of Ω^* . To check this last assertion, we invoke again the level set formulation of the Inverse Mean Curvature Flow, and the minimising properties of such level sets discovered in [HI01]. Let indeed $B=\{\rho<1\}$, and consider on $M\setminus B$ the function

 $w = (n-1)\log\left(\frac{\tanh\rho}{\tanh 1}\right)$

extended with continuity at 0 on B. The function w is immediately seen to satisfy the level set equation (3.3.6) in $M\setminus \overline{B}$. Since the level set of w sweep out the whole M as $\rho\to +\infty$, [HI01, Smooth Flow Lemma 2.3] implies that w is also a Weak Inverse Mean Curvature Flow in the sense recalled in Subsection 3.3.2. Observe also that w is proper, although $w\to +\infty$ as $\rho\to +\infty$ (compare with Remark 3.20). In particular by [HI01, Minimizing Hull Property 1.4, (ii)], the bounded sets $\mathrm{Int}\{w\le t\}=\{w< t\}$ are strictly outward minimising for any $t\ge 0$. Theorem 3.13 applies and yields a well posed strictly outward minimising hull to any bounded open set $\Omega\subset M$ with smooth boundary, but its area cannot be recovered by p-capacities, that vanish for any p>1.

Appendix A

Comparison with Colding's monotonicity formulas

In this section, we provide a comparison between our monotonicity formulas and those obtained by Colding and by Colding-Minicozzi in [Col12] and [CM13], respectively. To start with, let u be a solution of (8) in a nonparabolic Riemannian manifold (M, g) with Ric ≥ 0 , for a bounded subset $\Omega \subset M$ with smooth boundary, and set

$$b = u^{-\frac{1}{n-2}}. (A.1)$$

Note that b=1 on $\partial\Omega$ and that $b\to +\infty$ at infinity. Associated with the level sets of b, consider the family of functions $\{A_{\beta}\}$, where $A_{\beta}:[1,+\infty)\to [0,+\infty)$ is defined for every $\beta\geq 0$ as

$$A_{\beta}(r) = \frac{1}{r^{n-1}} \int_{\{b=r\}} |\mathrm{D}b|^{\beta+1} \,\mathrm{d}\sigma.$$

Now, replacing u be a minimal Green's function $G(O,\cdot)$, for some pole $O \in M$, the above defined A_{β} is exactly the quantity considered in [CM13, formula (1.1)]. Note that in Colding's setting, the level sets $\{b=r\}$ are considered for every r>0, since $b(q)\to 0$ as $d(O,q)\to 0$.

Our aim is to see how the monotonicity of our functions Φ_{β} translates in terms of that of A_{β} . First of all, it is straightforward from (1.3.9)–(1.3.11) and (A.1) that

$$b = e^{\frac{\varphi}{n-2}}, \qquad |Db| = \frac{|\nabla \varphi|_{\tilde{g}}}{n-2}, \qquad d\sigma = b^{n-1} d\sigma_{\tilde{g}}, \qquad (A.2)$$

and in turn that

$$\Phi_{\beta}(s) = (n-2)^{\beta+1} A_{\beta}(e^{\frac{s}{n-2}}), \quad \text{for every } s \ge 0.$$
(A.3)

We look at the derivative (1.3.22) of Φ_{β} and at its equivalent expression (1.3.23). In particular, the volume integral (1.3.23) contains the following terms.

$$\operatorname{Ric}(\nabla \varphi, \nabla \varphi) = \left(\frac{n-2}{2}\right)^2 \operatorname{Ric}(\mathrm{D}b^2, \mathrm{D}b^2), \tag{A.4}$$

and

$$\begin{split} \left| \nabla \nabla \varphi \right|_{\tilde{g}}^{2} + (\beta - 2) \left| \nabla |\nabla \varphi|_{\tilde{g}} \right|_{\tilde{g}}^{2} &= \left(\frac{n - 2}{2} \right)^{2} \left\{ \left| \mathrm{DD} \, b^{2} - \frac{\Delta b^{2}}{n} \, g \right|^{2} + (\beta - 2) \left| \mathrm{D}^{\top} |\mathrm{D} b| \right|^{2} + (\beta - 2) \left| \mathrm{D} b^{2} \right|^{2} \left[\mathrm{H} - (n - 1) \left| \mathrm{D} \log b \right| \right]_{A}^{2} \right\}.5) \end{split}$$

which have been expressed in terms of the function b and of the metric g via some computations (compare with the proof of (1.3.4)). Differentiating both sides of (A.3) and writing expression (1.3.23) in terms of b and g through formulas (A.2), (A.4) and (A.5), we obtain

$$\frac{dA_{\beta}}{dr}(r) = \frac{(n-2)^{-\beta}}{r} \frac{d\Phi_{\beta}}{ds} ((n-2)\log r)
= -\frac{\beta}{4} r^{n-3} \int_{\{b>r\}} |Db|^{\beta-2} \left\{ \operatorname{Ric}(Db^{2}, Db^{2}) + \left| DDb^{2} - \frac{\Delta b^{2}}{n} g \right|^{2} \right.
+ (\beta - 2) \left| D^{\top} |Db| \right|^{2}
+ (\beta - 2) \left| Db^{2} \right|^{2} \left[H - (n-1) |D\log b| \right]^{2} \right\} b^{2-2n} d\mu \leq 0.$$

Setting b=2 in the above formula, we obtain exactly the integrand of the right hand side of [Col12, (2.106)], that, arguing as in the conclusion of the present Theorem 1.25, leads to the monotonicity of A_2 . For a general $\beta \geq (n-2)/(n-1)$, in [CM14b, Theorem 1.3] the monotonicity of A_β is inferred grouping the terms in (A.6) in a different way. Observe indeed that for $\beta < 2$ the volume integral in (A.6) does not evidently carry a sign. On the other hand, (A.5) combined with Kato's inequality immediately show the nonnegativity of the expression.

We close this appendix by showing how our methods can be applied also to obtain the Monotonicity-Rigidity Theorem for the Green's function, obtaining a new (conformal) proof of Colding-Minicozzi's [CM14b, Theorem 1.3]. Indeed, let (M,g) be a nonparabolic Riemannian manifold with Ric ≥ 0 , let G be its minimal Green's function and consider the new metric on $M \setminus \{O\}$

$$\tilde{g} = G(O, \cdot)^{\frac{2}{n-2}}g,$$

for some point $O \in M$. Set

$$\varphi = -\log G(O, \cdot).$$

Then, we have that the triple M, \tilde{g} , φ satisfies the system

$$\begin{cases} \Delta_{\tilde{g}}\varphi = 0 & \text{in } M \setminus \{O\} \\ \operatorname{Ric}_{\tilde{g}} - \nabla \nabla \varphi + \frac{d\varphi \otimes d\varphi}{n-2} = \frac{|\nabla \varphi|^2_{\tilde{g}}}{n-2} \tilde{g} + \operatorname{Ric} & \text{in } M \setminus \{O\} \\ \varphi(y) \to +\infty & \text{as } d(O,y) \to +\infty \\ \varphi(y) \to -\infty & \text{as } d(O,y) \to 0. \end{cases}$$

We denote by d the distance with respect to g. Define the function $\Phi_{\beta} : \mathbb{R} \to \mathbb{R}$ given by

$$\Phi_{eta}(s) = \int\limits_{\{arphi=s\}} |
abla arphi|_{ ilde{g}}^{eta+1} \, \mathrm{d}\sigma_{ ilde{g}}.$$

All the theory developed in Section 3 holds with trivial modification for Φ_{β} as above, and immediately yields a conformal Monotonicity-Rigidity Theorem for the Green's function.

Theorem A.7. Let (M, g) be a nonparabolic Riemannian manifold with Ric ≥ 0 . Let G be its minimal Green's function. Then, with the notations above, we have

$$\frac{\mathrm{d}\Phi_{\beta}}{\mathrm{d}s}(s) \ = - \ \beta \ \mathrm{e}^{s} \int\limits_{\{\varphi \geq s\}} \frac{\left|\nabla\varphi\right|_{\tilde{g}}^{\beta-2} \left(\left. \mathrm{Ric}(\nabla\varphi, \nabla\varphi) + \left|\nabla\nabla\varphi\right|_{\tilde{g}}^{2} + \left. (\beta-2) \left|\nabla|\nabla\varphi\right|_{\tilde{g}}^{2} \right) \right.}{\mathrm{e}^{\varphi}} \ \mathrm{d}\mu_{\tilde{g}}$$

In particular, Φ'_{β} is alway nonpositive. Moreover, $(d\Phi_{\beta}/ds)(s_0) = 0$ for some $s_0 \in \mathbb{R}$ and some $\beta \geq (n-2)/(n-1)$ if and only if $\{\varphi \geq s_0\}$ is isometric to the Riemannian product $([s_0,\infty)\times \{\varphi=s_0\}, d\rho\otimes d\rho + \tilde{g}_{\{|\varphi=s_0\}})$.

The above Theorem clearly translates in terms of (M, g) and G, exactly as Theorem 1.19 was deduced from Theorem 1.25.

Appendix B

Existence of p-capacitary potentials

The following comprehensive statement contains both the thesis of the existence theorem for the p-capacitary potential on p-nonparabolic complete noncompact Riemannian manifolds recalled in Theorem 3.26 as well as the identification of the variational p-capacity of a bounded set with smooth boundary with a boundary integral. These properties hold in particular on \mathbb{R}^n and they were largely employed in Chapter 2. In particular, we furnish a proof of Theorem 3.26 and Lemma 2.4. Despite seemingly well known, it is not easy to find in literature a complete and in particular self-contained proof of these basic facts. Anyway, no original arguments appear below, and we are collecting stuff mainly from [Hol90], [HKM06] and [PST14].

Theorem B.1. Let (M,g) be a complete noncompact p-nonparabolic Riemannian manifold. Let $\Omega \subset M$ be an open bounded set with smooth boundary, and let $O \in M$. Assume also that the p-Green's function G_p satisfies $G_p(O,x) \to 0$ as $d(O,x) \to \infty$. Then, there exists a unique weak solution u_p to (3.4.3) attaining smoothly the boundary value on $\partial\Omega$. Moreover, it realises

$$\operatorname{Cap}_{p}(\Omega) = \int_{M \setminus \overline{\Omega}} |\operatorname{D} u|^{p} \, \mathrm{d}\mu = \int_{\{u_{p} = t\}} |\operatorname{D} u|^{p-1} \, \mathrm{d}\sigma \tag{B.2}$$

for almost any $t \in (0,1]$ including any regular value t. In particular, (B.2) holds at t=1, that is

$$\operatorname{Cap}_{p}(\Omega) = \int_{\partial \Omega} |\operatorname{D} u|^{p-1} \, \mathrm{d}\sigma \tag{B.3}$$

Proof. Let B(O,R) be a geodesic ball containing Ω . Let $\psi \in \mathscr{C}_c^{\infty}(B(O,R))$ satisfy $\psi = 1$ on Ω . Then, we can find a solution to the Dirichlet problem for the p-laplacian with boundary values 1 on $\partial\Omega$ and 0 on $\partial B(O,R)$, namely a weakly p-harmonic function u_R on $B(O,R)\setminus\Omega$ satisfying $u-\psi\in W_0^{1,p}(B(O,R)\setminus\overline{\Omega})$. It is immediately deduced from Tolksdorf's Comparison Principle for p-harmonic functions (see for example [HKM06, Lemma 3.18] that if \tilde{u} is another such function relative to another $\tilde{\psi}\in W_0^{1,p}(B(O,R))$ then $u=\tilde{u}$ on $B(O,R)\setminus\overline{\Omega}$. Defining the p-capacity of Ω relative to B(O,R) as

$$\operatorname{Cap}_{p}(\Omega, B(O, R)) = \inf \left\{ \int_{M} |\mathrm{D}f|^{p} \, \mathrm{d}\mu \, \left| \, f \geq \chi_{\Omega}, \, f \in \mathscr{C}^{\infty}_{c}(B(O, R)) \right. \right\}, \tag{B.4}$$

we immediately see through a very standard argument that u_R can be approximated in $W^{1,p}$ by a sequence of admissible competitors in (B.4), and thus

$$\operatorname{Cap}_{p}(\Omega, B(O, R)) \leq \int_{B(O, R)\setminus \overline{\Omega}} |\operatorname{D} u_{R}|^{p} d\mu.$$
(B.5)

On the other hand, observing that in the definition of weakly *p*-harmonic functions on $B(O,R)\setminus\overline{\Omega}$ the class of tests can easily be relaxed to $W_0^{1,p}(B(O,R)\setminus\overline{\Omega})$, we get

$$\int_{B(O,R)\setminus\overline{\Omega}} |\mathrm{D}u_R|^p \,\mathrm{d}\mu = \int_{B(O,R)\setminus\overline{\Omega}} \left\langle |\mathrm{D}u_R|^{p-2} \mathrm{D}u_R \,|\, \mathrm{D}u_R \right\rangle \,\mathrm{d}\mu = \int_{B(O,R)\setminus\overline{\Omega}} \left\langle |\mathrm{D}u_R|^{p-2} \mathrm{D}u_R \,|\, \mathrm{D}\psi \right\rangle \,\mathrm{d}\mu,$$
(B.6)

where the last equality is deduced from the definition of weak p-harmonicity with $u - \psi \in W_0^{1,p}(B(O,R) \setminus \overline{\Omega})$ as test function. Applying the Hölder inequality to the right hand side of (B.6), we are immediately left with

$$\int_{B(O,R)\setminus\overline{\Omega}} |Du_R|^p d\mu \le \int_{B(O,R)\setminus\overline{\Omega}} |D\psi|^p d\mu$$
(B.7)

for any $\psi \in \mathscr{C}^{\infty}_{c}(B(O,R))$ satisfying $\psi = 1$ on Ω . Since, clearly, a sequence of functions satisfying the same assumptions on ψ suffices to realise the relative p-capacity of Ω (see for example [Maz11, (2.2.1)] for details), we get passing to the limit through such a sequence in (B.7) that we can also take the opposite inequality sign in (B.5), obtaining

$$\operatorname{Cap}_{p}(\Omega, B(O, R)) = \int_{B(O, R)\setminus \overline{\Omega}} |\operatorname{D}u_{R}|^{p} \, \mathrm{d}\mu. \tag{B.8}$$

Let us finally show that passing to the limit as $R \to +\infty$ yields a solution to (3.4.3). In what follows, we are frequently extending u_R to 0 outside B(O,R) without explicitly mentioning it The deep $\mathscr{C}^{1,\alpha}_{loc}$ -estimates for p-harmonic functions give, for any compact set K of $M \setminus \Omega$, a constant C so that

$$|u_R(x) - u_R(y)| \le C d(x, y)$$

and

$$\Big||\mathrm{D}u_R|(x)-|\mathrm{D}u_R|(y)\Big|\leq \mathrm{C}\,d(x,y).$$

The constant C depends on the dimension, on p, on smooth quantities related to the underlying metric g on K and on the $W^{1,p}$ norm of u_R on K. By the Maximum Principle for p-harmonic functions, $0 \le u_R \le 1$ on K. Moreover, by (B.8) and the obvious monotonicity of the $\operatorname{Cap}_p(\Omega, B(O, R))$ as R increases, we have

$$\int_{K} |Du_{R}|^{p} d\mu \le \int_{B(O,R_{0})} |Du_{R_{0}}|^{p} d\mu$$
(B.9)

for any $R_0 > 0$ such that $K \in B(O, R_0)$ and any $R > R_0$. In particular, the constant C is uniform in R, and by Arzelà-Ascoli applied to the sequence of u_R and the sequence of the gradients we deduce that, up to a subsequence, the sequence u_R converges in \mathscr{C}^1_{loc} to a \mathscr{C}^1 function u. Moreover, as it is immediately seen from the weak formulation of p-harmonicity, such u is p-harmonic. Observe that, since the $\mathscr{C}^{1,\alpha}_{loc}$ -estimates of [Lie88] are valid up to the boundary, K was allowed to contain portions of $\partial\Omega$, and thus the convergence takes place also on $\partial\Omega$, where then the sequence obviously converges to 1. Moreover, since by Tolksdorf's Hopf lemma for p-harmonic functions |Du| > 0 on $\partial\Omega$, the continuity up to the boundary of the gradient ensures that |Du| > 0 in a neighbourhood of $\partial\Omega$, where thus the solution is a smooth classical solution by quasilinear

elliptic regularity, and in particular it attains smoothly the datum on $\partial\Omega$. To complete the existence part of the proof, we just have to check that u vanishes at infinity. This is immediately deduced from the vanishing at infinity of G_p . Indeed, since G_p is positive, a straightforward comparison argument for the approximators u_R yields a constant C independent of R so that $0 < u_R \le C G_p(O, \cdot)$, that by the convergence shown above, implies, sending $R \to +\infty$,

$$0 < u_R \le C G_p(O, \cdot). \tag{B.10}$$

This last estimate clearly implies the vanishing at infinity of *u*.

Briefly, we show uniqueness. Let v be any other solution to (3.4.3), and let $k \in \mathbb{N}$. Let R_k be such that $u \leq v+1/k$ on (a smoothed out approximation of) $\partial B(O,R_k)$. Such a radius surely exists by the vanishing at infinity of u. Then, since u and v achieve the same value on $\partial \Omega$, the Comparison Principle for p-harmonic functions applied to u and v+1/k on $B(O,R_k)\setminus\overline{\Omega}$ shows that $u\leq v+1/k$ on this set. Letting $k\to +\infty$ we obtain $u\leq v$ on $M\setminus\Omega$. Exchanging the roles of u and v gives the opposite inequality, showing that u=v, that is uniqueness.

Now, we check that u realises the p-capacity of Ω . This again will come from the properties of the approximators. Since $u_R \to u$ as $R \to \infty$ pointwise, and $\int_K |\mathrm{D} u_R|^p \,\mathrm{d} \mu$ is uniformly bounded in R for any compact $K \subseteq M \setminus \Omega$ by (B.9), we can invoke the basic but very useful [HKM06, Lemma 1.33] to infer that $u \in L^p(M \setminus \overline{\Omega})$ and that $\mathrm{D} u_R$ converges weakly to $\mathrm{D} u$. In particular, by lower semicontinuity of the norm we have

$$\int_{M\setminus\overline{\Omega}} |Du|^p d\mu \leq \liminf_{R\to\infty} \int_{M\setminus\overline{\Omega}} |Du_R|^p d\mu,$$

and since the righthand side is uniformly bounded again by (B.9), we conclude that $u \in W^{1,p}(M \setminus \overline{\Omega})$. Since it vanishes at infinity, it actually belongs to $W_0^{1,p}(M \setminus \overline{\Omega})$. We can thus argue exactly as done for (B.8) to show that u realises the p-capacity of Ω , that is the first equality in (B.2).

It remains to show the second equality in (B.2). We first show that

$$\int_{\{u=t_2\}} |Du|^{p-1} d\sigma = \int_{\{u=t_1\}} |Du|^{p-1} d\sigma$$
 (B.11)

for almost any $0 < t_1 < t_2 < 1$. To see this, observe first that obviously, by density, if u is p-harmonic we have

$$\int_{M\setminus\overline{\Omega}} \left\langle |\mathrm{D}u|^{p-2} \mathrm{D}u \, \Big| \, \mathrm{D}\psi \right\rangle \mathrm{d}\mu = 0 \tag{B.12}$$

for any $\psi \in \operatorname{Lip}_{\varepsilon}(M \setminus \overline{\Omega})$. Consider then $\psi_{\varepsilon}(u_p)$ with

$$\psi_{\varepsilon}(s) = \begin{cases} 0 & \text{for } s \in (0, t_1 - \varepsilon) \cup (t_2 + \varepsilon, 1), \\ \frac{s - (t_1 - \varepsilon)}{2\varepsilon} & \text{for } s \in [t_1 - \varepsilon, t_1 + \varepsilon) \\ \frac{(t_2 + \varepsilon) - s}{2\varepsilon} & \text{for } s \in (t_2 - \varepsilon, t_2 + \varepsilon] \\ 1 & \text{for } s \in [t_1 + \varepsilon, t_2 - \varepsilon]. \end{cases}$$

Then, since the function $u_p \in \mathscr{C}^{1,\alpha}_{loc}(M \setminus \overline{\Omega})$, we have $\psi_{\varepsilon}(u_p) \in \operatorname{Lip}_{\varepsilon}(M \setminus \overline{\Omega})$. Plugging $\psi_{\varepsilon}(u_p)$ into (B.12), we get

$$(2\varepsilon)^{-1} \int_{\{t_1 - \varepsilon \le u \le t_1 + \varepsilon\}} |Du|^p d\sigma - (2\varepsilon)^{-1} \int_{\{t_2 - \varepsilon \le u \le t_2 + \varepsilon\}} |Du|^p d\sigma = 0.$$

Since *u* is Lipschitz, the coarea formula applies and yields

$$(2\varepsilon)^{-1}\int\limits_{t_1-\varepsilon}^{t_1+\varepsilon}\int_{\{u=t\}}|\mathrm{D} u|^{p-1}\,\mathrm{d}\sigma\,\mathrm{d}t-(2\varepsilon)^{-1}\int\limits_{t_2-\varepsilon}^{t_2+\varepsilon}\int_{\{u=t\}}|\mathrm{D} u|^{p-1}\,\mathrm{d}\sigma\,\mathrm{d}t=0.$$

Letting $\varepsilon \to 0^+$ we obtain (B.11) for almost any $0 < t_1 < t_2 < 1$ by Lebesgue's Differentiation Theorem. Observe that, by the continuity of |Du|, this holds for any regular value t_2 , where it suffices to apply the Fundamental Theorem of Calculus. In particular, by combinining by Hopf's Lemma for p-harmonic functions with the $\mathscr{C}^{1,\alpha}$ regularity of u up to the boundary as done above, any value close enough to 1 is regular. Passing to the limit as $t \to 1^-$ we get, again by the Hölder gradient boundary regularity of u, that (B.11) holds for $t_2 = 1$. We then get, again by coarea formula,

$$\operatorname{Cap}_{p}(\Omega) = \int_{M \setminus \overline{\Omega}} |\operatorname{D} u|^{p} d\mu = \int_{0}^{1} \int_{\{u=t\}} |\operatorname{D} u|^{p-1} d\sigma dt = \int_{\{u=t\}} |\operatorname{D} u|^{p-1} d\sigma,$$

for almost any $0 < t \le 1$, that is the second equality in (B.2), for any regular value and in particular for t = 1, that is (B.3), completing the proof.

We conclude with one last remark concerning an even more general situation.

Remark B.13. If (M,g) is just assumed to be p-nonparabolic, without knowing that the p-Green's function vanishes at infinity, the very same argument used in the proof above to show existence of a solution to (3.4.3) yields a p-harmonic function attaining smoothly the value 1 on the smooth boundary of an open set Ω together with a sequence of points x_j with $d(O,x_j) \to +\infty$ as $j \to \infty$ such that $\lim_{j\to +\infty} u(x_j) \to 0^+$ as $j \to +\infty$. Everything works unchanged, except for this last point. It comes from the fact that on any p-nonparabolic manifold $G_p(O,\cdot)$ vanishes at infinity along such a sequence x_j . This immediately follows from the construction of G_p carried out in [Hol90], implying, by the basic barrier argument (B.10), that so does u.

On the other hand, we remark how, in spite of the finiteness of the p-Dirichlet energy of u, at least apparently (B.2) does not follow immediately from the arguments above, since $u - \psi$ does not in general belong to $W_0^{1,p}(M \setminus \overline{\Omega})$ if ψ is a test function vanishing at infinity and assuming unit value on Ω .

- [ABC11] G. Alberti, S. Bianchini, and G. Crippa. "Structure of level sets and Sardtype properties of Lipschitz maps". In: SISSA;51/2011/M (2011). URL: http://hdl.handle.net/1963/4657.
- [AFM18] V. Agostiniani, M. Fogagnolo, and L. Mazzieri. Sharp geometric inequalities for closed hypersurfaces in manifolds with nonnegative Ricci curvature. 2018. arXiv: 1812.05022 [math.DG].
- [AFM19] V. Agostiniani, M. Fogagnolo, and L. Mazzieri. *Minkowski Inequalities via Non-linear Potential Theory*. 2019. arXiv: 1906.00322 [math.AP].
- [Ale37] A. D. Aleksandrov. "Zur Theorie der gemischten Volumina von konvexen Körpern, II. Neue Ungleichungen zwischen den gemischten Volumina und ihre Anwendungen (in Russian)". In: *Mat. Sbornik N.S.* 2 (1937), pp. 1205–1238.
- [Ale38] A. D. Aleksandrov. "Zur Theorie der gemischten Volumina von konvexen Körpern, III. Die Erweiterung zweier Lehrsätze Minkowskis über die konvexen Polyeder auf beleibige konvexe Flächen (in Russian)". In: *Mat. Sbornik N.S.* 3 (1938), pp. 27–46.
- [Alm66] F. J. Almgren Jr. "Some interior regularity theorems for minimal surfaces and an extension of Bernstein's theorem". In: *Ann. of Math.* (2) 84 (1966), pp. 277–292. ISSN: 0003-486X. DOI: 10.2307/1970520. URL: https://doi.org/10.2307/1970520.
- [Alm86] F. Almgren. "Optimal Isoperimetric Inequalities". In: *Indiana Univ. Math. J.* 35 (3 1986), pp. 451–547. ISSN: 0022-2518.
- [AM15] V. Agostiniani and L. Mazzieri. "Riemannian aspects of potential theory". In: J. Math. Pures Appl. 104.3 (2015), pp. 561 –586. ISSN: 0021-7824. DOI: http://dx.doi.org/10.1016/j.matpur.2015.03.008. URL: http://www.sciencedirect.com/science/article/pii/S0021782415000410.
- [AM17] V. Agostiniani and L. Mazzieri. "On the geometry of the level sets of bounded static potentials". In: *Commun. Math. Phys.* 355 (2017), pp. 261 –301. DOI: https://doi.org/10.1007/s00220-017-2922-x.
- [AM20] V. Agostiniani and L. Mazzieri. "Monotonicity formulas in potential theory". In: *Calc. Var. Partial Differential Equations* 59.1 (2020), Art. 6, 32. ISSN: 0944-2669. DOI: 10.1007/s00526-019-1665-2. URL: https://doi.org/10.1007/s00526-019-1665-2.
- [BC12] V. Bour and G. Carron. "Optimal Integral Pinching Results". In: *Annales Scientifiques de l'Ecole Normale Superieure* 48 (2012). DOI: 10.24033/asens.2238.
- [BC18] C. Bianchini and G. Ciraolo. "Wulff shape characterizations in overdetermined anisotropic elliptic problems". In: *Comm. Partial Differential Equations* 43.5 (2018), pp. 790–820. ISSN: 0360-5302. DOI: 10.1080/03605302.2018. 1475488. URL: https://doi.org/10.1080/03605302.2018.1475488.

[Bes08] A. L. Besse. *Einstein manifolds*. Classics in Mathematics. Reprint of the 1987 edition. Berlin: Springer-Verlag, 2008, pp. xii+516. ISBN: 978-3-540-74120-6.

- [BH16] S. Brendle and G. Huisken. "Mean curvature flow with surgery of mean convex surfaces in R³". In: *Invent. Math.* 203.2 (2016), pp. 615–654. ISSN: 0020-9910. DOI: 10.1007/s00222-015-0599-3. URL: https://doi.org/10.1007/s00222-015-0599-3.
- [BH18] S. Brendle and G. Huisken. "Mean curvature flow with surgery of mean convex surfaces in three-manifolds". In: *J. Eur. Math. Soc. (JEMS)* 20.9 (2018), pp. 2239–2257. ISSN: 1435-9855. DOI: 10.4171/JEMS/811. URL: https://doi.org/10.4171/JEMS/811.
- [BHW16] S. Brendle, P.-K. Hung, and M.-T. Wang. "A Minkowski inequality for hypersurfaces in the Anti-de Sitter-Schwarzschild manifold". In: *Communications on Pure and Applied Mathematics* 69.1 (2016), pp. 124–144. ISSN: 1097-0312. DOI: 10.1002/cpa.21556. URL: http://dx.doi.org/10.1002/cpa.21556.
- [Bia+18] B. Bianchini et al. *On the interplay among maximum principles, compact support principles and Keller-Osserman conditions on manifolds.* 2018. arXiv: 1801.02102 [math.AP].
- [Bis63] R. L. Bishop. "A relation between volume, mean curvature, and diameter". In: *Amer. Math. Soc. Not* 10 (1963), p. 364.
- [BKN89] S. Bando, A. Kasue, and H. Nakajima. "On a construction of coordinates at infinity on manifolds with fast curvature decay and maximal volume growth." In: *Invent. Math.* 97.2 (1989), pp. 313–350. URL: http://eudml.org/doc/143705.
- [BM08] H. L. Bray and P. Miao. "On the capacity of surfaces in manifolds with non-negative scalar curvature". English. In: *Inventiones mathematicae* 172.3 (2008), pp. 459–475. ISSN: 0020-9910. DOI: 10.1007/s00222-007-0102-x. URL: http://dx.doi.org/10.1007/s00222-007-0102-x.
- [BM17] S. Borghini and L. Mazzieri. *On the mass of static metrics with positive cosmological constant: II.* 2017. arXiv: 1711.07024 [math.AP].
- [BM18] S. Borghini and L. Mazzieri. "On the mass of static metrics with positive cosmological constant: I". In: *Classical Quantum Gravity* 35.12 (2018), pp. 125001, 43. ISSN: 0264-9381. DOI: 10.1088/1361-6382/aac081. URL: https://doi.org/10.1088/1361-6382/aac081.
- [BM82] P. Bérard and D. Meyer. "Inégalités isopérimétriques et applications". In: *Ann. Sci. École Norm. Sup.* (4) 15.3 (1982), pp. 513–541. ISSN: 0012-9593. URL: http://www.numdam.org/item?id=ASENS_1982_4_15_3_513_0.
- [BMM19] S. Borghini, G. Mascellani, and L. Mazzieri. "Some sphere theorems in linear potential theory". In: *Trans. Amer. Math. Soc.* 371.11 (2019), pp. 7757–7790. ISSN: 0002-9947. DOI: 10.1090/tran/7637. URL: https://doi.org/10.1090/tran/7637.
- [BT84] R. C. Bassanezi and I. Tamanini. "Subsolutions to the least area problem and the «minimal hull» of a bounded set in *R*ⁿ". In: *Annali dell'Università di Ferrara* 30.1 (1984), pp. 27–40. ISSN: 1827-1510. DOI: 10.1007/BF02853267. URL: https://doi.org/10.1007/BF02853267.
- [Car11] D. G. Caraballo. "Local simplicity, topology, and sets of finite perimeter". In: *Interfaces Free Bound*. 13.2 (2011), pp. 171–189. ISSN: 1463-9963. DOI: 10.4171/IFB/253. URL: https://doi.org/10.4171/IFB/253.

[Car94] G. Carron. "Inégalités isopérimétriques sur les variétés riemanniennes". PhD thesis. Université Joseph Fourier, Grenoble, 1994.

- [Car95] G. Carron. "Inégalités isopérimétriques et inégalités de Faber-Krahn". In: Séminaire de Théorie Spectrale et Géométrie, No. 13, Année 1994–1995. Vol. 13. Sémin. Théor. Spectr. Géom. Univ. Grenoble I, Saint-Martin-d'Hères, 1995, pp. 63–66. DOI: 10.5802/tsg.152. URL: https://doi.org/10.5802/tsg.152.
- [CC96] J. Cheeger and T. H. Colding. "Lower bounds on Ricci curvature and the almost rigidity of warped products". In: Ann. of Math. (2) 144.1 (1996), pp. 189–237. ISSN: 0003-486X. DOI: 10.2307/2118589. URL: https://doi.org/10.2307/2118589.
- [CCM95] J. Cheeger, T. H. Colding, and W. P. Minicozzi. "Linear growth harmonic functions on complete manifolds with nonnegative Ricci curvature". In: *Geometric & Functional Analysis GAFA* 6 (1995), pp. 948–954. ISSN: 1420-8970. DOI: 10.1007/BF01902216. URL: https://doi.org/10.1007/BF01902216.
- [CCW12] S.-C. Chang, J.-T. Chen, and S. Wei. "Liouville properties for p-harmonic maps with finite q-energy". In: *Transactions of the American Mathematical Society* 368 (Nov. 2012). DOI: 10.1090/tran/6351.
- [CEV17] O. Chodosh, M. Eichmair, and A. Volkmann. "Isoperimetric structure of asymptotically conical manifolds". In: *J. Differential Geom.* 105.1 (Jan. 2017), pp. 1–19. DOI: 10.4310/jdg/1483655857. URL: https://doi.org/10.4310/jdg/1483655857.
- [CG71] J. Cheeger and D. Gromoll. "The splitting theorem for manifolds of nonnegative Ricci curvature". In: *J. Differential Geom.* 6 (1971/72), pp. 119–128. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214430220.
- [CG72] J. Cheeger and D. Gromoll. "On the structure of complete manifolds of non-negative curvature". In: *Ann. of Math.* (2) 96 (1972), pp. 413–443. ISSN: 0003-486X. DOI: 10.2307/1970819. URL: https://doi.org/10.2307/1970819.
- [CGT82] J. Cheeger, M. Gromov, and M. Taylor. "Finite propagation speed, kernel estimates for functions of the Laplace operator, and the geometry of complete Riemannian manifolds". In: *J. Differential Geom.* 17.1 (1982), pp. 15–53. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214436699.
- [Cha84] I. Chavel. *Eigenvalues in Riemannian geometry*. Vol. 115. Pure and Applied Mathematics. Including a chapter by Burton Randol, With an appendix by Jozef Dodziuk. Academic Press, Inc., Orlando, FL, 1984, pp. xiv+362. ISBN: 0-12-170640-0.
- [Che17] G. Chen. "Classification of gravitational instantons with faster than quadratic curvature decay". PhD thesis. Stony Brook University, 2017.
- [Che71] B.-Y. Chen. "On a theorem of Fenchel-Borsuk-Willmore-Chern-Lashof". In: *Mathematische Annalen* 194.1 (1971), pp. 19–26. ISSN: 1432-1807. DOI: 10.1007/BF01351818. URL: http://dx.doi.org/10.1007/BF01351818.
- [CK92] C. B. Croke and B. Kleiner. "A warped product splitting theorem". In: *Duke Math. J.* 67.3 (1992), pp. 571–574. ISSN: 0012-7094. DOI: 10.1215/S0012-7094-92-06723-8. URL: http://dx.doi.org/10.1215/S0012-7094-92-06723-8.
- [CLN06] B. Chow, P. Lu, and L. Ni. *Hamilton's Ricci flow*. Vol. 77. Graduate Studies in Mathematics. Providence, RI: American Mathematical Society, 2006, pp. xxxvi+608. ISBN: 978-0-8218-4231-7; 0-8218-4231-5.

[CM13] T. H. Colding and W. P. Minicozzi. "Monotonicity and its analytic and geometric implications". In: *Proceedings of the National Academy of Sciences* 110.48 (2013), pp. 19233–19236. DOI: 10.1073/pnas.1203856109. eprint: http://www.pnas.org/content/110/48/19233.full.pdf. URL: http://www.pnas.org/content/110/48/19233.abstract.

- [CM14a] T. H Colding and W. P. Minicozzi. "On uniqueness of tangent cones for Einstein manifolds". In: *Invent. math.* 196 (2014), pp. 515–588.
- [CM14b] T. H. Colding and W. P. Minicozzi. "Ricci curvature and monotonicity for harmonic functions". In: *Calculus of Variations and Partial Differential Equations* 49.3 (2014), pp. 1045–1059. ISSN: 1432-0835. DOI: 10.1007/s00526-013-0610-z. URL: http://dx.doi.org/10.1007/s00526-013-0610-z.
- [CM97a] T. H Colding and W. P. Minicozzi. "Harmonic functions with polynomial growth". In: *J. Differential Geom.* 45 (1997), pp. 1–77.
- [CM97b] T. H. Colding and W. P. Minicozzi. "Large scale behaviour of kernels of Shroedinger operators". In: American Journal of Mathematics 119 (1997), pp. 1355– 1398.
- [CMM12] G. Catino, C. Mantegazza, and L. Mazzieri. "On the global structure of conformal gradient solitons with nonnegative Ricci tensor". In: *Commun. Contemp. Math.* 14.6 (2012), pp. 1250045, 12. ISSN: 0219-1997. DOI: 10.1142/S0219199712500459. URL: https://doi.org/10.1142/S0219199712500459.
- [CNV15] J. Cheeger, A. Naber, and D. Valtorta. "Critical sets of elliptic equations". In: Comm. Pure Appl. Math. 68.2 (2015), pp. 173–209. ISSN: 0010-3640. DOI: 10.1002/cpa.21518. URL: https://doi.org/10.1002/cpa.21518.
- [Col12] T. H. Colding. "New monotonicity formulas for Ricci curvature and applications. I". In: *Acta Mathematica* 209.2 (2012), pp. 229–263. ISSN: 1871-2509. DOI: 10.1007/s11511-012-0086-2. URL: http://dx.doi.org/10.1007/s11511-012-0086-2.
- [Col97] Tobias H. Colding. "Ricci Curvature and Volume Convergence". In: *Annals of Mathematics* 145.3 (1997), pp. 477–501. ISSN: 0003486X. URL: http://www.jstor.org/stable/2951841.
- [Cro84] C. B. Croke. "A sharp four-dimensional isoperimetric inequality". In: *Comment. Math. Helv.* 59.2 (1984), pp. 187–192. ISSN: 0010-2571. DOI: 10 . 1007/BF02566344. URL: https://doi.org/10.1007/BF02566344.
- [CS03a] A. Colesanti and P. Salani. "The Brunn-Minkowski inequality for *p*-capacity of convex bodies". In: *Math. Ann.* 327.3 (2003), pp. 459–479. ISSN: 0025-5831. DOI: 10.1007/s00208-003-0460-7. URL: https://doi.org/10.1007/s00208-003-0460-7.
- [CS03b] A. Colesanti and P. Salani. "The Brunn-Minkowski inequality for *p*-capacity of convex bodies". In: *Math. Ann.* 327.3 (2003), pp. 459–479. ISSN: 0025-5831. DOI: 10.1007/s00208-003-0460-7. URL: https://doi.org/10.1007/s00208-003-0460-7.
- [CW13] S.-Y. A. Chang and Y. Wang. "Some Higher Order Isoperimetric Inequalities via the Method of Optimal Transport". In: *International Mathematics Research Notices* 2014 (May 2013). DOI: 10.1093/imrn/rnt182.
- [Dal+16] J. Dalphin et al. "On the minimization of total mean curvature". In: *J. Geom. Anal.* 26.4 (2016), pp. 2729–2750. ISSN: 1050-6926. DOI: 10.1007/s12220-015-9646-y. URL: https://doi.org/10.1007/s12220-015-9646-y.

[DG58] E. De Giorgi. "Sulla proprietà isoperimetrica dell'ipersfera, nella classe degli insiemi aventi frontiera orientata di misura finita". In: *Atti Accad. Naz. Lincei. Mem. Cl. Sci. Fis. Mat. Nat. Sez. I (8)* 5 (1958), pp. 33–44.

- [DG61] E. De Giorgi. *Frontiere orientate di misura minima*. Seminario di Matematica della Scuola Normale Superiore di Pisa, 1960-61. Editrice Tecnico Scientifica, Pisa, 1961, p. 57.
- [DiB83] E. DiBenedetto. " $C^{1+\alpha}$ local regularity of weak solutions of degenerate elliptic equations". In: *Nonlinear Anal.* 7.8 (1983), pp. 827–850. ISSN: 0362-546X. DOI: 10.1016/0362-546X(83)90061-5. URL: https://doi.org/10.1016/0362-546X(83)90061-5.
- [DLM05] C. De Lellis and S. Müller. "Optimal rigidity estimates for nearly umbilical surfaces". In: *J. Differential Geom.* 69.1 (2005), pp. 75–110. ISSN: 0022-040X. DOI: 10.4310/jdg/1121540340. URL: https://doi.org/10.4310/jdg/1121540340.
- [EH79] T. Eguchi and A. J. Hanson. "Self-dual solutions to Euclidean gravity". In: *Ann. Physics* 120.1 (1979), pp. 82–106. ISSN: 0003-4916. DOI: 10.1016/0003-4916(79)90282-3. URL: https://doi.org/10.1016/0003-4916(79)90282-3.
- [EH91] K. Ecker and G. Huisken. "Interior estimates for hypersurfaces moving by mean curvature". In: *Inventiones mathematicae* 105.1 (1991), pp. 547–569. ISSN: 1432-1297. DOI: 10.1007/BF01232278. URL: https://doi.org/10.1007/BF01232278.
- [ES91] L. C. Evans and J. Spruck. "Motion of level sets by mean curvature. I". In: *J. Differential Geom.* 33.3 (1991), pp. 635–681. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214446559.
- [Eva10] L.C. Evans. *Partial differential equations*. Second. Vol. 19. Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2010, pp. xxii+749. ISBN: 978-0-8218-4974-3. DOI: 10.1090/gsm/019. URL: https://doi.org/10.1090/gsm/019.
- [Eva82] L. C. Evans. "A new proof of local $C^{1,\alpha}$ regularity for solutions of certain degenerate elliptic p.d.e". In: *J. Differential Equations* 45.3 (1982), pp. 356–373. ISSN: 0022-0396. DOI: 10.1016/0022-0396(82)90033-X. URL: https://doi.org/10.1016/0022-0396(82)90033-X.
- [Fed58] H. Federer. "A note on the Gauss-Green theorem". In: Proc. Amer. Math. Soc. 9 (1958), pp. 447–451. ISSN: 0002-9939. DOI: 10.2307/2033002. URL: https://doi.org/10.2307/2033002.
- [Fen29] W. Fenchel. "Über Krümmung und Windung geschlossener Raumkurven." In: *Math. Ann.* 101 (1929), pp. 238–252.
- [FF60] H. Federer and W. H. Fleming. "Normal and integral currents". In: *Ann. of Math.* (2) 72 (1960), pp. 458–520. ISSN: 0003-486X. DOI: 10.2307/1970227. URL: https://doi.org/10.2307/1970227.
- [FMP19] M. Fogagnolo, L. Mazzieri, and A. Pinamonti. "Geometric aspects of p-capacitary potentials". In: *Ann. Inst. H. Poincaré Anal. Non Linéaire* 36.4 (2019), pp. 1151–1179. ISSN: 0294-1449. DOI: 10.1016/j.anihpc.2018.11.005. URL: https://doi.org/10.1016/j.anihpc.2018.11.005.

[FMV13] A. Farina, L. Mari, and E. Valdinoci. "Splitting theorems, symmetry results and overdetermined problems for Riemannian manifolds". In: *Comm. Partial Differential Equations* 38.10 (2013), pp. 1818–1862. ISSN: 0360-5302. DOI: 10.1080/03605302.2013.795969. URL: http://dx.doi.org/10.1080/03605302.2013.795969.

- [FS14] A. Freire and F. Schwartz. "Mass-Capacity Inequalities for Conformally Flat Manifolds with Boundary". In: Communications in Partial Differential Equations 39.1 (2014), pp. 98–119. DOI: 10.1080/03605302.2013.851211. eprint: https://doi.org/10.1080/03605302.2013.851211. URL: https://doi.org/10.1080/03605302.2013.851211.
- [Ge+15] Yuxin Ge et al. "A Penrose inequality for graphs over Kottler space". In: *Calc. Var. Partial Differential Equations* 52.3-4 (2015), pp. 755–782. ISSN: 0944-2669. DOI: 10.1007/s00526-014-0732-y. URL: https://doi.org/10.1007/s00526-014-0732-y.
- [Ger90] C. Gerhardt. "Flow of nonconvex hypersurfaces into spheres". In: *J. Differential Geom.* 32.1 (1990), pp. 299–314. DOI: 10.4310/jdg/1214445048. URL: https://doi.org/10.4310/jdg/1214445048.
- [GHL04] S. Gallot, D. Hulin, and J. Lafontaine. *Riemannian geometry*. Third. Universitext. Berlin: Springer-Verlag, 2004, pp. xvi+322. ISBN: 3-540-20493-8. DOI: 10.1007/978-3-642-18855-8. URL: http://dx.doi.org/10.1007/978-3-642-18855-8.
- [Gib73] G. W. Gibbons. "Thesis". PhD thesis. Cambridge University, 1973.
- [GJ19] G. J. Galloway and H. C. Jang. *Some scalar curvature warped product splitting theorems*. To appear on Proc. of the AMS. 2019. DOI: 10.1090/proc/14922. arXiv: 1907.09396 [math.DG].
- [GL09] P. Guan and J. Li. "The quermassintegral inequalities for k-convex starshaped domains". In: *Advances in Mathematics* 221.5 (2009), pp. 1725 –1732. ISSN: 0001-8708. DOI: https://doi.org/10.1016/j.aim.2009.03.005. URL: http://www.sciencedirect.com/science/article/pii/S0001870809000796.
- [Gri99] A. Grigor'yan. "Analytic and geometric background of recurrence and non-explosion of the Brownian motion on Riemannian manifolds". In: *Bull. Amer. Math. Soc. (N.S.)* 36.2 (1999), pp. 135–249. ISSN: 0273-0979. DOI: 10 . 1090/S0273-0979-99-00776-4. URL: https://doi.org/10.1090/S0273-0979-99-00776-4.
- [Gro80] M. Gromov. Paul Levý 's isoperimetric inequality. 1980. URL: https://www.ihes.fr/~gromov/wp-content/uploads/2018/08/1133.pdf.
- [Gro81] M. Gromov. Structures métriques pour les variétés riemanniennes. Vol. 1. Textes Mathématiques [Mathematical Texts]. Edited by J. Lafontaine and P. Pansu. CEDIC, Paris, 1981, pp. iv+152. ISBN: 2-7124-0714-8.
- [GS19] M. Ghomi and J. Spruck. *Total curvature and the isoperimetric inequality in Cartan-Hadamard manifolds*. 2019. arXiv: 1908.09814 [math.DG].
- [GS99] N. Garofalo and E. Sartori. "Symmetry in exterior boundary value problems for quasilinear elliptic equations via blow-up and a priori estimates". In: *Adv. Differential Equations* 4.2 (1999), pp. 137–161. ISSN: 1079-9389.
- [GT01] D. Gilbarg and N. S. Trudinger. *Elliptic partial differential equations of second order*. Classics in Mathematics. Reprint of the 1998 edition. Berlin: Springer-Verlag, 2001, pp. xiv+517. ISBN: 3-540-41160-7.

[Gua+10] P. Guan et al. "A form of Alexandrov-Fenchel inequality". In: *Pure Appl. Math. Q.* 6.4, Special Issue: In honor of Joseph J. Kohn. Part 2 (2010), pp. 999–1012. ISSN: 1558-8599. DOI: 10.4310/PAMQ.2010.v6.n4.a2. URL: https://doi.org/10.4310/PAMQ.2010.v6.n4.a2.

- [GWW13] Y. Ge, G. Wang, and J. Wu. "Hyperbolic Alexandrov-Fenchel quermassintegral inequalities I". In: *Journal of Differential Geometry* 98 (2013).
- [GWW14] Y. Ge, G. Wang, and J. Wu. "Hyperbolic Alexandrov-Fenchel quermassintegral inequalities II". In: *J. Differential Geom.* 98.2 (2014), pp. 237–260. DOI: 10. 4310/jdg/1406552250. URL: https://doi.org/10.4310/jdg/1406552250.
- [Ham88] R. S. Hamilton. "The Ricci flow on surfaces". In: *Mathematics and general relativity (Santa Cruz, CA, 1986)*. Vol. 71. Contemp. Math. Amer. Math. Soc., Providence, RI, 1988, pp. 237–262. DOI: 10.1090/conm/071/954419. URL: https://doi.org/10.1090/conm/071/954419.
- [Haw77] S. W. Hawking. "Gravitational instantons". In: *Phys. Lett. A* 60.2 (1977), pp. 81–83. ISSN: 0375-9601. DOI: 10.1016/0375-9601(77)90386-3. URL: https://doi.org/10.1016/0375-9601(77)90386-3.
- [HE73] S. W. Hawking and G. F. R. Ellis. *The large scale structure of space-time*. Cambridge Monographs on Mathematical Physics, No. 1. London: Cambridge University Press, 1973, pp. xi+391.
- [Heb99] E. Hebey. *Nonlinear analysis on manifolds: Sobolev spaces and inequalities*. Vol. 5. Courant Lecture Notes in Mathematics. New York University, Courant Institute of Mathematical Sciences, New York; American Mathematical Society, Providence, RI, 1999, pp. x+309. ISBN: 0-9658703-4-0; 0-8218-2700-6.
- [Hel73] W Helfrich. "Elastic properties of lipid bilayers: theory and possible experiments". In: Z. Naturforsch 28.11 (1973), pp. 693–703. URL: http://ludfc39.u-strasbg.fr/pdflib/membranes/Elasticity/1973_Helfrich_b.pdf.
- [Her00] M. Herzlich. "Refined Kato inequalities in Riemannian geometry". In: *Journées* "Équations aux Dérivées Partielles" (La Chapelle sur Erdre, 2000). Univ. Nantes, Nantes, 2000, Exp. No. VI, 11.
- [HI01] G. Huisken and T. Ilmanen. "The Inverse Mean Curvature Flow and the Riemannian Penrose Inequality". In: *J. Differential Geom.* 59.3 (2001), pp. 353–437. URL: http://projecteuclid.org/euclid.jdg/1090349447.
- [HI08] G. Huisken and T. Ilmanen. "Higher regularity of the inverse mean curvature flow". In: *J. Differential Geom.* 80.3 (Nov. 2008), pp. 433–451. DOI: 10.4310/jdg/1226090483. URL: https://doi.org/10.4310/jdg/1226090483.
- [Hir76] M.W. Hirsch. *Differential Topology*. Graduate Texts in Mathematics. Springer, 1976. ISBN: 9780387901480. URL: https://books.google.it/books?id=uETvAAAAMAAJ.
- [HKM06] J. Heinonen, T. Kilpeläinen, and O. Martio. *Nonlinear potential theory of degenerate elliptic equations*. Unabridged republication of the 1993 original. Dover Publications, Inc., Mineola, NY, 2006, pp. xii+404. ISBN: 0-486-45050-3.
- [Hol90] I. Holopainen. "Nonlinear potential theory and quasiregular mappings on Riemannian manifolds". English. In: *Annales Academiae Scientiarum Fennicae*. *Series A 1. Mathematica. Dissertationes* 74 (1990), pp. 1–45. ISSN: 0355-0087.

[Hol99] I. Holopainen. "Volume growth, Green's functions, and parabolicity of ends". In: *Duke Math. J.* 97.2 (1999), pp. 319–346. ISSN: 0012-7094. DOI: 10.1215/S0012-7094-99-09714-4. URL: https://doi.org/10.1215/S0012-7094-99-09714-4.

- [HP99] G. Huisken and A. Polden. "Geometric evolution equations for hypersurfaces". In: *Calculus of variations and geometric evolution problems* (*Cetraro*, 1996). Vol. 1713. Lecture Notes in Math. Berlin: Springer, 1999, pp. 45–84. DOI: 10. 1007/BFb0092669. URL: http://dx.doi.org/10.1007/BFb0092669.
- [HS74] D. Hoffman and J. Spruck. "Sobolev and isoperimetric inequalities for Riemannian submanifolds". In: *Comm. Pure Appl. Math.* 27 (1974), pp. 715–727. ISSN: 0010-3640. DOI: 10.1002/cpa.3160270601. URL: https://doi.org/10.1002/cpa.3160270601.
- [HS89] R. Hardt and L. Simon. "Nodal sets for solutions of elliptic equations". In: *J. Differential Geom.* 30.2 (1989), pp. 505–522. URL: http://projecteuclid.org/euclid.jdg/1214443599.
- [Hui] G. Huisken. *An isoperimetric concept for the mass in general relativity*. Video, available at https://video.ias.edu/node/234.
- [Hui86] G. Huisken. "Contracting convex hypersurfaces in Riemannian manifolds by their mean curvature." In: *Inventiones mathematicae* 84 (1986), pp. 463–480. URL: http://eudml.org/doc/143347.
- [JL17] J. Jauregui and D. Lee. "Lower semicontinuity of mass under C^0 convergence and Huisken's isoperimetric mass". In: *Journal für die reine und angewandte Mathematik* (Crelles Journal) (Jan. 2017). DOI: 10.1515/crelle-2017-0007.
- [Joy00] D. D. Joyce. *Compact manifolds with special holonomy*. Oxford Mathematical Monographs. Oxford University Press, Oxford, 2000, pp. xii+436. ISBN: 0-19-850601-5.
- [Kas83] A. Kasue. "Ricci curvature, geodesics and some geometric properties of Riemannian manifolds with boundary". In: *J. Math. Soc. Japan* 35.1 (Jan. 1983), pp. 117–131. DOI: 10.2969/jmsj/03510117. URL: https://doi.org/10.2969/jmsj/03510117.
- [Kle92] B. Kleiner. "An isoperimetric comparison theorem". In: *Invent. Math.* 108.1 (1992), pp. 37–47. ISSN: 0020-9910. DOI: 10.1007/BF02100598. URL: https://doi.org/10.1007/BF02100598.
- [KN09] B. Kotschwar and L. Ni. "Local gradient estimates of p-harmonic functions, 1/H-flow, and an entropy formula". In: *Ann. Sci. Éc. Norm. Supér.* (4) 42.1 (2009), pp. 1–36. ISSN: 0012-9593. DOI: 10.24033/asens.2089. URL: https://doi.org/10.24033/asens.2089.
- [Kro89a] P. B. Kronheimer. "A Torelli-type theorem for gravitational instantons". In: *J. Differential Geom.* 29.3 (1989), pp. 685–697. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214443067.
- [Kro89b] P. B. Kronheimer. "The construction of ALE spaces as hyper-Kähler quotients". In: *J. Differential Geom.* 29.3 (1989), pp. 665–683. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214443066.
- [KV86] S. Kichenassamy and L. Véron. "Singular solutions of the *p*-Laplace equation". In: *Math. Ann.* 275.4 (1986), pp. 599–615. ISSN: 0025-5831. DOI: 10.1007/BF01459140. URL: https://doi.org/10.1007/BF01459140.
- [Lev22] P. Levý. Leçons d'analyse fonctionnelle. Gauthier-Villars et Cie, Paris, 1922.

[Lew77] J. L. Lewis. "Capacitary functions in convex rings". In: *Arch. Rational Mech. Anal.* 66.3 (1977), pp. 201–224. ISSN: 0003-9527. DOI: 10.1007/BF00250671. URL: https://doi.org/10.1007/BF00250671.

- [Lew83] J. L. Lewis. "Regularity of the derivatives of solutions to certain degenerate elliptic equations". In: *Indiana Univ. Math. J.* 32.6 (1983), pp. 849–858. ISSN: 0022-2518. DOI: 10.1512/iumj.1983.32.32058. URL: https://doi.org/10.1512/iumj.1983.32.32058.
- [LG16] L. L. de Lima and F. Girão. "An Alexandrov–Fenchel-Type Inequality in Hyperbolic Space with an Application to a Penrose Inequality". In: *Annales Henri Poincaré* 17.4 (2016), pp. 979–1002. ISSN: 1424-0661. DOI: 10 . 1007 / s00023-015-0414-0. URL: https://doi.org/10.1007/s00023-015-0414-0.
- [Li] P. Li. "Lectures on Harmonic Function". In: (). Lectures at UCI.
- [Lie88] G. M. Lieberman. "Boundary Regularity for Solutions of Degenerate Elliptic Equations". In: *Nonlinear Anal.* 12.11 (1988), pp. 1203–1219. ISSN: 0362-546X. DOI: 10.1016/0362-546X(88)90053-3. URL: https://doi.org/10.1016/0362-546X(88)90053-3.
- [Lin17] P. Lindqvist. *Notes on the p-Laplace equation*. Second. University of Jyväskylä, 2017.
- [Liu13] G. Liu. "3-manifolds with nonnegative Ricci curvature". In: *Invent. Math.* 193.2 (2013), pp. 367–375. ISSN: 0020-9910. DOI: 10.1007/s00222-012-0428-x. URL: https://doi.org/10.1007/s00222-012-0428-x.
- [LN15] Dan A. Lee and André Neves. "The Penrose Inequality for Asymptotically Locally Hyperbolic Spaces with Nonpositive Mass". In: *Communications in Mathematical Physics* 339.2 (2015), pp. 327–352. ISSN: 1432-0916. DOI: 10.1007/s00220-015-2421-x. URL: https://doi.org/10.1007/s00220-015-2421-x.
- [LS84] P. Li and R. Schoen. "L p and mean value properties of subharmonic functions on Riemannian manifolds". In: *Acta Math.* 153 (1984), pp. 279–301. DOI: 10.1007/BF02392380. URL: https://doi.org/10.1007/BF02392380.
- [LT87] P. Li and L.-F. Tam. "Symmetric Green's functions on complete manifolds". In: *Amer. J. Math.* 109.6 (1987), pp. 1129–1154. ISSN: 0002-9327. DOI: 10.2307/2374588. URL: https://doi.org/10.2307/2374588.
- [LT92] P. Li and L.-F. Tam. "Harmonic functions and the structure of complete manifolds". In: *J. Differential Geom.* 35.2 (1992), pp. 359–383. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214448079.
- [LT95] P. Li and L.-F. Tam. "Green's functions, harmonic functions, and volume comparison". In: *J. Differential Geom.* 41.2 (1995), pp. 277–318. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1214456219.
- [LTW97] P. Li, L.-F. Tam, and J. Wang. "Sharp bounds for the Green's function and the heat kernel". In: *Math. Res. Lett.* 4.4 (1997), pp. 589–602. ISSN: 1073-2780. DOI: 10.4310/MRL.1997.v4.n4.a13. URL: https://doi.org/10.4310/MRL.1997.v4.n4.a13.
- [LU68] O. A. Ladyzhenskaya and N. N. Ural'tseva. *Linear and quasilinear elliptic equations*. Translated from the Russian by Scripta Technica, Inc. Translation editor: Leon Ehrenpreis. Academic Press, New York-London, 1968, pp. xviii+495.

[LW17] H. Li and Y. Wei. "On inverse mean curvature flow in Schwarzschild space and Kottler space". In: *Calc. Var. Partial Differential Equations* 56.3 (2017), Art. 62, 21. ISSN: 0944-2669. DOI: 10.1007/s00526-017-1160-6. URL: https://doi.org/10.1007/s00526-017-1160-6.

- [LY86] P. Li and S.-T. Yau. "On the parabolic kernel of the Schrödinger operator". In: *Acta Math.* 156.3-4 (1986), pp. 153–201. ISSN: 0001-5962. DOI: 10.1007/BF02399203. URL: https://doi.org/10.1007/BF02399203.
- [Mag12] F. Maggi. Sets of Finite Perimeter and Geometric Variational Problems: An Introduction to Geometric Measure Theory. Cambridge Studies in Advanced Mathematics. Cambridge University Press, 2012. ISBN: 9781107021037. URL: https://books.google.it/books?id=Qc2L02PD10gC.
- [Mal55] B. Malgrange. "Existence et approximation des solutions des équations aux dérivées partielles et des équations de convolution". In: *Ann. Inst. Fourier, Grenoble* 6 (1955–1956), pp. 271–355. ISSN: 0373-0956. URL: http://aif.cedram.org/item?id=AIF_1955__6__271_0.
- [Man] C. Mantegazza. *Notes on the Distance Function from a Submanifold-V3*. URL: http://cvgmt.sns.it/media/doc/paper/1182/distancenotes.pdf.
- [Man11] C. Mantegazza. *Lecture notes on mean curvature flow*. Vol. 290. Progress in Mathematics. Birkhäuser/Springer Basel AG, Basel, 2011, pp. xii+166. ISBN: 978-3-0348-0144-7. DOI: 10.1007/978-3-0348-0145-4. URL: https://doi.org/10.1007/978-3-0348-0145-4.
- [Mar09] M. Mars. "Present status of the Penrose inequality". In: Classical Quantum Gravity 26.19 (2009), pp. 193001, 59. ISSN: 0264-9381. DOI: 10.1088/0264-9381/26/19/193001. URL: https://doi.org/10.1088/0264-9381/26/19/193001.
- [Maz11] V. Maz'ya. Sobolev spaces with applications to elliptic partial differential equations. augmented. Vol. 342. Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer, Heidelberg, 2011, pp. xxviii+866. ISBN: 978-3-642-15563-5. DOI: 10.1007/978-3-642-15564-2. URL: https://doi.org/10.1007/978-3-642-15564-2.
- [McC17] S. McCormick. "On a Minkowski-like inequality for asymptotically flat static manifolds". In: *Proceedings of the American Mathematical Society* 146 (Sept. 2017). DOI: 10.1090/proc/14047.
- [Men00] X. Menguy. "Noncollapsing examples with positive Ricci curvature and infinite topological type". In: Geom. Funct. Anal. 10.3 (2000), pp. 600–627. ISSN: 1016-443X. DOI: 10.1007/PL00001632. URL: https://doi.org/10.1007/PL00001632.
- [Min03] H. Minkowski. "Volumen und Oberfläche". In: *Mathematische Annalen* 57.4 (1903), pp. 447–495. ISSN: 1432-1807. DOI: 10.1007/BF01445180. URL: https://doi.org/10.1007/BF01445180.
- [Min09a] V. Minerbe. "A mass for ALF manifolds". In: Comm. Math. Phys. 289.3 (2009), pp. 925–955. ISSN: 0010-3616. DOI: 10.1007/s00220-009-0823-3. URL: https://doi.org/10.1007/s00220-009-0823-3.
- [Min09b] V. Minerbe. "Weighted Sobolev inequalities and Ricci flat manifolds". In: *Geom. Funct. Anal.* 18.5 (2009), pp. 1696–1749. ISSN: 1016-443X. DOI: 10.1007/s00039-009-0701-3. URL: https://doi.org/10.1007/s00039-009-0701-3.

[Min10] V. Minerbe. "On the asymptotic geometry of gravitational instantons". In: *Ann. Sci. Éc. Norm. Supér.* (4) 43.6 (2010), pp. 883–924. ISSN: 0012-9593. DOI: 10.24033/asens.2135. URL: https://doi.org/10.24033/asens.2135.

- [Min11] V. Minerbe. "Rigidity for multi-Taub-NUT metrics". In: *J. Reine Angew. Math.* 656 (2011), pp. 47–58. ISSN: 0075-4102. DOI: 10.1515/CRELLE.2011.042. URL: https://doi.org/10.1515/CRELLE.2011.042.
- [Mir64] M. Miranda. "Distribuzioni aventi derivate misure insiemi di perimetro localmente finito". In: *Ann. Scuola Norm. Sup. Pisa* (3) 18 (1964), pp. 27–56.
- [Mir97] M. Miranda. "Diseguaglianze isoperimetriche e diseguaglianze di Sobolev". it. In: Rendiconti del Seminario Matematico della Università di Padova 97 (1997), pp. 187–191. URL: http://www.numdam.org/item/RSMUP_1997__97__187_0.
- [MN16] A. Mondino and S. Nardulli. "Existence of isoperimetric regions in non-compact Riemannian manifolds under Ricci or scalar curvature conditions". In: *Comm. Anal. Geom.* 24.1 (2016), pp. 115–138. ISSN: 1019-8385. DOI: 10.4310/CAG.2016.v24.n1.a5. URL: https://doi.org/10.4310/CAG.2016.v24.n1.a5.
- [Mos07] R. Moser. "The inverse mean curvature flow and p-harmonic functions". In: *J. Eur. Math. Soc. (JEMS)* 9.1 (2007), pp. 77–83. ISSN: 1435-9855. DOI: 10.4171/JEMS/73. URL: https://doi.org/10.4171/JEMS/73.
- [MP19] R. Magnanini and G. Poggesi. "On the stability for Alexandrov's Soap Bubble theorem". In: *Journal d'Analyse Mathématique* (2019). ISSN: 1565-8538. DOI: 10. 1007/s11854-019-0058-y. URL: https://doi.org/10.1007/s11854-019-0058-y.
- [MRS19] L. Mari, L. Rigoli, and A. G. Setti. *On the 1/H-flow by p-Laplace approximation:* new estimates via fake distances under Ricci lower bounds. 2019. arXiv: 1905. 00216 [math.DG].
- [MS16] M. Makowski and J. Scheuer. "Rigidity results, inverse curvature flows and Alexandrov-Fenchel type inequalities in the sphere". In: *Asian Journal of Mathematics* 20 (Nov. 2016), pp. 869–. DOI: 10.4310/AJM.2016.v20.n5.a2.
- [MS73] J. H. Michael and L. M. Simon. "Sobolev and mean-value inequalities on generalized submanifolds of Rn". In: Communications on Pure and Applied Mathematics 26.3 (1973), pp. 361–379. DOI: 10.1002/cpa.3160260305. eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1002/cpa.3160260305. URL: https://onlinelibrary.wiley.com/doi/abs/10.1002/cpa.3160260305.
- [Pen73] R. Penrose. "Naked singularities". In: *Annals N. Y. Acad. Sci.* 224 (1973), pp. 125–134. DOI: 10.1111/j.1749-6632.1973.tb41447.x.
- [Per11] D. R. Perez. "On nearly umbilical hypersurfaces". PhD thesis. University of Zurich, 2011.
- [Pet06] P. Petersen. *Riemannian geometry*. Second. Vol. 171. Graduate Texts in Mathematics. New York: Springer, 2006, pp. xvi+401. ISBN: 978-0387-29246-5; 0-387-29246-2.
- [Phi90] G. A. Philippin. "On a free boundary problem in electrostatics". In: *Math. Methods Appl. Sci.* 12.5 (1990), pp. 387–392. ISSN: 0170-4214. DOI: 10.1002/mma.1670120503. URL: https://doi.org/10.1002/mma.1670120503.
- [Pip16] G. Pipoli. *Inverse mean curvature flow in complex hyperbolic space*. To appear on Annales scientifiques de l'ENS. 2016. arXiv: 1610.01886 [math.DG].

[Pog18] G. Poggesi. "Radial symmetry for p-harmonic functions in exterior and punctured domains". In: *Applicable Analysis* 0.0 (2018), pp. 1–14. DOI: 10 . 1080/00036811 . 2018 . 1460819. eprint: https://doi.org/10.1080/00036811 . 2018 . 1460819. URL: https://doi.org/10.1080/00036811 . 2018 . 1460819.

- [PP79] L. E. Payne and G. A. Philippin. "On some maximum principles involving harmonic functions and their derivatives". In: *SIAM J. Math. Anal.* 10.1 (1979), pp. 96–104. ISSN: 0036-1410. DOI: 10.1137/0510012. URL: http://dx.doi.org/10.1137/0510012.
- [PST14] S. Pigola, A. G. Setti, and M. Troyanov. "The connectivity at infinity of a manifold and $L^{q,p}$ -Sobolev inequalities". In: *Expo. Math.* 32.4 (2014), pp. 365–383. ISSN: 0723-0869. DOI: 10.1016/j.exmath.2013.12.006. URL: https://doi.org/10.1016/j.exmath.2013.12.006.
- [PW84] M. H. Protter and H. F. Weinberger. *Maximum principles in differential equations*. Corrected reprint of the 1967 original. Springer-Verlag, New York, 1984, pp. x+261. ISBN: 0-387-96068-6. DOI: 10.1007/978-1-4612-5282-5. URL: https://doi.org/10.1007/978-1-4612-5282-5.
- [Qiu15] G. Qiu. "A family of higher-order isoperimetric inequalities". In: *Communications in Contemporary Mathematics* 17.03 (2015), p. 1450015. DOI: 10.1142/S0219199714500151. eprint: https://doi.org/10.1142/S0219199714500151. URL: https://doi.org/10.1142/S0219199714500151.
- [Rei15] M. Reiris. "On Ricci curvature and volume growth in dimension three". In: *J. Differential Geom.* 99.2 (2015), pp. 313–357. ISSN: 0022-040X. URL: http://projecteuclid.org/euclid.jdg/1421415565.
- [Rei96] W. Reichel. "Radial symmetry for an electrostatic, a capillarity and some fully nonlinear overdetermined problems on exterior domains". In: *Z. Anal. Anwendungen* 15.3 (1996), pp. 619–635. ISSN: 0232-2064. DOI: 10.4171/ZAA/719. URL: http://dx.doi.org/10.4171/ZAA/719.
- [Sal05] P. Salani. "Starshapedness of level sets of solutions to elliptic PDEs". In: *Appl. Anal.* 84.12 (2005), pp. 1185–1197. ISSN: 0003-6811. DOI: 10.1080/00036810412331297262. URL: https://doi.org/10.1080/00036810412331297262.
- [Sar98] E. Sartori. "On a free boundary problem for the *p*-Laplacian". In: *J. Math. Anal. Appl.* 218.1 (1998), pp. 117–126. ISSN: 0022-247X. DOI: 10.1006/jmaa. 1997.5755. URL: https://doi.org/10.1006/jmaa.1997.5755.
- [Sch08] F. Schulze. "Nonlinear evolution by mean curvature and isoperimetric inequalities". In: *J. Differential Geom.* 79.2 (June 2008), pp. 197–241. DOI: 10. 4310/jdg/1211512640. URL: https://doi.org/10.4310/jdg/1211512640.
- [Sch15] T. Schmidt. "Strict interior approximation of sets of finite perimeter and functions of bounded variation". In: *Proc. Amer. Math. Soc.* 143.5 (2015), pp. 2069–2084. ISSN: 0002-9939. DOI: 10.1090/S0002-9939-2014-12381-1. URL: https://doi.org/10.1090/S0002-9939-2014-12381-1.
- [Sch18] F. Schulze. Optimal isoperimetric inequalities for surfaces in any codimension in Cartan-Hadamard manifolds. 2018. arXiv: 1802.00226 [math.DG].
- [Sim67] J. Simons. "A note on minimal varieties". In: Bull. Amer. Math. Soc. 73 (1967), pp. 491–495. ISSN: 0002-9904. DOI: 10.1090/S0002-9904-1967-11790-7. URL: https://doi.org/10.1090/S0002-9904-1967-11790-7.

[SWZ93] P. Sternberg, G. Williams, and W. P. Ziemer. "The constrained least gradient problem in Rn". In: *Trans. Amer. Math. Soc.* 339.1 (1993), pp. 403–432. ISSN: 0002-9947. DOI: 10.2307/2154226. URL: https://doi.org/10.2307/2154226.

- [SY94] R. Schoen and S.-T. Yau. *Lectures on differential geometry*. Conference Proceedings and Lecture Notes in Geometry and Topology, I. Lecture notes prepared by Wei Yue Ding, Kung Ching Chang [Gong Qing Zhang], Jia Qing Zhong and Yi Chao Xu, Translated from the Chinese by Ding and S. Y. Cheng, With a preface translated from the Chinese by Kaising Tso. International Press, Cambridge, MA, 1994, pp. v+235. ISBN: 1-57146-012-8.
- [SZW91] P. Sternberg, W. P. Ziemer, and G. Williams. "C1,1-regularity of constrained area minimizing hypersurfaces". In: Journal of Differential Equations 94.1 (1991), pp. 83 –94. ISSN: 0022-0396. DOI: https://doi.org/10.1016/0022-0396(91) 90104-H. URL: http://www.sciencedirect.com/science/article/pii/002203969190104H.
- [Tol83] P. Tolksdorf. "On the Dirichlet problem for quasilinear equations in domains with conical boundary points". In: *Comm. Partial Differential Equations* 8.7 (1983), pp. 773–817. ISSN: 0360-5302. DOI: 10.1080/03605308308820285. URL: https://doi.org/10.1080/03605308308820285.
- [Top98] P. Topping. "Mean curvature flow and geometric inequalities". In: *J. Reine Angew. Math.* 503 (1998), pp. 47–61. ISSN: 0075-4102. DOI: 10.1515/crll. 1998.099. URL: https://doi.org/10.1515/crll.1998.099.
- [Tru94] N. S. Trudinger. "Isoperimetric inequalities for quermassintegrals". en. In: Annales de l'I.H.P. Analyse non linéaire 11.4 (1994), pp. 411–425. URL: http://http://www.numdam.org/item/AIHPC_1994__11_4_411_0.
- [Ura68] N. N. Ural'tseva. "Degenerate quasilinear elliptic systems". In: *Zap. Naučn. Sem. Leningrad. Otdel. Mat. Inst. Steklov. (LOMI)* 7 (1968), pp. 184–222.
- [Urb90] J. I.E. Urbas. "On the expansion of starshaped hypersurfaces by symmetric functions of their principal curvatures." In: *Mathematische Zeitschrift* 205.3 (1990), pp. 355–372. URL: http://eudml.org/doc/174181.
- [Val13] D. Valtorta. "On the p-Laplace operator on Riemannian manifolds". PhD thesis. Università degli Studi di Milano, 2013. URL: http://hdl.handle.net/2434/217559.
- [Var82] N. Th. Varopoulos. "Green's functions on positively curved manifolds. II". In: J. Funct. Anal. 49.2 (1982), pp. 170–176. ISSN: 0022-1236. DOI: 10.1016/0022-1236(82)90077-5. URL: https://doi.org/10.1016/0022-1236(82)90077-5.
- [Var95] N. Th. Varopoulos. "Hardy-Littlewood theory on unimodular groups". In: *Ann. Inst. H. Poincaré Probab. Statist.* 31.4 (1995), pp. 669–688. ISSN: 0246-0203. URL: http://www.numdam.org/item?id=AIHPB_1995__31_4_669_0.
- [Wei18] Y. Wei. "On the Minkowski-type inequality for outward minimizing hypersurfaces in Schwarzschild space". In: *Calculus of Variations and Partial Differential Equations* 57.2 (2018), p. 46. ISSN: 1432-0835. DOI: 10.1007/s00526-018-1342-x. URL: https://doi.org/10.1007/s00526-018-1342-x.
- [Whi00] B. White. "The Size of the Singular Set in Mean Curvature Flow of Mean-Convex Sets". In: *Journal of the American Mathematical Society* 13.3 (2000), pp. 665–695. ISSN: 08940347, 10886834. URL: http://www.jstor.org/stable/2646124.

[Wil68] T. J. Willmore. "Mean curvature of immersed surfaces". In: *An. Şti. Univ. "All. I. Cuza" Iaşi Secţ. I a Mat. (N.S.)* 14 (1968), pp. 99–103.

- [Xia17] J. Xiao. "P-capacity vs surface-area". In: *Advances in Mathematics* 308 (2017), pp. 1318 -1336. ISSN: 0001-8708. DOI: https://doi.org/10.1016/j.aim. 2017.01.007. URL: http://www.sciencedirect.com/science/article/pii/S0001870815304989.
- [Xu96] X. Xu. "Some results on functional capacity and their applications to P-Laplacian problems involving measure data". In: *Nonlinear Analysis: Theory, Methods & Applications* 27.1 (1996), pp. 17 –36. ISSN: 0362-546X. DOI: https://doi.org/10.1016/0362-546X(94)00357-N. URL: http://www.sciencedirect.com/science/article/pii/0362546X9400357N.
- [Yau75] S.-T. Yau. "Harmonic functions on complete Riemannian manifolds". In: *Comm. Pure Appl. Math.* 28 (1975), pp. 201–228. ISSN: 0010-3640. DOI: 10.1002/cpa. 3160280203. URL: https://doi.org/10.1002/cpa.3160280203.
- [Yau76] S.-T. Yau. "Some function-theoretic properties of complete Riemannian manifold and their applications to geometry". In: *Indiana Univ. Math. J.* 25.7 (1976), pp. 659–670. ISSN: 0022-2518. DOI: 10.1512/iumj.1976.25.25051. URL: https://doi.org/10.1512/iumj.1976.25.25051.
- [Yeg09] N. Yeganefar. "On manifolds with quadratic curvature decay". In: *Compositio Mathematica* 145.2 (2009), pp. 528–540. DOI: 10.1112/S0010437X09003947.
- [Zhu97] S. Zhu. "The comparison geometry of Ricci curvature, in Comparison geometry". In: *Fakultät für Mathematik, Universität*. Press, 1997, pp. 221–262.